Corn and Coconut Oil

Antioxidant Properties, Uses and Health Benefits

Constantin Apetrei

Nutrition and Diet Research Progress

NOVA

Free ebooks ==> www.Ebook777.com

www.Ebook777.com

CORN AND COCONUT OIL

ANTIOXIDANT PROPERTIES, USES AND HEALTH BENEFITS

No part of this digital document may be reproduced, stored in a retrieval system or transmitted in any form or by any means. The publisher has taken reasonable care in the preparation of this digital document, but makes no expressed or implied warranty of any kind and assumes no responsibility for any errors or omissions. No liability is assumed for incidental or consequential damages in connection with or arising out of information contained herein. This digital document is sold with the clear understanding that the publisher is not engaged in rendering legal, medical or any other professional services.

NUTRITION AND DIET RESEARCH PROGRESS

Additional books in this series can be found on Nova's website under the Series tab.

Additional e-books in this series can be found on Nova's website under the e-book tab.



NUTRITION AND DIET RESEARCH PROGRESS

CORN AND COCONUT OIL

ANTIOXIDANT PROPERTIES, USES AND HEALTH BENEFITS

CONSTANTIN APETREI Editor



New York

www.Ebook777.com

Copyright © 2015 by Nova Science Publishers, Inc.

All rights reserved. No part of this book may be reproduced, stored in a retrieval system or transmitted in any form or by any means: electronic, electrostatic, magnetic, tape, mechanical photocopying, recording or otherwise without the written permission of the Publisher.

We have partnered with Copyright Clearance Center to make it easy for you to obtain permissions to reuse content from this publication. Simply navigate to this publication's page on Nova's website and locate the "Get Permission" button below the title description. This button is linked directly to the title's permission page on copyright.com. Alternatively, you can visit copyright.com and search by title, ISBN, or ISSN.

For further questions about using the service on copyright.com, please contact: Copyright Clearance Center Phone: +1-(978) 750-8400 Fax: +1-(978) 750-4470 E-mail: info@copyright.com.

NOTICE TO THE READER

The Publisher has taken reasonable care in the preparation of this book, but makes no expressed or implied warranty of any kind and assumes no responsibility for any errors or omissions. No liability is assumed for incidental or consequential damages in connection with or arising out of information contained in this book. The Publisher shall not be liable for any special, consequential, or exemplary damages resulting, in whole or in part, from the readers' use of, or reliance upon, this material. Any parts of this book based on government reports are so indicated and copyright is claimed for those parts to the extent applicable to compilations of such works.

Independent verification should be sought for any data, advice or recommendations contained in this book. In addition, no responsibility is assumed by the publisher for any injury and/or damage to persons or property arising from any methods, products, instructions, ideas or otherwise contained in this publication.

This publication is designed to provide accurate and authoritative information with regard to the subject matter covered herein. It is sold with the clear understanding that the Publisher is not engaged in rendering legal or any other professional services. If legal or any other expert assistance is required, the services of a competent person should be sought. FROM A DECLARATION OF PARTICIPANTS JOINTLY ADOPTED BY A COMMITTEE OF THE AMERICAN BAR ASSOCIATION AND A COMMITTEE OF PUBLISHERS.

Additional color graphics may be available in the e-book version of this book.

Library of Congress Cataloging-in-Publication Data

ISBN: ; 9: /3/856: 5/684/; (eBook) Library of Congress Control Number: 2015945688

Published by Nova Science Publishers, Inc. † New York

CONTENTS

Preface		vii
Chapter 1	Chemical Composition of Corn Oil Constantin Apetrei and Irina Mirela Apetrei	1
Chapter 2	Biosynthesis, Analysis and Antioxidant Properties of Tocols and Carotenoids from Corn Oil Vasile Robert Gradinaru and Luiza Madalina Gradinaru	29
Chapter 3	Supercritical Fluid Extraction of Corn Germ Oil Sara Rebolleda, María Teresa Sanz, Sagrario Beltrán, Rodrigo Melgosa and Angela García Solaesa	53
Chapter 4	Uses and Applications of the Corn Oil Hakan Temur	79
Chapter 5	Corn Oil and Aging: Insights from Basic Research Hongwei Si	93
Chapter 6	Production Methods and Coconut Oil Quality Kapila N. Seneviratne and Nimanthi Jayathilaka	103
Chapter 7	Quality Analyses and Authentication of Coconut Oil Irina Mirela Apetrei and Constantin Apetrei	131
Chapter 8	The Effects of Fatty Acid Derivates from Corn and Coconut Oils on Microbial Physiology Leontina Gurgu, Georgiana Horincar and Gabriela Bahrim	159
Chapter 9	Nutritional and Health Benefits of Coconut Oil Ratheesh Mohanan, Asha Sukumarapillai and Sandya Sukumaran	187
Chapter 10	Beneficial Effects of Fatty Acids from Coconut Oil on Human Metabolism and Health Leontina Gurgu	197
Index		219

PREFACE

This book presents the properties, uses and health benefits of corn and coconut oils, which are important edible oils in different countries around the world. Corn and coconut oils are found in the market in two forms: virgin (crude, raw) or refined. Virgin coconut oil is a relative newcomer in the oil market and could be valuable oil, comparable with virgin olive oil. Furthermore, corn oil is a beneficial natural food product with excellent nutritional properties and health benefits.

In this book, we provide important data about the chemical composition, production, uses, and benefices of corn and coconut oils. The book comprises ten chapters and an index including numerous technical or scientific terms founds in all the chapters. The book is organized in the following way:

Chapter 1 deals with the description of the chemical composition of corn oil, highlighting the structure and properties of biologic active compounds present in this natural food product.

Chapter 2 describes the biosynthesis, analysis and antioxidant properties of tocopherols, tocotrienols and carotenoids from corn oil.

Chapter 3 presents technological and practical aspects related to supercritical fluid extraction of corn germ oil, a method that permits the efficient extraction of the oil and the preservation of biologic active compounds.

Chapter 4 describes different aspects related to uses and applications of the corn oil in different fields, such as foods, biodiesel production, etc.

Chapter 5 deals with the role of the corn oil in the aging process, presenting the long-term beneficial effects of corn oil in animals.

Chapter 6 present the coconut oil production methods highlighting the differences in quality of the oil produced in different manners.

Chapter 7 describes the most important methods involved in quality analyses and authentication of coconut oil.

Chapter 8 presents the antimicrobial activity of fatty acid derivates from corn and coconut oils against a large spectrum of Gram-positive and Gram-negative bacteria, yeasts, fungi, viruses and parasites, which cause spoilage or various infections in both animals and humans.

Chapter 9 describes the principal aspects related to the nutritional and health benefits of coconut oil.

The book closes with Chapter 10, which presents the beneficial effects of fatty acids from coconut oil on human metabolism and health.

Free ebooks ==> www.Ebook777.com

Constantin Apetrei

I gratefully acknowledge the help of the distinguished specialists who contributed to this book. I consider myself lucky to have had the opportunity to collaborate with colleagues from universities from India, Spain, Sri Lanka, Turkey, the United States, and Romania, who have elaborated the chapters based on their research experience. Finally, I want to thank Nova Publishers for the opportunity offered to contribute, assemble and edit this book.

Constantin Apetrei

Associate Professor, PhD Department of Chemistry, Physics and Environment Faculty of Sciences and Environment "Dunarea de Jos" University of Galati 47 Domneasca Street 800008 Galati, Romania E-mail: apetreic@ugal.ro

www.Ebook777.com

Chapter 1

CHEMICAL COMPOSITION OF CORN OIL

Constantin Apetrei^{1,*} and Irina Mirela Apetrei²

¹Faculty of Sciences and Environment, "Dunarea de Jos" University of Galati, Galati, Romania ²Faculty of Medicine and Pharmacy, "Dunarea de Jos" University of Galati, Galati, Romania

ABSTRACT

Corn or maize (Zea mays L.) originates from both North and South America. It is one of the most important cereal crops cultivated all over the world next to rice (Oryza sativa) and wheat (Triticum spp.). Corn belongs to Maydae tribe that is included in the subfamily Panicoideae, Gramineae family. The main cultivators are The United States of America, China, Brazil, Mexico, Romania, Argentina and France. A large percent of the corn grains is processed and fed to animals. Corn is also used for human consumption, fresh or dried and grounded into flour, being an important ingredient in various dishes around the world. Only a small fraction of the corn is used for obtaining corn oil. Usually, oil is extracted from the wet milled germ using a heating process, followed by mechanical expelling and hexane extraction. The major components of the crude corn germ oil are triacylglycerols. The crude oil also contains other minor nonpolar and polar lipid components such as free fatty acids, phytosterol fatty acyl esters, free phytosterols, phytosterol ferulate esters, tocopherols, tocotrienols, phospholipids, pigments, volatile compounds, waxes etc. The major undesirable components of the crude corn oil (the free fatty acids, the pigments, the volatile compounds, the phospholipids, and the waxes) are removed by appropriate refining steps. Corn oil is alsoused in cooking and in different industrial processes, i.e., margarine fabrication and biodiesel production. Among vegetable oils, corn oil is important because of its special characteristics, properties, nutritional and health benefits.

Keywords: corn oil, triacylglycerol, tocopherol, phytosterol, carotenoid

^{*} Corresponding author: Department of Chemistry, Physics and Environment, Faculty of Sciences and Environment, "Dunărea de Jos" University of Galați, 47 Domneasca Street, 800008 Galați, Romania, E-mail address: apetreic@ugal.ro.

INTRODUCTION

Maize oil (corn oil) is obtained from maize germ (the embryos of *Zea mays L*.) (Codex Alimentarius, 1999). Unlike most other vegetable oils, corn oil (maize oil) is obtained from seeds (kernels) that only contain 3-5% oil. Obtaining oil directly from the kernels is technically possible, but 'corn kernel oil' would be costly to produce, because of the low levels of oil in the kernels. Corn germ is rich in oil (>30%), and is the source of all the commercial corn oil, which could more accurately be called *corn germ oil* (Gunstone, 2011).

Corn oil is obtained from the germ, which is recovered during starch production. The germ from the corn flour industry contains 20% oil. The quality of the corn oil depends on the pretreatment and the separation of the germ. Traditionally, corn oil is obtained from the wet milled germ using a heating process, followed by mechanical press and hexane extraction. Extrusion is employed as a means of germ preparation for solvent extraction, producing a crude corn oil of high quality and high yield (Williams et al., 1997). Other scientists have demonstrated that corn germ can be effectively extracted by supercritical fluid extraction (King & List, 1996).

Crude corn oil contains principally triglycerides and minor nonpolar and polar lipid components such as free fatty acids, phytosterol fatty acyl esters, free phytosterols, phytosterol ferulate esters, tocols, tocopherols, tocotrienols; glycolipids, phospholipids, pigments, volatile compounds, waxes etc. (Gunstone, 2011).

Corn oil consists of about 95% triacylglycerols with 1.5% phospholipids, 1.7% free fatty acids, and 1.2% sterols (Fennema, 1996).

Free fatty acids, pigments, volatile compounds, phospholipids, and waxes are the major undesirable components of the crude corn oil and are removed by appropriate refining processes (Nagaraj, 2009).

In corn oil refining processes, the removing of the free fatty acids takes place by alkali refining, which involves adding base and subsequently neutralizing and separating the free fatty acid soaps. Otherwise, free fatty acids can be removed by physical refining, which involves treating the oil at high temperature and vacuuming, in order to volatilize the free fatty acids. The removing of the phospholipids is carried out by water degumming. If the phospholipids are not adequately removed, they result in a corn oil that will form dark colors and off-flavors when heated (Nagaraj, 2009).

The corn oil deodorization involves treatment at high temperature and vacuuming and the removing of the undesirable odors and flavor components (O'Brien, 2008).Pigments are usually removed by treating the oil with acid-activated bleaching clay. Another refining step that ensures the physical stability of the oils at low temperature is dewaxing or winterization. This involves cooling the oil to 5-10°C, and removing the precipitates by filtration (Antoniassi et al., 1998.

GENERAL COMPOSITION AND PHYSICAL PROPERTIES OF CORN OIL

Corn oil is primarily consists of triacylglycerols (> 95%) and secondarily free fatty acids, mono- and diacylglycerols, and several lipid compounds such as hydrocarbons, sterols, aliphatic alcohols, tocopherols, and pigments (Gunstone, 2011).

Property	Values
Relative density (20°C/water at 20°C)	0.917-0.925
Refractive index (at 25°C)	1.470-1.474
Refractive index (at 40°C)	1.465-1.468
Saponification value (mg KOH/g oil)	187-195
Unsaponifiable matter (g×kg ⁻¹)	10-28
Iodine value (Wijs method)	103-135
Viscosity (cP)	30.80 at 40°C
	18.15 at 60°C
Dielectric constant	3.954 at 26°C
Surface tension at 25°C (dyn/cm)	34.80
Thermal conductivity at 130°C (J/s/cm ² /°C)	4.2017×10 ⁻⁵
Stability	6.2 ^a
Melting point (°C)	-18 to -10
Ignition point (°C)	393
Flash point (°C)	321
Smoke point (°C)	230 to 238
Fire point (°C)	366 to 371
Cloud point (°C)	-14 to -11
Stable carbon isotope ratio	13.71 to 16.36

	Tabl	le 1.	. Phy	vsical	and	chemical	characteristics	of	corn	oil	(Maize)
--	------	-------	-------	--------	-----	----------	-----------------	----	------	-----	--------	---

^a Stability calculated as decimal fraction fatty acids multiplied by relative rate of reaction with oxygen of each fatty acid

Corn oil contains triglycerides plus phospholipids, glycolipids, phosphoinositides (phospholipids containing inositol), many isomers of sitosterol and stigmasterol (plant steroids), several tocopherols (vitamin E), waxes, unsaturated hydrocarbons such as squalene, carotenoids and chlorophyll compounds, as well as many products of decomposition, hydrolysis, oxidation, and polymerization of any of the natural constituents.

The commercial corn oil has been recognized as containing the highest levels of unsaponifiables (1.3–2.3%) of all the commercial vegetable oils (Strecker et al., 1996). The three main chemical components in the unsaponifiable fraction of corn oil are phytosterols, tocopherols and squalene.

The important properties of corn oil include its pleasing flavor, its high levels of polyunsaturated (essential) fatty acids, and its low levels of saturated fatty acids and linolenic acid.

The main physical and chemical characteristics of corn oil are presented in Table 1 (Woodbury et al., 1995a; Woodbury et al., 1995b; Woodbury et al., 1998).

FATTY ACID COMPOSITION OF CORN OIL

The fatty acid composition of oils is determined by the gas-liquid chromatography of the methyl esters of the fatty acids. Saponification (hydrolysis) followed by methylation is a

typical method for preparing the fatty acid methyl esters from glycerolipids and sterol esters (Sheppard & Iverson, 1975; Liu, 1994; Christie, 2003).

The fatty acids present in corn oil are palmitic (C16:0), palmitoleic (C16:1), stearic (C18:0), oleic (C18:1), linoleic (C18:2), and linolenic (C18:3). Other fatty acids are found in small or trace amounts (Dubois et al., 2007).

The fatty acid compositional limits adopted in the most recent editions of Codex Alimentarius are presented in Table 4.

	Typical Value (%)		
	Crude	Refined	
Triglycerides	95.6	98.8	
Free fatty acids	2.5	0.05	
Phospholipids	1.5	0.0	
Unsaponifiables			
Phytosterols	1.2	1.1	
Tocopherols	0.12	0.08	
Waxes	0.01	0.0	
Color variable:	very dark to yellow	pale yellow	
Odor and flavor	strong corn/feed	slight corn slight nutty/buttery	

Table 2. Comparative composition of crude & refined corn oil (adapted from Blanchard
et al., 1992; http://www.corn.org/)

Table 3. Approximate composition of refined corn oil (adapted
from Sayed & El Goud, 2010)

Amount in 100 g of oil				
Moisture	None			
Protein (N×6.25)	None			
Fats, total (g)	100.0			
Triglycerides (g)	98.8			
Polyunsaturated (g)	59.7			
Monounsaturated (g)	26.0			
Saturated (g)	13.1			
Phytosterols (mg)	1000			
Tocopherols (mg)	88			
Carbohydrates (s)	None			
Ash (g)	None			

		-
Fatty Acid	Symbol	Percentage (%)
Lauric acid	C12:0	ND-0.3
Myristic acid	C14:0	ND-0.3
Palmitic acid	C16:0	8.6-16.5
Palmitoleic acid	C16:1	ND-0.5
Heptadecanoic acid	C17:0	ND-0.1
Heptadecenoic acid	C17:1	ND-0.1
Stearic acid	C18:0	ND-3.3
Oleic acid	C18:1	20.0-42.2
Linoleic acid	C18:2	34.0-65.6
Linolenic acid	C18:3	ND-2.0
Arachidic acid	C20:0	0.3-1.0
Eicosenoic acid	C20:1	0.2-0.6
Eicosadienoic acid	C20:2	ND-0.1
Behenic acid	C22:0	ND-0.5
Erucic acid	C22:1	ND-0.3
Lignoceric acid	C24:0	ND-0.5

Table 4. Fatty acid composition of corn as determined by gas-liquid chromatography (expressed as percentage of the total fatty acids) (adapted from Codex Alimentarius)

The chemical structures of the most important fatty acids in the corn oil are presented in the Figure 1.

Edible oils are often compared by examining their fatty acid profiles. Corn oil has a high content of the essential fatty acid linoleic acid (18:2). As is well known, an essential fatty is one that is required by humans but cannot be biosynthesized and must, therefore, be consumed from food (Das, 2006). Linoleic acid (also called *cis, cis,*-9, 12-octadecadienoic acid) is an ω -6 fatty acid. The numerous benefits of linoleic acid include cancer prevention, rheumatoid arthritis management, heart protection, etc. (Rainer & Heiss, 2004).

Another desirable characteristic of corn oil is that it contains low levels (<15%) of saturated fatty acids and very low levels of linolenic acid (18:3). It is well known that linolenic acid is especially susceptible to oxidation, which leads to rancidity. Another important free fatty acid present in corn oil is the oleic acid. The oleic acid ((9Z)-octadec-9-enoic acid) is a monounsaturated ω -9 fatty acid, associated with decreased low-density lipoprotein cholesterol, blood pressure reducing effects, decreased risk of breast cancer, etc. (Sales-Campos et al., 2013).

Besides the main fatty acids, corn oil contains up to 0.3% each of myristic and palmitoleic acid as well as 0.4% each of behenic and of lignoceric acid. The oils produced from the corn cultivated in the northern hemisphere contain 10-13% more linoleic acid than those in the southern hemisphere (Gunstone, 2011).

Furan fatty acids are part of the corn oil minor constituents. Two of these acids present the chemical structures exposed in Figure 2.



Figure 1. Chemical structures of a) palmitic acid (16:0); b) stearic acid (18:0); c) arachidic acid (20:0); d) oleic acid (18:1); e) linoleic acid (18:2); f) linolenic acid (18:3).



Figure 2. Chemical structures of furan fatty acids found in corn oil.

Triacylglycerol	Number of double bonds
LLO	5
LLL	6
LLP	4
OOL	4
PLO	3
PPL	2
OOP	2
LLS	4
LOS	3
000	3
PPO	1
PLS	2
LLLn	7
LnLO	6
OOS	2
POS	1
PLnL	5
PPP	0
OOLn	5
PLnO	4
PPS	0
SSL	2
LnLS	5
SSO	1
PPLn	3
SSP	0
SSS	0

Table 5. Main triacylglycerols from corn oil and the number of double bonds per molecule

Abbreviations: Ln, linolenic acid; L, linoleic acid; O, oleic acid; S, stearic acid; P, palmitic acid. Symbols such as LOS refer to all the possible triacylglycerols containing these three acids. (adapted from Strecker et al., 1990; Byrdwell et al., 2001).

The compound I (10, 13-epoxy-11, 12-dimethyloctadeca-10, 12-dienoic acid) is found in a concentration of 8-11 mg×kg⁻¹ and the compound II (12, 15-epoxy-13, 14-dimethyleicosa-12, 14-dienoic acid) is found in a concentration of 9-13 mg×kg⁻¹ (Belitz et al., 2009).

TRIACYLGLYCEROLS FROM CORN OIL

Triacylglycerols form the major part of corn oil, and they largely determine its main characteristics. The molecular structure of each individual triacylglycerol species has three basic characteristics: (a) the total carbon number; (b) the degree of unsaturation in each fatty acid; and (3) the position and configuration of the double bonds in each fatty acid (Buchgraber et al., 2004). Triacylglycerols are esters of the glycerol (a trihydroxyalcohol) with three molecules of one or more different fatty acids. Triacylglycerols are named in relation to the fatty acid components. For example, tristearin contains three molecules of stearic acid, and oleo-dipalmitin, one of oleic acid and two of palmitic acid. The types and relative proportions of triacylglycerols in oils vary with the species from which the oils have been obtained. The large number of triacylglycerols makes the analysis of the triacylglycerol composition a very challenging task. Reversed-phase HPLC techniques are currently used to quantitatively analyze the triacylglycerol molecules of plant and animal origin oils (Nikolova-Damyanova, 1997).

Reports on the triacylglycerol molecules from the refined corn oil have led to the identification of 19-27 individual molecules (Table 5), from which oleate-linoleate-linoleate and linoleate-linoleate are the two most abundant molecules (Figure 3).

The fatty acids distribution in different triacylglycerols from corn oil shows that more than 65% of the triacylglycerols contain four or more double bonds and that more than 40% even contain five or more (Figure 4). Corn oil is considered a very stable oil.



Figure 3. Chemical structures of 1-oleo -2-linoleo-3-linolein.



Figure 4. Triacylglycerols composition of corn oil (Strecker et al., 1990).

In Figure 3 presents the chemical structure of 1-oleo -2-linoleo-3-linolein, one of the most abundant triacylglycerol found in corn oil.

Monoacylglycerols and diacylglycerols are partial esters of glycerol and have one or two fatty-acid radicals, respectively. They are naturally associated with oils and fats. They can be produced by the reaction of fats and oils with glycerol or by the hydrolysis of the tryacylglicerols. Monoacylglycerol and diacylglycerol act as prooxidants, which increase oxidation at moderate temperature in the dark (Chaiyasit et al., 2007). Their surface-active properties that stabilize the emulsion are due to the hydrophilic properties of the glycerol-OH residue and to the hydrophobic properties of the fatty acid chains of the ester.

Monoglycerols and diacylglycerols should be removed from the oil during the oilrefining process, in order to improve the oxidative stability of the edible oils (Waraho, 2011). Monoglycerols and diacylglycerols are used as emulsifiers (Gulik-Krzywicki & Larsson, 1984; Byrdwell et al., 2001).

PHOSPHOLIPIDS

Crude corn oil contains lecithins, whose role is to stabilize the oil and to deposit it as energy reserve in the seeds. During the oil extraction, a part of these disperses in the oil; usually, they are separated and sold as special products (List, 1989).

Constantin Apetrei and Irina Mirela Apetrei



Basic structure of phospholipids R₁, R₂ - fatty acid chains.



Figure 5. Chemical structure of the most important glycerophospholipids.

The term "lecithin" includes those phospholipids that are found in the oil after extraction. There are three main classes of compounds: phosphatidylcholine, phosphatidylethanolamine, and phosphatidylinosite (Figure 5). The family of phospholipids includes a great variety of substances because of the very different fatty acid residues (R_1 , R_2) and of the possibility of binding to two different positions, 1 or 2 (Shahidi, 1997).

In crude corn oil, phosphatides can be found at levels of 1-2%. The different varieties of crude oil had phospholipid-phosphorus values between 145 and 361 mg×kg⁻¹. For oils at further stages of refining, the values dropped to 4.9 - 15.5 mg×kg⁻¹. Degummed corn oils contained phosphatidylcholine (1.1 - 22 mg×kg⁻¹ phosphorus), phosphatidylethanolamine (1.5 - 2 mg×kg⁻¹ phosphorus), phosphatidylinositol (1.6 - 10 mg×kg⁻¹ phosphorus) and phosphatidic acid (trace - 5 mg×kg⁻¹ phosphorus). Further refined oils contained no phospholipids (Shahidi, 1997).

The level of phospholipids is important because these compounds carry out an antioxidant activity. The phospholipids may act as synergists (regenerating antioxidants such as tocopherols or phenolic compounds) or as metal scavengers (Pokorny & Korczak, 2001). At high levels, however, phospholipids may cause foaming or darkening during frying.

www.Ebook777.com

PHYTOSTEROLS FROM CORN OIL

Sterols are polycyclic alcohols derived from sterane (cyclopentanoperhydrophenantrene) as a base structure. One of the most important sterols is cholesterol, the first discovered sterol. Corn oils are cholesterol free, following the general agreement according to which all substances containing <50 ppm cholesterol are regarded as cholesterol free. Cholesterol, as a constituent of animal fat, is important as far as it can be linked to heart diseases (Dutta, 2003).

In general, vegetable oils have much higher concentrations of phytosterols than nonfatty vegetable foods. Among the commonly used commercial oils containing 0.77% phytosterols by weight (Ostlund et al., 2002), corn oil is one of the richest sources of phytosterol. Phytosterols are not absorbed by the human body (Weirauch & Gardner, 1987).

Corn germ oil contains two phytosterol lipid classes, free phytosterols and phytosterol fatty acid esters. Phytosterols have been recognized as one of the most important classes of phytonutrients (Ostlund et al., 2002). Most chemical identification processes of phytosterols in vegetable oils have been carried out by saponifying the oil and by measuring the resulting free phytosterols, usually trough GLC (Table 6).

The chemical structures of sterols present in corn oil are presented in the Figure 6.

Sterol	Composition (%)
Cholesterol	0.2-0.6
Brassicasterol	ND-0.2
Campesterol	16.0-24.1
Stigmasterol	4.3-8.0
β–sitosterol	54.8-66.6
Δ^5 -Avenasterol	1.5-8.2
Δ^7 -Stigmastenol	0.2-4.2
Δ^7 -Avenasterol	0.3-2.7
Others	ND-2.4
Stigmasta-8,22-dien-3β-ol	
5α-Stigmasta-7,22-dien-3β-ol	
$\Delta^{7,25}$ -Stigmastadienol	
Δ^7 -Campesterol	
Δ^7 -Ergosterol	
$\Delta^{7,25}$ -Stigmasterol	
Sitostanol	
Spinasterol	
24-Methylene cholesterol	
Total sterols ($mg \times kg^{-1}$)	7000-22100

Table 6. Levels of sterols in crude corn oils as a percentage of total sterols (the values represent free and esterified phytosterols, measured after saponification)













 Δ^7 -Campesterol - (24R)-24-Methyl- Δ^7 -cholesten-3 β -ol.







24-Methylene Cholesterol - 24- Methylene- Δ^5 -cholesten-3 β -ol.

Figure 6. Chemical structures of sterols from corn oil.

The examination of the total phytosterols in corn oil has revealed that the major phytosterol was β -sitosterol (Table 6). The major phytosterols in corn germ oil are β -sitosterol > campesterol > stigmasterol (Table 6). β -sitosterol is also found in the form of ferulate ester, displaying antioxidant properties.

TOCOPHEROLS AND TOCOTRIENOLS

Corn oil has long been recognized as a rich source of tocopherols, γ -tocopherol being the most abundant tocopherol, followed by α -tocopherol and δ -tocopherol (Table 7)(Codex Alimentarius, 1997; Fennema, 1996; Kamal-Eldin & Andersson, 1997).

Tocopherol	Composition (mg×kg ⁻¹)	
α-Tocopherol	23-573	
β-Tocopherol	ND-356	
γ-Tocopherol	268-2468	
δ-Tocopherol	23-75	
Total, mg×kg ⁻¹	330-3720	

Table 7. Levels of tocopherols in crude corn oils



Figure 7. Chemical structure of tocopherols.

Tocopherol	R ₁	R ₂	R ₃
α-Tocopherol	CH ₃	CH ₃	CH ₃
β-Tocopherol	CH ₃	Н	CH ₃
γ-Tocopherol	Н	CH ₃	CH ₃
δ-Tocopherol	Н	Н	CH ₃

Tocotrienols	Composition (mg×kg ⁻¹)
α-Tocotrienol	ND-239
β-Tocotrienol	ND-52
γ-Tocotrienol	ND-450
δ-Tocotrienol	ND-20
Total, mg×kg ⁻¹	ND -709

Table 8. Levels of tocotrienols in crude corn oils



Figure 8. Chemical structures of tocotrienols.

Tocotrienol	R ₁	R ₂	R ₃
α-Tocotrienol	CH ₃	CH ₃	CH ₃
β-Tocotrienol	CH ₃	Н	CH ₃
γ-Tocotrienol	Н	CH ₃	CH ₃
δ-Tocotrienol	Н	Н	CH ₃

High performance liquid chromatography (HPLC) is currently the preferred method for determining the tocols (Shahidi, 1997; Gimeno et al., 2000).

Tocopherols are the most important antioxidants present in corn oil. Among the tocopherols, α -tocopherol has received the most attention because of its vitamin E activity; however, other isomers also are known to display valuable antioxidant properties. Some studies suggest the fact that γ -tocopherol may be superior to α -tocopherol in preventing the oxidation of lowdensity lipoproteins and in delaying the thrombus formation. Tocopherols are used as antioxidants because they trap hydroperoxide intermediates both in vitro and in vivo, thus breaking the chain reaction. Tocopherols decrease corn oil oxidation under light by singlet oxygen ($^{1}O_{2}$) quenching (Burton & Ingold, 1981; Rizvi et al., 2014; Di Mascio et al., 1991).



Figure 9. Chemical structure of squalene.

Significant levels of tocotrienols (the most abundant was γ -tocotrienol followed by α -tocotrienol) were found in corn oil (Dolde, 2009). Tocotrienols have antioxidant properties and in addition, it is currently believed that tocotrienols also possess cholesterol-lowering properties, which are related to their ability to inhibit the cholesterol biosynthesis (Palozza et al., 2006).

The refining process reduces tocopherol contents (Ergönül & Köseoğlu, 2014).

TERPENOIDS AND DERIVATIVES FROM CORN OIL

Squalene (2,6,10,15,19,23-hexamethyl-2,6,10,14,18,22-tetracosahexaene) is the last metabolite preceding sterol ring formation. Squalene consists of six isoprene-units and contains six (all-trans) double bonds. On the basis of GC examination it was reported that squalene was the major hydrocarbon in corn germ oil (Worthington & Hitchcock, 1984). Squalene is a compound beloging to terpenoid family.

Squalene is contained in corn oil in percentages between 0.016-0.042 and is a marker substance that proves the adulteration of the olive oil.

Squalene presence is regarded as partially responsible for the beneficial health effects of corn oil and its chemopreventive action against certain types of cancer (Rao et al., 1998).

Carotenoids. Carotenoids are terpenoids, the majority having a central skeleton consisting of 40 carbon atoms, formed of eight isoprenoid units. Carotenoids are widely found among living beings, plants and animals. Higher concentrations and greater varieties are found within the plant kingdom (Sajilata et al., 2008).

The amounts of carotenoids in the commercial corn oil are relatively low, partly due to their low concentrations in the germ and partly to their removal during bleaching (Moreau et al., 2007). The most abundant carotenoids in corn oil are lutein and zeaxanthin (Figures 10 and Figure 11).



Figure 10. Chemical structure of lutein.



Figure 11. Chemical structure of zeaxanthin.

Lutein and zeaxanthin are yellow to red pigments founded broadly in vegetables, fruits, and cereals, including corn. Lutein and zeaxanthin are found in high quantities in the macula of the human eye (Sommerburg et al., 1998). Lutein is a yellow pigment which, in elevated concentrations, is orange-red. Lutein and zeaxanthin are detected and quantified by high-performance liquid chromatographic, using C18 column to increase the resolution of the peaks (Gilmore & Yamamoto, 1991).

Lutein and zeaxanthin are important in nature because they absorb the light energy excess, avoiding damaging the plants from too much sunlight. In the human body, lutein and zeaxanthin play significant antioxidant roles. These pigments protect the body cell from the destructive effects of the free radicals. However, in certain conditions, carotenoids in vegetable oils and certain other food matrices may serve as pro-oxidants, especially at higher concentrations (Fiedor & Burda, 2014).

Lutein and zeaxanthin may reduce the risk of macular degeneration and cataracts. Besides the significant eye and vision benefits, lutein and zeaxanthin protect the body against atherosclerosis, the disease closely related to heart attacks (Sommerburg et al., 1998; Fiedor & Burda, 2014).

Chlorophylls and pheophytins. Chlorophyll pigments are widely spread in the natural environment, chlorophylls a and b being the most abundant. Chlorophylls are included in the porphyrin group. They have a basic common structure, the porphin, with four units of pyrrole linked at the alpha positions by methylene bridges in a planar aromatic system, which is highly stable and amenable to the formation of chelates with metal ions. For the chlorophylls, the metal ion that forms the complex is Mg²⁺ (Shahidi, 1997).



Figure 12. Chemical structures of chlorophylls.



Figure 13. Chemical structure of phytol.

Chlorophyll and pheophytin contents have been determined in refined corn oils trough HPLC and fluorescence detection (Usuki et al., 1984). The total of the contents varied from 95 to 162 μ g×kg⁻¹ oil. The chemical structures of chlorophyll a and chlorophyll b are presented in Figure 12.

Pheophytin is a bluish-black waxy pigment that can be obtained from chlorophyll by replacing the magnesium with two hydrogen atoms by treatment with a weak acid (as oxalic acid). Chlorophylls are generally removed during oil processing through the bleaching process (Akoh & Min, 2008).

Chlorophylls and pheophytins act as sensitizers to produce ${}^{1}O_{2}$ in the presence of light and atmospheric ${}^{3}O_{2}$, and accelerate the oxidation of the oil (Choe & Min, 2006). Chlorophylls act as antioxidants in the dark probably by donating hydrogen to free radicals (Endo et al., 1985).

Phytol (3, 7, 11, 15-tetramethylhexadec-2-en-1-ol) is a diterpene, a member of the group of branched-chain unsaturated alcohols (Figure 13). Being the product of chlorophyll metabolism in plants, phytol is widely spread in nature. Phytol probably originates in corn oil from chlorophylls.

D Differentiang nonrefined and refined oils is an important task in food control. In chlorophyll only trans-phytol is to be found. The relative abundance of *cis*-phytol versus *trans*-phytol has been used as a marker in authenticating the nonrefined edible oils. The absence of cis-phytol in the nonrefined edible oils can be used as a marker for authentication (Vetter et al., 2012).

PHENOLIC COMPOUNDS FROM CORN OIL

Polar phenol fraction is a complex mixture of compounds with different chemical structures obtained from the oil by extraction with methanol-water. The methanol-water extract of corn oil mainly contains phenolic acids (Shahidi & Naczk, 2003).

In corn oil, p-hydroxybenzoic, vanillic, caffeic, syringic, p-coumaric and ferulic acid have been detected; out of which the ferulic acid was the most abundant at 5.79 mg×kg⁻¹, followed by the p-coumaric acid at 1.63 mg×kg⁻¹ (Daigle et al., 1988; Shahidi, 1997). Figure 14 presents the chemical structures of the phenolic compounds from the corn oil.



Figure 14. Chemical structures of the phenolic compounds from corn oil.

The spectrophotometric method commonly applied in determining the phenols in the water-methanol extract is based on the use of Folin-Ciocalteu reagent. Results are usually expressed as gallic acid equivalents (mg gallic acid/kg oil) (Rover & Brown, 2013).

Sophisticated gas chromatographic and liquid chromatographic methods have been developed for the analysis of polar phenols. These methods are useful in elucidating the complex phenolic composition of the polar fraction of the oils, but they cannot be easily applied in the routine analysis because of its high overall cost (Snyder et al., 2011).

Corn oil phenols have been studied with respect to their potential to scavenge the free radicals, the peroxyl radicals, the superoxide radicals, etc. The free radical scavenging activity of the phenolic acids using model free radicals such as 1,7-diphenyl-2-picrylhydrazyl radical (DPPH[•]), or ABTS^{•+} (2,2'-azinobis(3-ethylbenzothiazoline-6-sulfonic acid) diammonium salt cation radical) is often measured. TEAC (Trolox equivalent antioxidant capacity), FRAP (Ferric-reducing antioxidant capacity) and TRAP (total-radical-trapping antioxidant parameter) of the corn oil are also used in order to obtain additional information, necessary to investigate the relation between the antioxidant intake and the oxidative stress related diseases (Stratil et al., 2008; Cetó et al., 2014; Apak et al., 2007).

Phenolic compounds are related to the stability of the corn oil but also to its biological properties (Shahidi, 1997).



Figure 15. Chemical structure of ubiquinol, semiquinone radical and ubiquinone (n=9, 10).

Corn oil also contains ubiquinone (also known as co-enzyme Q), which in its turn, possess an anti-oxidant property. The presence of the oxidized and reduced forms of the Q (9) and Q (10) ubiquinones has been determined in the commercial corn oil (Cabrini et al., 2001). Very high concentrations of ubiquinones have been found in corn oils (Shahidi, 1997; Cabrini et al., 2001; Kokate et al., 2008; Guad et al., 2006).

Corn oil generally is considered to be fairly stable to oxidation. The quality of its flavor is very good. Corn oil is the only oil containing a significant amount of ubiquinone (200 mg \times kg⁻¹; a benzoquinone with a side chain of 6-10 prenyl units), an excellent antioxidant that helps protect oils against oxidative rancidity (Petillo & Hultin, 2008; Shahidi, 1997).

VOLATILE COMPOUNDS FROM CORN OIL



Figure 16. Chemical structure of volatile compounds from corn oil .

Several compounds belonging to different chemical classes have been identified and quantified in corn oil. Volatile compounds including 2,4-decadienal, nonanal, hexanoic acid, 2-pentyl-furan, pentanal, 2-decenal, hexanal, and pentane identified in corn oils from supercritical fluid-extracted oils have been analyzed with the help of the headspace gas chromatography (GC) (Figure 16) (Snyder & King, 1994).

2-Pentyl furan, which has been was identified in corn oil is a volatile compound obtained from the oxidation of the corn oil. The flavor threshold of this compound in oil, at room temperature, is 1 ppm. At concentrations of 1-10 ppm, it imparts, the oil a characteristic beany odor (Krishnamurthy et al., 1967; Steenson et al., 2002)

FUTURE TRENDS AND PERSPECTIVES

Corn oil's desirable properties include: a mild nutty flavor, high levels of unsaturated fatty acids, low levels of saturated fatty acids and very low levels of linolenic acid, high levels of desirable unsaponifiables (including phytosterols, tocopherols, phenolic compounds, etc.), and stability during frying. Nutritional properties and health benefits of the biological active compounds from corn oil have been proved by numerous studies published in literature. All these features make corn oil an excellent dietary choice that supports good overall health.

ACKNOWLEDGMENT

This work was supported by a grant of the Romanian National Authority for Scientific Research, CNCS-UEFISCDI, project number PN-II-ID-PCE-2011-3-0255.

REFERENCES

- Akoh, C. C. & Min, D. B. (2008). *Food Lipids: Chemistry, Nutrition, and Biotechnology,* Third Edition, CRC Press.
- Antoniassi, R., Esteves, W. & de Almeida Meirelles, A. J. (1998). Pretreatment of corn oil for physical refining, J. Am. Oil Chem. Soc., 75, 1411-1415.
- Apak, R., Güçlü, K., Demirata, B., Özyürek, M., Esin Çelik, S., Bektaşoğlu, B., Berker, K. I. & Özyurt, D. (2007). Comparative Evaluation of Various Total Antioxidant Capacity Assays Applied to Phenolic Compounds with the CUPRAC Assay. *Molecules*, 12, 1496-1547.
- Belitz, H. D., Grosch, W., Schieberle, P. (2009). *Food Chemistry*, Springer Science & Business Media.
- Blanchard, P. H. (1992). Technology of Corn Wet Milling and Associated Processes, Elsevier Science Ltd.
- Buchgraber, M., Ulberth, F., Emons, H. & Anklam, E. (2004). Triacylglycerol profiling by using chromatographic techniques. *Eur. J. Lipid Sci.Tech.*, 106, 621-648.

Free ebooks ==> www.Ebook777.com

- Burton, G. W. & Ingold, K. U. (1981). Autoxidation of biological molecules: 1. The antioxidant activity of vitamin E and related chain-breaking phenolic antioxidants in vitro. J. Am. Chem. Soc., 103, 6472–6477.
- Byrdwell, W. C., Neff, W. E. & List, G. R. (2001). Triacylglycerol analysis of potential margarine base stocks by high-performance liquid chromatography with atmospheric pressure chemical ionization mass spectrometry and flame ionization detection. J. Agric. Food Chem., 49, 446-457.
- Cabrini, L., Barzanti, V., Cipollone, M., Fiorentini, D., Grossi, G., Tolomelli, B., Zambonin, L. & Landi, L. (2001). Antioxidants and total peroxyl radical-trapping ability of olive and seed oils. J. Agric. Food Chem., 49, 6026-6032.
- Cetó, X., Apetrei, C., del Valle, M. & Rodríguez-Méndez, M. L. (2014). Evaluation of red wines antioxidant capacity by means of a voltammetric e-tongue with an optimized sensor array. *Electrochim. Acta*, 120, 180-186.
- Chaiyasit, W., Elias, R. J., Mcclements, D. J. & Decker, E. A. (2007). Role of Physical Structures in Bulk Oils on Lipid Oxidation, *Crit. Rev. Food Sci. Nutr.*, 47, 299-317.
- Choe, E. & Min, D. B. (2006). Mechanisms and Factors for Edible Oil Oxidation. *Compr. Rev. Food Sci. F.*, *5*, 169-186.
- Christie, W. W. (2003). Lipid Analysis. 3rd edition The Oily Press, Bridgwater, UK.
- Codex Alimentarius, Codex standard for named vegetable oils, CODEX STAN 210-1999, http://www.codexalimentarius.org/download/standards /336/CXS_210e.pdf
- Daigle, D. J., Conkerton, E. J., Sanders, T. H. & Mixon, A. C. (1988). Peanut hull flavonoids: their relationship with peanut maturity. J. Agric. Food Chem., 36, 1179.
- Das, U. N. (2006). Essential fatty acids a review. Curr. Pharm. Biotechnol., 7, 467-482.
- Di Mascio, P., Murphy, M. E. & Sies, H. (1991). Antioxidant defense systems: The role of carotenoids, tocopherols, and thiols. *Am. J. Clin. Nutr.*, 53, 194S–200S.
- Dolde, D. A. (2009). *Oxidative stability of corn oil with elevated tocotrienols*. Graduate Theses and Dissertations. Paper 11035. http://lib.dr.iastate.edu/ etd/11035
- Dubois, V., Breton, S., Linder, M., Fanni, J. & Parmentier, M. (2007). Fatty acid profiles of 80 vegetable oils with regard to their nutritional potential. *Eur. J. Lipid. Sci. Technol.*, 109, 710-732.
- Dutta, P. C. (2003). Phytosterols as Functional Food Components and Nutraceuticals, CRC Press.
- Endo, Y., Usuki, R. & Kaneda, T. (1985). Antioxidant effects of chlorophyll and pheophytin on the autoxidation of oils in the dark. II. The mechanism of antioxidative action of chlorophyll. J. Am. Oil Chem. Soc., 62, 1387-1390.
- Ergönül, P. G. & Köseoğlu, O. (2014). Changes in α-, β-, γ- and δ-tocopherol contents of mostly consumed vegetable oils during refining process, CyTA - Journal of Food, 12, 199-202.
- Fennema, O. R. (Ed.), (1996). Food Chemistry, Third Edition, Marcel Dekker Inc.
- Fiedor, J. & Burda, K. (2014). Potential Role of Carotenoids as Antioxidants in Human Health and Disease. *Nutrients*, *6*, 466-488.
- Gilmore, A. M. & Yamamoto, H. Y. (1991). Resolution of lutein and zeaxanthin using a nonendcapped, lightly carbon-loaded C18 high-performance liquid chromatographic column. *J. Chromatography A*, 543, 137-145.

www.Ebook777.com

- Gimeno, E., Castellote, A. I., Lamuela-Ravento, R. M. & Lopez-Sabater, M. C. (2000). Rapid determination of vitamin E in vegetable oils by reversed phase high-performance liquid chromatography. J. Chromatography, 881, 251-254.
- Guad, R. S., Surana, S. J., Talele, G. S., Talele S. G. & Gokhale S. B. (2006). *Natural Excipients*, Pragati Books Pvt. Ltd.
- Gulik-Krzywicki, T., Larsson, K. (1984). An electron microscopy study of the L2-phase (microemulsion) in a ternary system: triglyceride/ monoglyceride/ water. *Chem. Phys. Lipids*, 35, 127–132.
- Gunstone, F. (2011). Vegetable Oils in Food Technology: Composition, Properties and Uses, John Wiley & Sons. http://www.corn.org/
- Kamal-Eldin, A. & Andersson, R. (1997). A multivariate study of the correlation between tocopherol content and fatty acid composition in vegetable oils. J. Am. Oil Chem. Soc., 74, 375-380.
- King, J. W. & List, G. R. (1996). *Supercritical Fluid Technology in Oil and Lipid Chemistry*, The American Oil Chemists Society.
- Kokate, C. K., Purohit, A. P., Gokhale, S. B. (2008). Pharmacognosy, Nirali Prakashan.
- Krishnamurthy, R. G., Smouse, T. H., Mookherjee, B. D., Reddy, B. R., Chang, S. S. (1967). Identification of 2-Pentyl Furan in Fats and Oils and Its Relationship to the Reversion Flavor of Soybean Oil. J. Food Sci., 32, 372-374.
- List, G. R. (1989). *Commercial manufacture of lecithin*, in Lecithins: Source, Manufacture & Uses. Szuhaj B. F. (ed). AOCS Press.
- Liu, K. S. (1994). Preparation of fatty acid methyl esters for gas-chromatographic analysis of lipids in biological materials. *J. Am. Oil Chem. Soc.*, *71*, 1179-1187.
- Moreau, R. A., Johnson, D. B. & Hicks, K. B. (2007). A comparison of the levels of lutein and zeaxanthin in corn germ oil, corn fiber oil, and corn kernel oil. J. Am. Oil Chem. Soc., 84, 1039-1044.
- Nagaraj, G. (2009). *Oilseeds: Properties, Products, Processing and Procedures*, New India Publishing.
- Nikolova-Damyanova, B. (1997). *Reversed-phase high-performance liquid chromatography:* general principles and applications to the analysis of fatty acids and triacylglycerols. In Advances in Lipid Methodology - Four, W.W. Christie (ed.), Oily Press.
- O'Brien, R. D. Fats and Oils: Formulating and Processing for Applications, Third Edition, CRC Press, 2008.
- Ostlund, R. E. Jr., Racette, S. B., Okeke, A. & Stenson, W. F. (2002). Phytosterols that are naturally present in commercial corn oil significantly reduce cholesterol absorption in humans. *Am. J. Clin. Nutr.*, 75, 1000-1004.
- Palozza, P., Verdecchia, S., Avanzi, L., Vertuani, S., Serini, S., Iannone, A. & Manfredini, S. (2006). Comparative antioxidant activity of tocotrienols and the novel chromanylpolyisoprenyl molecule FeAox-6 in isolated membranes and intact cells. *Mol. Cell. Biochem.*, 287, 21-32.
- Petillo, D. & Hultin, H. O. (2008). Ubiquinone-10 as an antioxidant. J. Food Biochem., 32, 173-181.
- Pokorny, J. & Korczak, J. (2001). Preparation of natural antioxidants. In Pokony, J., Yanishlieva, N., Gordon, M. (eds). Antioxidants in food, Practical applications. Woodhead Publishing Limited, Cambridge.
- Rainer, L. & Heiss, C. J. (2004). Conjugated linoleic acid: health implications and effects on body composition (Review). J. Am. Diet. Assoc., 104, 963-968.
- Rao, C. V., Newmark, H. L. & Reddy, B. S. (1998). Chemopreventive effect of squalene on colon cancer. *Carcinogenesis*, 19, 287-290.
- Rizvi, S., Raza, S. T., Ahmed, F., Ahmad, A., Abbas, S. & Mahdi, F. (2014). The Role of Vitamin E in Human Health and Some Diseases. *Sultan Qaboos Univ. Med. J.*, 14, e157e165.
- Rover, M. R. & Brown, R. C., (2013). Quantification of total phenols in bio-oil using the Folin–Ciocalteu method. J. Anal. Appl. Pyrol., 104, 366-371. J ANAL APPL PYROL
- Sajilata, M. G., Singhal, R. S. & Kamat, M. Y. (2008). The Carotenoid Pigment Zeaxanthin -A Review. *Compr. Rev. Food Sci. F.*, 7, 29-49.
- Sales-Campos, H., Souza, P. R., Peghini, B. C., da Silva, J. S. & Cardoso, C. R. (2013). An overview of the modulatory effects of oleic acid in health and disease. *Mini Rev. Med. Chem.*, 13, 201-210.
- Sayed, A. K. & El Goud, A. A. (2010). Effect of Endomycorrhizal Fungi and Compost on the Yield and Quality of Maize and Sunflower Plants in Poor Nutrients Soil, Kassel University Press GmbH.
- Shahidi, F. & Naczk, M. (2003). Phenolics in food and nutraceuticals. CRC Press.
- Shahidi, F. (1997). *Natural Antioxidants: Chemistry, Health Effects, and Applications*, The American Oil Chemists Society.
- Sheppard, A. J. & Iverson, J. L. (1975). Esterification of fatty acids for gas-liquid chromatographic analysis. J. Chromatogr. Sci., 13, 448-452.
- Snyder, J. M. & King, J. W. (1994). Oilseed Volatile Analysis by Supercritical Fluid and Thermal Desorption Methods. J. Am. Oil Chem. Soc., 71, 261-265.
- Snyder, L. R., Kirkland, J. J. & Dolan, J. W. (2011). *Introduction to Modern Liquid Chromatography*, John Wiley & Sons.
- Sommerburg, O., Keunen, J., Bird, A. & van Kuijk F. J. G. M. (1998). Fruits and vegetables that are sources for lutein and zeaxanthin: the macular pigment in human eyes. *Br. J. Ophthalmol.*, 82, 907-910.
- Steenson, D. F., Lee, J. H. & Min, D. B. (2002). Solid Phase Microextraction of Volatile Soybean Oil and Corn Oil Compounds. J. Food Sci., 67, 71-76.
- Stratil, P., Kubáň, V. & Fojtová, J. (2008). Comparison of the Phenolic Content and Total Antioxidant Activity in Wines as Determined by Spectrophotometric Methods. *Czech J. Food Sci.*, 26, 242-253.
- Strecker, L. R., Bieber, M. A., Maza, A., Grossberger, T. & Doskoczynski, W. J. (1996). Corn oil. In Bailey's Industrial Oil and Fat Products, Edible Oil and Fat Products., John Wiley & Sons. New York.
- Strecker, L. R., Maza, A. & Winnie, F. G. (1990). Proceedings of the World Conference on Edible Fats and Oils Processing: Basic Principles and Modern Practices, Erickson D. R. (ed.), American Oil Chemists' Society, Champaign, IL, 309-325.
- Usuki, R., Suzuki, T., Endo, Y. & Kaneda, T. (1984). Residual amounts of chlorophylls and pheophytins in refined edible oils. J. Am. Oil Chem. Soc., 61, 785-788.
- Vetter, W., Schröder, M & Lehnert, K. (2012). Differentiation of refined and virgin edible oils by means of the trans- and cis-phytol isomer distribution. J. Agric. Food Chem., 60, 6103-6107.

Waraho, T. (2011). Effects of Free Fatty Acids, Mono- and Diacylglycerols on Oxidative Stability of Soybean Oil-In-Water Emulsions, 2011 Dissertations. Paper 376, http://scholarworks.umass.edu/cgi/

viewcontent.cgi?article=1379&context=open_access_dissertations

- Weirauch J. L. & Gardner, J. M. (1978). Sterol content of foods of plant origin. J. Am. Diet. Assoc., 73, 39-47.
- Williams, J. P., Khan, M. U. & Wan Lem, N. (1997). Physiology, Biochemistry and Molecular Biology of Plant Lipids. Springer Science & Business Media.
- Woodbury, S. P, Evershed, R. P. & Rossell, J. B. (1998). Purity assessments of major vegetable oils based on gamma ¹³C values of individual fatty acids. J. Am. Oil Chem. Soc., 75, 371-379.
- Woodbury, S. P., Evershed, R. P. & Rossell, J. B. (1998). Gamma ¹³C analysis of vegetable oil, fatty acid components, determined by gas chromatography combustion-isotope ratio mass spectrometry, after saponification or regiospecific hydrolysis. *J. Chromatogr. A* 805, 249-257.
- Worthington, R. E. & Hitchcock, H. L. (1984). A method for the separation of seed oil steryl esters and free sterols: Application to peanut and corn oils. J. Am. Oil Chem. Soci., 61, 1085-1088.

Chapter 2

BIOSYNTHESIS, ANALYSIS AND ANTIOXIDANT PROPERTIES OF TOCOLS AND CAROTENOIDS FROM CORN OIL

Vasile Robert Gradinaru^{1,*} and Luiza Madalina Gradinaru²

¹Department of Chemistry, "Alexandru Ioan Cuza" University, Iasi, Romania ²"Petru Poni" Institute of Macromolecular Chemistry, Iasi, Romania

ABSTRACT

Corn oil is an edible oil extracted from the maize germ and has multiple applications. This chapter focuses on biosynthesis, analysis and antitoxidant activity of two classes of compounds, tocols and carotenoids, which are present in this vegetal oil. The demand to obtain corn oil with high levels of antioxidants should be considered. Tremendous work has been carried out in order to increase the level of these antioxidants by switching or modifying biochemical pathways. However, new approaches should be taken into consideration in order to increase their concentration in corn oil. A whole range of analysis or estimation methods of antioxidant activity are presented herein. The data suggest that tocopherols, as well as carotenoids, from corn oil are important antioxidant molecules. Moreover, development of more specific detection assays to distinguish their individual contribution to the total antioxidant activity should be taken into account.

Keywords: corn oil, antioxidant, tocol, carotenoid, biosynthesis

INTRODUCTION

Corn oil, the second most produced oil in the United States, gains a well-deserved status in our nutrition. Nowadays its production is ranked in the top ten worldwide. Maize oil can be found in foodstuffs or used in chemical industry and less commonly in pharmaceutical industry. Corn oil is characterized by elevated levels of unsaturated fatty acids and

^{*} Corresponding author: E-mail: robert.gradinaru@uaic.ro.

Vasile Robert Gradinaru and Luiza Madalina Gradinaru

antioxidants. Particularly, tocols and carotenoids, biochemical compounds widely spread in plants, are valuable fat-soluble antioxidants for mammals (Ball, 2004; Kamal-Eldin & Appelqvist, 1996). In corn, these compounds are localized in the endosperm (tocopherols, tocotrienols, carotenoids), in the pericarp and germ (only tocopherols), and in the aleurone layer (carotenoids) (Balz & Schulte, 1992; Grams et al., 1970; Ndolo & Beta, 2013).

Corn oil, like soybean oil, presents a significant content (100-2000 mg/kg oil) of total tocopherol (Cerretani et al., 2010; Gliszczyńska-Świgłoet al., 2007; Grusak, 1999; Ndolo & Beta, 2013). However, the estimated tocopherol content in high-oil maize lines was higher than that in normal lines (Zhou et al., 2009). Furthermore, corn oil displays elevated levels of γ -tocopherol instead of its isomers (Eitenmiller & Lee, 2004; Lerma-Garcia, 2007; Shahidi, 2000). For this reason the use of corn oil induces an increase of γ -tocopherol in human serum (Lemcke-Norojärvi et al., 2001) and reduces cancer risk by minimizing DNA damage (Elmadfa & Park, 1999). γ -tocotrienol, an unsaturated variant of vitamin E, predominates in corn (Panfili et al., 2003). Conversely, moderate levels of α -tocotrienol were detected in maize germ oil (Speek et al., 2006).

By contrast, the concentration of β -carotene in corn oil is quite small (0.6 - 2 ppm) (Bootsma, 2012; E, Dauqan et al., 2011; Moreau et al., 2007). Among carotenoids, lutein and zeaxanthin are most abundant in corn kernels or corn products (de Oliveira & Rodriguez-Amaya, 2007; Gunstone, 2011) (more than 4 ppm). The analysis of six commercial corn hybrids reveals moderate levels of β -cryptoxanthin and β -carotene (1-3 ppm) (Kljak & Grbeša, 2015).

The reaction of homogentisic acid with phytyl-diphosphate (or geranyl geranyldiphosphate) constitutes an important metabolic step for the biosynthesis of tocopherols (or tocotrienols). Conversely, autocondensation of geranyl geranyl-diphosphate leads to important precursors required for carotenoids biosynthesis. All these metabolic steps are assisted by specific enzymes. The attempts to modify or manipulate genes encoding these enzymes have revealed that these metabolic pathways can be targeted to achieving the desired metabolic product. The elevated levels of antioxidants could bring multiple benefits to consumers' life quality.

Initially, simple methods such as ultraviolet–visible (UV-Vis), fluorescence or infrared (IR) spectroscopy were proposed in order to estimate the total amount of antioxidants in various food samples. However, coupled methods such as gas chromatography-mass spectrometry (GC-MS), high performance liquid chromatography with ultraviolet-visible (HPLC-UV-Vis) or nuclear magnetic resonance detector (HPLC-NMR) and liquid chromatography-mass spectrometry (LC-MS) are employed more to quantify individual tocols or carotens from various corn-based food samples. In this chapter all these methods of analysis will be briefly described.

The last part of this chapter is focused on the antioxidant activity of tocols and carotenoids. Their antioxidant potency seems to be correlated with such factors as:

- chemical composition of oil;
- antioxidant conformation whithin membrane;
- physical parameters (temperature, intensity of light);
- type of system used (a heterogeneous system is prefered).



Figure 1. Tocopherol biosynthesis in plants. Enzyme abbreviation: E1-homogentisate phytyltransferase; E2-2-methyl-6-phytylplastoquinol methyltransferase; E3-tocopherol cyclase; E4- γ -tocopherol methyltransferase. γ -tocopherol (γ -T) is the major component (>70%) found in corn seeds oil.

Various methods for determining the antioxidant capacity are referred here. These assays are based on antioxidant capacity to scavange various radicals, reduce certain ions or compete with other compounds. However, this activity can be indirectely measured by monitorizing various compouds derived from radical-induced processes.

BIOSYNTHESIS

Biosynthetic Pathway of Tocols

In plants, the biosynthesis of tocols head group begins in the cytosol with the homogentisic acid (1), a product resulted from the shikimic acid pathway. Seven main steps are distinguished in the homogentisic acid biosynthesis. Shikimate (A) biosynthesis is based on intermediates from two pathways (glycolysis and pentose phosphate). A key intermediate, the chorismate (B), can be further used to obtain aromatic amino acids, vitamins and antibiotics. Moreover, during the homogentisic acid biosynthesis, tyrosine and p-hydroxyphenyl pyruvate were identified (Garcia et al., 1997). Conversely, the hydrophobic tail (phytyl) biosynthesis occurs in plastids (Ischebeck et al., 2006). Polyprenyltransferase (E1) catalyzes the coupling reaction of the homogentisic acid with the phytyl-diphosphate (phytyl-DP). The resulted compound, 2-methyl-6-phytylbenzoquinol (2) is and intermediate in δ - and β -tocopherol biosynthesis (Soll & Schultz, 1980). Similarly, the same enzyme (E1)

assists the initial reaction of homogentisic acid with geranyl geranyl-diphosphate, an essential step that precedes tocotrienols formation. On the other hand, 2-methyl-6-phytylbenzoquinol (2) can be further methylated. A methyltransferase (E2) assists its convertion to another intermediate, 2,3-dimethyl-6-phytylbenzoquinol (5) (Shintani et al., 2002). These two intermediates (2 and 5) are required during the cyclisation step catalyzed by tocopherolcyclase (E3). In parallel, two chromane derivatives are formed, namely δ - and γ -tocopherol (δ -T and γ -T) (Hunter & Cahoon, 2007). The last-mentioned compounds can be readily methylated using a γ -T-methyltransferase (E4) (Shintani & DellaPenna, 1998). Thus, two final tocols, β - and α -tocopherol (β -T and α -T) may also appear in this pathway.

Regarding tocotrienols biosynthesis (Figure 2), at first, geranyl geranyl transferase (E1') catalyzes the reaction of geranyl geranyl-diphosphate with homogentisic acid. The condensation product, 2-methyl-6-geranylgeranyl-1,4-benzoquinone (MGGBQ) is converted using an identical set of enzymes (E2, E3 and E4) to δ - and γ -tocotrienols (δ -T3 and γ -T3) respectively β - and α -tocotrienols (β -T3 and α -T3) (Abbasi, 2007; Cahoon et al., 2003; Herbers, 2003; Porfirova et al., 2002; Soll & Schultz, 1979).

There are several factors that influence tocols biosynthesis. The tocopherol level could be correlated with precursor (phytol, Phytyl-PPor, GGPP, HGA) concentration (Soll et al., 1980; Whistance & Threlfall, 1970). Many target genes encode enzymes such as γ -methyltransferase (d'Harlingue & Camara, 1985; Shintani & DellaPenna, 1998), GG reductase (Keller et al., 1998), GGPP synthase, HRP dioxygease (Falk et al., 2003; Tsegaye et al., 2002), prenyl transferase (Dada et al., 1968), HGGT-homogentisic acid transferase (Cahoon et al., 2003), cyclase (Porfirova et al., 2002), phytyltransferase (Schledz et al., 2001), homogentisate prenyltransferases (Dörmann, 2003; Sadre et al., 2006), phytyl kinase (DellaPenna & Last, 2006; Valentin et al., 2006). Among these enzymes, γ -tocopherol methyltransferase was overexpressed in *Arabidopsis sp.* inducing a higher α -T level in plant oils.



Figure 2. Tocopherol biosynthesis in plants. Enzyme abbreviation: E1'- geranyl geranyl transferase; E2-2-methyl-6-phytylplastoquinol methyltransferase; E3-tocopherol cyclase; E4-γ-tocopherol methyltransferase (Adapted from Kamal-Eldin et al., 1997).

Instead, tocochromanol levels are impaired in sucrose export defective 1 maize mutant (Provencher et al., 2001). Conversely, regulation and function of plants tocochromanols were demonstrated by simultaneous transgenic expression and genetic engineering (Hoa et al., 2003; Karunanandaa et al., 2005; Mène-Saffrané & DellaPenna, 2010). ROS (Reactive oxygen species) and phytohormones (salicylic, abscisic and jasmonic acids) (Falk et al., 2002; Sandorf & Holländer-Czytko, 2002) also regulate tocopherol biosynthesis. Abscisic acid was found to control genes expression of p-hydroxyphenylpyruvate dioxygenase respectively, the tocopherol amount (Munné-Bosch, 2007). Tocopherol level increases under stress conditions and can be correlated with jasmonic acid (Sandorf & Holländer-Czytko, 2002) or salyciclic acid concentration (Munné-Bosch & Falk, 2004) or lack of water (Ali et al., 2010). In this respect, the tocopherol is a signal molecule within ROS-antioxidant network (Shao et al., 2008).

Biosynthesis of Carotenoids

Tocols and carotenoids biosynthesis pathways have a key intermediate, geranyl geranylpyrophosphate (GGPP). The second pathway starts with the condensation of two GGPP molecules and is assisted by phytoene synthase (E1") (Shumskaya et al., 2012). This is a rate limiting step which dictates the carotenoid level in maize tissues. In the first instance, a long hydrocarbonated compound, 15-cis-phytoene (P1) possessing 9 double (C=C) bonds, is formed. Coupled desaturation-isomerization steps are necessary to generate all-trans lycopene. In the first sequence, 9,15-cis-phytoene and 9,15,9'-tricis-ζ-carotene are formed. These reactions are catalyzed by phytoene desaturase (E2") (Li et al., 1996). The last intermediate is further isomerized to 9,9'-di-cis- ζ -carotene (C1) by ζ -carotene isomerase (E3") (Chen et al., 2010). In the next step, ζ-carotene desaturase (E4") assists sequential conversion C1 to all-trans-lycopene (C2) via 7,9,7',9'-tetra-cis-neurosporene (Matthews et al., 2003). Lycopene cyclization constitutes a branching point in maize carotenoid biosynthesis. This reaction is catalyzed by lycopene cyclases (ε and β or E5" and E6") (Cunningham et al., 1996). The resulted compounds, δ - and γ -carotene (C3 and C4) possess an ionone moiety at the edge of the molecule. A similar strategy is used for α - and β -carotene (C5 and C6) biosynthesis. At this stage lycopene β -cyclase (E6") introduces the second ionone ring. Furthermore, mono and dihydroxylated carotens are resulted by grafting of one or two hydroxyl groups on ionone ring. Thus, lutein is obtained from α -carotene via a monohydroxylated intermediate, α -cryptoxanthin or zeinoxanthin. An enzyme, ε -carotene hydroxylase (E7"), assists the previous biochemical reaction (Tian et al., 2004). Analogously, zeaxanthin results from β -carotene via a mono-hydroxylated intermediate, β -cryptoxanthin (Figure 3).

Approaches in plants engineering to increase carotenoids concentrations have been reviewed (Hirschberg, 1999). Overexpression of 1-deoxy-5-D-xylulose-5-phosphate synthase (DXS) and phytoene synthase (PSY) genes was correlated with elevated levels of carotenoids (Aluru et al., 2008; Burkhardt et al., 1997; Ducreux et al., 2005; Lindgrenet et al., 2003; Shewmaker et al., 1999). Combinatorial nuclear transformation was successfully used to generate maize transgenic plants with high carotenoids levels (Zhu et al., 2008).



Figure 3. Carotenoid biosynthesis in plants. Enzyme abbreviation: E1"- phytoene synthase; E2"phytoene desaturase; E3"- ζ -carotene isomerase; E4"- ζ -carotene desaturase; E5"- lycopene ε -cyclase; E6"- lycopene β -cyclase; E7"- ε -carotene hydroxylase; E8"- β -ring hydroxylase; E9"- β -carotene hydroxylase; E10"- ε -ring hydroxylase; E11"- β -carotene ketolase.

ANALYSIS METHODS OF TOCOLS AND CAROTENOIDS

Analysis of Tocols

Separation methods are most widely used for tocopherols and tocotrienols analysis from vegetable oils. Thin-layer chromatography (TLC) technique was earlier used to separate and estimate tocols and carotenoids (Jáky, 1967; Mishra & Singh, 2010) from vegetal samples. Two-dimensional TLC was successfully used for tocopherol and tocotrienol separation in several vegetable oils (Whittle & Pennock, 1967). The separated compounds could be developed using Emmerie-Engel reaction (Emmerie & Engel, 2010).

Derivative UV-Vis spectroscopy allows the estimation of individual or total tocopherol levels from their mixtures (Bukovits & Alberto, 1987). The UV spectra of different oils dissolved in n-hexan reveal characteristic signals of tocols (Figure 4). The concentration of tocopherol decreases when corn or soybean oil is heated to 50°C (Gonçalves et al., 2014). Tocopherols concentration was also estimated by chemiluminescence (Belyakov et al., 2004).

Free ebooks ==> www.Ebook777.com



Figure 4. UV spectra of corn oil (1), linseed oil (2); sunflower oil (3) and olive oil (4) in n-hexane. 100 mg oil was dissolved in 500 μ l n-hexan; the samples were diluted 1:20 in n-hexane with one exception - linseed oil sample was diluted 1:50. The spectrum of α -tocopheryl acetate (T; Biopharm pill; dilution of 1:8000) was recorded in hexane.

Isomers of tocopherols in edible oils were quantified using electrochemical methods: polarography (Smith et al., 1941), and differential pulse voltammetry (Galeano et al., 2004; Robledo et al., 2013). A couple of spectrophotometric assays were used to quantify tocopherols: copper(II)-neocuproin system (Tütem et al., 1997). GC-MS analysis reveals the instability at heating of γ -tocopherol from corn oil. This isoform is converted to α -tocopherol and later degraded when temperature is increased (Sim et al., 2014).

Quantification of tocopherols in edible oils was successfully achieved by HPLC with UV or fluorescent detection forty years ago (Abe et al., 1975). The typical UV absorption band of tocopherol has a characteristic peak at around 280-290 nm. This band attributed to vitamin E variant(s) is distinguished in various oils (Figure 4). The emission spectra of various oils are also illustrated in Figure 5. Thus, the corn oil is distinguished by higher emission intensity.

A lot of methods for tocols analysis by HPLC were reported in the literature (Bele et al., 2013; Cerretani et al., 2010; Ergönül & Köseo, 2013; Gimeno et al., 2000; Górnaś, 2015; Knecht et al., 2015; Ortíz et al., 2006; San Andrés et al., 2011; Taylor & Barnes, 1981; Thompson & Hatina, 1979). Tocopherol analysis by gas liquid chromatography was also reported (De Greyt et al., 1998). Tocopherols from rape oilseed were quantified by gas chromatography with flame ionization detector (Hussain et al., 2013). However, the sample preparation or detection procedures may substantially differ. Oils saponification was commonly used before HPLC separation (Ryynänen et al., 2004; Shammugasamy et al., 2013). Simultaneous analysis of both carotenoids and tocopherols was carried out by solid-liquid extraction followed by HPLC (Valdivielso et al., 2015). Similarly, RP-HPLC was used for simultaneous determination of α -tocopherol and β -carotene (Gimeno et al., 2000). Some analysts, including tochopherols, were analyzed by liquid chromatography coupled to mass spectrometry (Zarrouk et al., 2009).



Figure 5. Emission spectra of corn (1), sunflower (2); linseed (3) and olive (4) oil in n-hexane (the excitation wavelength, $\lambda ex = 295$ nm). 100 mg oil was dissolved in 500 µl n-hexan; the samples were diluted 1:50.

Non-aqueous capillary electrophoresis with fluorimetric detection was also successfully used for the determination of tocopherols in vegetable oils (Galeano-Díaz et al., 2012). Infrared (IR) spectroscopy was also employed for tocols quantification (Ahmed et al., 2005; Zahir et al., 2014). In order to complete this overview, various methods for quantitative analysis of tocopherols were compared (Koswig & Mörsel, 1990).

Analysis of Carotenoids

The relative content of carotenoids in maize samples was estimated by TLC after extraction-saponification-extraction steps (Mishra & Singh, 2010). TLC separation of ethanolic extracts components followed by UV detection or iodine staining was used to estimate carotenoids content in plants (Natividad & Rafael, 2014). Recently, carotenoids concentration was determined by densitometry after their separation on high performance TLC (Rodić et al., 2012)

Total carotenoid content of corn oil could be assessed at 450 nm (Wolf, 1968). In order to strengthen this information several visible absorption spectra of some edible oils are displayed in Figure 6. Near-infrared reflectance spectroscopy (NIRS) technique was used to estimate carotenoids content in maize (Brenna & Berardo, 2004).

Various organic solvent mixtures have been used to extract carotenoids from maize. A recent study reveals that these compounds are easily extracted using a methanol (MeOH)-less polar solvent mixture. Acetone-water (5:1) was also used for initial extraction of carotenes from ground maize. The filtrate was further used and mixed with petroleum ether. The

concentrated sample was analyzed on a C30 YMC column (Howe & Tanumihardjo, 2006). A similar approach was used to extract carotenoids from cereal grain products using 75% acetone and hexane. The separation was performed on a reverse-phase HPLC column (C18 Vvdac) (Luterotti et al., 2013). Corn xanthophylls were easily extracted using an 80% ethanolic alkaline solution and further separated by HPLC (C30 column). The elution profile was monitored at 450 and 445 nm (Moroset al., 2002). Rehydration of dry maize was recommended prior to MeOH:THF (1:1) extraction that precedes carotenoids separation on C18 or C30 column (Kimura et al., 2007). A hexane:ethyl acetate (9:1) extraction mixture was also used after saponification and the resulted carotenoids mixture was analyzed by HPLC on a Spherisorb ODS2 column (Scott & Eldridge, 2005). Boiling of maize samples (at 100 °C for 30 minutes) proved to be a better choice to increase the concentration of extracted carotenoids (Muzhingi et al., 2008). Solid phase extraction (on de-activated alumina column) was successfully used for non- and monohydroxylated carotenes separation from maize kernel prior to a RP-HPLC separation (Hulshof et al., 2007). A suitable extraction mixture (MeOH:ethyl acetate or MeOH:THF 1:1) has turned out to be a better choice prior to carotenoids analysis by UHPLC-Vis (Rivera & Canela, 2012). More than 11 carotenoids were resolved by UHPLC-UV (Ultra high pressure liquid chromatographic system coupled with a UV detector) (Maurer et al., 2014). However, beside high levels of lutein and zeaxanthin, β apo-8'-carotenal was also detected.



Figure 6. Spectra of n-hexane extracts of: corn (1), olive (2); linseed (3) and sunflower (4) oils (1,2 and 3-100 mg oil / 900 μ l n-hexan; 4-250 mg oil / 750 μ l n-hexan). Inset: A-corn oil; B-sunflower oil (cold pressed); C-refined sunflower oil; D-linseed oil; E-olive oil.



Figure 7. Reaction of γ -tocopherol with lipid peroxyl radicals (LOO•).

Since the carotenoids display similar absorption spectra it is recommended to identify these compounds after HPLC separation using a mass spectrometer or a NMR detector. Structural and geometrical isomers or epoxides were distinguished by this technique (Aman et al., 2005; Matsubara et al., 2012). An accelerated solvent extraction step in hexane-based solvent mixtures was successfully used prior to LC-ESI MS analysis of lutein and β -carotene from orange carrot (Saha et al., 2015).

ANTIOXIDANT PROPERTIES OF TOCOLS AND CAROTENOIDS

General Aspects

Corn oil contains natural antioxidants such as tocols, carotenoids and polyphenols. These compounds prevent protein oxidation, lipid peroxidation (Figure 7) and reactive oxygen species formation.

In this section, we focus on the antioxidant activity of lipophilic compounds such as tocols and carotenoids.

During the refining process the amount of antioxidants is reduced substantially. However, tocopherols are antioxidants which have a much higher stability compared with other antioxidants which are components of refined oils. The tocols manifest their antioxidant activity via their chromanol phenolic moiety that easily transfers hydrogen to a lipid-free radical (Kamal-Eldin & Appelqvist, 1996). The compounds obtained during these reactions could be reduced back to tocopherol by glutathione and vitamin C.

Earlier studies failed to establish correlations between the content of vitamin E and the formation of peroxide in corn and soybean oil (Chow & Draper, 1974). Additionally, it should be noted that the antioxidant capacity of α -tocopherol is greater than that of β -carotene in nonpolar environment (Tsuchihashi et al., 1995).

While the antioxidant-prooxidant switch seems to be dictated by α -tocopherol concentration (Huang et al., 2005), γ -tocopherol constitutes the main determinant of the total antioxidant capacity and this parameter is affected by other chemical components of the oil (Castelo-Branco & Torres, 2012). A recent study reveals that supplementation of four different vegetable oils with β and γ -tocopherol induces a slight increase of corn oil antioxidant activity (Hamdo et al., 2014).

Tocotrienols are distinguished by a higher antioxidant level than tocopherols. For example, α -tocotrienol is distinguished by superior antioxidant potency to that of α -tocopherol, having a pronounced disordering effect on membrane bilayer (Serbinova et al., 1991; Suzuki et al., 1993).

Electrochemical and Electron paramagnetic resonance (EPR) studies have been focused on neutral carotenoid radicals. These species are derived from carotenoid radical cations deprotonation (Focsan et al., 2015). At lower radical concentration this behaves like a scavenger for reactive species (single oxygen, hydroxyl or peroxyl radicals) (Foot, 1976). Although the antioxidant activity of carotenoids is higher than that of α -tocopherol, a modest contribution (less than 5%, due to modest concentration levels of carotenoids in oil) to the total activity is expected (Müller et al., 2011).

Compared to other tocopherols, α -tocopherol manifests a higher quenching ability for molecular oxygen, radical scavenging reactivity and biological potency (Fryer, 1992; Zingg & Azzi, 2004). Conversely, γ -tocopherol has a greater ability to scavenge nitrogen dioxide and peroxynitrite than the aforementioned isomer (Christen et al., 1997). It should be stressed here that the elevated levels of reactive oxygen species may be linked to several pathological states (Dreher et al., 1999; Jurkiewicz et al., 1995; Rahimi et al., 2005). Therefore, vitamin E was used as prophylactic and therapeutic agent for cancer, cardiovascular, chronic or autoimmune diseases (Borek, 2004; Hercberg et al., 1998; Ratnam et al., 2006; Suantawee et al., 2013). Moreover, γ -tocopherol specifically inhibits cyclooxygenase activity in macrophages and epithelial cells (Jiang et al., 2000).

The antioxidant activity is affected by the antioxidant structure (Lien et al., 1999), temperature (Réblová, 2006), light, relative humidity (Kimet al., 2015), oil physical state and presence of other compounds. Anyway, lipophilic antioxidants such as α -tocopherol were more effective in a heterogeneous system (Frankel et al., 1994; Losada-Barreiro et al., 2013a; Schwarz et al., 2000). In corn oil, tocopherol emulsion is localized toward oil-interfacial regions and for this reason it manifests a good efficiency in the inhibition of lipid oxidation (Losada-Barreiro et al., 2013b).

Measuring Antioxidant Activity of Oils

Trolox equivalent antioxidant capacity (TEAC) spectrophotometric assay was used to estimate antioxidant capacity (Miller et al., 1993; Re et al., 1999) of edible vegetable oils. This method relies on the capacity of antioxidants to scavenge a non-biological radical, 2,2'-azinobis(3-ethylbenzothiazoline-6-sulphonate) (ABTS) (Figure 8).

Free ebooks ==> www.Ebook777.com

Vasile Robert Gradinaru and Luiza Madalina Gradinaru



Figure 8. Chemical reaction steps that lie behind tocols antioxidant capacity. In the first instance a nonbiological radical is generated. In the second reaction this radical species were counteracted by –OH phenolic moiety of tocols.

Two ferric-ion spectrophotometric methods were used to estimate the antioxidant capacity of the methanolic extract of corn oil (Szydłowska-Czerniak et al., 2008). One of these spectroscopic methods is exemplified in Figure 9.

A spectrophotometric method based on reduction of Mo(VI) to Mo(V) has been developed for the quantitative determination of the total antioxidant capacity of hexane extracts of corn and soybean samples (Prieto et al., 1999). The antioxidant activity of methanolic corncobs extracts was measured based on their ability to scavenge the DPPH (2,2-diphenyl-1-picrylhydrazyl) radical (Sultana et al., 2007). This assay is based on radical disappearance according with sample antioxidant capacity (Figure 10).

Corn oil displays a modest DPPH radical scavenging activity when compared with olive or Perilla oils (Lee et al., 2015). However, the antioxidant activity of phenols from edible oils was estimated after a column extraction step (Siger et al., 2008).

Oxygen radical absorbance capacity (ORAC) and total trapping antioxidant parameter (TRAP) are two assays based on competition between an antioxidant and a substrate for peroxyl radicals (Cao & Prior, 1998; Cabrini et al., 2001).

Malondialdehyde (MDA) is a secondary carbonyl compound that arises from lipid peroxidation pathway. Gas chromatographic analysis of pentafluorophenyl-hydrazine derivatized carbonyl compounds was successfully used to assess antioxidant activity of vegetable oils (Stashenko et al., 1997).



Figure 9. Reduction of ferric complexes by antioxidant (AOH) species.



Figure 10. Capacity of antioxidants (AOH) to scavenge an organic radical.

CONCLUSION

In this chapter, a literature overview on corn oil antioxidants was performed. Corn oil displays a significant amount of tocols and low levels of carotenoids. Their biosynthesis routes have been well-understood. More attention should be paid to the metabolic engineering of antioxidants in corn. A significant increase in the concentration of one of these antioxidants will result in a higher nutritional value of corn oil. Such oils are rich in antioxidants and therefore are recommended as adjuvants in various therapies. An overview of the methods of tocols or carotenoid analysis has been one of main goals of this chapter. There were presented both simple and sophisticated methods of analysis and detection of these compounds.

The antioxidants have different structural features and hence they manifest distinct antioxidant properties. Most of the antioxidant assays presented are rather complicated and cannot distinguish the contribution of each antioxidant to the whole corn oil antioxidant power. Future studies should aim to design new more efficient methods to quantify their specific antioxidant capacity. This will allow us to create an overall picture of the way antioxidants trapped in this complex corn oil matrix action.

ACKNOWLEDGMENTS

This work was supported by a grant from the Romanian National University Research Council (RP-7/2007).

REFERENCES

- Abbasi, A-R. (2007). Functional analysis of tocopherol biosynthesis in plants. Erlangen-Nürnberg.
- Abe, K., Yuguchi, Y. & Katsui, G. (1975). Quantitative determination of tocopherols by highspeed liquid chromatography. J. Nutr. Sci. Vitaminol., 21, 183–188.
- Ahmed, M. K., Daun, J. K. & Przybylski, R. (2005). FT-IR based methodology for quantitation of total tocopherols, tocotrienols and plastochromanol-8 in vegetable oils. J. Food Compos. Anal., 18, 359–364.
- Ali, Q., Ashraf, M. & Anwar, F. (2010). Seed composition and seed oil antioxidant activity of maize under water stress. J. Am. Oil Chem., Soc. 87, 1179–1187.
- Aluru, M., Xu, Y., Guo, R., Wang, Z., Li, S., White, Wang, K. & Rodermel, S. (2008). Generation of transgenic maize with enhanced provitamin A content. J. Exp. Bot., 59, 3551–3562.
- Aman, R., Biehl, J., Carle, R., Conrad, J., Beifuss, U. & Schieber, A. (2005). Application of HPLC coupled with DAD, APcI-MS and NMR to the analysis of lutein and zeaxanthin stereoisomers in thermally processed vegetables. *Food Chem.*, 92, 753–763.
- Ball, G. F. M. (2004). *Vitamins: their role in the Human Body*. Blackwell Publishing Ltd, Oxford, UK.
- Balz, V. M. & Schulte, E. H. (1992). Tkennung von Tocopherolen und Tocotrienolen durch HPLC, Eur. J. Lipid Sci. Technol., 209–213.
- Bele, C., Matea, C. T., Raducu, C., Miresan, V. & Negrea, O. (2013). Tocopherol Content in Vegetable Oils Using a Rapid HPLC Fluorescence Detection Method. *Not. Bot. Horti. Agrobo.*, 41(1), 93–96.
- Bootsma, J. (2012). Oil composition and method for producing the same. Google Patents. Retrieved from https://www.google.co.in/patents/ WO2012033843A1?cl=en
- Borek, C. (2004). Dietary antioxidants and human cancer. Integr. Cancer Ther., 3, 333-341
- Brenna, O. V. & Berardo, N. (2004). Application of near-infrared reflectance spectroscopy (NIRS) to the evaluation of carotenoids content in maize. J. Agric. Food. Chem., 52, 5577–5582.
- Bukovits G. J. & Alberto, L. (1987). Determination of Individual Tocopherols by Derivative Spectrophotometry. J. Am. Oil Chem. Soc. 64(4), 517-520.
- Burkhardt, P. K., Beyer, P., Wünn, J., Klöti, A., Armstrong, G. A., Schledz, M., Von Lintig, J. & Potrykus, I. (1997). Transgenic rice (Oryza sativa) endosperm expressing daffodil

(Narcissus pseudonarcissus) phytoene synthase accumulates phytoene, a key intermediate of provitamin A biosynthesis. *Plant J.*, *11*, 1071–1078.

- Cabrini, L., Barzanti, V., Cipollone, M., Fiorentini, D., Grossi, G., Tolomelli, B., Zambonin, L. & Landi, L. (2001). Antioxidants and total peroxyl radical-trapping ability of olive and seed oils. J. Agric. Food. Chem., 49, 6026–6032.
- Cahoon, E. B., Hall, S. E., Ripp, K. G., Ganzke, T. S., Hitz, W. D. & Coughlan, S. J. (2003). Metabolic redesign of vitamin E biosynthesis in plants for tocotrienol production and increased antioxidant content. *Nat. Biotechnol.*, 21, 1082–1087.
- Cao, G. & Prior, R. L. (1998). Measurement of oxygen radical absorbance capacity in biological samples. *Methods Enzymol.*, 299, 50–62.
- Castelo-Branco, V. N. & Torres, A. G. (2012). Generalized linear model describes determinants of total antioxidant capacity of refined vegetable oils. *Eur. J. Lipid Sci. Technol.*, 114, 332–342.
- Cerretani, L., Lerma-García, M. J., Herrero-Martínez, J. M., Gallina-Toschi, T. & Simó-Alfonso, E. F. (2010). Determination of tocopherols and tocotrienols in vegetable oils by nanoliquid chromatography with ultraviolet-visible detection using a silica monolithic column. J. Agric. Food. Chem., 58, 757–761.
- Chen, Y., Li, F. & Wurtzel, E. T. (2010). Isolation and characterization of the Z-ISO gene encoding a missing component of carotenoid biosynthesis in plants. *Plant Physiol.*, 153(1), 66–79.
- Christen, S., Woodall, A. A., Shigenaga, M. K., Southwell-Keely, P. T., Duncan, M. W. & Ames, B. N. (1997). Gamma-tocopherol traps mutagenic electrophiles such as NO(X) and complements alpha-tocopherol: physiological implications. *Proc. Natl. Acad. Sci.*, USA 94, 3217–3222.
- Chow, C. K. & Draper, H. H. (1974). Oxidative stability and antioxidant activity of the tocopherols in corn and soybean oils. *Int. J. Vitamin. Nutr. Res.*, 44, 396–403.
- Cunningham, F. X., Pogson, B., Sun, Z., McDonald, K. A., DellaPenna, D. & Gantt, E. (1996). Functional analysis of the beta and epsilon lycopene cyclase enzymes of Arabidopsis reveals a mechanism for control of cyclic carotenoid formation. *Plant Cell*, 8(9), 1613–1626.
- d'Harlingue, A. & Camara, B. (1985). Plastid enzymes of terpenoid biosynthesis. J. Biol. Chem., 260, 15200–15203.
- De Greyt, W. F., Petrauskaite, V., Kellens, M. J. & Huyghebaert, A. D. (1998). Analysis of tocopherols by gas-liquid and high-performance liquid chromatography: a comparative study. *Eur. J. Lipid Sci. Technol.*, 100, 503–507.
- Dada, O. A., Threlfal, D. R. & Winstance, G. R. (1968). Stereospecific Biosynthesis of the Polyprenyl Side Chains of Terpenoid Quinones and Chromanols in Maize Shoots. *Eur. J. Biochem.*, 4(3), 329–333.
- Dauqan, E., Sani, H. A., Abdullah, H., Muhamad, H. & Top, A. G. M. (2011). Vitamin E and Beta Carotene Composition in Four Different Vegetable Oils. A. J. A. S., 8, 407–412.
- De Oliveira, G. P. & Rodriguez-Amaya, D. B. (2007). Processed and prepared corn products as sources of lutein and zeaxanthin: compositional variation in the food chain. *J. Food Sci.*, 72 (1), S79-S85.
- DellaPenna, D. & Last, R. L. (2006). Progress in the dissection and manipulation of plant vitamin E biosynthesis. *Physiol. Plant.*, 126, 356–368.

- Dörmann, P. (2003). Corn with enhanced antioxidant potential. *Nature Biotechnol.*, 21(9), 1015–1016.
- Dreher, F., Denig, N., Gabard, B., Schwindt, D. A. & Maibach, H. I. (1999). Effect of topical antioxidants on UV-induced erythema formation when administered after exposure. *Dermatology*, 198, 52–55.
- Ducreux, L. J. M., Morris, W. L., Hedley, P. E., Shepherd, T., Davies, H. V., Millam, S. & Taylor, M. A. (2005). Metabolic engineering of high carotenoid potato tubers containing enhanced levels of β-carotene and lutein. *J. Exp. Bot.*, *56*, 81–89.
- Eitenmiller, R. R. & Lee, J. (2004). Vitamin E: Food Chemistry, Composition, and Analysis, CRC Press, New York, USA.
- Elmadfa, I. & Park, E. (1999). Impact of diets with corn oil or olive/sunflower oils on DNA damage in healthy young men. *Eur. J. Nutr.*, *38*(6), 286–292.
- Emmerie, A. & Engel, C. (2010). Colorimetric determination of α-tocopherol (vitamin E). Recl. Trav. *Chim. Pays-Bas.*, *57*, 1351–1355.
- Ergönül, P. G. & Köseo, O. (2014). Changes in α -, β -, γ and δ -tocopherol contents of mostly consumed vegetable oils during refining process. *CyTA J. Food*, *12*(2), 199-202.
- Falk, J., Andersen, G., Kernebeck, B. & Krupinska, K. (2003). Constitutive overexpression of barley 4-hydroxyphenylpyruvate dioxygenase in tobacco results in elevation of the vitamin E content in seeds but not in leaves. *FEBS Lett.*, 540, 35–40.
- Falk, J., Krauß, N., Dähnhardt, D. & Krupinska, K. (2002). The senescence associated gene of barley encoding 4-hydroxyphenylpyruvate dioxygenase is expressed during oxidative stress. J. Plant Phys., 159, 1245–1253.
- Foot, C. S. (1976). *Photosensitized oxidation and singlet oxygen*: consequences in biological systems, Prior, W. A., Ed., Academic Press, NY
- Frankel, E. N., Huang, S., Kanner, J. & German, J. B. (1994). Interfacial Phenomena in the Evaluation of Antioxidants: Bulk Oils vs Emulsions. J. Agr. Food Chem., 42, 1054–1059.
- Fryer, M. J. (1992). The antioxidant effects of thylakoid Vitamin E (alpha-tocopherol). *Plant Cell Environ.*, 15, 381–392.
- Diaz G. T., Merás I. D., Cabanillas, A. G. & Alexandre F. M. F. (2004). Voltammetric behavior and determination of tocopherols with partial least squares calibration: Analysis in vegetable oil samples. *Anal. Chim. Acta*, 511, 231–238.
- Galeano-Díaz, T., Acedo-Valenzuela, M. I. & Silva-Rodríguez, A. (2012). Determination of tocopherols in vegetable oil samples by non-aqueous capillary electrophoresis (NACE) with fluorimetric detection. J. Food Comp. Anal., 25, 24–30.
- Garcia, I., Rodgers, M., Lenne, C., Rolland, A., Sailland, A. & Matringe, M. (1997). Subcellular localization and purification of a p-hydroxyphenylpyruvate dioxygenase from cultured carrot cells and characterization of the corresponding cDNA. *Biochem. J.*, 325 (3), 761–769.
- Gimeno, E., Castellote, A. I., Lamuela-Raventós, R. M., De La Torre, M. C. & López-Sabater, M. C. (2000). Rapid determination of vitamin E in vegetable oils by reversedphase high-performance liquid chromatography. J. Chromatog., A 881, 251–254.
- Gliszczyńska-świgło, A., Sikorska, E., Khmelinskii, I. & Sikorski, M. (2007). Tocopherol content in edible plant oils. *Pol. J. Food Nutr. Sci.*, 57(4), 157–161.
- Gonçalves, R. P., Março, P. H. & Valderrama, P. (2014). Thermal edible oil evaluation by UV-Vis spectroscopy and chemometrics. *Food Chem.*, 163, 83–86.

- Górnaś, P. (2015). Unique variability of tocopherol composition in various seed oils recovered from by-products of apple industry: Rapid and simple determination of all four homologues (α , β , γ and δ) by RP-HPLC/FLD. *Food Chem.*, *172*, 129–134.
- Grams, G. W., Blessin, C. W. & Inglett, G. E. (1970). Distribution of tocopherols within the corn kernel. J. Am. Oil Chem. Soc., 47, 337–339.
- Grusak, M. A. (1999). Genomics-assisted plant improvement to benefit human nutrition and health. *Trends Plant Sci.*, *4*, 164–166.
- Gunstone, F. D. (2011). Vegetable Oils in Food Technology Composition, Properties and Uses, Blackwell Publishing Ltd, Iowa, USA.
- Hamdo, H. H., Al-Assaf, Z. & Khayata, W. (2014). Determination of the antioxidant activity (AOA) by studying the changes in content of fames in some vegetable oils. J. Curr. Chem. Phar. Sc., 4(3), 110–117.
- Herbers, K. (2003). Vitamin production in transgenic plants. J. Plant Physiol., 160, 821-829.
- Hercberg, S., Galan, P., Preziosi, P., Alfarez, M. J. & Vazquez, C. (1998). The potential role of antioxidant vitamins in preventing cardiovascular diseases and cancers. *Nutrition*, 14, 513-520.
- Hirschberg, J. (1999). Production of high-value compounds: Carotenoids and vitamin E. *Curr. Opin. Biotechnol.*, 10, 86-191.
- Hoa, T. T. C., Potrykus, I. & Beyer, P. (2003). Increase the level of γ -oryzanol, tocopherol and tocotrienols in rice by isoprenoid-pathway engineering. *Omorice*, *11*, 28–34.
- Howe, J. A. & Tanumihardjo, S. A. (2006). Evaluation of analytical methods for carotenoid extraction from biofortified maize (Zea mays sp.). J. Agr. Food Chem., 54, 7992–7997.
- Huang, D., Boxin, O. U. & Prior, R. L. (2005). The chemistry behind antioxidant capacity assays. J. Agr. Food Chem., 53, 1841-1856.
- Hulshof, P. J. M., Kosmeijer-Schuil, T., West, C. E. & Hollman, P. C. H. (2007). Quick screening of maize kernels for provitamin A content. J. Food Comp. Anal., 20, 655–661.
- Hunter, S. C. & Cahoon, E. B. (2007). Enhancing vitamin E in oilseeds: Unraveling tocopherol and tocotrienol biosynthesis. *Lipids*, 42, 97-108.
- Hussain, N., Jabeen, Z., Li, Y., Chen, M., Li, Z., Guo, W., Shamsi, I. H., Chen X_Y. & Jiang, L. (2013). Detection of Tocopherol in Oilseed Rape (Brassica napus L.) Using Gas Chromatography with Flame Ionization Detector. J. Integr. Agr., 12(5), 803–814.
- Ischebeck, T., Zbierzak, A. M., Kanwischer, M. & Dörmann, P. (2006). A salvage pathway for phytol metabolism in Arabidopsis. *J. Biol. Chem.*, 281(5), 2470–2477.
- Jáky, M. (1967). Neue Tocopherol-Bestimmungs-Methoden und deren Anwendung bei Pflanzenölen. Fett. Wiss. *Technol.*, 69, 507–511.
- Jiang, Q., Elson-Schwab, I., Courtemanche, C. & Ames, B. N. (2000). Gamma-tocopherol and its major metabolite, in contrast to alpha-tocopherol, inhibit cyclooxygenase activity in macrophages and epithelial cells. *Proc. Natl. Acad. Sci.*, USA 97, 11494–11499.
- Jurkiewicz, B. A., Bissett, D. L. & Buettner, G. R. (1995). Effect of topically applied tocopherol on ultraviolet radiation-mediated free radical damage in skin. J. Invest. Dermatol., 104, 484–488.
- Kamal-Eldin, A. & Appelqvist, L. A. (1996). The chemistry and antioxidant properties of tocopherols and tocotrienols. *Lipids*, 31, 671–701.

- Karunanandaa, B., Qi, Q., Hao, M., Baszis, S. R., Jensen, P. K., Wong, Y. H. H., Jiang, J., Venkatramesh, M., Gruys, K. J., Moshiri, F., Post-Beittenmiller, D., Weiss, J. D. & Valentin, H. E. (2005). Metabolically engineered oilseed crops with enhanced seed tocopherol. *Metab. Eng.*, 7, 384–400.
- Keller, Y., Bouvier, F., D'Harlingue, A. & Camara, B. (1998). Metabolic compartmentation of plastid prenyllipid biosynthesis-evidence for the involvement of a multifunctional geranylgeranyl reductase. *Eur. J. Biochem.*, 251, 413–417.
- Kim, J. Y., Kim, M. J., Yi, B., Oh, S. & Lee, J. (2015). Effects of relative humidity on the antioxidant properties of α-tocopherol in stripped corn oil. *Food Chem.*, 167, 191–196.
- Kimura, M., Kobori, C. N., Rodriguez-Amaya, D. B. & Nestel, P. (2007). Screening and HPLC methods for carotenoids in sweetpotato, cassava and maize for plant breeding trials. *Food Chem.*, 100, 1734–1746.
- Kljak, K. & Grbeša, D. (2015). Carotenoid content and antioxidant activity of hexane extracts from selected Croatian corn hybrids. *Food Chem.*, *167*, 402–408.
- Knecht, K., Sandfuchs, K., Kulling, S. E. & Bunzel, D. (2015). Tocopherol and tocotrienol analysis in raw and cooked vegetables: A validated method with emphasis on sample preparation. Food Chem. 169, 20–27.
- Koswig, S. & Mörsel, J-T. (1990). Vergleichende Untersuchungen zur quantitativen Bestimmung von Tocopherolen. *Nahrung*, *34*, 89–91.
- Kreps, F., Vrbiková, L., Schmidt, Š., Sekretár, S. & Híreš, O. (2014). Chemical changes in microwave heated vegetable oils. *Eur. J. Lipid Sci. Tech.*, 116(12), 1685–1693.
- Lee, A. Y., Lee, M. H., Lee, S. & Cho, E. J. (2015). Comparative Study on Antioxidant Activity of Vegetable Oils under in vitro and Cellular System. J. Agr. Sci., 7(3), 58-64.
- Lemcke-Norojärvi, M., Kamal-Eldin, A., Appelqvist, L. A., Dimberg, L. H., Ohrvall, M. & Vessby, B. (2001). Corn and sesame oils increase serum gamma-tocopherol concentrations in healthy Swedish women. J. Nutr., 131, 1195-1201.
- Lerma-Garcia, M. J., Simo-Alfonso, E. F., Ramis-Ramos, G. & Herrero-Martinez, J. M. (2007). Determination of tocopherols in vegetableoils by CEC using methacrylate esterbasedmonolithic columns. *Electrophoresis*, 28, 4128–4135.
- Li, Z. H., Matthews, P. D., Burr, B. & Wurtzel, E. T. (1996). Cloning and characterization of a maize cDNA encoding phytoene desaturase, an enzyme of the carotenoid biosynthetic pathway. *Plant Mol. Biol.*, 30(2), 269–279
- Lien, E. J., Ren, S., Bui, H. H. & Wang, R. (1999). Quantitative structure-activity relationship analysis of phenolic antioxidants. *Free Radical Bio. Med.*, 26, 285–294.
- Ligia Focsan, A., Magyar, A. & Kispert, L. D. (2015). Chemistry of carotenoid neutral radicals. *Arch. Biochem. Biophys.*, *18*(848), 359-364
- Lindgren, L. O., Stålberg, K. G. & Höglund, A-S. (2003). Seed-specific overexpression of an endogenous Arabidopsis phytoene synthase gene results in delayed germination and increased levels of carotenoids, chlorophyll, and abscisic acid. *Plant Physiol.*, 132, 779– 785.
- López Ortíz, C. M., Prats Moya, M. S. & Berenguer Navarro, V. (2006). A rapid chromatographic method for simultaneous determination of β-sitosterol and tocopherol homologues in vegetable oils. *J. Food Compos. Anal.*, *19*, 141–149.

- Losada-Barreiro, S., Sánchez-Paz, V. & Bravo-Díaz, C. (2013a). Effects of emulsifier hydrophile-lipophile balance and emulsifier concentration on the distributions of gallic acid, propyl gallate, and α -tocopherol in corn oil emulsions. *Colloid Interface Sci.*, 389, 1–9.
- Luterotti, S., Marković, K., Franko, M., Bicanic, D., Madžgalj, A. & Kljak, K. (2013). Comparison of spectrophotometric and HPLC methods for determination of carotenoids in foods. *Food Chem.*, 140, 390–397.
- Matsubara, A., Uchikata, T., Shinohara, M., Nishiumi, S., Yoshida, M., Fukusaki, E. & Bamba, T. (2012). Highly sensitive and rapid profiling method for carotenoids and their epoxidized products using supercritical fluid chromatography coupled with electrospray ionization-triple quadrupole mass spectrometry. J. Biosci Bioeng., 113, 782-787.
- Matthews, P. D., Luo, R. & Wurtzel, E. T. (2003). Maize phytoene desaturase and ??carotene desaturase catalyse a poly-Z desaturation pathway: Implications for genetic engineering of carotenoid content among cereal crops. J. Exp. Bot., 54(391), 2215–2230.
- Maurer, M. M., Mein, J. R., Chaudhuri, S. K. & Constant, H. L. (2014). An improved UHPLC-UV method for separation and quantification of carotenoids in vegetable crops. *Food Chem.*, 165, 475–482.
- Mène-Saffrané, L. & DellaPenna, D. (2010). Biosynthesis, regulation and functions of tocochromanols in plants. *Plant Physiol. Biochem.*, 48(5), 301–309.
- Miller, N. J., Rice-Evans, C., Davies, M. J., Gopinathan, V. & Milner, A. (1993). A novel method for measuring antioxidant capacity and its application to monitoring the antioxidant status in premature neonates. *Clin. Sci.*, 84(4), 407–412.
- Mishra, P. & Singh, N. (2010). Spectrophotometric and TLC based characterization of kernel carotenoids in short duration maize. *Maydica*, 55, 95–100.
- Moreau, R. A., Johnston, D. B. & Hicks, K. B. (2007). A comparison of the levels of lutein and zeaxanthin in corn germ oil, corn fiber oil and corn kernel oil. J. Am. Oil Chem. Soc., 84, 1039–1044.
- Moros, E. E., Darnoko, D., Cheryan, M., Perkins, E. G. & Jerrell, J. (2002). Analysis of xanthophylls in corn by HPLC. J. Agr. Food Chem., 50, 5787–5790.
- Müller, L., Fröhlich, K. & Böhm, V. (2011). Comparative antioxidant activities of carotenoids measured by ferric reducing antioxidant power (FRAP), ABTS bleaching assay (αTEAC), DPPH assay and peroxyl radical scavenging assay. *Food Chem.*, 129(1), 139–148.
- Munné-Bosch, S. (2007). α-Tocopherol: A Multifaceted Molecule in Plants. Vitam. *Horm.*, 76, 375-392.
- Munné-Bosch, S. & Falk, J. (2004). New insights into the function of tocopherols in plants. *Planta*, 218, 323–326.
- Muzhingi, T., Yeum, K. J., Russell, R. M., Johnson, E. J., Qin, J. & Tang, G. (2008). Determination of carotenoids in Yellow Maize, the effects of saponification and food preparations. *Int. J. Vitam .Nutr. Res.*, 78, 112–120.
- Natividad, L. R. & Rafael, R. R. (2014). Carotenoid Analyses and Antibacterial Assay of Annato (Bixa orellana L.), Carrot (Daucus carota L.), Corn (Zea mays L.) and Tomato (Solanum lycopersicum L.) Extracts. Res. J. Recent Sci., 3(3), 40–45.
- Ndolo, V. U. & Beta, T. (2013). Distribution of carotenoids in endosperm, germ, and aleurone fractions of cereal grain kernels. *Food Chem.*, *139*, 663–671.

- Taylor, P. & Barnes, P. (1981). Analysis of vitamin E in edible oils by high performance liquid chromatography. *Chem. Ind.*, *17*, 722–726.
- Tian, L., Musetti, V., Kim, J., Magallanes-Lundback, M. & DellaPenna, D. (2004). The Arabidopsis LUT1 locus encodes a member of the cytochrome p450 family that is required for carotenoid epsilon-ring hydroxylation activity. *Proc. Natl. Acad. Sci. USA*, 101(1), 402–407.
- Panfili, G., Fratianni, A. & Irano, M. (2003). Normal phase high-performance liquid chromatography method for the determination of tocopherols and tocotrienols in cereals. J. Agr. Food Chem., 51, 3940–3944.
- Porfirova, S., Bergmuller, E., Tropf, S., Lemke, R. & Dormann, P. (2002). Isolation of an Arabidopsis mutant lacking vitamin E and identification of a cyclase essential for all tocopherol biosynthesis. *Proc. Natl. Acad. Sci. USA*, 99, 12495–12500.
- Prieto, P., Pineda, M. & Aguilar, M. (1999). Spectrophotometric quantitation of antioxidant capacity through the formation of a phosphomolybdenum complex: specific application to the determination of vitamin E. *Anal. Biochem.*, *269*, 337–341.
- Provencher, L. M., Miao, L., Sinha, N. & Lucas, W. J. (2001). Sucrose export defective1 encodes a novel protein implicated in chloroplast-to-nucleus signaling. *Plant Cell*, 13, 1127–1141.
- Rahimi, R., Nikfar, S., Larijani, B. & Abdollahi, M. (2005). A review on the role of antioxidants in the management of diabetes and its complications. Biomed. *Pharmacother.*, 59, 365–373.
- Ratnam, D. V., Ankola, D. D., Bhardwaj, V., Sahana, D. K. & Kumar, M. N. V. R. (2006). Role of antioxidants in prophylaxis and therapy: A pharmaceutical perspective. J. Control. Release, 113, 189-207.
- Re, R., Pellegrini, N., Proteggente, A., Pannala, A., Yang, M. & Rice-Evans, C. (1999). Antioxidant activity applying an improved ABTS radical cation decolorization assay. *Free Radical Biol. Med.*, 26, 1231–1237.
- Réblová, Z. (2006). The effect of temperature on the antioxidant activity of tocopherols. *Eur. J. Lipid Sci. Tech.*, 108, 858–863.
- Rivera, S. & Canela, R. (2012). Influence of sample processing on the analysis of carotenoids in Maize. *Molecules*, *17*, 11255–11268.
- Robledo, S. N., Zachetti, V. G. L., Zon, M. A. & Fernández, H. (2013). Quantitative determination of tocopherols in edible vegetable oils using electrochemical ultramicrosensors combined with chemometric tools. *Talanta*, 116, 964–971.
- Rodić, Z., Simonovska, B., Albreht, A. & Vovk, I. (2012). Determination of lutein by highperformance thin-layer chromatography using densitometry and screening of major dietary carotenoids in food supplements. J. Chromatog. A, 1231, 59–65.
- Ryynänen, M., Lampi, A. M., Salo-Väänänen, P., Ollilainen, V. & Piironen, V. (2004). A small-scale sample preparation method with HPLC analysis for determination of tocopherols and tocotrienols in cereals. J. Food Compos. Anal., 17, 749–765.
- Sadre, R., Gruber, J. & Frentzen, M. (2006). Characterization of homogentisate prenyltransferases involved in plastoquinone-9 and tocochromanol biosynthesis. *FEBS Lett.*, 580, 5357–5362.
- Saha, S., Walia, S., Kundu, A., Sharma, K. & Paul, R. K. (2015). Optimal extraction and fingerprinting of carotenoids by accelerated solvent extraction and liquid chromatography with tandem mass spectrometry. *Food Chem.*, 177, 369–375.

- San Andres, M. P., Otero, J. & Vera, S. (2011). High performance liquid chromatography method for the simultaneous determination of α-, γ- and δ-tocopherol in vegetable oils in presence of hexadecyltrimethylammonium bromide/n-propanol in mobile phase. *Food Chem.*, *126*, 1470–1474
- Sandorf, I., Holländer-Czytko, H. (2002). Jasmonate is involved in the induction of tyrosine aminotransferase and tocopherol biosynthesis in Arabidopsis thaliana. *Planta*, 216, 173– 179.
- Schledz, M., Seidler, A., Beyer, P. & Neuhaus, G. (2001). A novel phytyltransferase from Synechocystis sp. PCC 6803 involved in tocopherol biosynthesis. *FEBS Lett.*, 499, 15– 20.
- Schwarz, K., Huang, S. W., German, J. B., Tiersch, B., Hartmann, J. & Frankel, E. N. (2000). Activities of antioxidants are affected by colloidal properties of oil-in-water and water-inoil emulsions and bulk oils. J. Agr. Food Chem., 48, 4874–4882.
- Scott, C. E. & Eldridge, A. L. (2005). Comparison of carotenoid content in fresh, frozen and canned corn. J. Food Compos. Anal., 18, 551–559.
- Serbinova, E., Kagan, V., Han, D. & Packer, L. (1991). Free radical recycling and intramembrane mobility in the antioxidant properties of alpha-tocopherol and alphatocotrienol. *Free Radical Biol. Med.*, 10, 263–275.
- Shahidi, F. (2000). Antioxidants in food and food antioxidants. Nahrung, 44, 158-163.
- Shammugasamy, B., Ramakrishnan, Y., Ghazali, H. M. & Muhammad, K. (2013). Combination of saponification and dispersive liquid-liquid microextraction for the determination of tocopherols and tocotrienols in cereals by reversed-phase highperformance liquid chromatography. J. Chromatog., A, 1300, 31–37.
- Shao, H-B., Chu, L-Y., Lu, Z-H. & Kang, C-M. (2008). Primary antioxidant free radical scavenging and redox signaling pathways in higher plant cells. *Int. J. Biol. Sci.*, *4*, 8–14.
- Shewmaker, C. K., Sheehy, J. A., Daley, M., Colburn, S. & Ke, D. Y. (1999). Seed-specific overexpression of phytoene synthase: Increase in carotenoids and other metabolic effects. *Plant J.*, 20, 401–412.
- Shintani, D. K., Cheng, Z. & DellaPenna, D. (2002). The role of 2-methyl-6phytylbenzoquinone methyltransferase in determining tocopherol composition in Synechocystis sp. PCC6803. *FEBS Lett.*, 511(1-3), 1–5.
- Shintani, D. & DellaPenna, D. (1998). Elevating the vitamin E content of plants through metabolic engineering. *Science*, 282, 2098–2100.
- Shumskaya, M., Bradbury, L. M. T., Monaco, R. R. & Wurtzel, E. T. (2012). Plastid Localization of the Key Carotenoid Enzyme Phytoene Synthase Is Altered by Isozyme, Allelic Variation, and Activity. *Plant Cell*, 24(9), 3725-3741.
- Siger, A., Nogala-Kalucka, M. & Lampart-Szczapa, E. (2008). The content and antioxidant activity of phenolic compounds in cold-pressed plant oils. J. Food Lipids 15(2), 137–149.
- Sim, S. F., Lee, T. Z. E., Mohd Irwan Lu, N. A. L. & Samling, B. (2014). Synchronized analysis of FTIR spectra and GCMS chromatograms for evaluation of the thermally degraded vegetable oils. J. Anal. Methods Chem., 1, 1-9.
- Smith, L. I., Kolthoff, I. M., Wawzonek, S. & Ruoff, P. M. (1941). The Chemistry of Vitamin E. XXIX. 1 Studies of the Behavior of Compounds Related to Vitamin E at the Dropping Mercury Electrode. J. Am. Chem. Soc., 63, 1018–1024.
- Soll, J., Kemmerling, M. & Schultz, G. (1980). Tocopherol and plastoquinone synthesis in spinach chloroplasts subfractions. Arch. Biochem. Biophys., 204, 544–550.

Free ebooks ==> www.Ebook777.com

Vasile Robert Gradinaru and Luiza Madalina Gradinaru

- Soll, J. & Schultz, G. (1979). Comparison of geranylgeranyl and phytyl substituted methylquinols in the tocopherol synthesis of spinach chloroplasts. *Biochem.Bioph. Res. Co.*, *91*, 715–720.
- Speek, A. J., Schrijver, J. & Schreurs, W. H. P. (2006). Vitamin E Composition of Some Seed Oils as Determined by High-Performance Liquid Chromatography with Fluorometric Detection. J. Food Sci. 50, 121–124.
- Stashenko, E. E., Ferreira, M. C., Sequeda, L. G., Martínez, J. R. & Wong, J. W. (1997). Comparison of extraction methods and detection systems in the gas chromatographic analysis of volatile carbonyl compounds. J. Chromatog. A, 779(1-2), 360–369.
- Suantawee, T., Tantavisut, S., Adisakwattana, S., Tanavalee, A., Yuktanandana, P., Anomasiri, W., Deeoaisamsakul, B. & Honsawek, S. (2013). Oxidative stress, vitamin E, and antioxidant capacity in knee Osteoarthritis. J. Clin. Diagn. Res. 7, 1855–1859.
- Sultana, B., Anwar, F. & Przybylski, R. (2007). Antioxidant potential of corncob extracts for stabilization of corn oil subjected to microwave heating. *Food Chem.*, 104, 997–1005.
- Suzuki, Y. J., Tsuchiya, M., Wassall, S. R., Choo, Y. M., Govil, G., Kagan, V. E. & Packer, L. (1993). Structural and dynamic membrane properties of alpha-tocopherol and alphatocotrienol: implication to the molecular mechanism of their antioxidant potency. *Biochemistry*, 32, 10692–10699.
- Szydłowska-Czerniak, A., Dianoczki, C., Recseg, K., Karlovits, G. & Szłyk, E. (2008). Determination of antioxidant capacities of vegetable oils by ferric-ion spectrophotometric methods. *Talanta*, 76, 899–905.
- Thompson, J. N. & Hatina, G. (1979). Determination of Tocopherols and Tocotrienols in Foods and Tissues by High Performance Liquid Chromatography. J. Liq. Chrom., 2(3), 327-344.
- Tsegaye, Y., Shintani, D. K. & DellaPenna, D. (2002). Overexpression of the enzyme phydroxyphenolpyruvate dioxygenase in Arabidopsis and its relation to tocopherol biosynthesis. Plant. Physiol. *Biochem.*, 40, 913–920.
- Tsuchihashi, H., Kigoshi, M., Iwatsuki, M. & Niki, E. (1995). Action of beta-carotene as an antioxidant against lipid peroxidation. *Arch. Biochem. Bioph.*, *323*, 137–147.
- Tütem, E., Apak, R., Günaydı, E. & Sözgen, K. (1997). Spectrophotometric determination of vitamin E (alpha-tocopherol) using copper(II)-neocuproine reagent. *Talanta*, 44, 249–55.
- Valdivielso, I., Bustamante, M. Á., Ruiz de Gordoa, J. C., Nájera, A. I., de Renobales, M. & Barron, L. J. R. (2015). Simultaneous analysis of carotenoids and tocopherols in botanical species using one step solid–liquid extraction followed by high performance liquid chromatography. *Food Chem.*, 173, 709–717.
- Valentin, H. E., Lincoln, K., Moshiri, F., Jensen, P. K., Qi, Q., Venkatesh, T. V, Karunanandaa, B., Baszis, S. R., Norris, S. R., Savidge, B., Gruys, K. J. & Last, R. L. (2006). The Arabidopsis vitamin E pathway gene5-1 mutant reveals a critical role for phytol kinase in seed tocopherol biosynthesis. *Plant Cell*, 18, 212–224.
- Whistance, G. R. & Threlfall, D. R. (1970). Biosynthesis of phytoquinones. Homogentisic acid: a precursor of plastoquinones, tocopherols and alpha-tocopherolquinone in higher plants, green algae and blue-green algae. *Biochem. J.*, 117, 593–600.
- Whittle, K. J. & Pennock, J. F. (1967). The examination of tocopherols by two-dimensional thin-layer chromatography and subsequent colorimetric determination. *Analyst*, 92, 423– 430. doi:10.1039/an9679200423
- Wolff, J. P. (1968). Fatty substances analysis manual, Azoulay, Paris.

- Yoshida, H., Hirooka, N. & Kajimoto, G. (1990). Microwave Energy Effects on Quality of Some Seed Oils. J. Food Sci., 55, 1412–1416.
- Zahir, E., Saeed, R., Hameed, M. A. & Yousuf, A. (2014). Study of physicochemical properties of edible oil and evaluation of frying oil quality by Fourier Transform-Infrared (FT-IR) Spectroscopy. *Arab. J. Chem.*
- Zarrouk, W., Carrasco-Pancorbo, A., Zarrouk, M., Segura-Carretero, A. & Fernández-Gutiérrez, A. (2009). Multi-component analysis (sterols, tocopherols and triterpenic dialcohols) of the unsaponifiable fraction of vegetable oils by liquid chromatographyatmospheric pressure chemical ionization-ion trap mass spectrometry. *Talanta*, 80, 924– 934.
- Zhou, Y., Zhi-Yuan, F., Li, Q., Shu-Tu, X., Subhash, C., Yan, X., Jianbing, Y., JianSheng, L. (2009). Comparative analysis of carotenoid and tocopherol compositions in high-oil and normal maize (Zea mays L.) inbreds. *Acta Agronomica Sinica*, 35(11), 2073–2084.
- Zhu, C., Naqvi, S., Breitenbach, J., Sandmann, G., Christou, P. & Capell, T. (2008). Combinatorial genetic transformation generates a library of metabolic phenotypes for the carotenoid pathway in maize. *Proc. Natl. Acad. Sci. USA*, 105, 18232–18237.
- Zingg, J-M. & Azzi, A. (2004). Non-antioxidant activities of vitamin E. *Cur. Med. Chem.*, 11, 1113–1133.

Free ebooks ==> www.Ebook777.com

Chapter 3

SUPERCRITICAL FLUID EXTRACTION OF CORN GERM OIL

Sara Rebolleda, María Teresa Sanz^{*}, Sagrario Beltrán, Rodrigo Melgosa and Angela García Solaesa

Industrial and Environmental Biotechnology, Chemical Engienering Section. University of Burgos, Burgos, Spain

ABSTRACT

In this chapter an environmental friendly extraction process of corn germ oil based on the use of supercritical CO_2 (SC- CO_2) is presented. The effect of important operating parameters in supercritical fluid extraction (SFE) processes such as pressure, temperature and flow rate on the extraction kinetics and the quality of the extracted oil is discussed. As for many SC- CO_2 extractions of vegetable oils, extraction curves of corn germ oil present an initially linear part with a slope close to the oil solubility value in CO_2 . Then, a second section of the extraction curve is determined by the diffusional resistance in the solid matrix. Characterization of supercritical crude corn oil is presented by showing some properties reported in the literature such as physical parameters, fatty acid composition, neutral lipids, content of tocopherols, acid index, peroxide value, antioxidant capacity and the oxidative stability.

The quality of supercritical corn oil is also compared with the oil obtained by using conventional extraction process.

Keywords: supercritical fluid extraction, corn germ oil

^{*} Corresponding author: María Teresa Sanz. E-mail: tersanz@ubu.es, tel.: +34 947 258810, fax: +34 947 258831.

NOMENCLATURE

as	specific area between the regions of intact and broken cells (m ⁻¹)		
C ₁ , C ₂	fitting parameters		
e	extraction yield, (kg extract×kg insoluble solid ⁻¹)		
E	extract (kg)		
k _s	solid-phase mass transfer coefficient (s ⁻¹)		
n	number of experimental data		
N _m	charge of insoluble solid (kg)		
O.F.	objective function		
Q	solvent flow rate $(kg \times h^{-1})$		
q	relative amount of the passed solvent (kg solvent×kg insoluble solid ⁻¹)		
q _c	relative amount of the passed solvent when all the solute in broken cells has		
	been extracted (kg solvent×kg insoluble solid ⁻¹)		
r	grinding efficiency (fraction of broken cells)		
t	extraction time (h)		
Xu	concentration in the untreated solid (kg solute×kg solid insoluble ⁻¹)		
y _s	solubility (kg solute×kg solvent ⁻¹)		

INTRODUCTION

Corn (*Zea mays*) is mainly used for starch, sugar syrups and oil production. In the corn wet milling process, the corn kernel is separated into 3 main parts: the endosperm (gluten and starch), the outer skin, called the bran or hull, and the germ (around 8% of raw corn), which is the most valuable part of the kernel because of its high concentration of linoleic acid and its bland taste (Dijkstra et al., 2007). As depicted in Figure 1, after a cleaning process, corn undergoes a steeping stage that conditions the grain for subsequent germ separation. Corn germ, after removal of the starch adhered along the separation process from the whole corn, is dried to approximately a 4% moisture content to improve shelf life. The dry milling process does not start with a steeping step; instead, the corn kernels are directly ground and the germ. Extracting the kernel is limited to high-oil corn with an oil content of at least 8%. Corn oil concentrates in the germ with a content lying between 40% and 52% (Dijkstra, 2013).

Most of the corn germ oil is traditionally obtained from the germ using a conditioning process followed by mechanical expelling (prepress) and, in some cases, ending up with solvent extraction of the expeller cake. The conditioning process is diverse and several procedures as milling, cracking, flaking, heating, etc. have been proposed (Dijkstra, 2013). The corn germ meal obtained after pressing still contains 6-10% of residual oil while it contains 1-3% after solvent extraction (AWARENET, 2004). Extrusion has also been employed to prepare the germ for solvent extraction producing a crude corn oil of high quality and high yield (Gunstone, 2002). The aim of the extraction method is to optimize the oil yield while maintaining the oil quality (Matthäus, 2012).

After extraction, corn oil usually undergoes a refining process that comprises removal of gums, free fatty acids, adsorbable compounds, waxes and malodorous compounds (Dijkstra

and Segers, 2007). In some cases, specialty oil producers manufacture unrefined, 100% expeller pressed corn oil, with lower yield, thus more expensive and smaller market share.

Corn germ oil is mainly used in food and cosmetic applications and corn germ meal for animal feed production.



Figure 1. Flow diagram of the wet milling process for obtaining corn germ oil.

In the last decades, some more environment friendly procedures have been suggested for corn germ oil production, among them, the use of enzymes to enhance the oil recovery and/or extraction of oil by using supercritical fluid extraction (SFE) (Matthäus, 2012). This chapter is focused on the SFE process of corn germ oil, paying special attention to the description of the process and the influence of process parameters on the quality of the resulting oil.

SUPERCRITICAL FLUID EXTRACTION OF CORN GERM OIL

A supercritical fluid (SCF) is any fluid at a temperature and pressure above its critical point, where the liquid and vapor phases merge in one only phase because their densities

become equal. The physical properties of a supercritical fluid are thus intermediate to those of the respective liquid and gas, with good transport properties (viscosities and diffusivities) like gases, and high solvent power, like liquids. In addition, small changes in pressure or temperature result in large changes in density, allowing many properties of a SCF to be "tuned."

Specifically CO₂ shows mild critical conditions (Tc = 304.15 K and Pc = 7.38 MPa); therefore, although operation at high pressure is usually costly, pressures above 7.38 MPa can be cost effective. Analogously, temperatures above 31°C can be mild enough for processing thermo-sensible products. Carbon dioxide, together with water, are the only solvents that are not considered as solvents requiring process revaluation by the US FDA. In the cases where water cannot be used because solutes are non-polar, carbon dioxide can be used as a nonpolar solvent. Carbon dioxide is non-toxic, not flammable, naturally abundant and inexpensive. In addition, at the critical point, surface tension approaches zero and there is no formation of a liquid-vapor meniscus, reasons for which supercritical fluids can easily penetrate into solid matrices and can be separated from the processed products by simple depressurization resulting products free from processing residues.

The number of processing plants that employ supercritical carbon dioxide (SC-CO₂) technologies, operating worldwide, is growing steadily. Most of the current plants are related to food processing, yet other type of processes are using these technologies. Some examples are fluoropolymer synthesis by DuPont, hydrogenation by Thomas Swan, coatings by Union Carbide or polyurethane processing by Crain Industries (Beckman, 2012).

In the food industry, the most widely commercialized SCF technology is SC-CO₂ extraction. SC-CO₂ is able to selectively solve caffeine from coffee, certain acids from hops, oil from seeds, various components from spices, etc. This ability has been fundamental to design different extraction processes that are today widely commercialized. The attractive properties of SC-CO₂ as solvent are currently being exploited in processes such as coffee decaffeination (Hag AG, Evonik, Kraft General Foods, etc.) or hops fractionation (NATECO2, Carlton, United Breweries, Hops Extraction Corp. of America, etc.) displacing the conventional solvents.

Also companies designing SCF processes and equipment are steadily growing. Companies such as NATEX (Natex, 2015) or Uhde High Pressure Technologies (Uhde, 2015) have designed supercritical fluid industrial plants for extraction of spices, herbs, hops, nutraceuticals or edible oil as well as for cork purification or rice treatment among other.

SFE of corn germ oil, although not yet at an industrial scale, has been studied by several authors, including some recent patents related to SFE of corn germ oil and SF fractionation of the extracted oil to obtain a refined corn oil (DeLine et al., 2009, DeLine et al., 2013, Marentis, 2013).

One of the first studies about SFE of corn germ oil was performed by the Agricultural Research Service of the United States Department of Agriculture (Peoria, Illinois). List et al., (List et al., 1984, List and Friedrich, 1989) and Christianson et al., (Christianson et al., 1984) compared the quality of crude oils obtained from dry and wet milled corn germ using SFE at 50-90°C and 55-83 MPa and conventional extraction methods. They concluded that the oil obtained by SC-CO₂ extraction exhibited lower refining loss and lighter color.

List et al., (List et al., 1984) concluded that, in general, the quality of crude oil obtained by SFE was unaffected by extraction conditions. The levels of tocopherols present in SC-CO₂ extracted corn oil were similar to those obtained by conventional methods (List et al., 1984, List and Friedrich, 1989). However the phospholipids were almost absent in the SC-CO₂ oils, which is an advantage from a processing point of view.

Wilp and Eggers (Wilp and Eggers, 1991) proposed a SFE process followed by a fractionation separation step to improve the oil quality, since low polar triglycerides could be separated from high polar compounds such as water and free fatty acids.

Vigh et al., (Vigh et al., 1993) reported some extraction curves of wet corn germ oil with SC-CO₂ in an interval of pressure (27.0-33.0 MPa) and temperature (42-78 $^{\circ}$ C) for corn germ with a particle size < 0.8 mm or 0.8-1.4 mm. Based on a second order orthogonal design, Vigh et al., (Vigh et al., 1993) concluded that, for the smaller particle size range, application of low pressures and high temperatures should be avoided. Ronyai et al., (Rónyai et al., 1998) studied the SC-CO₂ extraction of wet-milled corn germ oil with cosolvents (ethyl alcohol: from 0% to 10% by weight in CO2) at constant pressure and temperature (30 MPa and 42° C). The extraction time was reduced due to the higher solubility of corn oil the higher the ethanol concentration, although the amount of phospholipids in the oil was larger, since they are soluble in ethanol. Ronyai et al., (Rónyai et al., 1998) found that the emulsifying, foaming and absorption properties of the defatted meal and its protein isolates were better when using ethyl alcohol as cosolvent. Recently, Rebolleda et al., (Rebolleda et al., 2012) studied the SC- CO_2 extraction kinetics of oil from milled corn germ at different pressures (20.0-53.0 MPa), temperatures $(35-86^{\circ}C)$ and solvent flow rates $(4-9 \text{ kg CO}_{2}/h)$ concluding that the extraction process may be controlled by the solubility of the oil in SC-CO2 in the first stages of the extraction and that the external mass transfer resistance was not important. They also studied the oil quality concluding that the extraction temperature affects the oil quality, yielding oils with higher tocopherol content, and therefore higher antioxidant activity and lower oxidation level, when the extraction temperature increased. In addition, the oil they obtained by SC-CO2 extraction had lower values of acidity than oils obtained by conventional extraction methods. Rebolleda et al., (Rebolleda et al., 2012) also studied on-line fractionation of the oil extracted in two separators installed in series and separated the co-extracted water improving the oil stability against oxidation. In addition to oil, SFE of corn germ results in interesting defatted corn germ flour, due to its low fat content and lower peroxidase activity compared to hexane extracted corn germ flour, SFE flour having larger storage stability and better flavor, which are important economic factors. The bound lipid content of hexane extracted meal was nearly four times that of the SC-CO₂ extracted flour. Regarding the peroxidase activity, it is normally difficult to inactivate the enzyme by the normal toasting process, but it has been found that a sevenfold reduction in peroxidase activity can be achieved under SCF extraction (Christianson et al., 1984, Friedrich and Pryde, 1984, Christianson and Friedrich, 1985). Another important consideration is that deterioration of proteins and amino acids do not take place under the conditions usually employed in SFE in food, and a good balance of essential amino acids has been found compared to the published FAO/ALTO standards for highly nutritious proteins (Favati et al., 1996). Figure 2 provides a visual comparison of the corn germ meal obtained after SC-CO₂ extraction at the University of Burgos and corn germ bagasse obtained after mechanical expelling. In this chapter, the influence of different parameters on the extraction yield and oil quality obtained when using $SC-CO_2$ extraction as a process to obtain corn germ oil is presented by showing different results reported in the literature.

SUPERCRITICAL FLUID EXTRACTION EQUIPMENT AND PROCEDURE

The elements of a supercritical fluid extraction plant have to be designed taking into account the safety factors needed to operate safely under high pressure.



Figure 2. Samples of corn germ, ground corn germ, and corn germ meal obtained after $SC-CO_2$ extraction at the University of Burgos. Corn germ bagasse obtained after mechanical expelling is also included for comparison.

The most important elements to be assembled in a SFE plant are pump or compressor, extractor, one or more consecutive separators, heating and cooling systems, closures, tubing, valves, safety elements such as rupture disks and safety valves, and the instrumentation needed for measurement and control of the process parameters. As an example, the P&I diagram of the semi-pilot SFE plant of the University of Burgos is presented in Figure 3. This plant uses a pump for pressurizing liquid CO₂ after being cooled to conditions away from saturation. The maximum specifications of the SFE plant are: $T = 200^{\circ}$ C, p = 65.0 MPa and solvent flow, F = 20 kg/h.

In order to illustrate the operation of a SFE plant, the procedure followed by Rebolleda et al., (Rebolleda et al., 2012) is presented. In a SFE experiment, a certain amount of corn germ is placed in the extractor (350 g). At a laboratory scale, the material to be extracted is sometimes mixed with inert fillings in order to avoid bed compaction; however bed compaction is usually avoided at industrial scale by using special designs or subtract pretreatments. The extractor was later pressurized with CO_2 up to the extraction pressure. Then, the solvent was circulated at the desired extraction pressure and temperature, with a certain solvent flow, F, and during a specific time, t. The solvent was continuously recycled

to the extractor after removing the solute in the separator. Solvent recycling is always used at an industrial scale.

Moisture content of the raw material is an important factor in SC-CO₂ extraction of natural products (Brunner, 1994) although many authors have not found a strong effect when moisture is lower than 20% (Rubio-Rodríguez et al., 2008). Christianson et al., (Christianson et al., 1984) reported that tempering the germ to 8% moisture prior to flaking did not influence extraction solubility or recovery. For instance, in the experiments carried out by Rebolleda et al., (Rebolleda et al., 2012) the corn germ was milled in a coffee grinder to a particle size ranging from 0.5 mm to 1 mm and had an average moisture content of $6.0 \pm 0.4\%$ and $46 \pm 3\%$ average fat content.



Figure 3. Flow-sheet diagram of the supercritical fluid extraction semi-pilot plant at the University of Burgos. CO_2 storage (1), Purge valves (P1, P2, P3), Shutoff valves (S1, S2, ..., S8), Check valve (C1), Cooling bath (2), Temperature controllers (TC), Total mass flow meter (3), Pump (4), Rupture disks (5, 8), Heating bath (6), Extractor (7), Temperature transducers (TT), Pressure transducers (PT), Expansion valve (E1), Separator (9).

ANALYTICAL METHODS FOR CHARACTERIZATION OF CORN GERM OIL

Lipid analysis comprises a vast range of methods, approaches and analyses that serve different purposes. A large amount of analytical methods have been developed and published by learned societies, and national or international bodies that issue standards. For source materials, meals and oils and fats alone, the AOCS (http://onlinemethods.aocs.org) has published more than 250 different Official Methods. A compilation of the most important

Sara Rebolleda, María Teresa Sanz, Sagrario Beltrán et al.

parameters and methods used for oil analyses is provided by Dijkstra et al., (Dijkstra et al., 2007).

Different parameters can be employed to characterize the extracted corn germ oil such as physical parameters (e.g.: refractive index, density, color) and chemical parameters, including composition in terms of fatty acids profile by gas chromatography (AOAC, 1995) and neutral lipids, that can be determined, among other analytical methods, by liquid chromatography (Schaefer et al., 2003).

Other important quality parameters are related to the antioxidant capacity of the extracted oil. Total antioxidant capacity can be evaluated through assays such as ABTS (Rebolleda et al., 2014), DPPH (Brand-Williams et al., 1995) or FRAP (Rebolleda et al., 2013) among others. Quantifying the main compounds that provide the antioxidant capacity is a different way of inferring such parameter for a substrate; Tocopherols for instance are important antioxidants that are found in many vegetable oils. Tocopherols have four isomers (α -, β -, γ -, δ -) differing in the number and position of methyl groups on the chromanol ring (Table 1). Tocopherols are well recognized for their effective inhibition of lipid oxidation in foods and biological systems. The antioxidant activity of tocopherols is mainly due to their ability to donate their phenolic hydrogens to lipid free radicals (Kamal-Eldin and Appelqvist, 1996). In the literature (Rebolleda et al., 2012) tocopherols have been successfully determined by HPLC-DAD after isolation by solid phase extraction (SPE). Individual compounds of α -, β -, γ - and δ -tocopherols can be identified and quantified using a calibration curve of the corresponding standard compound (Figure 4).



$\begin{array}{c} R1 \\ HO \\ 6 \\ 5 \\ 4 \\ R2 \\ \hline Me \\ \hline \hline Me \\ \hline Me \\ \hline Me \\ \hline \hline \hline Me \\ \hline \hline \hline \hline Me \\ \hline \hline \hline \hline Me \\ \hline \hline \hline \hline \hline Me \\ \hline \hline \hline \hline \hline Me \\ \hline $					
α-tocopherol	$\mathbf{R}_1 = \mathbf{C}\mathbf{H}_3 \ \mathbf{R}_2 = \mathbf{C}\mathbf{H}_3$				
β-tocopherol	$\mathbf{R}_1 = \mathbf{C}\mathbf{H}_3 \ \mathbf{R}_2 = \mathbf{H}$				
γ-tocopherol	$\mathbf{R}_1 = \mathbf{H} \ \mathbf{R}_2 = \mathbf{C} \mathbf{H}_3$				
δ- tocopherol	$\mathbf{R}_1 = \mathbf{H} \ \mathbf{R}_2 = \mathbf{H}$				

Kamal-Eldin and Appelqvist, 1996.

Other parameters that are usually determined to assess the oil stability and quality are the acid index (AI) and the peroxide value (Ca 5a-40 and Cd 8-53 AOCS Methods). The oxidative stability is also an important parameter in the characterization of fats and oils that can be determined among other methods by using the Rancimat test.

Further details for most of these analytic procedures can be found in the work published by Rebolleda et al., (Rebolleda et al., 2012).



Figure 4. HPLC-DAD chromatogram of tocopherols in the SC-CO₂ extracted corn oil.

INFLUENCE OF PROCESS PARAMETERS ON THE EXTRACTION YIELD

The most important parameters that influence the supercritical fluid extraction process of solid substrates are extraction pressure and temperature, which determine solvent density, solvent ratio and solid substrate pretreatmente and moisture (Brunner, 1994).

By determining the influence of such parameters on the extraction kinetics, the mechanism controlling the extraction process can be elucidated. SFE can be governed by one or all of the following resistances to the process: internal mass transfer, equilibrium solute solubility and external mass transfer. The optimization of the process requires the knowledge of the stage controlling the process under different process parameters.

In order to illustrate the influence of the different process parameters on the extraction yield, recent results obtained by Rebolleda et al., (Rebolleda et al., 2012) under different extraction conditions (see Table 2) are presented in this chapter and compared with other results previously reported in the literature.

Similar to other vegetable oils, extraction curves present an initial section controlled by equilibrium constraints and therefore characterized by a linear relationship between the mass of oil extracted and the quantity of CO_2 employed. The characteristics of this curve depend on the pretreatment of the oilseeds which accelerates the release of the oil bound to the cells. In the second section of the curve, mass transfer is determined by diffusional resistance in the solid matrix (Eggers, 1996).

Run	P (MPa)	T (°C)	Solvent flow rate, (kg CO ₂ /h)
R1	45 ± 2	40 ± 2	8.5 ± 0.8
R2	45 ± 2	63 ± 2	8.8 ± 0.8
R3	46 ± 2	79 ± 2	9.5 ± 0.9
R4	45 ± 2	85 ± 3	8.0 ± 0.9
R5	30 ± 2	80 ± 2	10.0 ± 0.7
R6	21 ± 1	39 ± 1	9.0 ± 0.7
R7	53 ± 2	80 ± 2	6.0 ± 0.5
R8	44 ± 2	78 ± 2	3.9 ± 0.5

 Table 2. Process parameters used in the experiments of SFE of corn germ oil with

 SC-CO2

Effect of extraction temperature. In SFE, an increase of the extraction temperature, although implying a decrease of the fluid density, could be also responsible for an increase in the extraction yield due to an increase of the solute vapour pressure.

As an example, Figure 5 shows the effect of the extraction temperature on the extraction yield reported by Rebolleda et al., (Rebolleda et al., 2012) from 40°C to 85°C (Runs 1-4) at 45.0 MPa and an average SC-CO₂ flow rate around 9 kg/h. It can be observed that the higher the temperature the higher the extraction rate. Therefore, this behavior may indicate that, at this pressure, the increase of oil vapor-pressure with temperature is more important than the decrease in SC-CO₂ density.

As it is well established in the literature (Salgın et al., 2006), the increase of seed oil solubility with temperature can be significant when the process is performed at pressures higher than 40 MPa, pressure at which a crossover behavior is usually observed in seed oils (Salgın et al., 2006).



Figure 5. Influence of extraction temperature on corn germ oil extraction yield (\Box 40°C; \triangle 63°C; \bigcirc 79°C; * 85°C), at around 45.0 MPa and 9 kg CO₂/h.. Solid lines correspond to the model of Sovová (Sovová, 2005).
Free ebooks ==> www.Ebook777.com

Supercritical Fluid Extraction of Corn Germ Oil

Effect of extraction pressure. The CO₂ pressure is probably one of the main parameters that affect the extraction process. At a given temperature, an increase of the extraction pressure means an increase in the CO₂ density and consequently a higher solvent capacity. To illustrate the effect of extraction pressure, Figures 6a and 6b present the effect of pressure from 21 MPa to 53 MPa at two different operating temperatures of 40°C and 80°C, reported by Rebolleda et al., (Rebolleda et al., 2012) (Runs 1, 6 and 3, 5 and 7). In both cases, the extraction curves indicate that, at constant temperature, the higher the pressure the higher the extraction rate. This is mainly due to the higher density of SC-CO₂, and therefore, to its higher solvent power.



Figure 6. Influence of extraction pressure on corn germ oil extraction yield at constant extraction temperature of (a) 40°C (+ 30 MPa; \bigcirc 45 MPa; \bigcirc 53 MPa) (b) 80°C (\blacksquare 21 MPa; \square 45 MPa). SC-CO₂ flow was around 9 kg CO₂/h in all cases. Solid lines correspond to the model of Sovová (Sovová, 2005).

In any case, Figure 2b indicates that the extraction at the lowest pressure (Run 6, 21 MPa) is very slow, which may be due to the low solubility of seed oil in SC-CO₂. An increase of oil solubility when extraction pressure is increased has been also reported for dry-milled corn germ (Christianson et al., 1984) and other seed oils (Friedrich and Pryde, 1984). As it will be explained later in this chapter, the first part of the extraction can be considered to be controlled by this thermodynamic parameter.

Effect of solvent flow rate. The effect of solvent flow rate has been studied by Rebolleda et al., (Rebolleda et al., 2012) and their results are presented in Figure 7 at a constant pressure and temperature. Extraction curves, expressed in dependence on the solvent-to-feed ratio, are not significantly affected by SC-CO₂ flow rate. This behavior supports the fact that solubility, not external mass transfer, is controlling the SC-CO₂ extraction process of corn germ oil.



Figure 7. Influence of solvent flow rate on corn germ oil extraction yield at around 45 MPa and 79°C: ● 4 kg CO₂/h; O 9.5 kg CO₂/h. Solid lines correspond to the model of Sovová (Sovová, 2005).

MODELLING OF THE SUPERCRITICAL FLUID EXTRACTION OF SOLIDS

Mathematical modeling of processes is a powerful tool to easily describe and optimize such processes and for scaling them up. However, modelling of extraction of natural materials is a difficult task since they have irregular structure and complicated geometry that change during the extraction process (Eggers, 1996).

For the specific process of supercritical fluid extraction (SFE) from solid substances, several models, from relatively simple theoretical models to more rigorous ones, have been developed (Brunner, 1994, Oliveira et al., 2011, Huang et al., 2012, Rai et al., 2014). When SFE is specifically applied to obtain extracts from plant materials, it should be considered that the solutes of interest are usually inside cell structures.

Plant materials are thus usually pretreated to break the cell structures and facilitate the contact of the solvent and the solute, thus increasing the extraction kinetics. Taking this into account, Sovová (Sovová, 1994) introduced the concept of broken and intact cells (BIC) to mathematically describe the extraction process from natural materials.

This concept has been since then extensively applied for modeling of SFE processes of natural substances in terms of the overall extraction curves (Huang et al., 2012). A BIC type mathematical model developed later by Sovová (Sovová, 2005) was the one chosen by Rebolleda et al., (Rebolleda et al., 2012) for modeling the experimental extraction curves previously presented This type of model assumes that the solute is regarded as a single pseudo compound. This simplification may lead to some errors since several components are generally involved in the extraction of seed oil (Reverchon and Marrone, 2001), but still keep models very useful for process description and scaling up. In this model (Sovová, 2005), the extraction yield is expressed as:

$$e = \frac{E}{N_m}$$
(1)

where E is the amount of extract (kg) and N_m the charge of insoluble solid (kg) in the extractor. The dimensionless amount of solvent consumed is obtained by:

$$q = \frac{Qt}{N_m}$$
(2)

where Q is the solvent flow rate (kg/h) and t the extraction time (h). Based on this model, the extraction curves consist of two parts. During the first one, the easily accessible solute from broken cells is transferred directly to the fluid phase, obtaining an initial part with a slope close to the solubility of oil in CO_2 , while in the second one, the solute from intact cells diffuses first to broken cells and then to the fluid phase (Sovová, 2005).

Keeping this in mind, Rebolleda et al., (Rebolleda et al., 2012) calculated the initial slopes of different extraction curves of corn germ oil and compared them with data of solubility of oil in CO_2 . To estimate the solubility of corn germ oil in SC-CO₂ these authors followed the general equation proposed by del Valle et al., (del Valle et al., 2012) to predict the solubility of many vegetable oils in high-pressure CO_2 :

$$c_{sat}(g\cdot kg^{-1}) = 8.07 \left(\frac{\rho}{910}\right)^{\left[9.59-8.45\left(\frac{\rho}{910}-1\right)-23.0\left(\frac{\rho}{910}-1\right)^{2}\right]} \exp\left\{-4182 \left[1-259\left(\frac{1}{T}-\frac{1}{313}\right)\right]\left(\frac{1}{T}-\frac{1}{313}\right)\right\}$$
(3)

The reason for this behaviour is that the fatty acid composition of each oil is slightly different (Eggers, 1996). Del Valle et al., (del Valle et al., 2012) stated that Equation (3) can be applied to systems pure oil + high-pressure CO_2 as well as to oil containing vegetable substrates, since the initial stages of the extraction process is typically solubility-controlled. Del Valle et al., (del Valle et al., 2012) indicated that data obtained at a process temperature

different from 40°C (T = 313.15 K) must be divided by a temperature-correction term (TCT) in order to compare the initial slopes of the extraction curves:

$$TCT = \exp\left\{-4182\left[1 - 259\left(\frac{1}{T} - \frac{1}{313}\right)\right]\left(\frac{1}{T} - \frac{1}{313}\right)\right\}$$
(4)

The corrected (at 40°C) initial slope values calculated by Rebolleda et al., (Rebolleda et al., 2012) have been plotted in Figure 8 as a function of CO_2 density together with the prediction of the oil solubility from the General Model proposed by del Valle et al., (del Valle et al., 2012) and the error limit of this model. It can be observed that the values of the slope of the first part of the extraction are within the error limits for solubility of vegetable oils in CO_2 .

In other studies of SFE of corn germ oil using ethanol as co-solvent (Rónyai et al., 1998) a linear increase at the beginning of the process has been also observed.

The value obtained in this study (Rónyai et al., 1998) has been also plotted in Figure 8. In this Figure, the solubility values of corn germ oil at 40°C obtained by Soares et al., (Soares et al., 2007) have been also included.



Figure 8. Corrected (at 40°C) experimental solubility values of corn germ oil as function of pure CO2 density.(\times) experimental initial slope values of the extraction curves obtained by Rebolleda et al., (Rebolleda et al., 2012); (—) prediction of del Valle et al., (del Valle et al., 2012) General Model; (- -) error limits of the General Model; (\Box) solubility data from Soares et al., (Soares et al., 2007); (O) slope datum from Rónyai et al., (Rónyai et al., 1998).

Since the slope of the first part of the extraction curves can be considered of the same order as the oil solubility data, the model proposed by Sovová (Sovová, 2005) can be used to evaluate the first and second part of the extraction curves by using the following equations:

$$\mathbf{e} = \mathbf{q} \, \mathbf{y}_{s} \,, \, \text{for } 0 \le \mathbf{q} \le \mathbf{q}_{c} \tag{5}$$

$$e = x_{u} \left[1 - C_{1} \exp(-C_{2}q) \right]_{, \text{ for } q > q_{c}}$$
(6)

 C_1 and C_2 are model parameters, y_s is the experimental solubility datum (slope of the extraction curve), q_c the crossing point and x_u is the solute concentration in the untreated solid that can be obtained from the oil content in the corn germ.

The values of such constants obtained from the experimental extraction curves obtained by Rebolleda et al., (Rebolleda et al., 2012) are presented in Table 3. Model calculations have been represented along with the experimental data points in Figures 5-7 and good agreement can be observed between experimental data and model correlation. Following the model proposed by Sovová (Sovová, 2005), important parameters can be estimated from constants C_1 , C_2 and co-ordinate q_c at the crossing point such as the volumetric fraction of broken cells in the particles, called grinding efficiency, r, and the solid-phase mass transfer coefficient, k_sa_s (Table 3):

$$r = 1 - C_1 \exp(-C_2 q_c/2)$$
⁽⁷⁾

$$\mathbf{k}_{s}\mathbf{a}_{s} = (1-r)(1-\varepsilon)\dot{\mathbf{Q}}\mathbf{C}_{2}/\mathbf{N}_{m}$$
(8)

In equation (8) solvent flow rate is expressed in kg·s⁻¹. The estimated grinding efficiency was similar in all the extractions runs and it can be concluded that the volumetric fraction of broken cells in the corn germ particles is nearly 0.5. The crossing point, q_c , was found to increase with a decrease in the solubility value, specially marked at low operating pressure.

Experiment	C ₁	C ₂	q _c	r	k _s a _s
R1	0.6049	0.0054	25	0.43	$1.9 \cdot 10^{-5}$
R2	0.6728	0.0144	26	0.44	$5.0 \cdot 10^{-5}$
R3	0.6546	0.0188	23	0.47	$6.6 \cdot 10^{-5}$
R4	0.6586	0.0214	22	0.48	$6.2 \cdot 10^{-5}$
R5	0.6419	0.0050	55	0.44	$2.0 \cdot 10^{-5}$
R6	0.7478	0.0023	75	0.32	$1.0 \cdot 10^{-5}$
R7	0.7234	0.0297	17	0.44	$7.1 \cdot 10^{-5}$
R8	0.6509	0.0158	28	0.48	$2.3 \cdot 10^{-5}$

 Table 3. Model parameters C1, C2, qc, estimated grinding efficiency r and solid-phase mass transfer coefficient, ksas

Obtained by Rebolleda et al., (Rebolleda et al., 2012).

QUALITY AND STABILITY OF CORN GERM OIL

Rebolleda et al., (Rebolleda et al., 2012) evaluated the quality of corn germ oil in two samples extracted with SC-CO₂ at a constant pressure near 45.0 MPa and two different temperatures of 56 and 84°C (runs 9-10, Table 4).

The physical parameters evaluated (refraction index, density and color) for these authors showed no difference between the oils extracted at different temperatures. The average values they reported were a refraction index of 1.472 ± 0.001 at 25°C, a density of 0.92 ± 0.01 kg/L at 20°C and a color expressed as 100.07, -0.007, -0.012 corresponding to lightness (L*), redness (a*) and yellowness (b*), respectively.

Table 5 presents the fatty acid composition in different SFE extracts analyzed by Rebolleda et al., (Rebolleda et al., 2012) for the experiments presented in Table 4. It can be observed that the extraction temperature does not influence the fatty acid profile. Corn germ oil only contains 15% of saturated fatty acids, while it is very rich in polyunsaturated fatty acids (56%). A total of 8 fatty acids were identified by Rebolleda et al., (Rebolleda et al., 2012), being linoleic acid (more than 50%) the most abundant fatty acid, followed by oleic acid (28%) and palmitic acid (13%). Similar fatty acid profile of corn germ oil has been reported by other authors when using SFE (Vigh et al., 1993) and for crude germ oil obtained by pressing, followed in some cases by solvent extraction (Johnson, 2004).

Table 4. Experimental conditions in the SFE of corn germ oil with SC-CO₂ to evaluate the quality of corn germ oil

Run	P (MPa)	T (°C)	Solvent flow rate, (kg CO ₂ /h)
R9	45 ± 2	56 ± 2	10.0 ± 0.9
R10	43 ± 2	84 ± 2	5.0 ± 0.5

One of the most important parameters to assess the oil quality is its acidity. Oil acidity depends on several factors such as oil composition, extraction procedure and raw material freshness. The acidity is related to the presence of free fatty acids (FFA) mainly generated by hydrolysis reaction of triacylglycerides and other non-lipid acid compounds such as acetic acid, that can be generated during spoilage of the raw material.

For corn germ oil extracted by SC-CO₂, the acidity index (AI) reported by Rebolleda et al., (Rebolleda et al., 2012) was found to be of the same order at different extraction temperatures $(1.6 \pm 0.1\%)$ oleic acid at 84°C and $1.3 \pm 0.1\%$ oleic acid at 56°C), and close to the lowest value of AI described in the literature for crude corn germ oils obtained by conventional extraction (from 1.5 to 4.0% oleic acid) (Johnson, 2004).

The AI values obtained by Rebolleda et al., (Rebolleda et al., 2012) agree with the results reported by List et al., (List et al., 1984) and Christianson et al., (Christianson et al., 1984) who compared the free fatty acid content of corn oil processing by expeller with those oils resulting of the SFE of corn germ obtaining lower values of acidity in the latter. Fractionation experiments will be explained later in this chapter since oils with higher quality and less acidity can be obtained by this procedure (Rubio-Rodríguez et al., 2012).

Fatty acids (EA)	R9, 56°C	R10, 84°C	R9-R10%	Crude oil%
Fatty actus (FA)	mg/g oil	mg/g oil	(g/100 g FA)	(g/100 g FA)*
C16:0	107 ± 1	106 ± 1	12.5 ± 0.3	11.1-12.8
C18:0	20 ± 1	19 ± 1	2.3 ± 0.1	1.4-2.2
C18:1 n-9	245 ± 1	234 ± 1	28.1 ± 0.4	22.6-36.1
C18:1 n-7	5 ± 1	5 ± 1	0.6 ± 0.1	-
C18:2 n-6	475 ± 2	457 ± 2	54.8 ± 0.8	49.0-61.9
C18:3 n-3	8 ± 1	8 ± 1	0.9 ± 0.1	0.4-1.6
C20:0	4 ± 1	3 ± 1	0.4 ± 0.1	0.0-0.2
C20:1 n-9	3 ± 1	3 ± 1	0.4 ± 0.1	-
SFA	131 ± 3	128 ± 3	15.2 ± 0.5	12.5-15.2
MSFA	253 ± 3	242 ± 3	29.0 ± 0.7	22.6-36.1
PUFA	483 ± 3	465 ± 3	55.7 ± 0.9	49.4-63.5
Total FA	867 ± 9	835 ± 9	100 ± 2	84.5-114.8
and the second se				

 Table 5. Fatty acid profile of corn germ oil extracted with SC-CO2 and comparison with crude oil

^{*}Johnson, 2004.

SFA: saturated fatty acids; MSFA: monosaturated fatty acids; PUFA: polyunsaturated fatty acids.

The composition of corn germ oil extracted with SC-CO₂ by Rebolleda et al., (Rebolleda et al., 2012), in terms of neutral lipids, is listed in Table 6. Triacylglyceride, free fatty acid and sterol content was nearly the same for the two temperatures studied. List et al., (List et al., 1984) reported values slightly lower for the unsaponificable content of SC-CO₂ extracted corn oil, in the range of 1.2-1.3 wt%, being sterols the majority.

Corn germ oil is rich in linoleic acid (Table 5), a polyunsaturated fatty acid, and therefore can suffer oxidation during processing. Lipids oxidation involves three different stages, initiation, propagation and termination, which rate depends on the substrates and reaction conditions. A common parameter to evaluate the oil oxidation is the Peroxide Value (PV) which indicates the total level of hydroperoxides in oil and is related to the primary compounds in the oxidation. Hydroperoxides are very unstable and decompose to different secondary compounds, especially at high temperatures. Therefore to assess the oxidative stability of SC-CO₂ extracted corn germ oil, the measurement of secondary oxidation compounds such as the anisidine value should be also considered. Unfortunately in the revised literature no data of the anisidine value or similar parameters related to the oxidation secondary products have been found for SC-CO₂ extracted corn germ oil and no comparison can be done with conventional corn germ oil extraction process.

Table 6. Neutral lipids profile of corn germ oil extracted with SC-CO2	

	% wt	in oil
Neutral lipids	R9, 56°C	R10, 84°C
Triacylglycerides (TAG)	95.1 ± 0.8	95.4 ± 0.4
Free fatty acids (FFA)	0.8 ± 0.2	1.0 ± 0.1
Sterols (St)	2.3 ± 0.4	2.2 ± 0.1
Others	1.7 ± 0.2	1.4 ± 0.2

	R9, 56°C	R10, 84°C
Peroxide Value (meq/kg oil)	25 ± 2	20 ± 1
DPPH (mmol BHT/kg oil)	22 ± 3	32 ± 3
Tocopherol content (ppm)	1082 ± 14	1397 ± 17
α-tocopherol (ppm)	71 ± 5	86 ± 2
β-tocopherol (ppm)	65 ± 1	64 ± 1
γ-tocopherol (ppm)	945 ± 8	1232 ± 11
δ-tocopherol (ppm)	nd	15 ± 3

Table 7. Peroxide value, antioxidant capacity (DPPH) and tocopherol content of SC-CO2 extracted oils

nd: not detected.

Temperature is one of the factors that could affect the initiation stage and, therefore, the oxidation of the oil (Anna, 2002), although results reported by Rebolleda et al., (Rebolleda et al., 2012) presented a slight decrease in the peroxide content of SC corn germ oil obtained at the highest extraction temperature studied (see Table 7). In any case, it must be highlighted that in the work carried out by Rebolleda et al., (Rebolleda et al., 2012), considerable high values of the peroxide content have been obtained (Table 7), especially when comparing with other values reported in literature for SC-CO₂ extraction of dry, freshly milled corn germ, which rarely exceed 0.5 meq/kg (List et al., 1989).

According to the specifications for corn oil by the Committee on Food Chemicals Codes of the National Academy of Sciences/National Research Council PV should be not more than 10 meq·kg⁻¹ (NRC, 2003). Tocopherols have been described as effective antioxidants present in vegetable oils. Therefore, its presence contributes favorably to the conservation and the quality of oils. The tocopherol content of the oil obtained by Rebolleda et al., (Rebolleda et al., 2012) (Table 7) is even higher than the values reported for hexane prepress extracted wet corn germ (1000 µg/g) (List et al., 1984). The tocopherol profile of SC-CO₂ extracted corn germ oil obtained by Rebolleda et al., (Rebolleda et al., 2012) is also listed in Table 7; they found that γ -tocopherol was the major one followed by α -tocopherol and β -tocopherol, along with a small amount of δ -tocopherol. In Table 7, it can also be seen that, when extraction temperature increases from 56°C to 84°C, the yield percentage of tocopherols extracted increase by almost 30%. The solvent power of carbon dioxide decreases with increasing temperature at constant pressure because of its decreasing density.

Therefore, the higher tocopherol content found at the highest temperature might be due to an increase of vapor pressure of the tocopherols with temperature. No solubility data of tocopherols in CO₂ were found in the literature for the pressure and temperature range presented in Table 4. However, Chrastil (Chrastil, 1982) reported solubility data of α tocopherol in the pressure and temperature range from100 to 250 bar and from 40 to 80°C, respectively. These data show that α -tocopherol solubility decreases with temperature at low operating pressure (lower than 120-130 bar) while increases with temperature at operating pressure higher than 130-135 bar (Figure 9). Although no solubility data for other tocopherol isomers have been found in the literature, similar solubility behavior can be expected. Therefore, at pressure higher than 130 bar an increase in solubility is expected with increasing temperature. The increase of the tocopherol content with temperature can explain the

70

reduction of the peroxide content and the increase of the antioxidant level of oils (see value of DPPH in Table 7) at the highest extraction temperature. However, different results have been also found in the literature about tocopherol content in SC-CO₂ extracted corn germ oil, e.g.: List et al., (List et al., 1984) obtained SC-CO₂ extracted corn germ oil with less amount of tocopherols when temperature was increased from 70 to 90°C at a constant pressure of 82.7 MPa (ranged from 1.840 to 1.180 μ g/g at 70°C and 90°C respectively), although these authors concluded that the reason for this drop was unclear.



Figure 9. Solubility data of α -tocopherol in carbon dioxide reported by Chrastil (Chrastil, 1982). 40°C, \Box 60°C, Δ 80°C.

List and Friedrich (List and Friedrich, 1989) and Rebolleda et al., (Rebolleda et al., 2012) found that $SC-CO_2$ extracted seed oils undergo rapid deterioration. Many factors can affect the oxidative stability of oils, including the different pre-treatments before oil extraction (corn steeping, germ separation, germ storage...), fatty acid composition, concentration and stability of antioxidants and the presence of prooxidant compounds (FFA, lipid peroxides, or prooxidant metals) (List and Friedrich, 1989, Winkler-Moser and Breyer, 2011).

List and Friedrich (List and Friedrich, 1989) suggested that the absence of phospholipids in the SC-CO₂ extracted corn germ oil could cause a decrease in the positive synergistic effects of tocopherols with phospholipids.

This synergic effect of phospholipids and tocopherols has been described in the literature as responsible for the oxidative stability in vegetable oils. In contrast, Calvo et al., (Calvo et al., 1994) suggested that oil instability may be related to the trace amounts of oxygen in the CO_2 . In this case, during SC-CO₂ extraction of corn germ oil, oxidation would take place without mass transfer limitation since solvent and oil are in the same phase.

Sara Rebolleda, María Teresa Sanz, Sagrario Beltrán et al.

SUPERCRITICAL FLUID FRACTIONATION

Oils extracted by conventional solvents require refining processes to remove phospholipids, free fatty acids and oxidation products. Solubility of phospholipids and glycolipids in CO_2 is very low and therefore it is possible to eliminate the degumming step in the refining process, which is an advantage compared with the conventional process (Eggers, 1996). Another important advantage of SFE is that oil fractionation can be carried out under supercritical conditions since solubility of triacylglycerides in CO_2 below 16 MPa is fairly low, leading to save energy (Eggers, 1996). As an example, solubility data of trilinolein and linoleic acid, as the major fatty acid compound, have been represented in Figure 10 where higher values of solubility are observed for the FFA than for the corresponding triglyceride in SC-CO₂. Therefore, a simple process to obtain refined corn germ oil after SCCO₂ extraction can be proposed (Figure 11).



Figure 10. Solubility data of trilinolein (---- correlation of solubility data obtained by Chrastil (Chrastil, 1982)) and linoleic acid (Güçlü-Üstündağ and Temelli, 2000).

Rebolleda et al., (Rebolleda et al., 2012) reported results of an extraction-fractionation process in two separators installed in series, aiming to improve the oil quality. They maintained the first separator at 10.0 MPa and 40°C. These authors evaluated the influence of extraction pressure and temperature on the tocopherol content of the oil and on their stability to oxidation in different experiments (runs 11-13), which conditions are here reported in Table 8.



Figure 11. Obtaining refine corn germ oil by using supercritical CO₂ technology.

Table 8. Experimental conditions in the SFE of corn germ oil with SC-CO ₂
to evaluate the effect of fractionation after extraction on oil quality
(10 MPa and 40°C in the first separator)

Run	P (MPa)	T (°C)	Solvent flow rate, (kg CO ₂ /h)
R11	50 ± 2	35 ± 2	11.0 ± 0.7
R12	48 ± 2	86 ± 2	7.0 ± 0.9
R13	26 ± 2	85 ± 2	9.0 ± 0.9

These authors found that the fraction collected in the first separator was mostly oil whereas the fraction recovered in the second one was mostly an aqueous emulsion, what is probably due to the low solubility of triacylglycerides in CO₂ below 16 MPa (see Figure 10). The relative oxidative stability of the two fractions determined by the Rancimat test reported, show that the induction time in the oil fraction recovered in separator 1 (1.9 ± 0.3 h) was significantly higher than the induction time determined in the fraction recovered in separator 2 (0.5 ± 0.3 h). Water and free fatty acids, with a high tendency to oxidation, were mostly removed in separator 2, which agrees with the higher stability found in the oil fraction recovered in the first separator by Rebolleda et al., (Rebolleda et al., 2012) was lower than values reported by other authors (Winkler-Moser and Breyer, 2011) for hexane Soxhlet extracts of corn germ (3.91 ± 0.4 h). As previously explained, this could be because SC-CO₂ extracted oils suffer rapid deterioration. Another important result found by Rebolleda et al., (Rebolleda et al., 2012) is that total concentration of tocopherols in the oil fraction recovered in separator 1 is

almost double when temperature increased form 35°C to 86°C at a constant extraction pressure of around 49.0 MPa (see Figure 12).

This result agrees with the values of tocopherol content presented in Table 7 where the effect of extraction temperature on the oil quality was analyzed. However, different results are also found in the literature. Wilp and Eggers (Wilp and Eggers, 1991) found that an increase in extraction temperature from 50°C to 80°C at constant extraction pressure of 50 MPa results in a decrease in tocopherols concentration in the first separator of more than 30%. On the contrary, the effect of extraction pressure, at constant extraction temperature on tocopherol content has been found negligible in different studies.

Rebolleda et al., (Rebolleda et al., 2012), at constant temperature, around 85°C, when the extraction pressure is increased from 26 MPa to 49 MPa, found that tocopherol concentration in the oil fraction recovered in the first separator was only slightly increased (less than by 10%). Similar results were obtained by Wilp and Eggers (Wilp and Eggers, 1991) for a extraction pressure range from 50 MPa to 32 MPa at a constant extraction temperature of 50°C. Based on these results, it could be concluded that the effect of temperature on the extraction of tocopherols is more notable than the effect of pressure.



Figure 12. Tocopherol concentration of fractionated oils in the first separator (10.0 MPa and 40°C) at different extraction temperatures (a) and pressures (b).

CONCLUSION

Supercritical carbon dioxide extraction has been presented as an alternative procedure to obtain oil of high quality from milled corn germ. Typical extraction curves of vegetables oils, including corn germ oil, present two different sections.

In the first stages of the extraction, the process may be controlled by the solubility of the oil in SC-CO₂ since the initial slope calculated for SC-CO₂ extracted corn germ oil is of the same order as the solubility of vegetable oils in CO₂. While, a second section of the extraction curve is determined by the diffusional resistance in the solid matrix.

SC-CO₂ extraction of corn germ permits to obtain oil with good quality with lower values of acidity than for oils obtained by conventional extraction. High tocopherol content and therefore high antioxidant activity has been reported in the literature for SC-CO₂ extracted corn germ oil. However, SC-CO₂ extracted seed oils undergo rapid deterioration. Nevertheless, fractionation of the extracted oil in two separators installed in series allows separating the co-extracted water and seems to improve the oil stability against oxidation.

It should also be highlighted that the defatted corn germ meal obtained after removing the oil with $SC-CO_2$ could be used in food formulation due to its good stability and protein quality.

REFERENCES

- Anna, K. (2002). Lipid Oxidation in Food Systems. Chemical and Functional Properties of Food Lipids. A. K. Zdzisław E. Sikorski. CRC Press.
- AOAC (1995). Fatty acids in encapsulated fish oils and fish oil methyl and ethyl esters. Gas chomatographic method. AOAC.
- AWARENET (2004). Handbook for the prevention and minimization of waste and valorization of by-products in european agro-food industries.
- Beckman, E. J. (2012). Supercritical and near-critical CO2 processing. Green Technologies in Food Production and Processing. J. I. Boye and Y. Arcand. Springer, New York: 239-271.
- Brand-Williams, W., M. E. Cuvelier and C. Berset (1995). Use of a free radical method to evaluate antioxidant activity. *Food Science and Technology* 28, 25-30.
- Brunner, G. (1994). Gas extraction: an introduction to fundamentals of supercritical fluids and the application to separation processes. Steinkopff, New York (US).
- Calvo, L., M. Cocero and J. Díez (1994). Oxidative stability of sunflower oil extracted with supercritical carbon dioxide. *Journal of the American Oil Chemists' Society* 71, 1251-1254.
- Chrastil, J. (1982). Solubility of solids and liquids in supercritical gases. *The Journal of Physical Chemistry* 86, 3016-3021.
- Christianson, D. D. and J. P. Friedrich (1985). *Production of food grade corn germ product* by supercritical fluid extraction. US.
- Christianson, D. D., J. P. Friedrich, G. R. List, K. Warner, E. B. Bagley, A. C. Stringfellow, and G. E. Inglett (1984). Supercritical Fluid Extraction of dry-milled corn germ with carbon dioxide. *Journal of Food Science* 49, 229-232.
- del Valle, J. M., J. C. de la Fuente and E. Uquiche (2012). A refined equation for predicting the solubility of vegetable oils in high-pressure CO2. *The Journal of Supercritical Fluids* 67, 60-70.
- DeLine, K. E., D. L. Claycamp, D. Fetherston, and R. T. Marentis (2009). Carbon Dioxide extraction of corn germ oil from corn germ. US, MOR Technology, LLC. US 7,612,220 B2: 1-22.
- DeLine, K. E., D. L. Claycamp, D. Fetherston, and R. T. Marentis (2013). *Carbon Dioxide* corn germ oil extraction system. US, MOR Supercritica LLLC. US 8,603,328 B2: 1-21.
- Dijkstra, A. J. (2013). Edible Oil Processing from a Patent Perspective. Springer, New York.

Free ebooks ==> www.Ebook777.com

Sara Rebolleda, María Teresa Sanz, Sagrario Beltrán et al.

- Dijkstra, A. J., W. W. Christie and G. Knothe (2007). *Analysis. The Lipid Handbook.* F. D. Gunstone, J. L. Harwood and A. J. Dijkstra. CRC Press, Boca Raton, FL, US.
- Dijkstra, A. J. and J. C. Segers (2007). Production and refining of oils and fats. *The Lipid Handbook*. F. D. Gunstone, J. L. Harwood and A. J. Dijkstra. CRC Press, Boca Raton, FL, US.
- Eggers, R. (1996). Supercritical Fluid Extraction (SFE) of oilseeds/lipids in natural products. Supercritical Fluid Technology in oil and lipid chemistry. J. W. King and G. R. List. AOCS Press, Champaign, Illinois, US: 35-64.
- Favati, F., J. W. King and G. R. List (1996). Effect of supercritical fluid extraction on residual meals and protein functionality. Supercritical Fluid Technology in oil and lipid chemistry. J. W. King and G. R. List. AOCS Press, Champaign, Illinois, US: 267-287.
- Friedrich, J. and E. Pryde (1984). Supercritical CO2 extraction of lipid-bearing materials and characterization of the products. *Journal of the American Oil Chemists' Society* 61, 223-228.
- Güçlü-Üstündağ, Ö. and F. Temelli (2000). Correlating the solubility behavior of fatty acids, mono-, di-, and triglycerides, and fatty acid esters in supercritical carbon dioxide. *Industrial and Engineering Chemistry Research* 39, 4756-4766.
- Gunstone, F. D. (2002). Vegetable oils in food technology: composition, properties and uses. Blackwell.
- Huang, Z., X.-h. Shi and W.-j. Jiang (2012). Theoretical models for supercritical fluid extraction. *Journal of Chromatography A* 1250, 2-26.
- Johnson, L. A. (2004). Corn: the mayor cereal of the Americas. Handbook of cereal science and technology. K. Kulp and J. G. Ponte. Marcel Dekker, New York.
- Kamal-Eldin, A. and L. A. Appelqvist (1996). The chemistry and antioxidant properties of tocopherols and tocotrienols. *Lipids* 31, 671-701.
- List, G. R. and J. P. Friedrich (1989). Oxidative stability of seed oils extracted with supercritical carbon dioxide. *Journal of the American Oil Chemists' Society* 66, 98-101.
- List, G. R., J. P. Friedrich and D. D. Christianson (1984). Properties and processing of corn oils obtained by extraction with supercritical carbon dioxide. *Journal of the American Oil Chemists' Society* 61, 1849-1851.
- List, G. R., J. P. Friedrich and J. W. King (1989). Supercritical CO₂ extraction and processing of oilseeds. *Oil Mill Gazetteer December*, 28-34.
- Marentis, R. T. (2013). *Method of supercritical fluid fractionation of oil seed extraction materials*. US, MOR supercritical, LLC. US 8,557,318 B2: 1-13.
- Matthäus, B. (2012). Oil Technology. Technological Innovations in Major World Oil Crops. Volume 2: Perspectives. S. K. Gupta. Springer Science + Business Media, New York. 2. Perspectives: 23-92.
- Natex. (2015). "Supercritical fluid Industrial extraction plants." Retrieved 25/02/2015, from http://www.natex.at/indusextractionplants.html.
- NRC (2003). *National Research Council. Food Chemicals Codex*. National Academy Press, Washington D.C.
- Oliveira, E. L. G., A. J. D. Silvestre and C. M. Silva (2011). Review of kinetic models for supercritical fluid extraction. *Chemical Engineering Research and Design* 89, 1104-1117.
- Rai, A., K. D. Punase, B. Mohanty, and R. Bhargava (2014). Evaluation of models for supercritical fluid extraction. *International Journal of Heat and Mass Transfer* 72, 274-287.

- Rebolleda, S., S. Beltrán, M. T. Sanz, and M. L. González-Sanjosé (2014). Supercritical fluid extraction of wheat bran oil. Study of extraction yield and oil quality. *European Journal* of Lipid Science and Technology 116, 319-327.
- Rebolleda, S., S. Beltrán, M. T. Sanz, M. L. González-Sanjosé, and Á. G. Solaesa (2013). Extraction of alkylresorcinols from wheat bran with supercritical CO₂. *Journal of Food Engineering* 119, 814-821.
- Rebolleda, S., N. Rubio, S. Beltrán, M. T. Sanz, and M. L. González-Sanjosé (2012). Supercritical fluid extraction of corn germ oil: Study of the influence of process parameters on the extraction yield and oil quality. *The Journal of Supercritical Fluids* 72 270-277.
- Reverchon, E. and C. Marrone (2001). Modeling and simulation of the supercritical CO2 extraction of vegetable oils. *The Journal of Supercritical Fluids* 19, 161-175.
- Rónyai, E., B. Simándi, S. Tömösközi, A. Deák, L. Vigh, and Z. Weinbrenner (1998). Supercritical fluid extraction of corn germ with carbon dioxide-ethyl alcohol mixture. *The Journal of Supercritical Fluids* 14, 75-81.
- Rubio-Rodríguez, N., S. M. de Diego, S. Beltrán, I. Jaime, M. T. Sanz, and J. Rovira (2008). Supercritical fluid extraction of the omega-3 rich oil contained in hake (Merluccius capensis-Merluccius paradoxus) by-products: Study of the influence of process parameters on the extraction yield and oil quality. *The Journal of Supercritical Fluids* 47, 215-226.
- Rubio-Rodríguez, N., S. M. de Diego, S. Beltrán, I. Jaime, M. T. Sanz, and J. Rovira (2012). Supercritical fluid extraction of fish oil from fish by-products: A comparison with other extraction methods. *Journal of Food Engineering* 109, 238-248.
- Salgın, U., O. Döker and A. Calımlı (2006). Extraction of sunflower oil with supercritical CO2: Experiments and modeling. *The Journal of Supercritical Fluids* 38, 326-331.
- Schaefer, A., T. Küchler, T. J. Simat, and H. Steinhart (2003). Migration of lubricants from food packagings: Screening for lipid classes and quantitative estimation using normalphase liquid chromatographic separation with evaporative light scattering detection. *Journal of Chromatography A* 1017, 107-116.
- Soares, B. M. C., F. M. C. Gamarra, L. C. Paviani, L. A. G. Gonçalves, and F. A. Cabral (2007). Solubility of triacylglycerols in supercritical carbon dioxide. *The Journal of Supercritical Fluids* 43, 25-31.
- Sovová, H. (1994). Rate of the vegetable oil extraction with supercritical CO₂-I. Modelling of extraction curves. *Chemical Engineering Science* 49, 409-414.
- Sovová, H. (2005). Mathematical model for supercritical fluid extraction of natural products and extraction curve evaluation. *The Journal of Supercritical Fluids* 33, 35-52.
- Uhde. (2015). "Supercritical Fluids Technologies and Plants." Retrieved 25/ 02/2015, from http://www.uhde-hpt.com/fileadmin/documents/publications/05-UHPT_Extract.pdf.
- Vigh, L., B. Simandi and A. Deak (1993). Optimization of supercritical fluid extraction of corn germ oil in a multipurpose extractor. *Proceedings of the World Conference on Oilseed Technology and Utilization*, Champaigne, IL, AOCS press.
- Wilp, V. C. and R. Eggers (1991). Hochdruckextraktion mit mehrstufiger fraktionierender Separation zur schonenden Gewinnung von Keimölen mit hochverdichtetem Kohlendioxid. *Fat Science Technology* 93, 348-352.
- Winkler-Moser, J. K. and L. Breyer (2011). Composition and oxidative stability of crude oil extracts of corn germ and distillers grains. *Industrial Crops and Products* 33, 572-578.

Chapter 4

USES AND APPLICATIONS OF THE CORN OIL

Hakan Temur*

Atatürk University, Engineering Faculty, Department of Chemical Engineering, Erzurum, Turkey

ABSTRACT

Corn or maize oil is an important vegetable oil obtained from the germ part of the corn kernel. It is used in food, biofuel and some other important industrial applications for a long time. Corn oil use is increasing gradually as a result of its production as a co-product in the ethanol manufacturing. Its use in food and biofuel markets grows as its price reduced as a result of increasing production rates. Corn oil is used as a frying oil, salad oil, margarine and spread oil etc. in food markets. It is also used to produce biodiesel which is an alternative biofuel obtained from vegetable oils and animal fats. Biodiesel is compatible in terms of many fuel properties with petroleum diesel fuel. Corn oil has also some other important applications in chemical industry such as polymers, coatings and corrosion inhibitors etc. In this chapter, main uses and applications of corn oil will be elucidated in order to show its growing importance in related sectors.

Keywords: corn oil, biodiesel feedstocks, vegetable oil

INTRODUCTION

World corn production was about 38,065 million bushels in 2013-2014 marketing year. Corn is mainly used as an animal feed, a food component and an ethanol feedstock (Anonymous 2014). Corn oil is one of the co-products of the corn seeds. Its capacity depends on the corn production. Corn oil production and consumption data show that the largest corn oil producers and consumers are United States and China respectively. The recent increase in

^{*} Corresponding Author address: Contact address: Ataturk University, Engineering Faculty, Department of Chemical Engineering, 25240 Erzurum, Turkey. E-mail: htemur@atauni.edu.tr.

the ethanol production particularly in the United States results in an increase in corn oil production (Menon 2014).

Corn kernel contains about 4% oil mainly in its germ parts (Moreau 2005). Corn oil is produced as a co-product in industrial ethanol production processes. There are two methods of ethanol production from the corn seeds: Dry-grind and wet milling. In dry-grind process, which is the common one, corn grains are crushed by the high speed rotating hammer tips into the fine particles with 3-5 mm diameter. After fermentation and distillation steps, corn oil is obtained by extracting the oil from the thin stillage portion of the distiller's dried grains with solubles (DDGS) (Anonymous 2011).

In wet milling process, which is more capital and energy intensive, wet mills separate the corn kernel into its components such as starch, fiber, gluten and germ. After germ is removed from the kernel, corn oil is extracted from the germ (Bothast & Schlicer 2005). In both processes, corn oil containing intermediate products are first pressed and then corn oil is extracted with using hexane or 2-methylpentane (isohexane) as a solvent (Anonymous 2006). Some of the physical properties of corn oil are given in Table 1.

Property	Value
Iodine value	127-133
Saponification number	187-193
Specific gravity 25°C/25°C	0.919
Viscosity, 40°C	30.80 cP
Surface tension, 25°C	34.80 dyn/cm
Thermal conductivity, 130°C	$4.2017 \times 10^{-5} \text{ J/s/cm}^{2/\circ}\text{C}$
Unsaponifiables	1-2%
Melting point	-11 to -8 C
Smoke point	230 to 238°C
Flash point	323 to 338°C
Cloud point	-14 to -11°C

Table 1. Some chemical and physical properties of corn oil (Gunstone 2002).

Corn oil is used in industry mainly in two applications: As a food and biofuel feedstock. Some other notable uses are soap making, rustproofing, medicine and insecticide production. Some important uses of corn oil are explained in detail as following.

FOOD USES OF CORN OIL

Corn oil has different uses in food sector such as cooking or salad oil, margarine oil, blends of butter (40%) and corn oil margarine (60%), mayonnaise and emulsion type salad dressings as either consumer or institutional products. It is also used as an oil ingredient in many packaged and restaurant foods, including spaghetti sauce, potato chips and snack foods, French fries and breaded fried foods, baking mixes, frosting, whipped toppings, crumb coatings for meat and poultry, and baked goods. Corn oil is often used without blending with other oils except for packaged foods (Anonymous 2006).

Salad and Cooking Oils

Approximately half of the food use of corn oil is as salad/cooking oil. It is preferred as a salad/cooking oil because of its mild flavor, good nutritional properties, highly digestible structure and good oxidative stability (because of low linolenate levels). It is also a better source of essential fatty acids (Meurant 1983, Dupont 1990). Corn oil is used as 100% corn oil or in blends with other vegetable oils.

Deep fat frying is one of the most common cooking methods because fried foods have desirable flavor, color and crispy texture (Boskou et al., 2006). Food is immersed in hot oil at high temperatures (150-190°C) in the cooking container where heat and mass transfer of oil, food and air are occurred simultaneously. As a result of favorable conditions, chemical reactions such as oxidation, hydrolysis and polymerization take place during the deep frying. Many factors such as frying time, temperature, composition of food, type of fryer and antioxidant content, quality of frying oil are significant parameters on the quality criteria of fried food (Choe & Min 2007). It is reported that (Warner & Knowlton 1997) oxidation rate of the oil increased as the unsaturated fatty acid content of the oil increased. That is why corn oil with less unsaturated fatty acid is better frying oil than soybean and canola oils with more unsaturated fatty acids.

In 69% of the popular chain fast food restaurants in the United States, corn oil is used as frying oil in French fries while by 20% of the small businesses use it as fry oil. This is attributed that major users have large-scale cost effective corporate agreements with the suppliers; whereas minor users find it more expensive than other plant oils such as soybean oil (Jahren & Schubert 2009). In addition, fast food companies have changed potato frying applications from animal fat to non-hydrogenated corn oil because of the *trans* fatty acid concerns (Anonymous 2006).

Vegetable oils and fats incline to form polar compounds and polymerized triglycerides. Corn oil has one of the fastest rates of production of polar compounds and polymerized triglycerides among the common oils at 190°C temperature as a result of its higher polyunsaturated fatty acid content (Takeoka et al., 1997).

Margarines and Spreads

Margarine is mainly classified into two types: stick and tub margarines. Stick margarine is formulated to approximate the butter properties. It is a little bit softer than butter under the ordinary storage conditions and can melt in the mouth. Tub margarine contains higher unsaturated fatty acids and easily spreadable at lower temperatures (McGee 2004).

Corn oil use is expanded over the years especially after the invention of that corn oil has a favorable effect on serum cholesterol in 1950s. It became the highest consumed oil among the polyunsaturates of all important margarines (Anonymous 2006). *Trans* fatty acid formation during the hydrogenation process turned out to be a major concern in margarine production. Several methods were suggested in order to avoid *trans* fatty acid formation (List et al., 1995; Sundram et al., 1999). Corn oil can be interesterified with fully hydrogenated vegetable oil to produce *trans* free margarines (Anonymous 2006). The corn oil margarines and other food products containing corn oil have a wide variety of applications in food industry (Gunstone 2002). Hakan Temur

Mayonnaise and Salad Dressings

Mayonnaise is defined as "an emulsion of oil droplets suspended in a base composed of egg yolk, lemon juice or vinegar, which provides both flavor and stabilizing particles and carbohydrates" (McGee 2004). Varieties of oils are used in mayonnaise and it can contain vegetable oil up to 80%. The vegetable oil used in mayonnaise purveys a medium for the rest of the ingredients. The pH of the mayonnaise is 3.6 to 4.0 because of the acetic acid. It contains 9-11% salt and 7-10% sugar.

Salad dressings have similar composition with mayonnaise except for finished product contains 30% vegetable oil. pH of the salad dressing is in the range of 3.2 to 3.9 as a result of acetic acid content. The aqueous phase contains 3.0 to 4.0% salt and 20 to 30% sugar (Jay 2000). Use of corn oil in mayonnaise improved sensory characteristics such as color, taste, texture and flavor. Corn oil use in the mayonnaise also provides good emulsion stability (Rasool 2013).

BIODIESEL FEEDSTOCK

Biodiesel Production

Biodiesel is an alternative diesel fuel obtained from vegetable oils and animal fats. The term of biodiesel is commonly referred for mono-alkyl esters obtained from vegetable oils and animal fats via transesterification even though it implies other lipid products having a potential to be used as a biofuel. Other plant oil based biofuels can be obtained via different methods such as direct use, blending, microemulsions and pyrolisis (Ma & Hanna 1999). Mono-alkyl esters and other liquid biomasses are used as a fuel usually after a blending step with petroleum diesel fuel in order to improve its operational characteristics such as viscosity and cold flow properties. Biodiesel, as a renewable diesel fuel, has mild exhaust emission characteristics and faster biodegradability along with a lubricating effect in diesel engines.

Fatty acid structure (composition and unsaturation degree) of vegetable oils used as feedstocks considerably affect the physical properties of biodiesel such as viscosity, cold flow properties, oxidation stability and exhaust emission profile (Knothe 2008). Even, there are many successful attempts to predict the biodiesel properties from the fatty acid composition of the plant oils. Fatty acid distribution of some common vegetable oils is given in Table 2. Higher saturated fatty acid based biodiesel has higher viscosity, cold flow temperatures, oxidation stability and calorific value while higher unsaturated fatty acid based biodiesel shows a combination of improved fuel properties as a whole (Knothe 2005; Ramos 2009; Kumar 2013).

Transesterification of vegetable oils into the methyl ester fatty acid mixture (alcholysis) is achieved at similar reaction conditions for almost all oil feedstocks. Biodiesel is produced with the reaction of a vegetable oil with an alcohol (mostly methanol) in the presence of a catalyst (a strong base such as NaOH or KOH) at about 50°C temperature in a short time (shorter than 1 hour). The overall chemical reaction is presented in Scheme 1.

Vegetable oils with higher amount of free fatty acid and water are needed to first introduce a neutralization step in order to prevent saponification between the base catalyst

and free fatty acids. Another method is to use acidic catalyst such as sulfuric acid in the transesterification. When the free fatty acid content of the triglycerides is high, use of acid catalyst is the best way of transesterification to avoid undesired neutralization of the base catalyst (Ma et al., 1998). Figure 1 shows the process steps in the biodiesel production via basic transesterification. Biodiesel is used after blending with a higher percentage of petroleum based diesel fuel in practice. The percentage of biodiesel is designated by terms such as B5 or B20 which means 5% or 20% of the blend is biodiesel respectively.

C atoms:	8.0	10.0	12.0	14.0	16.0	16.1	18.0	10.1	18.2	18.2	20.0	20.1	22.0	22.1	24.0
	0.0	10.0	12.0	14.0	10.0	10.1	10.0	10.1	10.2	10.5	20.0	20.1	22.0	22.1	24.0
double bond															
Canola oil	-	-	-	0.1	4.0	0.3	1.8	60.9	21.0	8.8	0.7	1.0	0.3	0.7	0.2
Castor oilb	-	-	-	-	2.0	-	1.0	7.0	3.0	-	-	-	-	-	-
Coconut oil	7.1	6.0	47.1	18.5	9.1	-	2.8	6.8	1.9	0.1	0.1	-	-	-	-
Corn oil	-	-	-	0.1	10.9	0.2	2.0	25.4	59.6	1.2	0.4	-	0.1	-	-
Cottonseed	-	-	0.1	0.7	21.6	0.6	2.6	18.6	54.4	0.7	0.3	-	0.2	-	-
oil															
Linseed oil	-	-	-	-	6.0	-	4.0	22.0	16.0	52.0	0.5	-	-	-	-
Olive oil	-	-	-	-	9.0	0.6	2.7	80.3	6.3	0.7	0.4	-	-	-	-
Palm oil	-	-	0.1	1.0	44.4	0.2	4.1	39.3	10.0	0.4	0.3	-	0.1	-	-
Peanut oil	-	-	-	0.1	11.1	0.2	2.4	46.7	32.0	-	0.1	0.1	-	-	-
Rapeseed oil	-	-	-	0.1	3.8	0.3	1.2	18.5	14.5	11.0	0.7	6.6	0.5	41.1	1.0
Safflower	-	-	-	0.1	6.8	0.1	2.3	12.0	77.7	0.4	0.3	0.1	0.2	-	-
oil															
Soybean oil	-	-	-	0.1	10.6	0.1	4.0	23.3	53.7	7.6	0.3	-	0.3	-	-
Sunflower	-	-	-	0.1	7.0	0.1	4.5	18.7	67.5	0.8	0.4	0.1	0.7	-	-
oil															

 Table 2. Fatty acid composition of some common plant oils^a (Kirk-Othmer 1991)

^a Some oil compositions may not add to 100% due to the presence of minor fatty acids.

^bContains 87% OH-bearing ricinoleic acid (C18:1).

Triglyceride Methanol

Mixture of fatty esters Glycerin

Scheme 1. Transesterification (methanolysis) of vegetable oils.

Enzymes can also be used as catalysts in transesterification. The enzymatic route can provide advantages such as elimination of organic solvents, chemical catalyst and waste water (Ranganathan 2008). On the other hand, it has several disadvantages such as high enzyme cost, low yield and longer reaction time.

Corn oil is used as a feedstock in biodiesel production via transesterification either as a raw vegetable oil or as waste frying oil. Advantages and specifications of biodiesel obtained from corn oil are examined in detail in the following subtopics.



Figure 1. Process flow diagram for (corn oil based) biodiesel production.

Reliability of Corn Oil as a Biodiesel Feedstock

Corn oil has been ignored as a biodiesel feedstock formerly because of its higher cost and value as edible oil. However, developments in the ethanol production processes such as dry grinding have made way for higher amount of corn oil production. Corn oil is the fastest growing biodiesel feedstock in biodiesel production in the United States (Kotrba 2014). Ethanol producers began to utilize corn oil as a biodiesel feedstock onsite biodiesel plants in order to take advantage of lower operational and transportation costs. Increases in ethanol production made the corn oil a reliable feedstock in biodiesel production.

Transesterification Kinetics

Transesterification of triglycerides has a stepwise reversible reaction route in which diglycerides and monoglycerides are intermediates (Scheme 2):

Free ebooks ==> www.Ebook777.com

Uses and Applications of the Corn Oil



Scheme 2. Stepwise transesterification of triglycerides.

An excess alcohol makes the forward reaction pseudo-first order whereas reverse reaction second order in the case of corn oil (Meher et al., 2006). When corn oil is transesterified in a pressurized batch reactor in the presence of sodium methoxide and methanol, higher conversion can be obtained. Kinetic constants of the stepwise reactions are increased in the direction of the progressing steps of the transesterification (Velazquez 2007).

Biodiesel production from corn oil can also be carried out in enzymatic catalysis. In a study using immobilized lipase enzyme (Novozym 435) as a catalyst, it was reported that 81.3% fatty acid methyl ester content was obtained at 15% enzyme load, 60°C temperatures and 10 MPa pressure in 4 hours (Ciftci & Temelli 2013). Corn oil was also transesterified with methanol by injecting it into a supercritical carbon dioxide stream in the presence of immobilized lipase enzyme. Fatty acid methyl ester yield was observed as greater than 98% (Meher et al., 2006).

Cold Flow Temperatures

Behavior of biodiesel from corn oil at lower ambient temperatures is specified by cloud point, pour point and cold filter plugging point temperatures. Cloud point is the temperature by which liquid material begins cloudy since the saturated esters become crystallized. If the temperature decreases more, it will stop flowing. The lowest temperature it can keep flowing is called pour point. Cold filter plugging point gives a more practical measure in terms of cold flow properties of liquid fuels. It is defined as the highest temperature at which a given amount of sample cannot pass through a standard filter in a settled time (Knothe et al., 2005). Cold flow properties of biodiesel samples obtained from corn oil were seen acceptable in terms of common biodiesel standards (Rasimoglu & Temur 2014). In this study, the cloud point of the B100 corn oil based biodiesel was reported to be -4°C while the pour point and cold filter plugging point to be -10°C and -12, respectively.

Kinematic Viscosity

Kinematic viscosity of corn oil biodiesel is strongly affected by the structure of fatty acids. Fatty acid properties such as chain length, position, double bonds (double bond location, number, and nature) have effects on the kinematic viscosity of biodiesel. Biodiesel products obtained from shorter fatty acids and longer alcohol moieties have lower viscosity than those have longer fatty acids and shorter alcohol moieties (Knothe & Stidley 2005; Knothe & Stidley 2007). Allen et al. (1999) developed an empirical model to predict the

viscosity of 15 biodiesel samples including corn oil based one from the fatty acid contents of the parent oils.

Engine Performance and Emission Characteristics

Biodiesel obtained from corn oil has been reported as having improved engine performance (Kumar & Kumar 2013). Emission characteristics of biodiesel-diesel fuel mixtures enhance as biodiesel ratio in the mixture increases (Gopinath & Suresh 2015). On the other hand, fuel consumption increased by about 10% while maximum engine power slightly decreased in the case of corn oil biodiesel (Pullen and Saeed, 2014). Tesfa at al. (2013) pointed out that there is no significant difference between the engine performances of different vegetable oils.

A Valuable Co-product: Glycerol

As a result of transesterification process, a mixture of fatty acid methyl esters (biodiesel) and glycerol are obtained. These products are in different phases and can be separated by gravitational settling (Figure 2).

Glycerol recovery plays an important role in biodiesel production in terms of making whole process more profitable. The glycerol phase separated from the biodiesel contains about 50% glycerol together with excess methanol, catalyst and soap and has a little economic value. The soaps formed during transesterification are decomposed again into free fatty acids and salts by adding acid (Gerpen 2005). Methanol is evaporated after free fatty acids are precipitated and separated. The glycerin purity can reach to 85% as a result of these purification steps.



Figure 2. Corn oil biodiesel and glycerol separation (upper phase is biodiesel).

Alternative Production Routes

The quality specifications of biodiesel fuel obtained from corn oil is changeable as a result of its broader fatty acid distribution since corn oil has a wide range of saturated-unsaturated fatty acid contents from region to region as well as time to time (Lofland & Quasckenbush 1956; Ericson 1989; Gunstone 2011; Vesna et al., 2013).

Mendes (2011) produced biodiesel from corn oil with using ethanol as an alcohol reactant and NaOH as a catalyst. He pointed out that 0.4% NaOH in weight, 40°C reaction temperature and 90 minutes transesterification time is enough to obtain a higher biodiesel yield. One step alkali transesterification is sufficient to obtain higher biodiesel yield when corn oil contains lower amount of free fatty acid (Patil & Deng 2009).

Thermal behavior of corn oil-based biodiesel obtained by using methanol and ethanol was investigated by (Dantas et al., 2010). They reported that corn oil was thermally stable up to 336°C, methyl biodiesel up to 145°C, and ethyl biodiesel up to 169°C in nitrogen atmosphere.

Biodiesel production from corn oil can also be carried out via enzymatic route as well. When the Lipozyme TL IM-catalyzed transesterification was performed in the presence of absolute ethyl alcohol, 98.95% fatty acid ethyl ester conversion was obtained at 35°C in 12 hours reaction period (Mata et al., 2012).

As a result, the corn oil biodiesel satisfies ASTM D6751 standard and it can firmly claimed that corn oil is a rational biodiesel feedstock (Alptekin et al. 2014; Rasimoglu & Temur 2014).

OTHER USES

Polymerization

Corn oil together with other vegetable oils is seen nowadays as important natural raw materials used in many chemical industries because of their availability, biodegradability, cheapness and harmless health effects. Vegetable oil based polymers and composites have a wide range of applications such as paints and coatings, adhesives and biomedicine (Ligadas et al., 2013). A mixture of vegetable oils including corn oil is polymerized with styrene and divinilbenzene in the presence of different fillers and fibers such as organic clays, glass, hemp, flax, wood flour, corn stover, wheat straw and sugarcane bagasse. These polymeric materials show improved mechanical and thermophysical performance (Frederick at al., 2004; Lu 2009; Samarth & Mahanwar 2014).

Paints-coatings and Corrosion Inhibition

Poly(ether amide urethane) resin produced from corn oil was used as a corrosion protective and environmental friendly coating material. The coatings have good scratch harness, flexibility, corrosion performance and temperature resistance up to 250°C (Alam & Alandis 2014). Corn oil can be used as a coating material for eggs to extent their shelf life.

Corn oil coating retains internal quality of eggs by 3 weeks longer than that of non-coated eggs with minimized weight loss (0.8%) at 25°C storage temperature (Ryu et al., 2011). Similar results can be observed with other vegetable oils.

Some surfactants based on corn oil and monoethanolamine exhibited better corrosion inhibition effect for the mild steel in 1% NaCl solution saturated with CO_2 at 323 K temperature (Ismailov 2013).

Insulating Applications

Petroleum based insulating liquids are generally used as a dielectric and coolant in electrical power applications. They are not only toxicant but also inflammable as a result of higher temperatures and required costly fire protection precautions. Al-Eshaikh and Qureshi (2012) offered the food grade corn oil to overcome these problems because it is non-toxic and has a higher fire point. Furthermore, corn oil has higher breakdown strength and less degradation than mineral oils.

CONCLUSION

As a result of increasing corn production in worldwide, corn oil is expected to gain more importance as a vegetable oil. In addition to this, corn oil consumption as a food source is increasing since it has several benefits as a valuable food component in terms of health effects. It has lower blood cholesterol and higher polyunsaturated fatty acid content. It has a potential to be one of the most reliable biodiesel feedstock in the near future. It can be also used in many other industrial applications such as medication, chemical and soap production. Finally, it is an important lipid product already having many important uses and applications. More researches on corn oil applications are needed to clarify its benefits extensively.

REFERENCES

- Alam, M. and Alandis, N. M. (2014), Corn oil based poly(ether amide urethane) coating material-Synthesis, characterization and coating properties. *Industrial Crops and Products*, 57, 17-28.
- Al-Eshaikh, M. A. and Qureshi, M. I., (2012), Evaluation of food grade corn oil for electrical applications. *International Journal of Green Energy*, 9, 441-455.
- Alptekin, E., Canakci, M. and Sanli, H. (2014), Biodiesel production from vegetable oil and waste animal fats in a pilot plant. *Waste Management*, 34, 2146-2156.
- Anonymous, (2006), Corn oil, Corn Refiners Association, 5th edition, 21.
- Anonymous, (2011), DDGS User Handbook, 3rd edition, US Grains Council, (2), p. 4.
- Anonymous, (2014), FAS Grain: World Markets and Trade, USDA. Jan. 10, 2014, Marketing Year Oct. 1, 2013.
- Bothast, R. J. and Schlicher, M. A. (2005), Biotechnological processes for conversion of corn into ethanol. *Appl. Microbial Biotechnol*, 67, 19-25.

- Choe, E. and Min, D. B. (2007), Chemistry of Deep-fat Frying Oils. 00, (0), *Journal of Food Science*. R1-R10.
- Ciftci, O. N. and Temelli, F. (2013), Enzymatic conversion of corn oil into biodiesel in a batch supercritical carbondioxide reactor and kinetic modeling. *The journal of Supercritical Fluids*. 75, 172-180.
- Dantas, M. B., Conceio, M. M., Fernandes, Jr. V., Santos J., Rosenhaim N. A., Marques R., Santos A. L. B., I. M. G. and Souza, A. G. (2007), Thermal and Kinetic Study of Corn Biodiesel Obtained by the Methanol and Ethanol Routes, *Journal of Thermal Analysis* and Calorimetry. 87 (3), 835-839.
- Dupont J., White, P. J., Carpenter, M. P., Schaefer, E. J., Meydani, S. N., Elson, C. E., Woods, M. and Gorbach, S. L. (1990). Food uses and health effects of corn oil, *J. of Am. Coll. Nutr*, 9 (5), 438-470.
- Ericson D. R. (1989), Edible Fats and Oils Processing: Basic Principles and Modern Practices, *AOCS.World Conference Preceedings*, 312.
- Fangrui, M. and Hanna, M. A. (1999), Biodiesel production: A review. *Bioresource Technology*. 70, 1-15.
- Gerpen, J. V., (2005), Biodiesel processing and production. *Fuel Processing Technology*, 86, 1097-1107.
- Gopinath, V. and Suresh, P. (2015), Experimental study on the emission characteristics of a diesel engine using corn oil as fuel. *International Journal of Renewable Energy Research*, 5(1), 99-102.
- Gunstone, F. D. (2002). Vegetable Oils in Food Technology: Composition, Properties and Uses, CRC Press, 300.
- Ismailov, I. T., Abd El-Lateef, H. M., Abbasov, V. M. Mamedxanova, S. A., Yolchuyeva, U. C. and Slamanova, C. K. (2013), Anti-corrosion ability of some surfactants based on corn oil and monoethanolamine. *American Journal of Applied Chemistry*, 1(5), 79-86.
- Jay, M. J. (2000), Modern Food Microbiology. 6th Edition, An Aspen Publication, 167.
- Kumar U. S. and Kumar K. R. (2013), Performance, combusiton and cmission characteristics of corn oil blended with diesel. *IJETT*, 4(9), 3904-3908.
- Kirk, R. E., Othmer D. F., Kroschwitz J. I., and Howe-Grant M. 1991, Encyclopedia of Chemical Technology, 4th ed., Wiley.
- Knothe, G. (2005), Dependence of biodiesel fuel properties on the structure of fatty acid alkyl esters, *Fuel Processing Technology*, 86, 1059-1070.
- Knothe, G. and Steidley, K. R. (2005), Kinematic viscosity of biodiesel fuel components and related compounds. Influence of compound structure and comparasion to petrodiesel fuel components. *Fuel*, 84.1059-1065.
- Knothe, G., Gerpen, J. V. and Krahl, J. (2005), The Biodiesel Hanbook, AOCS Press, 83.
- Knothe, G. and Steidley, K. R. (2007), Kinematic viscosity of biodiesel components (fatty acid alkyl esters) and related compounds at low temperatures, *Fuel*. 86, 2560-2567.
- Knothe, G. (2008). "Designer" biodiesel: Optimizing fatty ester composition to improve fuel properties, *Energy & Fuels*. 22, 1358-1364.
- Kostik, V., Memeti, S. and Bauer, B. (2013), Fatty acid composition of edible oils and fats, *Journal of Hygienic Engineering and Design*, 4, 112-116.
- Kotrba, Ron. (2014), "Time has come today." Biodiesel Magazine. May 2014. Print.

Free ebooks ==> www.Ebook777.com

- Kumar, N., Varun and Chauhan, S. R. (2013), Performance and emission characteristics of biodiesel from different origins: A review. *Renewable and Sustainable Energy Reviews*, 21, 633-658.
- List, G. R., Friedrich, J. P. and Christianson, D. D. (1984), Properties and processing of corn oils obtained by extraction with supercritical carbon dioxide. J. Am. Oil Chem. Soc., 61, 1849-1851.
- Lligadas, G., Ronda, J. C. Galia, M. and Cadiz, V. (2013), Renewable polymeric materials from vegetable oils: a perspective. *Materials today*, 16, (9), 337-343.
- Lofland, H. B. and Quasckenbush, F. W. 1956, Disribution of fatty acids in corn oil, *The Journal of American Oil Chemists' Society*. 31, 412-413.
- Ma, F., Clement, L. D. and Hanna, M. A. (1998), The effect of catalyst, free fatty acids, and water on transesterification of beef tallow. *Transactions of the ASAE*. 41(5), 1261-1264.
- Mata T. M., Sousa, I. R. B. G. and Caetano, N. S. (2012), Biodiesel Production from Corn Oil via Enzymatic Catalysis with Ethanol. *Energy and Fuels*. 26 (5), 3034-3041.
- McGee, H. (2004), On Food and Cooking, Simon & Schuster, p. 38.
- Meher, L. C., Sagar, D. V. and Naik, S. N. (2006), Technical aspects of biodiesel production by transesterification: A review. *Renewable and Sustainable Reviews*, 10, 248-268.
- Mendes, A. M. (2011), Production of biodiesel from corn oil and ethanol by homogeneous alkali catalyzed transesterification, MSc. Thesis. Royal Institute of Technology, Stockholm, Sweden.
- Menon, S. (2014), The problem with corn oil biodiesel. Biodiesel Magazine, Feb. 2014. Print.
- Meurant,G. (1983). Lipids in Cereal Technology: A Series of Monographs. Academic Press Inc. 386.
- Moreau, R. A. (2005). Corn oil, in Bailey's Industrial Oil & Fat Products, Vol. 2, Edible Oil and Fat Products: Edible Oils (ed. F. Shahidi), 6th edn, John Wiley & Sons, Inc., Hoboken, NJ, 149-172.
- Patil, D. P. and Deng S. (2009), Optimization of biodiesel production from edible and nonedible vegetable oils. *Fuel* 88, 1302-1306.
- Pullen, J. and Saeed, K., (2014), Factors effecting biodiesel engine performance and exhaust emissions – Part II: Experimental study. *Energy*. 72, 17-34.
- Ramos, J. M., Fernandez, C. M., Casa, A., Rodriguez, L. and Perez, A. (2009). Influence of fatty acid composition of raw materials on biodiesel properties. *Bioresource Technology*. 100, 261-268.
- Ranganathan, S. V., Narasimhan, S. L. and Muthukmar, K. (2008), An overview of enzymatic production of biodiesel. *Bio source Technology*. 99, 3975-3981.
- Rasimoglu N. and Temur, H. (2014), Cold flow properties of biodiesel obtained from corn oil. *Energy*. 68, 57-60.
- Rasool, G., Hussain, S., Alam, Z. and Ibrahim, M. S., (2013), The effect of corn oil on the quality characteristics of mayonnaise, *American Journal of Food Science and Technology*. 1(3), 45-49.
- Ryu, K. N., No, H. K. and Prinyawiwatkul, W. (2011), Internal quality and shelf life of eggs coated with oils from different sources. *J. Food Sci.*, 76(5), 325-329.
- Samarth, N. B. and Mahanwar P. A. (2014), Modified vegetable oil based additives as a future polymeric material-Review. *Open Journal of Organic Polymer Materials*. 5, 1-22.
- Sundram, K., Perlman, D. and Hayes, K. C. (1999), Blends of palm fat and corn oil provide oxidation-resistant shortening for baking and frying. *US Patent* 5, 874, 117.

- Takeoka, G. R., Full, G. H. and Dao, L. T. (1997). Effect of heating on the characteristics and chemical composition of selected frying oils and fats. J. Agric. Food. Chem. 45, 3244-3249.
- Tesfa, B., Mishra, R., Gu, F. and Ball, A., (2013), Combustion and Performance Characteristics of CI Engine Running with Biodiesel. *Energy*, 51 (1) 101-115.
- Velazquez, J. M., (2007). *Conversion of corn oil to alkyl esters*, Master Thesis, Iowa State University.
- Wallenberger, F. T. and Weston, N. (2004), *Natural Fibers, Plastics and Composites,* Springer Science + Business Media, LLC.
- Warner K. and Knowlton S. (1997), Frying quality and oxidative stability of high-oleic corn oils. *J Am. Oil. Chem. Soc.* 74, 1317-22.
- Yu, L. and Larock, R., (2009). Novel polymeric materials from vegetable oils and vinyl monomers: Preparation, properties and applications. *ChemSusChem.* 2, (2), 136-147.

Chapter 5

CORN OIL AND AGING: INSIGHTS FROM BASIC RESEARCH

Hongwei Si^{*}

Department of Family and Consumer Sciences, Tennessee State University, Nashville, TN, US

ABSTRACT

Corn oil, having one of the highest polyunsaturated (PUFA) levels after sunflower, safflower, walnut and wheat germ oil, is the second leading vegetable oil consumed in the United States. While corn oil is well known in reducing blood lipids in humans, there are very few studies on the long-term beneficial effects of corn oil in animals. In the present chapter, the beneficial effects of corn oil on survival rate, bodyweight, food intake and other physiological parameters were described in aging mice and rats fed with corn oil long time. The possible mechanisms of these corn oil effects were also discussed. Long-term corn oil intake reverses aging-increased blood lipids and circulating pro-inflammatory cytokines as well as aging-damaged rotarod performance test and liver function, and thus increases longevity of aging mice. These health benefits of corn oil may result from the combinations of the high levels of PUFA, vitamin E and plant sterols. Therefore, corn oil, even at a high energy percentage, is a favorable replacement of animal fats in the human diet if the total energy intake is controlled.

Keywords: corn oil, aging, mice, rats, mechanism

INTRODUCTION

High saturated fat diets are well associated with obesity prevalence and the increased risk of cardiovascular disease, diabetes and cancer. Reducing saturated fats from the diet is recommended to eliminate Western diet-induced health problems.

^{*} Corresponding author: Hongwei Si. Department of Family and Consumer Sciences, Tennessee State University, Nashville, TN 37209, US. E-mail: hsi@tnstate.edu.

The common alternatives of animal (saturated) fats for humans are plant oil, including soybean oil, peanut oil and corn oil because of the high percentage of unsaturated fat acids.

Corn oil, a co-product starch manufacture named corn wet milling process, is majorly (85%) from corn germ although germ is only about 11.9% dry weight of conrn kernel (Figure 1). However, corn oil was increasing demanded after it was reported in reducing human serum cholesterol in 1950s. Annual production of crude corn oil increased 5.75% each year from 1974 to 2002. In 2005, crude corn oil production in US was 2.48 billion pounds and ten years later increased to 4.45 billion pounds in 2015, and nearly all of it is refined into high-quality oil for the food industry and direct use by consumers (USDA, 2015; Association, 2006). Corn oil is now the second importance only to soybean oil in the United States (USDA, 2015; Corn Refiners Association, 2006).

Corn oil is composed mainly (99% of the refined or 96% of the crude oil) of acylglycerols (mono-, di- and primarily tri-), and has 59% polyunsaturated (PUFA), 24% monounsaturated (MUFA) and 13% saturated fatty acid (SFA). The PUFA to SFA ratio (P/S) is about 4.6 (Table 1).

Corn oil has one of the highest PUFA levels after sunflower, safflower, walnut and wheat germ oil (Landers and Rathmann, 1981) (Table 2). The primary PUFA is linoleic acid (C18:2n-6), with a small amount of linolenic acid (C18:3n-3) giving a n-6/n-3 ratio of 83.

Corn oil contains a significant amount of ubiquinone and high amounts of gammatocopherols (vitamin E). Corn oil contains a significant amount of ubiquinone and high amounts of gamma-tocopherols (vitamin E) (Table 1). These high contents of PUFA, phytosterols and vitamin E may contribute to the health benefits of corn oil consumption.

The beneficial effects of PUFA have been extensively investigated, however, there are very few studies investigating the effects on human health with long-term corn oil consumption, particularly on the older population.

This is very important because corn oil is the second leading vegetable oil consumed and produced 4.45 billion pounds in 2015 in the United States (USDA, 2015; USDA, 2014). Since US adults age 65 and older heavily consume this high fat oil, their population is rapidly increasing and is projected to reach 71 million by 2030. The objectives of the present study are to summarize the long-term health effect of high volume of corn oil consumption in aging animals and to understand the relevant mechanisms.



Figure 1. A kernel of corn.

Nutrient composition	Values
Protein (g)	0
Fat, total (g)	100
Triglycerides (g)	98.8
Polyunsaturates, total (g)	59.7
Cis, Cis only (g)	58.7
Monounsaturates (g)	26.0
Saturates, total (g)	13.1
Unsaponifiable Matter (g)	1.2
Cholesterol (mg)	0
Phytosterols (mg)	1000
Tocopherols, total (mg)	88
Alpha-tocopherol (mg)	19
Gamma-tocopherol (mg)	67
Delta-tocopherol (mg)	3
Carbonhydrate (g)	0
Ash (g)	0
Sodium (mg)	0
Energy (Cal)	885

Table 1. Nutrients values of refined corn oil (100 g)

Table 2. Composition comparison of major vegetable oils

	Saturated	Mono-	Polyu	nsaturated fa	Oleic	Smoke	
Туре	fatty acids	unsaturated	Total poly	Linolenic	Linolenic	acid	point
	fatty actus	fatty acids	rotai pory	acid (ω-3)	acid (ω-6)	(ω-9)	(°C)
Coconut	91.0	6.0	3.0	-	2.0	6.0	177
Corn	12.9	27.6	54.7	1.0	58.0	28.0	232
Cottonseed	25.9	17.8	51.9	1.0	54.0	19.0	216
Flaxseed	7.5	15.5	79.0	64.0	15.0	11.0	107
Peanut	16.9	46.2	32.0	-	32.0	48.0	225
Soybean	15.7	22.8	57.7	7.0	50.0	24.0	238
Sunflower	10.1	45.4	40.1	0.2	39.8	45.3	227

RESULTS OF ANIMAL STUDIES

Aging Mice

We recently conduct a study investigated whether dietary intake of phytochemical epicatechin in extending lifespan in mice, we unexpectedly found that high corn oil dietary intake improves health and longevity of aging mice (Si et al., 2014).

Twelve-month old male C57BL/6 mice purchased from the National Cancer Institute (Bethesda, MD) were randomly divided into two groups (n = 31) and given either a normal diet (N) or a high corn oil diet (HF).

Both diets are based on the AIN-93 produced by Dyets Inc. (Bethlehem, PA) with two exceptions 1) soybean oil was replaced with corn oil and 2) the percentage of corn oil in each diet. The normal diet has 10% fat derived calories (FDC) and the HF diet has 58% fat derived calories (FDC) (Reeves, et al., 1993). This dose of corn oil (58% FDC) was calculated based on previous studies using high fat diet (60% FDC, 6% from soybean oil and 54% from lard) (Sung et al., 2014). We also collected samples from young control mice (YC) to compare changes between YC and normal diet fed old mice (N).

Our results shown that at 25 months of age, 53.8% of mice had died in the N group, whereas the mortality rate was only 23.2% in the HF group (P = 0.02, Figure 2).

The median for the HF group was reached two months later at 27 months of age. Although the average body weight in the HF group was significantly higher than that in the N group, the food consumption in the HF group was significantly lower than that in the N group. This may be a result of the texture of the HF diet, which has the consistency of paste. Some days an oily liquid would form on the top of the rodent feeding jars, while the N diets were typical pellets. Interestingly, there was no significant difference in energy intake between these two groups. Physical activity and locomotor function continuously diminishing with aging (Yankner et al., 2008; Baur et al., 2006) and although corn oil increased longevity, it is important to know whether quality of life was maintained. We evaluated the ability to perform on a rotarod, a classic method of testing balance and motor coordination. While the time on the rod was significantly decreased as mice aged in the N group (decreased from 115 sec at month 14 to 80 sec at month 25), the HF group maintained their motor skills until they were 25 months of age. Histological examination of liver sections stained with haematoxylin and eosin revealed a loss of cellular integrity and the accumulation of large lipid droplets in the livers of the N group, but the HF group reduced the size of intracytoplasmic lipid vacuoles. Circulating lipid abnormalities are increasingly recognized as playing an important role in the aging process and age-related disorders such as diabetes and vascular dysfunction (Labinskyy et al., 2006). Compared to the YC group, serum triglyceride, total cholesterol and LDL-cholesterol were significantly increased in the N group; however, all these three serum lipids were reversed in the HF group.



^{*}P < 0.05. (Si et al., 2014).

Figure 2. Suvival curve in normal diet (N) and high corn oil diet (HF) mice for 13-15 months. There initially was 31mice/group.

Consistent with observations of the pathological alterations in liver and remarkably shortened lifespan of mice in the N group, circulating levels of cytokines including IL-1 β , IL-6, IFN- γ and MCP-1 were significantly elevated in the N group compared to those in the YC group. However, dietary consumption of corn oil significantly reduced these proinflammatory markers, indicating that corn oil may suppress chronic inflammation caused by aging. In addition, GSR activity in the livers of the N group was significantly decreased compared to that in the YC group, whereas the activity of this enzyme was reversed in the HF group.

Aging Rats

The National Toxicology Program (NTP) initiated studies to evaluate the role of several oils in altering cancer rates in male rats in early 1990s (Program, 1992). 7 weeks old male F344/N rats (50 rats/group) were administrated 2.5, 5, 10 mL corn oil/kg body weight or 10mL saline/kg body weight by gavage, 5 days a week, for 2 years.

The results shown that two-year survival was increased in male rates receiving corn oil (untreated control, 26/50; saline control, 32/50; 2.5mL/kg, 33/50; 5mL/kg, 38/50; 10mL/kg, 40/50). The major reason was the dose-related decreased incidence of mononuclear cell leukemia (untreated control, 27/50; 2.5mL/kg, 16/50; 5mL/kg, 11/50; 10mL/kg, 7/50).

Although the mean body weights of all dosed groups were at least 5% higher than those of the untreated and saline controls by week 48, the mean body weights of groups receiving 2.5mL/kg, or 5mL/kg corn oil/kg decreased during the final weeks of the study (after week 89) and were similar to those of the controls at the end of the study.

This study was further extended to explore whether sex and diets affect survival in rats (Rao and Haseman, 1993). Four to five weeks old male and female F344/N rats (50 rats/ group) fed either commercial diet or a NIH-07 diet were administrated with 5 mL corn oil/kg body weight or 10mL saline/kg body weight by gavage, 5 days a week, for 2 years. Corn oil intake with either commercial diets or NIH-07 diet significantly increased the body weight and survival but significantly lowered the incidence of leukemia compared with their corresponding diet control groups at 106 weeks (Table 3), however, these beneficial effects were not observed in female rats. Moreover, oil intake worked slightly better in reducing leukemia in NIH-07 diet (from 48.9% to 21%) than that in commercial diet (from 28.3% to 12.6%) in male rats although the control groups without corn oil in commercial diet had higher survival and lower leukemia than these in control group without corn oil in NIH-07 diet. Interestingly, the gross energy intake (per kilogram body weight) was essentially the same for male and female rats with or without corn oil intake but reduced food intake and increased PUFA in corn oil gavage groups comparing to the control groups (40 g/week vs. 32 g/week, 2.7 g/week vs. 1.1 g/week respectively), this is in line with our recently studies in mice.

MECHANISMS

Aging is well-known as an inevitable process that is physiologically characterized as a progressive, generalized systematic dysfunction of almost or all organs, giving rise to the escalated vulnerability to environmental challenges and resulting in increased risks of disease and death. Indeed, aging is associated with a greatly increased metabolic and oxidative stress, elevated chronic, low-grade inflammation, and accumulated DNA mutations as well as increased levels of its DNA damages (Heininger, 2000a; Heininger, 2000b; Frisard and Ravussin, 2006). It is established that calorie restriction delays age-associated organ disorders and increases longevity as well as improves inflammation and oxidative stress in a wide range of species, suggesting that targeting nutrient-sensing and energy metabolism pathways may be an effective approach to delay the aging process and age-related diseases.

Indeed, our recent study in mice and previous studies in rats provide evidence that longterm corn oil intake improved general health, decreased incidence of mononuclear cell leukemia and extended longevity. These health benefits of corn oil may be associated with reversing aging-increased blood lipids and pro-inflammatory makers as well as agingdamaged brain and liver functions. These results are very important for us to understand the health effects of corn oil consumption, the second largest vegetable oil consumption in the United States, although more studies, particular clinical studies are needed to evaluate the beneficial effects of long-term corn oil consumption.

Corn oil was first attended by its highly effective in lowering blood cholesterol, particularly LDL-cholesterol (Ahrens et al., 1957; Hill et al., 1979), one of the major risks of cardiovascular disease. Reducing blood LDL is recommended to lower cardiovascular disease as well as other aging-related chronic diseases (2002). Our results in mice are in line with these previous studies that corn oil lowers LDL-cholesterol, total cholesterol and triglyceride (Si et al., 2014).

This effect may be due to the high level of PUFA, which is supported by evidence that corn oil is more effective than olive oil in lowering LDL-cholesterol because corn oil has higher PUFA (58.7 g/100 g oil) than olive oil (8.4 g/100 g oil) (Howell et al., 1998; Dupont et al., 1990). Corn oil has a plant sterols content of 128 mg/1000kcal vs. 66 mg/1000kcal for olive oil, and these plant sterols can reduce cholesterol absorption from the gut which in turn lowers body pools and enhances synthesis rate through de-suppression of cellular hydroxy-methylglutaryl-CoA reductase activity (Howell et al., 1998; Maki, et al., 2015). This is supported by a recent human study that the hypocholesteremic effect of corn oil is connected with the high amount of phytosterols (Wagner et al., 2001).

Indeed, corn oil intake affected lipid metabolism genes including cholesterol 7α hydroxylase, hydroxyl-3-methylglutaryl-Coenzyme A reductase, fatty acid synthase and angiopoietin-like protein 4 at a circadian rhythm (decreased at hour 3, increased at hour 6 to 9, and decreased at hour 24) in rats (Takashima et al., 2006).
	Male				Female				
	Commercial diet		NIH-07 diet		Commercial diet		NIH-07 diet		
	Control	Corn oil	Control	Corn oil	Control	Corn oil	Control	Corn oil	
Body weight, g	446±20	473±21*	478±18	504±19*	320±22	$298 \pm 20^{*}$	348±15	330±16 [*]	
Suvival, %	74±7.4	82±5.1*	63±7.4	$71 \pm 8.4^{*}$	79±7.0	78±7.3	71±8.2	73±8.6	
Leukemia, %	28.3±10	$23.6 \pm 8.2^*$	48.9±10.3	$21.0\pm9.2^{*}$	18.7±6.2	18.7±10.3	25.2±8.1	22.7±7.6	

Table 3. Influence of corn oil on body weight, survival and leukemia in F344 rats

*Significant (p < 0.01) compared with corresponding control.

Rao and Haseman, 1993.

These genes/proteins changes may lead to the decrease of food intake (15% in weight) (Takashima et al., 2006), which is similar with our results (Si et al., 2014). These results suggest that the animals can self-control their total calorie intake to a constant level by changing these genes expressions. A large body of evidence indicates that increased generation of reactive oxygen species (ROS) which are chemically reactive molecules with most of them containing oxygen and unpaired electrons is one of the major triggers of aging. There is a strong correlation between chronological age and the levels of ROS generation and oxidative damage of tissues.

ROS are primarily produced by mitochondria during energy production (about 2% of total oxygen consumption was funneled to ROS) (Chance et al., 1979). Extra amounts of ROS induces oxidation of fatty acids and proteins and causes oxidative damage of DNA that may lead to cellular senescence, functional alterations, and pathological conditions (Harman, 1972; Linnane et al., 1989). This extra amount of ROS is deactivated to water and oxygen by endogenous enzymes including superoxide dismutase (SOD), catalase or glutathione peroxidases (Chang et al., 2004).

Endogenous antioxidant glutathione (GSH) and exogenous antioxidants including vitamins C and E are also important ROS scavengers. Reducing ROS is proposed as a leading strategy to delay aging and related degenerative diseases. Corn oil is one of the highest natural sources of vitamin E (62.01 mg/100 g oil) and is just a little less than cottonseed oil (62.37 mg/100g oil). This high content of vitamin E may prevent aging-increased ROS and extend the lifespan of aging mice. This is supported by our results showing decreased GSR activity, a critical endogenous antioxidant enzyme, was reversed by corn oil intake (Si et al., 2014). Aging-induced ROS also contributes to low-grade chronic inflammation (Brod, 2000), a crucial player of the process of aging and age-related diseases in older adults. Indeed, chronic pro-inflammatory markers including IL-6, MCP-1 and TNF- α are consistently elevated with age in the absence of acute infection or other physiological stress (Ferrucci, et al., 2005). Consequently, the sustained increases of these pro-inflammatory molecules impair the function and integrity of various tissues and organs and thus accelerate aging and agingrelated chronic diseases, although this increase is still in the sub-acute range (Chung, et al., 2006). Interestingly, calorie restriction significantly attenuates the increase of these proinflammatory markers while extending the lifespan (Chung et al., 2006; Zou et al., 2004), suggesting that anti-inflammatory agents may have the potential to extend a healthy lifespan. Results from our study shows that dietary intake of corn oil significantly reversed the increase of circulating pro-inflammatory markers including IL-1 β , IL-6, IFN- γ and MCP-1 in aging mice and therefore increased longevity (Si et al., 2014).

A lower ratio of omega-6/omega-3 fatty acids (n-6/n-3) is recommended to reduce the risk of many highly prevalent chronic diseases in Western societies (ratio of (n-6/n-3) in Western diets is 15/1-16.7/1) (Simopoulos, 2002). Mammalian cells cannot convert omega-6 to omega-3 fatty acids because they lack the converting enzyme, omega-3 desaturase.

These two classes of essential fatty acids (EFAs) are not interconvertible, are metabolically and functionally distinct and have opposing physiological functions. Therefore, too much omega-6 may be detrimental for cells. Corn oil is not a good source for EFAs because the ratio of omega-6/omega-3 fatty acids from corn oil is 83, which is much higher than the recommended ratio (1/1 to 4/1) (Simopoulos, 2002).

However, the present study and other studies show that high corn oil intake improves health and longevity in mice and rats (10 mg/kg/d by oral gavage) (1994).

One explanation is that majority of omega-6 PUFA from corn oil is used for energy, and is not used to produce thrombi and atheromas, which are required for cardiovascular disease development. Moreover, high levels of vitamin E (majorly γ -tocopherol) (Elmadfa and Park, 1999) and plant sterols (0.77% by weight) (Ostlund, et al., 2002) may counter the bad effects of omega-6 PUFA of corn oil. For example, γ -tocopherol, the major form of vitamin E in the corn oil, and its metabolite have more anti-inflammatory properties than α -tocopherol, the predominant form of vitamin E in the tissues and most supplements (Jiang and Ames, 2003).

Taken together, if total calorie intake is kept in a normal range, long-term corn oil intake reverses aging-increased blood lipids and circulating pro-inflammatory cytokines as well as aging-damaged brain and liver function, and thus increases longevity of aging mice and rats.

CONCLUSION

These health benefits of corn oil may result from the combinations of the high levels of PUFA, vitamin E and plant sterols. Therefore, corn oil, even at a high energy percentage, is a favorable replacement of animal fats in the human diet if the total energy intake is controlled.

REFERENCES

- (1994). NTP Comparative Toxicology Studies of Corn Oil, Safflower Oil, and Tricaprylin (CAS Nos. 8001-30-7, 8001-23-8, and 538-23-8) in Male F344/N Rats as Vehicles for Gavage. *Natl. Toxicol. Program Tech. Rep. Ser.* 426, 1-314.
- (2002). Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III) final report. *Circulation* 106, 3143-421.
- Ahrens, E. H., Jr., Insull, W., Jr., Blomstrand, R., Hirsch, J., Tsaltas, T. T., and Peterson, M. L., (1957). The influence of dietary fats on serum-lipid levels in man. *Lancet* 272, 943-53.
- Baur, J. A., Pearson, K. J., Price, N. L., Jamieson, H. A., Lerin, C., Kalra, A., Prabhu, V. V., Allard, J. S., Lopez-Lluch, G., Lewis, K., Pistell, P. J., Poosala, S., Becker, K. G., Boss, O., Gwinn, D., Wang, M., Ramaswamy, S., Fishbein, K. W., Spencer, R. G., Lakatta, E. G., Le Couteur, D., Shaw, R. J., Navas, P., Puigserver, P., Ingram, D. K., de Cabo, R.,

and Sinclair, D. A., (2006). Resveratrol improves health and survival of mice on a high-calorie diet. *Nature* 444, 337-42.

- Brod, S. A., (2000). Unregulated inflammation shortens human functional longevity. *Inflamm. Res.* 49, 561-70.
- Chance, B., Sies, H., Boveris, A., (1979). Hydroperoxide metabolism in mammalian organs. *Physiol. Rev.* 59, 527-605.
- Chang, T. S., Cho, C. S., Park, S., Yu, S., Kang, S. W., and Rhee, S. G., (2004). Peroxiredoxin III, a mitochondrion-specific peroxidase, regulates apoptotic signaling by mitochondria. J. Biol. Chem. 279, 41975-84.
- Chung, H. Y., Sung, B., Jung, K. J., Zou, Y., and Yu, B. P., (2006). The molecular inflammatory process in aging. *Antioxid. Redox Signal.* 8, 572-81.
- Corn Refiners Association. (2006). Corn oil. www.corn.org.
- Dupont, J., White, P. J., Carpenter, M. P., Schaefer, E. J., Meydani, S. N., Elson, C. E., Woods, M., and Gorbach, S. L., (1990). Food uses and health effects of corn oil. J. Am. Coll. Nutr. 9, 438-70.
- Elmadfa, I. and Park, E., (1999). Impact of diets with corn oil or olive/sunflower oils on DNA damage in healthy young men. *European Journal of Nutrition* 38, 286-292.
- Ferrucci, L., Corsi, A., Lauretani, F., Bandinelli, S., Bartali, B., Taub, D. D., Guralnik, J. M., and Longo, D. L., (2005). The origins of age-related proinflammatory state. *Blood* 105, 2294-9.
- Frisard, M. and Ravussin, E., (2006). Energy metabolism and oxidative stress: impact on the metabolic syndrome and the aging process. *Endocrine* 29, 27-32.
- Harman, D., (1972). The biologic clock: the mitochondria? J. Am. Geriatr. Soc. 20, 145-147.
- Heininger, K., (2000a). A unifying hypothesis of Alzheimer's disease. III. Risk factors. Hum. Psychopharmacol. 15,1-70.
- Heininger, K., (2000b). A unifying hypothesis of Alzheimer's disease. IV. Causation and sequence of events. *Rev. Neurosci.* 11 Spec. No, 213-328.
- Hill, P., Reddy, B. S. and Wynder, E. L., (1979). Effect of unsaturated fats and cholesterol on serum and fecal lipids. A study of healthy middle-aged men. J. Am. Diet. Assoc. 75, 414-20.
- Howell, T. J., MacDougall, D. E. and Jones, P. J., (1998). Phytosterols partially explain differences in cholesterol metabolism caused by corn or olive oil feeding. *J. Lipid Res.* 39, 892-900.
- Jiang, Q., Ames, B. N., (2003). Gamma-tocopherol, but not alpha-tocopherol, decreases proinflammatory eicosanoids and inflammation damage in rats. *FASEB J*. 17, 816-22.
- Labinskyy, N., Csiszar, A., Veress, G., Stef, G., Pacher, P., Oroszi, G., Wu, J., and Ungvari, Z., (2006). Vascular dysfunction in aging: potential effects of resveratrol, an antiinflammatory phytoestrogen. *Curr. Med. Chem.* 13, 989-96.
- Landers, R. and Rathmann, D., (1981). Vegetable oils: effects of processing, storage and use on nutritional values. J. Am. Oil Chem. Soc. 58, 255-259.
- Linnane, A. W., Marzuki, S., Ozawa, T., and Tanaka, M., (1989). Mitochondrial DNA mutations as an important contributor to ageing and degenerative diseases. *Lancet* 1, 642-5.
- Maki, K. C., Lawless, A. L., Kelley, K. M., Kaden, V. N., Geiger, C. J., Dicklin, M. R., (2015). Corn oil improves the plasma lipoprotein lipid profile compared with extra-virgin

olive oil consumption in men and women with elevated cholesterol: results from a randomized controlled feeding trial. J. Clin. Lipidol. 9, 49-57.

- Ostlund, R. E., Racette, S. B., Okeke, A., and Stenson, W. F., (2002). Phytosterols that are naturally present in commercial corn oil significantly reduce cholesterol absorption in humans. *American Journal of Clinical Nutrition* 75, 1000-1004.
- Program, N. T., (1992). Comparative Toxicology Studies of Corn Oil, Safflower Oil and Tricaprylin in Male F344/N Rats as Vehicles for Gavage.
- Rao, G. N., Haseman, J. K., (1993). Influence of corn oil and diet on body weight, survival, and tumor incidences in F344/N rats. *Nutr. Cancer.* 19, 21-30.
- Reeves, P. G., Nielsen, F. H. and Fahey, G. C., Jr., (1993). AIN-93 purified diets for laboratory rodents: final report of the American Institute of Nutrition ad hoc writing committee on the reformulation of the AIN-76A rodent diet. J. Nutr. 123, 1939-51.
- Si, H., Zhang, L., Liu, S., LeRoith, T., and Virgous, C., (2014). High corn oil dietary intake improves health and longevity of aging mice. *Exp. Gerontol.* 58, 244-9.
- Simopoulos, A. P., (2002). The importance of the ratio of omega-6/omega-3 essential fatty acids. *Biomed. Pharmacother*. 56, 365-79.
- Sung, Y. Y., Kim, D. S., Kim, H. K., (2014). Viola mandshurica ethanolic extract prevents high-fat-diet-induced obesity in mice by activating AMP-activated protein kinase. *Environ. Toxicol. Pharmacol.* 38, 41-50.
- Takashima, K., Mizukawa, Y., Morishita, K., Okuyama, M., Kasahara, T., Toritsuka, N., Miyagishima, T., Nagao, T., Urushidani, T., (2006). Effect of the difference in vehicles on gene expression in the rat liver-analysis of the control data in the Toxicogenomics Project Database. *Life Sci.* 78, 2787-96.
- USDA, (2014). *Corn: Market Outlook* http://webarchives.cdlib.org/sw15d8pg7m/http://ers. usda.gov/Briefing/corn/2005baseline.htm.
- USDA, (2015). *Oil Crops Yearbook* http://usda.mannlib.cornell.edu/MannUsda/view DocumentInfo.do?documentID=1290.
- Wagner, K. H., Tomasch, R. Elmadfa, I., (2001). Impact of diets containing corn oil or olive/ sunflower oil mixture on the human plasma and lipoprotein lipid metabolism. *Eur. J. Nutr.* 40, 161-7.
- Yankner, B. A., Lu, T. Loerch, P., (2008). The aging brain. Annu. Rev. Pathol. 3, 41-66.
- Zou, Y., Jung, K. J., Kim, J. W., Yu, B. P., and Chung, H. Y., (2004). Alteration of soluble adhesion molecules during aging and their modulation by calorie restriction. *FASEB J*. 18, 320-2.

Chapter 6

PRODUCTION METHODS AND COCONUT OIL QUALITY

Kapila N. Seneviratne^{} and Nimanthi Jayathilaka* Department of Chemistry, University of Kelaniya, Kelaniya, Sri Lanka

ABSTRACT

Among many edible oils, coconut oil can be considered a unique oil because of its unique short and medium chain saturated triglyceride composition. Coconut oil is extracted by dry and wet extraction methods. In dry extraction, coconut oil is extracted from dried coconut kernels by pressing or solvent extraction. In wet extraction, an aqueous emulsion of white coconut kernel (coconut milk) is either boiled or treated under cold conditions to separate coconut oil. The cold methods include freezing, enzymatic treatment or fermentation of coconut milk emulsion to break down proteins and separate oil. Copra oil is prepared by pressing dry coconut kernel known as copra. Virgin coconut oil is prepared by both wet and dry extraction methods. Coconut oil contains two fractions, fatty or lipid fraction and non-lipid fraction. The quality parameters such as acid value, peroxide value, anisidine value, iodine value and saponification value are originating from the lipid fraction. Among these quality parameters, iodine value and saponification value of coconut oil are not significantly affected by the method of extraction. Fatty acid composition of coconut oils extracted by any dry or wet methods remains reasonably similar. However, acid value, peroxide value and anisidine value vary with the extraction method of coconut oil. Non-lipid fraction also changes with the method of extraction. Health properties of coconut oil are associated with not only lipid fraction but also non-lipid fraction. As a result, health properties of coconut oil considerably vary with the method of extraction. Therefore, it is advisable to verify the origin of coconut oil when coconut oil is used for research purposes or for consumption.

Keywords: coconut oil, quality parameters, oil extraction, wet process, dry process, antioxidant properties

^{*}E-mail address: kapilas@kln.ac.lk.

INTRODUCTION

The average height of a coconut tree of ordinary tall coconut cultivars is about 12 meters. Coconuts become mature within 12-14 months. Coconuts can be harvested manually or by allowing the nuts to fall naturally. Manual harvesting can be done by hand picking after climbing the tree or by using a knife attached to a long bamboo pole. A coconut tree produces a single bunch of mature coconuts every month and the number of nuts harvested can be up to about 75 nuts a year. A mature fresh coconut with the husk weighs about 1.5 kg. The husk weighs about 0.5 to 0.8 kg and it can be removed by dehusking the coconut using a sharp iron spike fixed in the ground. About 1750 coconuts can be dehusked per eight-hour period by an experienced worker. Inside the husk there is a thick hard shell covering the coconut endosperm. Dehusked coconuts have to be cracked using a chopping knife or hatchet to open the coconut. This process is called shelling. During this shelling, coconuts are cracked into two equal halves. After opening the coconut by shelling, coconut kernel is still attached to the shell halves and can be further processed to obtain coconut oil. This coconut kernel, which weighs about 0.35 kg, is either dried to produce copra or grated to make desiccated coconut.

PROCESSING COCONUT KERNEL FOR OIL EXTRACTION

Coconuts have a solid as well as a liquid endosperm. Solid endosperm is the white coconut kernel which is used to extract coconut oil. The extraction process can be categorized based on the nature of coconut kernel and the methods used to obtain oil from coconut kernel. If dry coconut kernel is used for the extraction of coconut oil, the process is called dry process. Fresh coconut kernel is used in the wet extraction process. Here, a water emulsion of coconut kernel is prepared by mixing crushed or scraped coconut kernel with water. The resultant white-colored, milk-like emulsion, usually known as coconut milk is used for the extraction of coconut oil.

Dry Process

This process is used for the preparation of coconut oil in industrial scale for human consumption as cooking oil or as an ingredient for preparation of other foods. This oil is also used to prepare soap and other skin care products. The first step of the dry extraction of coconut oil is the drying of coconut kernel. Usually the fresh coconut kernel contains about 50% water, 34% oil, 7.3% carbohydrates, 3.5% proteins, 3.0% fiber and 2.2% ash (Banzon & Velasco, 1982). The water content in coconut kernels has to be reduced from 50% to about 6% during the drying process for the oil extraction by pressing. The dry coconut kernel or copra is prepared from mature coconut kernel. There are two types of copra, cup copra and ball copra, which reflect the shape of the copra. Ball copra is mainly used in making sweets or consumed directly. Ball copra is commonly prepared by mildly drying mature coconuts for 8-9 months. For this, it is not necessary to expose coconuts to direct sunlight. In this method, coconuts are stored in the shade. The coconut kernel inside the nut shrinks due to slow evaporation of moisture from the liquid endosperm of coconut and further drying of the

coconut kernel. During this process, the kernel separates from the shell as a ball. This detachment of copra ball from the coconut shell can be confirmed by shaking the nut to check for the rattling sound made by the detached kernel. At this stage, the husk can be removed and the shell can be broken to take copra out of the coconut shell as a ball. In the production of cup copra, coconuts are split-open into two halves and dried using various methods.

Sun Drying

Sun drying is the cheapest and probably the greenest way of drying copra. After removing the husk, coconuts are halved by breaking the thick shell of the coconut. When these coconut halves are dried for two days, the low moisture coconut kernels from the halves can be removed including the thin brown outer skin of the coconut kernel, known as testa. These separated coconut kernel halves are still not suitable for extraction of oil as they contain about 30% moisture. The moisture content has to be further reduced to 6% for the copra to be suitable for extraction of oil. Copra has to be dried in bright sunlight for 8 hours a day for at least 7-14 days to lower the moisture content down to 6%. Though this drying process takes a long time, this method is popular among small-scale copra producers, as it does not require any special instrument or additional expense for drying. However, the drying process can be easily disturbed by rain. In addition to getting copra wet during rain, the lack of sunlight delays the drying process.

As coconut kernel contains proteins and carbohydrates in addition to the moisture and oil, the kernel is prone to fungal attacks, which produce aflatoxins in fresh coconut kernel. Fresh coconut kernel is very susceptible to deterioration, which results in producing free fatty acids and rancidity due to fungal attacks and other reasons. To minimize fungal contaminations, the drying process should be started as soon as the nut is open. The drying process has to continue without any disruption due to rain or lack of sunlight in order to prepare good quality sundried copra. In some instances, coconut kernels or partially dried copra is cut in to small slices to accelerate the drying process. This slicing requires a lot of labor and as a result, the cost of production of copra increases. As fungal attacks occur on the surfaces that are exposed, slicing may promote deterioration of the kernel if proper sunlight is not provided. Therefore, it is safer to split the nuts into halves.

Usually the drying process takes about two weeks. With the evaporation of water, the oil to water ratio of the coconut kernel increases. It is easier to extract oil from copra, which has low moisture content compared to the extraction of coconut oil from fresh coconut kernel containing a lot of water. Properly drying copra to minimize water is important for the production of coconut oil by pressing copra. The water content of the coconut oil extracted using properly dried copra is negligible. Water content affects the quality of the coconut oil as the presence of water in coconut oil promotes the hydrolysis of triglycerides to give hydrolytic rancidity. In the production of desiccated coconut, the drying is done after removing the testa, which gives the light yellowish color to coconut oil. In the production of virgin coconut oil, the testa is removed and the white coconut kernel is used to produce the colorless oil.

Kiln Drying

Coconut kernel can be dried by heating the coconut halves in a special kiln or on an open stage as well. In this case, copra is smoked by a fire underneath. This process dries coconut kernel more quickly than the sun drying process. Kiln drying is important during the rainy season and this drying can also be combined with sun drying. The special platform used for this purpose can be set up using bamboo stems or some wood planks. This platform has many empty or open spaces. Coconut halves are spread on the platform and a fire is made using coconut shells on the floor under the platform. With slow burning of coconut shells, the coconut kernels on the platform are dried due to the contact of smoke generated by slow burning process underneath. The kiln where this platform drying takes place is covered by a brick structure. The temperature of the smoke generated in this drying process is about 40- 50° C and the drying time is reduced to 4-5 days due to the hot smoke, which is in contact with the copra. Coconut oil produced by this copra is inferior in quality and brownish in color. Of the different parameters, the width of the drying chamber and the distance between the floor where the burning takes place and the drying platform are critical in the drying process (Rodrigo et al., 1996). Two types of fuels are used in traditional kilns, coconut shells and coconut shell charcoal. Better quality copra based on commercial grading system can be obtained by using charcoal as fuel.

Hot Air Drying

This drying process needs fuels other than coconut-based fuels. Therefore, this is a more expensive drying process. However, the quality of copra produced by this method is superior to kiln dried copra because of the more efficient drying without exposing copra to smoke. The drying time of copra by hot air drying is also shorter compared to other drying methods. Hot air can be generated within the kiln or can be pumped from an outside source.

Desiccated Coconut

Desiccated coconut is obtained from small particles of coconut kernel by drying ground or shredded coconut kernel after the removal of the brown testa. It is an important ingredient in confectioneries, puddings and many other food preparations. Husk is removed from fresh matured coconut. Then the hard shell is removed and coconut meat is separated. The testa is removed and the white coconut kernel is washed with clean water. Then it is grated and sulfur dioxide is passed to obtain a white color. The grated coconut kernel mass is blanched with live steam for about 20 minutes to bring down the microbiological counts. The blanched mass is then dried in a hot air drier at a temperature of 80-90°C for about 10 hours to bring down the moisture content below 3%. The dried coconut kernel is tested for moisture, free fatty acids and microbial counts. This desiccated coconut kernel contains 2-3% moisture, 65-68% fat and 30-32% other solids such as proteins and carbohydrates.

EXTRACTION OF COCONUT OIL

Pressing Copra

The main method of extraction of coconut oil from copra is by pressing copra using expellers. Copra is pressed in large expeller presses that generate heat and pressure. The resultant crude coconut oil is brown and turbid in appearance. This oil can be further purified by filtering and refining to remove free fatty acids (a breakdown product from the oil), any remaining moisture, any bad flavor or smell. Coconut oil made this way is the least expensive of all coconut oils, which is used in food preparations. Pressing is also used to produce virgin coconut oil is of high quality and free of fungal contaminations. Special driers can be used to prepare copra for this purpose and the brown portion of copra is not used to prepare virgin coconut oil by pressing copra. Here, the pressing can be considered as cold pressing, since the temperature is controlled.

Solvent Extraction

A solvent can be used in the extraction of coconut oil from copra. *n*-Hexane is considered to be the most efficient solvent for oil extraction as oils easily dissolve in hexane. It is the most suitable solvent also because of the low boiling point, which makes it easier to remove from the oil. It is also a relatively low cost solvent. However, its flammability, mild toxicity, explosiveness and environmental impacts are the concerns of industrial scale solvent extraction of coconut oil. The solvent extraction leaves low levels of solvent residue in the oil, which is safe but undesirable for food purposes. During the extraction, the oil in copra is leached out with the solvent and the insoluble copra meal is retained unaffected. The efficiency of extraction depends upon the temperature of the solvent, the ratio of the solvent to copra meal, size and the porosity of the copra particles and contact time with the solvent. Oil extraction by solvent extraction is more suitable for oil seeds containing relatively low amounts of oil. As copra contains about 70% oil, mechanical extraction by pressing is more efficient and economical. In addition to the full solvent extraction, prepress solvent extraction can also be used. In the mechanical extraction by pressing under moderate pressure, the oil content of copra can be reduced to nearly 15%. This remaining oil can be further extracted by solvent extraction. The resultant mixture of solvent extraction includes solvent, oil solution and extracted coconut meal. The solvent in the meal is removed by heating to boil off the volatile solvent and the solvent is recovered by condensation. The remaining solvent in the oil solution is removed by distillation. Traces of solvent left in the meal and the oil are removed by steam-stripping under reduced pressure. Various high temperature steps in the solvent extraction process may thermally degrade the oil to a very small extent and about 500-1000 ppm concentrations of solvent will also remain in the oil after purification.

Extraction of Coconut Oil by Supercritical Carbon Dioxide

Extraction of seed oils by supercritical carbon dioxide (SC-CO₂) can overcome the problem of solvent residues present in oils extracted by solvent extraction. Different compounds in edible oils may be incorporated into seed oils by changing pressure and temperature of carbon dioxide. However, the establishment of a supercritical extraction unit needs high investment. Thus, the technology is not in widespread use, though the quality of oil extracted by this method is claimed to be high. Information about the operational costs of this method of extraction based on systematic studies is not available for coconut oil and several other vegetable oils. The SC-CO₂ extraction of soybean oil has been studied and the equilibrium solubility data for different temperatures and pressures have been reported. The operating costs for SC-CO₂ and solvent extraction of soybean oil have been studied and compared (Reverchon & De Marco, 1994). Though the initial costs of the establishment of SC-CO₂ system is higher, the operational costs fell in the same range as conventional hexane extraction plants.

Triglycerides in seed oils are readily soluble in SC-CO₂ at 40 °C and at pressures higher than 280 bars. The main parameters important for efficient oil extraction in this method are particle size, pressure and residence time. Small particles of about 1mm mean diameter or less and high pressures (300-500 bar) can strongly reduce the extraction time. After extraction, the SC-CO₂ tryglicerides solution is sent to a separator working at subcritical conditions. This operation reduces the solvent power of CO₂ and allows the recovery of oil. The complete elimination of gaseous CO₂ from oil is also accomplished in this step. The SC-CO₂ extraction of seed oils has been tested up to pilot scale for several seed oils.

Wet Process

In the wet process of extraction of coconut oil, fresh coconut kernel is scraped and pressed with water. Due to the presence of proteins in coconut kernels, an emulsion is made. Separation of oil from this milk-like emulsion is a difficult task. This separation of the oil from the emulsion can be done in various ways. As the solution is in emulsion form due to the presence of proteins, these proteins have to be broken down in order to separate oil. Prolonged heating is the traditional way of extracting coconut oil from a coconut oil emulsion. Modern techniques use centrifuges and pre-treatments including cold, heat, acids, salts, enzymes, electrolysis, shock waves, or some combination of them for the extraction of oil from the emulsion. Despite numerous variations and technologies, wet processing is less efficient in oil extraction than dry processing, resulting in a 10-15% lower yield, even after considering the losses due to spoilage and pests during dry processing. Wet processes also require investment of equipment and energy, incurring high capital and operating costs. High quality virgin coconut oil is prepared by some wet extraction methods where temperature can be maintained below 40-50°C. However, coconut oil prepared by any wet extraction method is sometimes called virgin coconut oil even though higher temperatures are involved in some wet extractions.

One traditional way of extracting coconut oil from the coconut milk emulsion is by prolonged heating of the emulsion. Heating breaks down and deposits proteins at the bottom of the container. When the heating is continued, water in the emulsion evaporates. Due to its high boiling temperature, coconut oil does not evaporate significantly during this process. Finally, the coconut oil can be separated by decanting from the residue containing proteins, carbohydrates and other substances. The resultant coconut oil gives a nice coconut aroma and the oil is free of water. This oil can be kept for a very long time without forming oxidation products that cause rancidity. However, due to caramelization and other reactions such as Maillard reaction, the coconut oil produced by this method has a light yellowish color. One disadvantage of this method is the high amount of energy needed and relatively longer period of time taken to evaporate water from the coconut milk emulsion. In addition, there are no machines designed to produce coconut oil in industrial scale using this method. Therefore, this method is limited to the preparation of coconut oil in small scale for household consumption.

Centrifugation Process

The coconut milk emulsion prepared by pressing fresh coconut kernels contains approximately 40% oil. In the production of high quality virgin coconut oil, the pressing should be done using a special machine of which both the pressing plate and the sleeve are cooled by chilled water. Using a centrifuge, the cream is then concentrated to yield a higher percentage of oil while the proteins and water-soluble substances are separated out. Coconut oil produced by this method has a very light coconut flavor and the texture of coconut oil is extremely mild and smooth. In some instances, coconut milk is chilled at 10°C for 10 h to solidify the lipids. Then the aqueous layer is discarded and the lipid block is allowed to stand at 30°C until it dissolves completely. Then the mixture is centrifuged and the oil layer is separated. The coconut oils. These oils are expensive and usually labeled as extra virgin coconut oil.

Fermentation Process

Fermentation method is the least consistent of all the coconut oil production processes. Therefore, the quality of coconut oil produced by this method varies for different producers. The oil has to be further purified for food purposes. In the preparation of coconut oil by this method, fresh coconut kernel is first grated and then coconut milk or cream is pressed out from the white flesh. This milk is placed into vats or buckets and allowed to ferment at about 37°C. The enzymes and bacteria break the proteins in emulsion and separate the milk into different layers which include a top protein curd layer, a coconut oil layer underneath, another curd layer and a layer of water. The protein curd on the top can be removed and then the oil layer can be siphoned.

Enzyme Assisted Extraction

Enzyme assisted extraction of coconut oil is an eco-friendly process based on isolation of coconut oil and proteins in an emulsion. Copra meal prepared by cutting copra and passing through a 1 mm mesh or grated fresh coconut kernels is mixed with water and the mixture is treated with enzymes. The enzymes can be crude mixtures of protease, cellulase, hemicellulase and amylase enzymes. The presence of enzymes facilitates oil recovery by breaking cell walls and oil bodies. The oil recovery can vary from 80-98%. A single enzyme is not sufficient to break the insoluble cell wall materials to release oil (Rosenthal et al., 1996).

The polysaccharides of fresh mature coconut kernels are galactomannans (61%), mannans (26%) and cellulose (13%) (Balasubramanium, 1996). When an aqueous medium of ground desiccated coconut kernel was treated at 50°C with a mixture of commercially available enzymes including a hemicellulase preparation, a pectinase preparation, a cellulase preparation and an enzyme complex with multiple activities, 84% of the coconut oil present in coconut kernel was released (Chen & Diosady, 2003). The incubation followed by centrifugation gave a clear oil phase, an emulsion layer, an aqueous phase, and a solid phase containing the remaining coconut meal. Proteins of the solid residue were extracted by freeze drying the solid and treating with aqueous alkali, which dissolved 88% of the proteins in the solid phase. The acidification of the protein solution towards the isoelectric region precipitated 93.5% of dissolved proteins. Therefore, enzymatic extraction has an additional advantage because of the recovery of proteins.

QUALITY PARAMETERS OF COCONUT OILS

Various quality parameters have been proposed in order to establish the quality of coconut oil. Asian Pacific Coconut Community (APCC) and Codex Alimentarius have established the values of quality control parameters in physical, chemical and microbiological aspects of coconut oil. Quality parameters established by Codex Alimentarius for coconut oil are acid value or free fatty acid percentage (none), peroxide value (< 15 meq/kg oil), iodine value (6.3-10.6) and volatile matter percentage at 105°C (0.2%). APCC standards of quality parameters for virgin coconut oil are given in the Table 1.

Color

Colorless coconut oil is considered as commercially more acceptable good quality coconut oil. Due to this reason, refined, bleached and deodorized (RBD) coconut oil is prepared. One source of the light yellow or light brown color of coconut oil is coconut testa. When processing copra, some colored substances get incorporated into coconut oil. If copra without coconut testa is used for the dry extraction of coconut oil, the color of coconut oil will be colorless. In both dry and wet extraction of virgin coconut oil, this brown thin skin of coconut kernel is avoided. As a result, the extracted virgin coconut oil is colorless. Coconut oil can be extracted by heating coconut milk emulsion during wet extraction. In this case,

light yellow or light brown colored coconut oil is resulted and this color is due to caramelization of non-lipid substances such as sugars present in coconut milk or due to various oxidation reactions of the non-lipid substances. Even though the color of coconut oil is important as a commercial parameter, the color does not reflect the nutritional quality of coconut oil.

Parameter	Maximum value or range
Moisture (%)	0.1
Volatile matter, 120°C (%)	0.2
Free fatty acid (%)	0.2
Peroxide value, meq/kg	3
Density g/mL	0.915-0.920
Refractive index, 40°C	1.4480-1.4492
Insoluble impurities (%)	0.05
Saponification value, mg KOH/g oil	250-260
Iodine value	4.1-11
Unsaponifiable matter (%)	0.2-0.5
Total plate count	0.5
Color	Colorless

Table 1. APCC standards for virgin coconut oil

Adapted from APCC, 2009 (APCC, 2009).

Acid Value

Acid value of an oil is a measure of the amount of free acids present in the oil. The major component of oil is the triglyceride portion. These triglycerides produce free fatty acids upon hydrolysis. The hydrolysis can occur due to the action of lipase enzyme. The source of enzyme can be from the broken cells or tissues during the extraction process of coconut oil or from microorganisms. The acidity is an indicator of aging oil. The free acid formation due to hydrolysis of oil can accelerate due to poor extraction and storing conditions of coconut oil. Acid value is defined as the weight of potassium hydroxide (KOH) in mg needed for the neutralization of free fatty acids from 1 g of the oil. Even though the acid value unit, mg KOH/g oil is frequently used by chemists, the acidity is also expressed as a percentage by weight of total fat. Here, the number of moles of free acids found by the titration can be converted to mass by multiplying the number of moles by the molecular weight of fatty acids. As it is a fatty acid mixture, the molecular weight of the major fatty acid of the oil is usually considered for the calculation. In coconut oil, free fatty acid content is given as lauric acid equivalents while in olive oil it is given as oleic acid equivalents. When the acid value percentages are given as the major fatty acids in different oils, such acid values for different oils cannot be easily compared. Therefore, the acid value expressed as mg KOH/g oil is always easier for comparison purposes.

As triglycerides get hydrolyzed to form free acids, glycerol is also formed. Due to volatility of free short chain fatty acids, a light rancid smell may be observed. Free fatty acids and glycerol also change the taste of the oil or oil based food. Rancidity due to the free fatty

acids and glycerol formed by hydrolysis of triglycerides is called hydrolytic rancidity. This rancidity is milder than the rancidity resulting from the aldehydes and ketones due to oxidation of fatty acids. Therefore, hydrolytic rancidity arising mainly from free fatty acids is not seriously objectionable to the consumers as to cause serious economic disadvantageous.

Peroxide Value

Peroxides are primary oxidation products of edible oils formed due to free radical reactions and these peroxides further break in to shorter chain secondary oxidation products such as aldehydes and ketones that are volatile giving the smell of rancidity, which is known as oxidative rancidity. Therefore, the peroxide value may increase in the beginning and then secondary oxidation products appear later. Virgin coconut oil has a lower peroxide value compared to copra oil, which indicates the freshness of virgin coconut oil. Recommended maximum value for peroxides in coconut oil is 3 meq/kg. Copra oil, even after freshly collected from a mill has a peroxide value higher than that of virgin coconut oil prepared by dry or wet methods. This indicates that longer drying period of copra may also contribute to the formation of peroxides. Peroxide formation is also catalyzed by the presence of metal impurities. Pressing and grinding of copra and storing coconut oil in metal barrels may introduce dissolved metals such as iron in to coconut oil. This dissolving of metals is supported by the acidity of coconut oil if the acid value of coconut oil is high.

The traditional coconut oil, prepared by prolonged boiling of coconut milk also has low peroxide values. Surprisingly, peroxide value of this high temperature-processed coconut oil is lower and comparable with the peroxide value of virgin coconut oil prepared under cold conditions, indicating that peroxide formation may not be induced by the heating of coconut milk during this wet extraction. High content of saturated fatty acids of coconut oil gives a high oxidative stability to coconut oil. Therefore, peroxide formation is minimal in coconut oil prepared by any method of extraction compared to other polyunsaturated oils. Then coconut oil can be considered as the most stable and, therefore, most suitable oil for deepfrying.

Iodine Value

Iodine value reflects the degree of unsaturation of the fatty acids of oils. This parameter does not change significantly with the method of extraction if same quality coconut kernel is used for the extraction of coconut oil. However, low quality coconut oil for nonfood purposes can be obtained by pressing coconut pairings and this oil is called pairing oil. Such pairing oil may have slightly higher iodine values compared to other coconut oils. However, quality of pairing oil is low not because of the slightly different iodine value but due to higher peroxide value and higher acidity.

Saponification Value

Saponification value reflects the number of fatty acid units in a given weight of oil. If the fatty acids present in an oil are relatively small, the saponification value is relatively high as more units of fatty acids are present in a given weight that is going to be saponified by potassium hydroxide. This means that if the average molecular weight of the oil is low, the saponification value is high. Fatty acid compositions of all common edible oils are well known. Therefore, average molecular weight, which reflects the average chain length of fatty acids present in triglycerides of an oil is shorter, the saponification value becomes higher. Table 2 shows this relation between saponification value and average chain length of fatty acids of four common edible oils. When comparing reported saponification values for different oils or fats, it is important to check the units and the base used for saponification. The saponification values are given as milligrams of KOH per gram of oil, milligrams of NaOH per gram of oil, grams of KOH per gram of oil or grams of NaOH per gram of oil.

Oil	Average chain length	Saponification value mg KOH/g oil
Canola	17.92	185-195
Olive	17.74	184-196
Sunflower	17.86	185-198
Coconut	12.86	250-264

Fable 2. Average chain length	h and saponification	n values for common	edible oils
--------------------------------------	----------------------	---------------------	-------------

Oil	C8	C10	C12	C14	C16	C18	C18:1	C18:2	C18:3
Canola	-	-	-	-	4	2	62	22	10
Olive	-	-	-	-	13	3	71	10	1
Sunflower	-	-	-	-	7	5	19	68	1
Coconut	8	7	48	18	9	2	6	2	-

Table 3. Fatty acid composition (%) of common edible oils

Coconut oil has a unique fatty acid composition of short and medium chain fatty acids as indicated in Table 3. Due to the presence of this unique composition of shorter and medium chain fatty acids, coconut oil has a higher saponification value and this value in coconut oil is the highest among almost all other edible oils. Saponification value also does not change significantly with the method of extraction of coconut oil. However, slightly high saponification values may be observed for oils with other non-lipid substances, which may react with potassium hydroxide.

Anisidine Value

Peroxide value is used to analyze primary oxidation products. It is important to evaluate the oxidative status of oils by both primary oxidation products as well as secondary oxidation products. If an oil initially has a high peroxide value, keeping the oil in stock for a long time in absence of oxygen, produces secondary oxidation products that determine the decrease of peroxide value but the increase of anisidine value. Over time, these peroxides get further oxidized to volatile compounds such as hexanal and other non-volatile aldehydes and ketones. Anisidine values reflect the amount of relatively non-volatile aldehydes and ketones.

Unsaponifiable Substances

Even if all the insoluble impurities are removed from coconut oil, the oil still has up to about 0.5% of unsaponifiable matter, which includes polyphenols, alcohols, terpenoids, and several other organic substances (Kirschner & Harris, 1961). Quality parameters of coconut oil are mainly and traditionally defined for the lipid fraction. However, many studies conducted in recent times show that unsaponifiable components are also important to evaluate the quality of oil.

COCONUT OIL QUALITY

Coconut oil is a stable oil due to its high saturated fatty acid content. Due to this thermal stability, coconut oil is considered probably the most suitable oil for frying purposes. The progress of lipid oxidation of coconut oil was assessed by measuring peroxide value, anisidine value and total oxidation value during a period of 12 months. The low peroxide value (0.24-0.49 meq/kg oil) signifies a high oxidative stability, while anisidine values were in the range 0.19-0.87. Fourier transform infrared (FTIR) spectroscopy was used to monitor the peak changes due to oxidation during storage. The results of this study suggest that coconut oil retains its good chemical properties during 12 months of storage (Moigradean et al., 2012). When the quality of coconut oil is evaluated, it is important to consider both nutritional quality and commercial quality. Acid value, peroxide value and anisidine value are commercially attractive parameters because these values are directly connected to the rancidity of oil. Iodine value and saponification value are important in checking the authenticity of an oil. The most crucial parameters important for the nutritional quality are the identity of fatty acids in the triglycerides of the oil and the non-lipid components.

Basic Quality Parameters and the Extraction Method

Quality of coconut oil is evaluated by basic parameters such as color, acid value, peroxide value, iodine number, saponification value and anisidine value, etc. These parameters change over a range of values not only because of the method of extraction but also because of cultivar and maturity of coconut. Most of the records in literature give the values for copra oil and virgin coconut oil. Copra oil refers to the coconut oil extracted by dry process by pressing sun dried or kiln-dried copra. Virgin coconut oil has been made either dry process or wet process. Careful observation of the reported data indicates that the quality parameters such as iodine value and saponification value do not change significantly in coconut oils extracted by any method. However, acid value and peroxide value can vary

significantly in oils extracted by different methods and also with the aging of oils. Wet extracted coconut oil by centrifugation or enzymatic degradation contains lower free fatty acid content and peroxide value. Coconut oil extracted by dry methods or copra oil has a relatively higher acidity and peroxide value. However, for the extraction of virgin coconut oil by dry method, high quality copra prepared by quick drying in better conditions is used. Therefore, acid values of such coconut oil are lower compared to normal commercial coconut oil.

Among drving methods of copra, hot air drving is used to produce high quality copra in a relatively shorter period of time compared to other drying methods. Studies have been conducted to assess the quality of copra and coconut oil prepared using copra, dried in laboratory scale hot air driers (Guarte et al., 1996). In a laboratory scale drier, at a constant air velocity of 0.5 m/s, the moisture reduction rate of copra (down to 7%) increases with the drying temperature from 40-90°C. At this temperature range, the quality of copra or coconut oil was not affected. At temperatures above 90°C browning of copra occur due to Maillard reaction. However, drying temperature up to 100°C does not affect the oil content, acid value, anisidine value or saponification value. Drying copra at 90°C in a hot air dryer can reduce the moisture down to 7% within one day without making any changes of color, smell or taste of copra. Drying under this condition does not affect the contents of fatty acids such as lauric and palmitic. Other fatty acids showed small differences with no definite trend in variation with the temperature. Coconut oil prepared by dry and wet methods have been compared for copra oil prepared by pressing dry copra in a small scale expeller and coconut oil prepared by prolonged boiling of an emulsion of coconut milk in the traditional method (Seneviratne & Dissanayake, 2005). The quality parameters are given in the Table 4 and fatty acid compositions are given in the Table 5.

Source	Acid value (mg of	Saponification	Iodine value (g	Peroxide value
	KOH/g)	value (mg of	of I ₂ /100 g)	(absorbance at
		KOH/g)		500 nm)
Traditional oil	$0.40\pm0.02^{\rm a}$	274 ± 2^{a}	$6.6\pm0.2^{\rm a}$	0.034 ± 0.002^a
Copra oil	3.5 ± 0.9^{b}	241 ± 3^{b}	6.2 ± 0.3^{a}	$0.62\pm0.04^{\rm b}$

Table 4. Quality parameters of traditional coconut oil and copra oil

Adapted from Seneviratne & Dissanayake, 2005 (Seneviratne & Dissanayake, 2005). Each data point represents the mean of eight replicates \pm standard deviation; Different superscript letters in a same column denote a significant difference (p < 0.05) by MINITAB 2-sample t test.

Table 5. Fatty acid compositions of traditional coconut oil and copra oil

		Fatty acid							
Source	6:0	8:0	10:0	12:0	14:0	16:0	18:0	18:1	18:2
Traditional	$0.63 \pm$	9.2 ±	$6.5 \pm$	$49.6\pm$	$19.2 \pm$	$7.2 \pm$	2.4 ±	4.5 ±	$1.1 \pm$
oil	0.04	0.1	0.1	0.2	0.1	0.1	0.1	0.1	0.1
Copra oil	$0.53 \pm$	$8.6 \pm$	6.4 ±	$48.7 \pm$	$18.9 \pm$	7.6±	2.4 ±	4.9 ±	$1.3 \pm$
	0.04	0.1	0.2	0.2	0.1	0.2	0.1	0.2	0.1

Adapted from Seneviratne & Dissanayake, 2005 (Seneviratne & Dissanayake, 2005). Each data point represents the mean of three replicates ± standard deviation.

According to this study, acid value, peroxide value and saponification value of the two types of oils are significantly different and the iodine values showed no significant difference. Fatty acid compositions of the two types of oils were not significantly different enough to cause any nutritional changes.

In wet processing, natural fermentation carried out by microbes present in the natural environment is well known for its traditional and industrial applications. In this process, the fermentation of coconut oil is usually carried out without applying any heat.

This natural fermentation process has contamination problems due to the presence of unwanted microorganisms and uncontrolled conditions. It leads to the production of poor quality coconut oil, usually yellow in color, with a rancid smell. Therefore, induced fermentation method, using certain species of probiotic microorganisms, can be used under semi-controlled conditions. Quality parameters for virgin coconut oil produced both by natural fermentation and induced fermentation under semi-controlled conditions have been studied and compared with the quality standards given by APCC (Neela & Prasad, 2012). In this experiment, coconut milk was allowed to ferment naturally in a laboratory scale experiment in a separating funnel or in a clean container and the mixture was allowed to stand for 48 hours at room temperature. During the production of virgin coconut oil by natural fermentation, contaminations by different microbes occurred, resulting in yellow-colored, spoiled, coconut oil. The fermentation was also carried out under controlled conditions. Here, coconut milk was sterilized and a pure culture of L. plantarum was used in fermentation. The yield of virgin coconut oil from the natural fermentation was $25.68 \pm 0.96\%$ while that of induced fermentation was $28.47 \pm 1.07\%$. Physico-chemical parameters such as specific gravity, refractive index, insoluble impurities, free fatty acid content, peroxide value, iodine value, saponification value, unsaponifiable matter, and fatty acid compositions of virgin coconut oil produced by natural fermentation and induced fermentation were within the limits of APCC standards. However, the moisture content of virgin coconut oil by natural fermentation (0.52%) and induced fermentation (0.56%) is little higher than the APCC values (0.1-0.5%). Total aerobic plate count of virgin coconut oil by natural fermentation was 31 colony forming units (CFU) per 0.1 mg while that of virgin coconut oil by induced fermentation was 49 CFU/0.1 mg. Both oils contained about 6 mg/100 g tocopherols. The impurities and free acids of the coconut oil prepared by fermentation process can be removed to some extent by diatomaceous filters and carbon filters. As the moisture content of this coconut oil is high, the oil has to be heated for several hours at 100°C to remove water. Due to the presence of moisture, free fatty acids and a high peroxide value, coconut oil produced by fermentation is of poor quality.

A detailed study has been conducted to differentiate virgin coconut oil, RBD coconut oil and copra oil (Dayrit et al., 2007). Virgin coconut oil prepared by centrifugation, expeller process, enzymatic treatment, fermentation with and without heat and settling methods were used for the study. According to this study, gas chromatography analysis done with virgin coconut oil, RBD coconut oil and copra oil indicates that these oils cannot be differentiated by their fatty acid composition due to similar fatty acid compositions. However, volatile organic matter, moisture content and peroxide values are different in these oils. Moisture contents of virgin coconut oil extracted under different conditions varied from 0.05 to 0.12%. RBD coconut oil had a lower moisture content ranging from 0.01 to 0.1%. Copra oil had a higher average moisture content at a range of 0.08 to 0.14%. Volatile organic matter contents of virgin coconut oils ranged from 0.00 to 0.08%. RBD coconut oil had no volatile

organic matter while copra oil had a level 1.77% of volatile organic matter by weight. APCC standards for free fatty acids expressed as lauric acid equivalents is 0.5%. Average free fatty acid contents of virgin coconut oil, RBD coconut oil and copra oil were 0.131%, 0.021% and 1.41% respectively. There were no significant differences in iodine values of the three types of oils tested. The average peroxide values were 0.56, 0.98 and 1.48 meq/kg for virgin coconut oil and copra oil respectively and these values are well below the Codex Alimentarius peroxide value limit of 15 meq/kg for virgin oils. In the assessment of microbial contamination, it was observed that the CFU/mL for most of the virgin coconut oil and RDB coconut oil samples were below 10 CFU/mL, which is the APCC standard value. However, there were randomly distributed samples with higher CFU values. The microbial contamination depends on the quality of production including the quality of copra but not on the method of preparation of virgin coconut oil. Production quality of copra oil is usually poorer than that of virgin coconut oil. The microbial contamination of the samples of copra oil used in this study was not less than 10 CFU/mL but was only less than 250 CFU/mL range.

Enzymes in the Extraction of Coconut Oil

In the wet extraction of coconut oil from a coconut milk emulsion, the most difficult part is the separation of oil from the proteins and cell wall matter. The yield of coconut oil prepared by wet extractions is usually less unless modifications are done to the extraction method. In a wet extraction method, conditions for destabilization of coconut milk emulsion have been studied by employing different treatments such as thermal, pH, enzyme, chilling and combination of enzymes and chilling treatments (Raghavendra & Raghavarao, 2010). The coconut milk emulsion was treated with protease enzyme at 0.1% concentration and incubated at 25°C and 37°C for 2 h. Out of these treatment methods, the most effective method for destabilization of coconut milk emulsion was found to be combination treatments, which gave a high yield of 94.5% oil. The oil obtained by combination of treatments contained lower free fatty acids and peroxides and higher lauric acid content compared to commercial coconut oil.

In addition to the separation of coconut oil from a coconut milk emulsion, enzyme treatment can also be used to extract coconut oil from finely divided copra meal. Coconut oil extraction based on the enzymatic action of polygalacturonases, α -amylase and proteases on a diluted coconut paste has been tested (McGlone et al., 1986). After the reaction with enzymes, the mixture gave three phases upon centrifugation. The upper phase contained high quality coconut oil and the middle layer and the lower layer contained water and coconut meal respectively. This process gave a yield of 80%, which is a much higher yield compared to other traditional wet extraction methods. Enzyme assisted extractions have also been used to extract coconut oil from powdered copra. The crude commercial enzyme used in this study contained α -amylase, neutral protease, acid protease, cellulase/hemicellulase, and pectinase. The enzyme treatment in this process can be considered as a pretreatment of copra prior to the oil extraction. The enzyme was added 1% rate of the copra and allowed to stand for 30 min. After enzyme treatment, the meal was extracted by hot water and the emulsion was boiled to evaporate water. This enzyme pretreatment of copra prior to the extraction improved the yield of coconut oil by 50% compared to the same extraction procedure without enzyme

pretreatment. However, the quality of oil extracted by enzyme pretreatment method was similar to the oil extracted without enzyme pretreatment (Tano-Debrah & Ohta, 1997).

Supercritical Carbon Dioxide in the Extraction of Coconut Oil

Extraction of coconut oil by supercritical carbon dioxide (SC-CO₂) extraction has been reported by several authors. CO_2 is a nonpolar solvent at supercritical conditions and triglycerides in coconut oil are easily soluble in CO_2 . In a study, coconut kernel has been dried by a microwave-assisted method and the resultant dried kernel material has been used to extract oil by supercritical extraction (Quitain et al., 2003). The sample of grated coconut kernel was dried using microwave-assisted low-temperature air-drying method. The microwave frequency used in the drier is 2.45 GHz and this is similar to the frequency of domestic microwave ovens. The typical operating temperature was set at 40°C. About 95% of oil recovery can be achieved within 20 h using this method while the cold press method of virgin coconut oil extraction gives a recovery of about 50%. Therefore, the supercritical extraction method seems to be more efficient than the cold press method. However, the economic aspects of the two methods have not been compared. In the supercritical extraction method, the drying method, whether it is microwave assisted, hot air or freeze drying have no effect on the yield of coconut oil. There is no significant difference in physical properties of super critically extracted oil compared to cold pressed oil. However, the vitamin E content of coconut oil extracted by SC-CO₂ extraction is not significant. Fatty acid compositions showed no significant difference in cold pressed virgin coconut oil, or supercritical oil or ordinary coconut oil prepared by pressing copra. However, phenolic contents of the super critically extracted coconut oil can be different. Due to the low polarity of CO_2 and the high polarity of phenolic compounds, solubility of phenolic substances in CO₂ is poor. Organic modifiers can be coupled with this extraction and the amount and the nature of organic modifier is important for incorporating phenolic substances into oil during this extraction (Palma et al., 2000).

Distinguishing between Virgin Coconut Oil and RBD Coconut Oil

Both virgin coconut oil and RBD coconut oil are colorless and they have low acid values and peroxide values. Therefore, it is difficult to distinguish between these two oils by simple experiments. Attempts have been made to distinguish between virgin coconut oil and RBD coconut oil using standard quality parameters such as free fatty acid content and peroxide value, ³¹P NMR spectroscopy and head space solid phase microextraction gas chromatography coupled to mass spectroscopy (SPME/GCMS) (Dayrit et al., 2011). Virgin coconut oil has higher free fatty acids, moisture and volatile matter and lower peroxide value. ³¹P NMR analysis has shown that virgin coconut oil and RBD coconut oil contained 4.10 w/w% diglycerides. Virgin coconut oil contained 1.55 w/w% and RBD coconut oil contained 4.10 w/w% diglycerides. Virgin coconut oils prepared by different methods such as expeller process, centrifugation process and fermentation with and without heat cannot be distinguished by standard quality parameters. However, SPME/GCMS analysis showed that virgin coconut oil produced by fermentation process with and without

heat can be distinguished from virgin coconut oil produced by expeller or centrifugation process based on higher levels of volatiles such as acetic acid and octanoic acid present in virgin coconut oil produced by fermentation process (Dayrit et al., 2011).

Coconut Hybrids and Coconut Oil Quality

Quality parameters of virgin coconut oil from different hybrids have been compared. For this study, virgin coconut oil was prepared by cold press method and fermentation method. The results showed that some hybrids produced significantly high lauric acid contents compared to their parents. However, there was no significant difference in the lauric acid content of virgin coconut oil extracted by cold pressing and fermentation. This study also compares the free fatty acid contents of virgin coconut oil prepared by cold press method and fermentation method. The results indicate that cold pressed coconut oil contains lower free fatty acids compared to coconut oil prepared by fermentation process. The lowest and highest acid values of the cold pressed coconut oils of the tested varieties were about 0.05 and 0.25 mg KOH/g oil respectively while the lowest and highest acid values for the coconut oils prepared by fermentation process for the tested cultivars were about 0.25 mg KOH/g oil respectively (Arlee et al., 2013).

Phenolic Substances and the Extraction Method

It is well-known that phenolic substances provide protection against the oxidation of unsaturated fatty acids in triglycerides. Phenolic substances are naturally present in seed oils and these phenolic substances improve the shelf life of oils. The naturally present phenolic substances of oils can be removed by passing oils through a column containing activated charcoal and silicic acid (Waraho et al., 2009). Such stripped oils are oxidatively less stable than unstripped oils. Phenolic substances can also be added to oils to improve the oxidative stability. Added phenolic compounds such as caffeic acid and *p*-coumaric acid in avocado and coconut oils improved the oxidative stability and nutritional quality. However, those phenolic acids facilitated the hydrolysis of triglycerides (Sun-Waterhouse et al., 2011). Adding the carotenoids and vitamin E fraction back to refined, bleached and deodorized palm oil reduced its atherogenicity (Wilson et al., 2005).

Total phenol contents of seed oils vary significantly with the extraction conditions. Solvent-extracted oils contain more phenolic substances than virgin oils (Gutfinger, 1981). The most striking difference between the traditional coconut oil extracted by prolonged heating of coconut milk and copra oil is in the phenol content. Total phenol content of traditional coconut oil was 618 ± 46 mg while that of copra oil was 91 ± 11 mg as gallic acid/kg oil (Seneviratne & Dissanayake, 2008). This clearly indicates that extraction methods have a remarkable impact on the total phenol content of coconut oil.

The effect of the use of cell-wall-degrading-enzyme preparations during the mechanical extraction process of virgin olive oil on the phenolic compounds and polysaccharides have been investigated. The use of the enzyme preparations increased the concentration of phenolic compounds in olive oil. Especially, the contents of secoiridoid derivatives such as the dialdehydic form of elenolic acid linked to 3, 4-dihydroxyphenylethanol (3, 4-DHPEA-EDA)

and an isomer of oleuropein aglycon (3, 4-DHPEA-EA), which have high antioxidant activities, increased significantly in the olive oil (Vierhuis et al., 2001).

According to a recent study, virgin coconut oil prepared by fermentation process contains higher amount of phenolic substances compared to virgin coconut oil prepared by other methods. Total phenol contents of virgin coconut oil prepared by fermentation process by leaving coconut milk standing for 12 h, chilling coconut milk and RBD coconut oils measured as gallic acid equivalents are about 250 μ g/g, 180 μ g/g and 120 μ g/g respectively (Marina et al., 2009). The total phenol contents of coconut oil extracted by chilling coconut milk and traditional coconut oil prepared by boiling coconut milk emulsion have been compared (Seneviratne et al., 2009). In the cold extraction, coconut milk was chilled at 10°C for 10 h to solidify the lipids. Then the aqueous layer was discarded and the lipid block was allowed to stand at 30°C until it dissolved completely. Then the oil layer was separated by centrifugation. In the traditional extraction, coconut milk emulsion was heated to 100-120°C and the water in the emulsion was evaporated. Finally, the oil was decanted from the deposits at the bottom of the container. According to this study, the maximum total phenol content as gallic acid equivalents of cold pressed coconut oil was 66 mg/kg oil and that of traditional coconut oil was 449 mg/kg oil.

The higher phenolic antioxidant contents of coconut oil prepared at high temperatures can be easily explained by considering the polarities of phenolic substances and coconut oil. Coconut oil can be considered as a non-polar substance and oil dissolves in nonpolar solvents such as hexane. On the other hand phenolic substances are polar substances and, therefore, the solubility of polar phenolic substances in nonpolar coconut oil is low. More phenolic substances are incorporated into coconut oil when oil is extracted by wet extraction with prolonged boiling of coconut milk emulsion. High temperatures improve the solubility of phenolic substances in coconut oil. In the cold extraction methods, the phenolic substances are not properly incorporated into coconut oil due to the mild temperature conditions.

In the wet extraction methods, the phenol content of coconut oil also depends on the nature of the endosperm components used to make coconut milk emulsion (Seneviratne et al., 2009). White coconut kernel is a poor source of phenolic substances with only 61 mg/kg of phenolic substances. Phenolic contents of copra, copra meal (coconut cake or poonac, which is the residue after the extraction of coconut oil by pressing copra) and coconut testa are 405 mg/kg, 2156 mg/kg and 3946 mg/kg respectively (Seneviratne & Dissanayake, 2006). Therefore, when white kernel is used in the extraction of virgin coconut oil under mild temperature conditions, high quantities of phenolic substances are not incorporated into coconut oil as in the high temperature methods. Even when coconut oil is produced by boiling coconut milk, the phenolic content of coconut oil will be lower if only white coconut kernel is used for the preparation of coconut milk. Coconut oil produced using only the white coconut kernel by any method contains lower amount of phenolic substances. Coconut oil from the pairings which contain the rind or testa has slightly higher phenolic content, higher iodine value and higher free fatty acid content compared to the coconut oil extracted from white coconut kernel or meat. Though these differences do not affect the nutritional quality of coconut oil, the oil extracted from pairings is considered as low quality coconut oil (Zajew, 1956).

Identified Phenolic Substances

Phenolic substances present in coconut oil have been identified and quantified. HPLC with diode array detection and mass spectroscopic methods have been used for this purpose (Seneviratne et al., 2009). The identified phenolic substances present in coconut oil extracted by boiling coconut milk are given in Table 6. The quantities of phenolic substances are compared for copra oil and traditional coconut oil prepared by boiling coconut milk extracted from white coconut kernel are given in Table 7.

Table 6. Quantities of phenolic substances present in coconut oil prepared by boiling coconut milk

Phenolic compound	Quantity mg/kg oil
Gallic acid	20.2 ± 10.1
(-)-Epigallocatechin	26.7 ± 1.7
(+)-Catechin	81.7 ± 22.7
<i>p</i> -Hydroxybenzoic acid	4.8 ± 1.0
(+)-Epicatechin	1.4 ± 0.6
Caffeic acid	4.6 ± 1.5
Syringic acid	4.1 ± 0.9
Ferulic acid	22.1 ± 8.9

Adapted from Seneviratne et al., 2009 (Seneviratne et al., 2009). Each data point represents the mean of five replicates ± standard deviation.

Table 7. Comparison of phenolic substances present in copra oil and traditional coconut oil prepared by boiling coconut milk based on white coconut kernel

Compound		Quantity (mg/kg)
	Copra oil	Traditional coconut oil
(+/-)-Catechin	0.87 ± 0.1	2.9 ± 0.4
Caffeic acid	0.13 ± 0.06	3.0 ± 1.1
<i>p</i> -Coumaric acid	0.34 ± 0.01	2.0 ± 0.2
Ferulic acid	0.31 ± 0.2	3.3 ± 1.2
Unidentified phenolic acids	11.2 ± 2.1	97.2 ± 5.7
Unidentified flavonoids	1.6 ± 0.3	10.8 ± 5.0

Adapted from Seneviratne & Dissanayake, 2008 (Seneviratne & Dissanayake, 2008). Each data point represents the mean of three replicates ± standard deviation.

Vitamin E Contents

Comparison of tocopherol contents of cold pressed and fermented coconut oils from different coconut cultivars indicate that the cold pressed method provides highest tocopherol content in coconut oil. According to this study, cold pressed coconut oil contains tocopherol contents ranging from 3.17 mg/100 g oil to 5.78 mg/100 g oil for different coconut cultivars. Coconut oil produced by fermentation process contained tocopherol contents ranging from

2.14 mg/100 g oil to 4.15 mg/100 g oil for different coconut cultivars. According to the same study, the total phenol contents of cold pressed coconut oils were slightly higher compared to fermented coconut oil for each cultivar tested. However, the differences are not very important nutritionally as the values are very close to each other (Arlee et al., 2013).

As a general statement, fatty acid composition of coconut oil is more or less same in all the coconut oils extracted by different methods. However, the unsaponifiable matter content varies significantly with the method of extraction. Vitamin E content in coconut oil is usually low and the quantities are negligible in coconut oils extracted by dry methods. However, virgin coconut oil extracted by cold wet methods contains higher amounts of vitamin E. Vitamin E and phytosterols in virgin coconut oil prepared by chilling coconut milk and copra oil have been compared and the results are given in Table 8 (Nevin & Rajamohan, 2009). Studies also show that not only vitamin E levels but also vitamin A levels are higher in virgin coconut oil compared to copra oil. When exposed to sunlight for several days, vitamin content of virgin coconut oil decreased sequentially, which confirms the loss of vitamins from copra when exposed to UV radiation from sunlight (Nevin & Rajamohan, 2006).

	Virgin coconut oil	Copra oil
Vitamin E (µg/100g)	30.87	12.76
Campesterol (ng/dL)	17.00	25.07
Stigmasterol (ng/dL)	63.13	57.05
β-Sitosterol (ng/dL)	73.03	57.00

Table 8. Vitamin E and phytosterol contents of coconut oils

Adapted from Nevin & Rajamohan, 2009 (Nevin & Rajamohan, 2009).

Antioxidant Capacities and Related Health Benefits

It is common opinion that saturated fat is bad for health and it increases cholesterol levels in blood. However, when considering health effects of saturated fat, it is also important to consider the nature of saturated fatty acids. In this case, chain lengths of fatty acids are important, as the metabolic pathways of fatty acids depend on the chain length of fatty acids. A study has been conducted with ¹⁴C labeled synthetic triglycerides containing saturated fatty acids such as butyric acid (C 4:0), caprylic acid (C 10:0), lauric acid (C 12:0) and palmitic acid (C16:0). Feeding Sprague-Dawley rats with these synthetic triglycerides indicated that the excretion rate of ¹⁴CO₂ as the end product of fatty acid metabolism decreases with the chain length of fatty acids. This observation is probably due to, at least in part, differences in the route of transport of the fatty acid in coconut oil is lauric acid and percentage of the administered radioactivity recovered as ¹⁴CO₂ in the breath of rats after 48 hours was 69% compared to 50% in longer chain palmitic acid (Kirschner & Harris, 1961). These results indicate that shorter chain fatty acids oxidizes more easily instead of entering the fat deposits compared to longer chain fatty acids.

Antioxidant capacity of coconut oil varies with the phenolic substances present in coconut oil. As the contents of phenolic substances and other antioxidants depend on the method of extraction of coconut oil, antioxidant capacities of coconut oils also vary with the

extraction method. The antioxidant properties of virgin coconut oil produced through chilling and fermentation have been investigated and compared with those of RBD coconut oil. Virgin coconut oil showed better antioxidant capacity than RBD coconut oil. The virgin coconut oil produced through the fermentation method had the strongest scavenging effect on 1,1diphenyl-2-picrylhydrazyl and the highest antioxidant activity based on the β -carotenelinoleate bleaching method. However, virgin coconut oil obtained through the chilling method had the highest reducing power (Marina et al., 2009). These antioxidant activities, which correlated with the phenolic contents were lower in RBD coconut oil. During the refining and bleaching steps natural phenolic antioxidants are removed and antioxidant capacities decrease accordingly.

In another study, rats were fed with diets containing virgin coconut oil prepared by chilling coconut milk and copra oil. After 45 days, antioxidant vitamin levels were higher in rats fed with virgin coconut oil. Low-density lipoprotein (LDL) isolated from the rats fed with virgin coconut oil was found to be more resistant to oxidation compared to LDL isolated from rats fed with copra oil or sunflower oil. Virgin coconut oil also showed beneficial effects on blood coagulation. The lower levels of fibrin and fibrinogen in virgin coconut oil fed rats compared to copra oil fed rats indicate that the blood clotting tendency decreased in virgin coconut oil fed rats (Nevin & Rajamohan, 2008).

Phenolic fractions of virgin coconut oil prepared by chilling coconut milk and copra oil have been compared for their potential to inhibit Cu²⁺ induced *in vitro* LDL oxidation (Nevin & Rajamohan, 2004). The results indicated that the phenolic fraction of virgin coconut oil prepared by chilling coconut milk has better potential to inhibit LDL oxidation compared to the phenolic fraction of copra oil as measured by carbonyl formation. Most of the studies on health effects of coconut oil, do not report the origin or method of extraction of coconut oil. As mentioned earlier, the fatty acid composition of coconut oil does not significantly vary with extraction method. However, due to the differences in non-lipid components, coconut oil prepared by chilling coconut milk has a better beneficial effect in lowering lipid components compared to copra oil. Lower levels of total cholesterol, triglycerides, phospholipids, LDL and VLDL cholesterol levels and higher high-density lipoprotein (HDL) cholesterol levels were observed in rats fed with virgin coconut oil compared to rats fed with copra oil (Nevin & Rajamohan, 2004).

The beneficial effects of virgin coconut oil extracted by wet extraction by chilling coconut milk and by boiling coconut milk have also been compared. The antioxidant activity of the phenolic fraction of coconut oil prepared by boiling coconut milk, tested by DPPH radical scavenging assay and deoxyribose degradation assay is at least two times higher than that of the phenolic fraction of coconut oil prepared by chilling coconut milk. Serum Trolox equivalent antioxidant activity studies of the rats fed with special diets containing traditional coconut oil prepared by boiling coconut milk and coconut oil prepared by chilling coconut milk indicate that traditional coconut oil improves the serum antioxidant status more pronouncedly compared to coconut oil prepared by chilling coconut milk (Seneviratne et al., 2009). These antioxidant activity studies indicate that most of the antioxidant substances of coconut oil are stable at high temperatures. The polyunsaturated oils have a lower thermal stability in the saponifiable fraction due to various oxidation processes of fatty acids possible at high temperatures. However, due to high saturated fat content of coconut oil and due to the thermal stability of phenolic substances in coconut oil, high temperature extraction of coconut oil appears to not cause any nutritional damage to coconut oil. A later study also further

verifies the findings on the properties of traditional and cold-extracted coconut oils. In this study reported in 2011, coconut oil was prepared from coconut milk emulsion by cold and hot extraction under similar conditions used by Seneviratne et al. (2009) (Siddalingaswamy et al., 2011). The results of this study indicated that coconut oil extracted by boiling coconut milk at high temperatures (393 K, or 120°C) reduced blood glucose, total cholesterol, triglycerides, LDL, and very low density lipoprotein. This oil also elevated activities of antioxidant enzymes such as superoxide dismutase, catalase, glutathione peroxidase, and glutathione while it decreased lipid peroxidation in the liver more effectively than coconut oil extracted by freezing the coconut milk emulsion. The improved antioxidant properties are due to the increased amounts of phenolic and other antioxidants present in coconut oil extracted at high temperatures from coconut milk emulsion. The results of these studies further prove that antioxidants present in coconut oil are stable at high temperatures and their activities remain unaffected even at 120°C. According to this study, the total phenol contents of commercial copra oil, cold-extracted coconut oil and hot-extracted coconut oil are 64, 75, and 242 $\mu g/g$ respectively. Some antioxidants may inactivate or destroy at high temperatures. For example, α -tocopherol was destroyed during the processing of raw corn and other selected foods (Wyatt, 1998). Hydroxytyrosol derivatives present in virgin olive oil were also destroyed due to thermal oxidation at high temperatures (Nissiotis & Margari, 2002). Even though high temperatures may destroy or inactivate certain phenolic compounds in coconut oil, the better incorporation of thermally stable phenolic substances during high temperature extractions may compensate for this loss.

Polycyclic Aromatic Hydrocarbons

Polycyclic aromatic hydrocarbons (PAHs) are a class of carcinogenic compounds produced due to incomplete combustion of organic compounds. At high temperatures, certain bonds in organic compounds are broken and small unstable fragments are formed due to a process called pyrolysis. These unstable fragments are mostly radicals, which recombine to give PAHs. The process, which occurs at high temperatures, is called pyrosynthesis. Due to their nonpolar nature, PAHs can easily dissolve in oils. Both light and heavy PAH levels have been tested in common edible oils such as coconut, groundnut, mustard, olive, palm, rice bran, safflower, sesame, soybean and sunflower. Among 296 samples tested from these oils, 88.5% of the samples contained total PAH levels above 40 μ g/kg and range was 40.3 to 624 μ g/kg indicating that PAHs are common contaminants of edible oils (Pandey et al., 2004). PAH contamination in seed oils can occur in various stages of oil preparation. One major source of contamination is the seed or kernel drying by direct fire. Refining process can remove PAH considerably. Levels of low molecular weight PAHs containing up to four aromatic rings can be reduced by deodorization. The main process of removing PAHs from edible oils is charcoal treatment.

In coconut oil, PAH contamination occurs during the drying of coconut kernel. Coconut oils prepared from kernels dried in the sun, in direct heating kilns and in indirect heating kilns have been tested for PAH levels (Wijeratne et al., 1996). According to this study, coconut kernels dehydrated inside the husk and shell by prolonged storage (ball copra) did not contain any PAH. Smoke dried coconut kernels contained in the order of 100 μ g/kg PAHs. Unrefined

coconut oil contained 359 μ g/kg of total PAHs and extremely low levels of PAHs (11 μ g/kg) were detected in desiccated coconut, dried using an indirect drying method. The results also indicate that modifying the drying method by avoiding direct fire or smoke during copra preparation lowers the PAH levels in coconut oil.

Among PAHs, fluoranthene, pyrene, and benzo(a)anthracene are light PAHs while benzo(b)fluoranthene, benzo(k)fluoranthene, benzo(a)pyrene, dibenzo(a,h) anthracene and benzo(g,h,i)perylene are considered as heavy PAHs, which are known to be carcinogenic. This study showed that 95% of PAHs in the tested coconut kernel products were light PAHs and heavy PAHs were only 5%. The PAH levels of soybean oil, mustard oil and coconut oil have been compared. For this purpose, PAHs in oils have been extracted by liquid-liquid extraction using acetonitrile–acetone 60:40 (v/v) (Hossain & Salehuddin, 2012). The study showed that soybean oil contained lowest PAH levels compared to mustard oil and coconut oil. As the study has been conducted for oil samples available in the market, the processing methods and conditions of oil preparation are not clear. However, results indicate that both coconut oil and soybean oil contain levels of PAHs that are below the accepted limits by the World Health Organization (WHO). Mustard oil has relatively high levels of PAH exceeding the WHO recommended maximum value for safety.

Various purification steps remove PAH from soybean oil. The main refining process includes degumming, neutralization, bleaching and deodorization. Because of the presence of PAHs in raw soybeans and solvents, PAH levels in solvent extracted soybean oils were high and reached 45.16µg/kg. Degumming can remove enzymes, microorganisms and other impurities to ensure the stability of the oil. The PAH content decreased to 12.00µg/kg with the removal of impurities in the degumming process because most of the light PAHs in extracted oils were removed. Deacidification is one of the important steps in the soybean oil refining process. Its main purpose is to remove free fatty acids in crude oils and to remove impurities such as pigments, phospholipids, hydrocarbons and mucilaginous materials. The deacidification of soybean oils is achieved by neutralization of free fatty acids with alkaline solution. The content of PAHs was further reduced after the deacidification process and the total amount of PAHs in neutralized soybean oils decreased to 6.64µg/kg. In the decolorization process, the content of light PAHs has a slight decrease showing that the bleaching clay has a little adsorption capacity for light PAHs. The purpose of the deodorization step is to remove unpleasant odors and remaining free fatty acids. It further reduced light PAHs and the total PAHs in the deodorized oil decreased to 1.88µg/kg. Degumming has the most prominent effect of decreasing the amount of light PAHs, followed by neutralization and deodorization (Yu et al., 2014).

Aflatoxin Levels

Many oil seeds such as groundnut, cottonseed, coconut, soybean and sunflower are susceptible to fungal attacks in the field or during storage. The natural mycotoxins produced by *Aspergillus flavus* and *Aspergillus parasiticus* are the main aflatoxins in seed oils. Humans can be exposed to aflatoxins by consumption of contaminated food, contributing to an increase in nutritional deficiencies, immunosuppression and hepatocellular carcinoma (Wagacha & Muthomi, 2008). Among the 18 different types of aflatoxins identified, the

major ones are, aflatoxin B1, aflatoxin B2, aflatoxin G1, aflatoxin G2, aflatoxin M1 and aflatoxin M2, which are produced by *Aspergillus flavus* and/or *Aspergillus parasiticus*.

The aflatoxin contamination in coconut oil occurs during the preparation of copra. Therefore, the quality of coconut oil depends on the quality of copra used in the extraction of coconut oil. According to a study conducted in Sri Lanka, the aflatoxin levels of coconut oil prepared by small-scale mills are higher than that of oils prepared in large-scale mills. The reason is, in small-scale mills low quality copra with fungal contaminations are used for the oil extraction while better-processed copra is used in large scale mills. According to this study, the tested coconut oils from the small-scale mills contained a mean value of 186 ppb of aflatoxin B1 in 115 samples while coconut oil from large-scale mills contained only about 50 ppb of aflatoxin B1.

In the dry extraction of coconut oil, copra is produced in various ways. During the rainy season, levels of aflatoxins in copra and coconut oil increases. Fungal growth occurs in copra when the moisture level of copra is above 6%. During any curing process, the moisture level in copra decreases to 6%. However, due to high humidity in rainy season, copra can absorb moisture from the atmosphere so that the moisture levels will be high enough for fungal growth. In addition, some mills store copra for some time before pressing for oil production. During this period, fungal contaminations occur (Samarajeewa & Arseculeratne, 1983).

CONCLUSION

In evaluating the quality of coconut oil, several quality parameters are considered. Parameters such as acid value, peroxide value, iodine number, saponification value, anisidine value and fatty acid composition in triglycerides are the parameters related to the lipid fraction of coconut oil. Among these quality parameters, acid value, peroxide value and anisidine value are important in quickly monitoring the commercial quality of coconut oil and these parameters vary significantly with the method of extraction. Variation of iodine value and saponification value with the extraction method of coconut oil is not significant even though iodine value varies with the quality of copra used for the extraction of coconut oil. Fatty acid composition of the triglycerides of coconut oil remains relatively unaffected by the method of extraction even though the content of some minor fatty acids vary by small percentages.

As the fatty acid composition of coconut oil is independent of the method of extraction, it can be assumed that any health effect purely connected to the fat content may not vary with the type of coconut oil. However, non-lipid components of coconut oil significantly vary with the method of extraction. As most of these minor compounds are polar in nature, their solubility in coconut oil is higher at high temperatures. Therefore, coconut oil prepared by boiling coconut milk contains higher amounts of phenolic antioxidants compared to other coconut oils. Virgin coconut oil produced under cold conditions can also retain thermally unstable non-lipid matter such as vitamins, some phenolic compounds and several other compounds such as sterols. Copra oil contains lower amount of important non-lipid compounds. Therefore, coconut oil prepared by wet extraction under cold or hot conditions seem to confer better beneficial health effects compared to copra oil.

REFERENCES

- APCC (2009). Asian and Pacific Coconut Community. *Apccsec.org*. Downloaded from <www.apccsec.org/standards.htm> on 5th February, 2015.
- Arlee, R., Suanphairoch, S., Pakdeechanuan, P., (2013). Differences in chemical components and antioxidant-related substances in virgin coconut oil from coconut hybrids and their parents. *Intl. Food Res. J.* 20(5), 2103-2109.
- Balasubramanium, K., (1996). Polysaccharides of the kernel of maturing and matured coconuts. J. Food Sci. 41, 1370-1373.
- Banzon, J. A., Velasco, J. R., (1982). Coconut production and utilization. PCRDF Manila.
- Chen, B.-K., Diosady, L. L., (2003). Enzymatic aqueous processing of coconuts. *Int. J. Appl. Sci. and Eng.* 1, 55-61.
- Dayrit, F. M., Buenafe, O. E. M., Chainani, E. T., de Vera, I. M. S., Dimzon, I. K. D., Gonzales, E. G., Santos, J. E. R., (2007). Standards for essential composition and quality factors of commercial virgin coconut oil and its differentiation from RBD coconut oil and copra oil. *Philippine J. Sci.* 136(2), 119-129.
- Dayrit, F. M., Dimzon, I. K. D., Valde, M. F., Santos, J. E. R., Garrovillas, M. J. M., Villarino, B. J., (2011). Quality characteristics of virgin coconut oil: Comparisons with refined coconut oil. *Pure Appl. Chem.* 33, 1789-1799.
- Guarte, R. C., Mühlbauer, W., Kellert, M., (1996). Drying characteristics of copra on quality of copra and coconut oil. *Postharvest Biol.Tech.* 9, 361-372.
- Gutfinger, T., (1981). T. Polyphenols in olive oils. J. Am. Oil Chem. Soc. 68, 966-968.
- Hossain, M. A., Salehuddin, S. M., (2012). Polycyclic aromatic hydrocarbons (PAHs) in edible oils by gas chromatography coupled with mass spectroscopy. *Arabian J. Chem.* 5, 391-396.
- Kirschner, S. L., Harris, R. S., (1961). The effect of chain length on the metabolism of saturated fatty acids by the rat. *J. Nutrition* 73, 397-402.
- Marina, A. M., Che Man, Y. B., Nazimahi, A. H., Amin, I., (2009). Antioxidant capacity and phenolic acids of virgin coconut oil. *Int. J. Food Sci. Nut.* 60(2), 114-123.
- McGlone, O. C., Canales, A. L. M., Carter, J. V., (1986). Coconut oil extraction by a new enzymatic process. J. Food Sci. 51, 695-697.
- Moigradean, D., Poiana, M-A., Gogoasa. I., (2012). Quality characteristics and oxidative stability of coconut oil during storage. *Journal of Agroalimentary Processes and Technologies*, 18(4), 272-276.
- Neela, S., Prasad, N. B. L., (2012). Induced fermentative production of virgin coconut oil. *As. J. Food Ag-Ind.* 5(05), 355-363.
- Nevin, K. G., Rajamohan, T., (2004). Beneficial effects of virgin coconut oil on lipid parameters and *in vitro* LDL oxidation. *Clin. Biochem.* 37, 830-835.
- Nevin, K. G., Rajamohan, T., (2006). Virgin coconut oil supplemented diet increases the antioxidant status in rats. *Food Chemistry* 99, 260-266.
- Nevin, K. G., Rajamohan, T., (2008). Influence of virgin coconut oil on blood coagulation factors, lipid levels and LDL oxidation in cholesterol fed Sprague-Dawley rats. *e-SPEN The Eur e-Journal of Clin. Nut. and Met.* 3, e1-e8.

- Nevin, K. G., Rajamohan, T., (2009). Wet and dry extraction of coconut oil: impact on lipid metabolic and antioxidant status in cholesterol coadministered rats. *Can. J. Physiol. Pharmacol.* 87, 610-616.
- Nissiotis, M., Margari, M. T., (2002). Changes in antioxidant concentration of virgin olive oil during thermal oxidation. *Food Chem.* 77, 371-376.
- Palma, M., Taylor, L. T., Zoecklein, B. W., Douglas, L. S., (2000). Supercritical fluid extraction of grape glycosides. J. Agric. Food Chem. 48, 775-779.
- Pandey, M. K., Mishra, K. K., Khanna, S. K., Das, M., (2004). Detection of polycyclic aromatic hydrocarbons in commonly consumed edible oils and their likely intake in the Indian population. J. Am. Oil Chem. Soc. 81, 1131-1136.
- Quitain, A. T., Moriyoshi, T., Goto, M., (2003), Coupling Microwave-Assisted Drying and Supercritical Carbon Dioxide Extraction for Coconut Oil Processing. *Chem. Eng. and Sci.* 1, 12-16.
- Raghavendra, S. N., Raghavarao, K. S. M. S., (2010). Effect of different treatments for the destabilization of coconut milk emulsion. J. Food Eng. 97, 341-347.
- Reverchon, E., Osseo, L. S., (1994). Comparison of processes for the supercritical carbon dioxide extraction of oil from soybean seeds. J. Am. Oil Chem. Soc. 71, 1007-1012.
- Reverchon, E., De Marco, I., (2006). Supercritical fluid extraction and fractionation of natural matter. J. Supercrit. Fluids 38, 146-166.
- Rodrigo, M. C. P., Amarasiriwardena, B. L., Samarajeewa, U., (1996). Some observations on copra drying in Sri Lanka. *Cocos* 11, 21-31.
- Rosenthal, A., Pyle, D. L., Niranjan, K., (1996), Aqueous and enzymatic process for edible oil extraction. *Enzyme Microb. Tech.* 19, 402-420.
- Samarajeewa, U., Arseculeratne, S. N., (1983). A survey of aflatoxin contamination of coconut products in Sri Lanka: Incidence, origins and recommendations. J. Natn. Sci. Coun, Sri Lanka 11, 225-235.
- Seneviratne, K. N., Dissanayake, D. M. S., (2005). Effect of method of extraction on the quality of coconut oil. J. Sci. Univ. of Kelaniya 2, 63-72.
- Seneviratne, K. N., Dissanayake, D. M. S., (2006). Analysis of phenolics antioxidants in coconut cake by high performance liquid chromatography (HPLC). *Herbal Medicine Phytopharmaceuticals and other Natural Products: Trends and Advances*. Colombo: Centre for Science and Technology of the Non-aligned and other Developing Countries (NAM S&T Centre), India and Institute of Chemistry, Ceylon, Sri Lanka 162-166.
- Seneviratne, K. N., Dissanayake D. M. S., (2008). Variation of phenolic content in coconut oil extracted by two conventional methods. *Int. J. Food Sci. Tech.* 43, 597-602.
- Seneviratne, K. N., Hapuarachchi, C. D., Ekanayake, S., (2009). Comparison of the phenolicdependent antioxidant properties of coconut oil extracted under cold and hot conditions. *Food Chem.* 114, 1444-1449.
- Siddalingaswamy, M., Rayaorth, A., Khanum, F., (2011). Anti-diabetic effects of cold and hot extracted virgin coconut oil. *J. Diabetes Mellitus* 1, 118-123.
- Sun-Waterhouse, D., Thakorlal, J., Zhou, J., (2011). Effects of added phenolics on the storage stability of avocado and coconut oils. *Int. J. Food Sci Tech.* 46, 1575-1585.
- Tano-Debrah, K., Ohta, Y., (1997). Aqueous extraction of coconut oil by an enzyme-assisted process. J. Sci. Food Agric. 74, 497-502.

- Vierhuis, E., Servili, M., Baldioli, M., Schols, H. A., Voragen, A. G. J., Montedoro, G. F., (2001). Effect of enzyme treatment during mechanical extraction of olive oil on phenolic compounds and polysaccharides. J. Agric. Food Chem. 49, 1218-1223.
- Wagacha, J. M., Muthomi, J. W., (2008). Mycotoxin problem in Africa: current status, implications to food safety and health and possible management strategies. *Int. J. Food Microbiol.* 124, 1-12.
- Waraho, T., Cardenia, V., Rodriguez-Estrada, M., McClements, D. J., Decker, E. A. (2009). Prooxidant mechanisms of free fatty acids in stripped soybean oil-in water emulsions. J. Agrc. Food Chem. 57, 7112-7117.
- Wijeratne, M. C. P., Samarajeewa, U., Rodrigo, C. P., (1996). Polycyclic aromatic hydrocarbons in coconut kernel products. J. Natn. Sci. Coun. Sri Lanka 24, 285-297.
- Wilson, T. A., Nicolosi, R. J., Kotyla, T. T., Sundram, K., Kritchevskyc, D., (2005). Different palm oil preparations reduce plasma cholesterol concentrations and aortic cholesterol accumulation compared to coconut oil in hypercholesterolemic hamsters. J. Nut. Biochem. 16, 633-640.
- Wyatt, C. J., Carballido, S. P., Mendez, R. O., (1998). Tocopherol content of selected foods in Maxican diet. J. Agrc. Food Chem. 46, 4657-4661.
- Yu, Y., Jin, Q., Wang, Y., Gu, Y., Wang, X., (2014). Sources of Polycyclic Aromatic Hydrocarbons in Soybean Oil and its Dynamic Changes Refining Processing. Adv. J. Food Sci. Tech. 6(1) 42-47.
- Zajew, M., (1956). The question of differences between iodine numbers of coconut oil and of the corresponding soapstock fatty acids. J. Am. Oil Chem. Soc. 33, 306-308.

Chapter 7

QUALITY ANALYSES AND AUTHENTICATION OF COCONUT OIL

Irina Mirela Apetrei^{1,*} and Constantin Apetrei²

¹Faculty of Medicine and Pharmacy, "Dunarea de Jos" University of Galati, Galati, Romania ²Faculty of Sciences and Environment, "Dunarea de Jos" University of Galati, Galati, Romania

ABSTRACT

Coconut oil contains mainly triacylglycerols and free fatty acids. Coconut oil is a very important source of medium chain fatty acids. Three important medium chain fatty acids exist in coconut oil: caprylic acid (C8:0), capric acid (C10:0) and lauric acid (C12:0), lauric acid being about 50% of the total fatty acids content. Medium chain fatty acids exhibit good properties in different metabolism pathways. The unsaponifiable matter consists principally of sterols, tocols, squalene, phenols, color compounds and odor compounds such as γ - and δ -lactones. Physico-chemical analysis can be used for quality detection and monitoring of coconut oils. The virgin coconut oils authentication is of central importance for consumers, food processors and food industries. Virgin coconut oil can be adulterated with other vegetable oils affecting the quality and physicochemical and organoleptic properties. Palm kernel oil, canola oils and others vegetable oils were used as adulterants in different studies. These oils were chosen because of their similarities with chemical composition of virgin coconut oil. The principal analytical techniques used for authentication of virgin coconut oil are refractometry, Fourier transform infrared spectroscopy, differential scanning calorimetry, electronic nose, fiber optical sensor and nuclear magnetic resonance spectroscopy. The analytical technique results combined with multivariate data analysis tools (principal component analysis, discriminant analysis, partial least square regressions, etc.) were able to detect adulteration down to 1% limit.

^{*} Corresponding Author address: Department of Pharmaceutical Sciences, Faculty of Medicine and Pharmacy, "Dunărea de Jos" University of Galați, 47 Domneasca Street, 800008 Galați, Romania. E-mail address: irina. apetrei@ugal.ro.

Keywords: coconut oil, quality, physico-chemical analysis, authentication

INTRODUCTION

Coconut oil is obtained from copra, which is the dried kernel of coconuts. The coconut palm is the species *Cocos nucifera L*, which grows well in the humid regions near to the equator. Usually, fresh coconut kernel contains (% weight) moisture (50%), oil (34%), ash (2.2%), fibre (3.0%), protein (3.5%) and carbohydrate (7.3%) (Hui 1996; APCC 2006; Gopala Krishna et al., 2010).

Virgin coconut oil is obtained from the mature kernel of the coconut by mechanical or physical means with or without the application of minimal heat (Dia et al., 2005). Coconut oil contains about 90% saturated fats. The oil contains predominantly triglycerides with 86.5% saturated fatty acids, 5.8% monounsaturated fatty acids and 1.8% polyunsaturated fatty acids (Gunstone 2011; Hui 1996; Marina et al., 2009).

Raw coconut oil (or virgin coconut oil) is colorless at or above 30°C and it is white when in its solid form, being an unctuous mass. Coconut oil melts at 25°Cand becomes solid below this temperature. The typical smell of coconut oil is that of coconuts (if not refined, bleached and deodorized). Coconut oil is insoluble in water at room temperature (APCC 2006; Gunstone 2011).

Refined coconut oil is the oil obtained from the dried, solid part of the endosperm of *Cocos nucifera L*., then refined. It appears as a white or almost white, unctuous mass, insoluble in water, easily soluble in methylene chloride and in light petroleum and very slightly soluble in alcohol (Baldwin 1986; European Pharmacopoeia).

Coconut oil contains mainly triacylglycerols and free fatty acids. Typical fatty acid composition of coconut oil is 16:0 (9 wt%); 18:1 (6 wt%); 18:2 (2 wt%); 8:0 (8 wt%); 10:0 (7 wt%); 12:0 (48 wt%); 14:0 (18 wt%). As can be seen, coconut oil is rich in medium chain fatty acids and exhibits good digestibility (Che Man & Marina 2006; Marina et al., 2009a).Coconut oil contains among 0.5%-0.8% of unsaponifiable matter (Malaysian standard (MS2943 2007; Codex 2003). The term *unsaponifiable matter* refers to those substances present in oils that are not saponified by alkali hydroxides and are extractable into ether. This unsaponifiable matter consists principally of sterols, tocols, squalene, pheols, color compounds, carbohydrates and odor compounds (such as lactones). The characteristic odor and taste of coconut oil is principally due to γ - and δ -lactones, which are present in trace amounts (Young 1983).

METHODS FOR THE ANALYSIS OF MAIN CONSTITUENTS OF COCONUT OIL

Determination of Triacylglycerols

The analysis of vegetable oils by their triacylglycerol (TAG) content has assumed a great importance as far as quality control and possible origin determination (Graciani-Constante et al., 1997; Nollet & Toldra 2012; Aitzetmuller 1993).

The various TAG of oils, may be separated and quantified by high-performance liquid chromatography (HPLC), according to their carbon numbers (Moreda et al., 2003; Aitzetmüller et al., 1988). The values obtained can then serve to distinguish them from other oils or to interpret their properties. For example, the percentage of trilinolein (LLL) was adopted as criterion to detect the presence of seed oils in olive oils (EEC 2568/91). TAGs are effectively separated byHPLC on reversed-phase columns, containing silica with chemically bonded to octadecyl groups (RP-18) as stationary phase. Using acetonitrile as mobilephase, separation of TAGs occurs according to the chain length and degree of unsaturation of the fatty acids in the glycerol moiety (Schulte 1981). Other solvents were used as mobile phase. One of the most frequently employed is acetone indifferent proportions. In the standard official methods, acetone/acetonitrile (1 volume: 1 volume) is used as mobile phase (IUPAC 1987; EEC 2472/97).

Regarding the differences between oils, only oils that are rich in one fatty acid contain much monoacid triacylglycerol, for example, olive oil, linseed oil and sunflower oil containing triolein (OOO), trilinolein (LLL) and trilinolenin (LnLnLn) (Reske et al., 1997).

The triacylglycerol compositions of several oils are presented in the Table 1.

Coconut oil		Olive oil		Soybean o	Soybean oil	
Triacylglycerol	wt %	Triacylglycerol	wt %	Triacylglycerol	wt %	
12, 12, 8	12	OOL	11	LnLL	7	
12, 12, 10	6	000	43	LnLO	5	
12, 12, 12	11	POP	3	LLL	15	
12, 12, 14	11	POL	4	LLO	16	
14, 12, 8	9	POO	22	LLS	13	
		StOO	5	LOO	8	
				LOS	12	
				OOS	5	

Table 1. Triacylglycerol composi	tion of several oils	(molecular speci-	es (wt%))
----------------------------------	----------------------	-------------------	-----------

where: L-linoleic acid; Ln-linolenic acid; O-oleic acid; P-palmitic acid; S-saturate acids; St-stearic; 8caprylic acid (8:0); 10-capric acid (10:0); 12-lauric acid (12:0); 14-myristic acid (14:0). (Laureles et al., 2002).

Determination of Fatty Acids

Fatty acids are usually analyzed by gas-liquid chromatography (GLC). This method is applicable to oils containing fatty acids with chain length in the range C14 to C24. GLC analysis of fatty acids is performed following their conversion to methyl ester derivatives. Columns with polar phases are used, for example polyethylene glycol stationary phase (Carbowax) (Jennings 1987).

When samples contain short-chain fatty acids, methods of derivatization have been proposed, for example benzylation and ter-butyldimethyl silylation (Monseur et al., 1981; Burger et al., 1990).

Analysis of free fatty acids can be carried out by Gas Chromatography: Flame Ionization Detection (GC-FID), this technique being an alternative method of analysis (Hajimahmoodi et

al., 2005; Regulation (EEC) N8 2568/ 91, annex X; Lerma García 2012). Furthermore, other analytical methods, such as High-Performance Liquid Chromatography (HPLC) (Shahidi 2005; Kotani et al., 2002), or ³¹P-NMR (Dayrit et al., 2008) were also developed for this purpose.

The use of pre-column HPLC method for the analysis of fatty acid with 2nitrophenylhydrazine hydrochloride was reported (Miwa & Yamamoto 1990). After alkaline hydrolysis of coconut oil free fatty acids are reacted with 2-nitrophenylhydrazine hydrochloride and then derivatized to corresponding fatty acid hydrazides. Each of the derivatives was separated on reversed-phase HPLC with isocratic elution and detected at VIS 400 nm (Miwa & Yamamoto 1996). In Table 2 is presented the typical fatty acid composition of coconut oil.

Table 2. Fatty acid composition of coconut oil as determined by gas liquid chromatography from coconut oil samples (expressed as percentage of total fatty acids)

Fatty acid	Fatty acid concentration (% wt)
C6:0	ND-0.7
C8:0	4.6-10.0
C10:0	5.0-8.0
C12:0	45.1-53.2
C14:0	16.8-21.0
C16:0	7.5-10.2
C18:0	2.0-4.0
C18:1	5.0-10.0
C18:2	1.0-2.5
C18:3	ND-0.2
C20:0	ND-0.2
C20:1	ND-0.2

Adapted from Codex 2003.

Determination of Sterols

The methods for the analysis of total sterols in coconut oil involve sample preparation steps such as saponification, extraction of the neutrals from the soap solution, pre-separation by preparative thin-layer chromatography, silylation of sterols and GC-FID analysis (ECCR N8 2568/1991, annex V; Ham et al., 2000).

The major disadvantage of GC is the requirement of thermally stable columns and the need of chemical derivatization prior to analysis. For this reason, alternative methods were described based on the use of liquid chromatography-mass spectrometry (HPLC-MS) (Santos et al., 2011; Itoh et al., 1973). In Table 3 are presented the levels of desmethylsterols in crude coconut oil samples.
Sterols	Fatty acid concentration (% wt)
Cholesterol	ND-3.0
Brassicasterol	ND-0.3
Campesterol	6.0-11.2
Stigmasterol	11.4-15.6
Beta-sitosterol	32.6-50.7
Delta-5-avenasterol	20.0-40.7 1
Delta-7-stigmastenol	ND-3.0
Delta-7-avenasterol	ND-3.0
Others	ND-3.6
Total sterols $(mg \times kg^{-1})$	400-1200

 Table 3. Levels of desmethylsterols in crude coconut oil samples as a percentage of total sterols

Adapted from (Firestone 2013).

In general, the desmethylsterols are useful markers to assess authenticity. Taking into account that beta-sitosterol is the most abundant sterol in the majority of oils, its value has only limited use for the authenticity assessment and differentiation of vegetable oils.

Determination of Tocopherols and Tocotrienols

Tocopherol itself is a term that constitutes eight different subtypes namely alpha-, beta-, gamma- and delta-tocopherol and alpha-, beta-, gamma-and delta-tocotrienols.

Tocopherols analysis is usually performed by using high performance liquid chromatography (HPLC) with fluorescence detection. Coconut oil samples are dissolved in n-hexane and n-hexane/tetrahydrofurane/2-propanol can be used as the mobile phase (Mansor et al., 2012; Nollet & Toldra 2012).

In Table 4 are presented the levels of tocopherols and tocotrienols in crude coconut oil samples.

Tocopherols and Tocotrienols	Concentration (mg×kg ⁻¹)
α-tocopherol	ND-17
β-tocopherol	ND-11
γ-tocopherol	ND-14
α-tocotrienol	ND-44
γ-tocotrienol	ND-1
Total (mg×kg ⁻¹)	ND-50

Adapted from Codex 2003.

Determination of Volatile Compounds

Volatile compounds of coconut oil were determined by GC-MS. Volatile compounds can be identified and quantified by analysis of the headspace using methods such as static head space, dynamic headspace (purge-and-trap), direct thermal desorption, and solid phase microextraction (SPME) techniques (Santos et al., 2011).

Different studies on coconut oil have identified hydrocarbons, alcohols, aldehydes, ketones and δ -lactones that have significant contributors to its aroma (Pai et al., 1970; Allen, 1965; Lin & Wilkins 1970).

The distinct coconut odor has been shown to be due to δ -octalactone (Padolina et al., 1987; Maarse 1991).

Other approach was proposed for the determination of volatile compounds, based on the use of gas sensor arrays with overlapping selectivities. The electronic nose consists of an array of gas sensors with different selectivity, a signal collecting unit and suitable pattern recognition software (Gardner and Bartlett 1999).

Different gas sensors are presently used for construction of electronic noses. They are based on different sensitive materials (i.e., metal oxides, conducting polymers etc.) and detection principles (resistive, piezoelectric, optical, electrochemical sensors, etc.) (Patel 2014).

Determination of Phenolic Compounds

Virgin coconut oil is a important source of phenolic compounds. Total phenolic contents of the virgin coconut oil are quantified by means of Folin-Ciocalteu method. For detection a quantification of of phenolic compounds several methods have been described. The most used technique has been HPLC, coupled to florescence detection (Seneviratne & Sudarshana Dissanayake 2008), diode array detector (Seneviratne et al., 2009), or UV-Vis detector (Santos et al., 2013).

CHEMICAL AND PHYSICAL METHODS OF ANALYSIS OF COCONUT OIL CHARACTERISTICS

The most important physico-chemical characteristics included in national and international standards and used for coconut oil quality control are as follows. In the case of virgin coconut oils the physico-chemical characteristics include the free fatty acids content, iodine value, saponification value, refractive index, specific gravity, unsaponifiable matter, moisture and impurities. All these characteristics are related to the authenticity of the coconut oil and the values are decreasing during refining. For refined coconut oils, the physico-chemical characteristics also usually include the color, peroxide value, slip melting point and solid fat content. All these characteristics are indications of quality and behavior during cooking or using as food (MMAF 2005).

Determination of the Free Fatty Acids content

The free fatty acids content (acid value or acidity index) is the number of mg of potassium hydroxide required to neutralize the free fatty acids in 1 g of the coconut oil.

The chemical reaction involved is:

 $R-COOH + KOH \longrightarrow R-COOK + H_2O$

The acidity is an expression of the content (in %, weight/weight; percentage) of free fatty acids as content of dominant or chosen fatty acid. In the case of coconut oil the chosen fatty acid is lauric acid that has the molecular weight of 200 g×mol⁻¹ (I.S. 1964)

Iodine Value

The iodine value gives primarily a measure of unsaturation of a lipid being directly proportional to the degree of unsaturation (Yasuda 1931). Therefore, a higher iodine value indicates a greater number of C = C double bonds. The iodine value is expressed as the grams of iodine bonded per 100g of lipid (A.O.A.C. 2000c).

The methods for determining the iodine value of lipids are Hanus method and Wijs method. The lipid to be analyzed is weighed and dissolved in a suitable organic solvent, to which a known excess of iodine monobromide (or the monochloride) is added. Some of the IBr reacts with the double bonds in the unsaturated lipids, whereas the rest remains unreacted:

$$R-CH=CH-R + IBr_{excess} \longrightarrow R-CHI-CHBr-R + IBr_{remaining}$$

The amount of IBr that has reacted is determined by measuring the amount of IBr remaining after the reaction has gone to completion ($IBr_{reacted} = IBr_{excess} - IBr_{remaining}$). The amount of IBr remaining is determined by adding excess potassium iodide to the solution to release iodine and then titrating with a sodium thiosulfate ($Na_2S_2O_3$) solution in the presence of starch to determine the concentration of iodine released:

$$IBr_{remaining} + 2KI \longrightarrow KBr + KI + I_2$$

$$I_2 + 2Na_2S_2O_3 \xrightarrow{\text{starch}} 2NaI + Na_2S_4O_6$$

In the case of unhydrogenated solid oils, the iodine value is a very good measure of consistency and it is inversely proportional to the melting point of lipid. An increase in iodine value indicates high susceptibility of lipid to oxidative rancidity. In addition, there is a very good correlation between the iodine value, the sleep melting point of unhydrogenated fats of the same species.

The Codex standard (Table 8) specifies the iodine value range of coconut oil as 6.3-10.6, while the Malaysian Standard (Table 7) and give the iodine value range of coconut oil as 7.5-

10.5. The Codex (2003) standard is shown in Table 5 and the Malaysian standard for crude and refined coconut oil (MS 2007) is shown in Table 5.

Oil	Grams of iodine absorbed per 100g of oil
Coconut oil	6.3-10.6
Palm kernel oil	14.1-21.0
Palm oil	50.0-55.0
Olive oil	75-94
Peanut oil	86-107
Cottonseed oil	100-123
Corn oil	103-135
Soybean oil	124-139
Safflower oil	136-148
Sunflowerseed oil	118-141
Rapeseed oil	94-120
Grapeseed oil	128-150

Table 5. Typical iodine values for several oils

Adapted from Codex 2003.

Saponification Value

Saponification value is a measure of the molecular weights of the triacylglycerides in coconut oils. The triacylglycerides with high value of saponification value are considered to make better quality soaps than those with low saponification value. As definition, saponification value is the number of milligrams of potassium hydroxide required to saponify completely the fatty material present and also to neutralize the free fatty acids present in one gram of the coconut oil.

A scheme of the saponification process is presented below:



The saponification value is an indirect measure of the amount of free acids present in coconut oils (I.S.I. 1984d).

Ester value is another important parameter. It is defined as the number of milligrams of KOH required to combine with fatty acids present in the triacylglyceride form in 1 g of oils. It is a measure of amount of triacylglyceride present in a sample of oil, which is saponificable.

Refractive Index

The refractive index (n) of coconut oil is the ratio of the velocity of light in a vacuum to its velocity in the coconut oil. The refractive index is also defined as the ratio of the sine of the angle of incidence (i) to the sine of the angle of refraction (r):

$$n = \frac{\sin i}{\sin r}$$

It varies with the wavelength of the light used in its measurement and with the temperature. Therefore, it is necessary to specify these conditions when the values are reported. Refractive indices are usually stated in terms of sodium light of wavelength 589.3 nm (line D) at a temperature of 20 ± 0.5 °C (n_D^{20}). The Abbe's refractometer is used for determination of refractive index with very good accuracy (Khanna & Gulati 1985; A.O.A.C. 2000b). In the case of coconut oil determination of refractive index is carried out at 40°C.

The measurement of the refractive index can be employed to establish the identity and authenticity of coconut oil.

Specific Gravity

The specific gravity (or relative density) of coconut oil is the ratio of the mass of the coconut oil in air at 30°C to that of an equal volume of water at the same temperature. Determination of relative density (d_{30}^{30}) is carried out using a hydrostatic balance (if it is necessary a precision with three decimal digits) or a pycnometer.

The specific gravity of coconut oils, containing only carbon, hydrogen and oxygen is less than one (A.O.A.C. 2000).

Unsaponifiable Matter

The term unsaponifiable matter is applied to the substances non-volatile at 100-105°C obtained by extraction with an organic solvent from the oil to be examined after it was saponified. The result is calculated as percent weight/weight (I.S.I. 1984e).

Saponification of the coconut oil is usually done with ethanolic potassium hydroxide solution. After that, extraction of the unsaponifiable matter is then carried out with petroleum ether for several times.

Unsaponifiable mater of coconut oil is a variable mixture of hydrocarbons, alcohols, aldehydes, ketones, pigments and fat-soluble vitamins which occur naturally or may be formed during processing or degradation of the oil (Moura Fe et al., 1975).

Moisture and Impurities

The determination of the water content is one of the most frequently used methods in laboratories for foods around the world. One method is the drying method using drying ovens Free ebooks ==> www.Ebook777.com

Irina Mirela Apetrei and Constantin Apetrei

or infrared lamps at temperature 105°C (I.S.I. 1984). By this method the loss on drying is determined and not just the water content because the method is not specific.

The titrimetric determination of water by the Karl Fischer method depends on the reaction that takes place quantitatively between water and a reagent consisting of sulfur dioxide and iodine in anhydrous pyridine and methanol. The reaction is carried out in methanol.

The chemical reaction is:

$H_2O + CH_3OH + I_2 + SO_2 + R_3N \rightarrow [RNH]SO_4CH_3 + 2[RNH]I$

In contrast to drying, this is a specific method. In the absence of side reactions only water will be determined (Pande 1974).

The term *insoluble impurities* refers to extraneous substances such as dirt, debris and fibers. They are defined as those substances which remain insoluble and can be filtered off after the oil is dissolved in a specific solvent such as petroleum.

Volatile Mater

The matter volatile at 105°C in oil sample is usually determined according to I.S.I. procedure (I.S.I. 1984). The test sample is heated at105°C until moisture and volatile matter were completely removed and the loss in mass was recorded. The result was expressed in percentage by mass.

Color

Coconut oil in its virgin form is clear in color. High heat exposure during deodorization process (230°C) normally will lead to polymerization and consequently the oil will turn yellow. Color of virgin coconut oil in this survey was in range of 0.1 Red and 0.5 Yellow max.

The liquid oil samples are placed in a Lovibond 5 $\frac{1}{4}$ " cell and the color is determined at 30°C by achieving the best possible match with the standard color slides of red and yellow indices using a Lovibond tintometer.

The method determines the color of oils by comparison with Lovibond glasses of known color characteristics. The color of coconut oil is expressed as the sum total of the yellow and red slides used to match the color of the oil in a cell of the specified size in the Lovibond tintometer (I.S.I. 1984b).

The color of the oil calculated in terms of Lovibond units as follows:

Color reading = (a Y + 5 b R) or (a Y + 10 b R)

in (5 1/4" cell) where

140

www.Ebook777.com

a: sum total of the various yellow slides (Y) used;b: sum total of the various red (R) slides used;

Y + 5R is the mode of expressing the color of light colored oils; Y + 10 R is for the dark-colored oils.

Peroxide Value

Peroxide value is a measure of the amount of peroxidesformed in fats and oils through autoxidation and oxidation processes (Rohman et al., 2011). It is a measure of the degree of initialoxidation of fats and oils.

The peroxide value is determined iodometrically according to standard methods for the oils analysis and the results are expressed in meq O_2/kg oil (AOCS 1998).

The sample is treated in solution with a mixture of acetic acid and a suitable organic solvent and then with a solution of potassium iodide. The liberated iodine is titrated with a standard solution of sodium thiosulfate in presence of starch.

Peroxides and similar products which oxidize potassium iodide under the conditions of the test will contribute to the peroxide value. Peroxide values are expressed either in milliequivalents of O_2/kg oil.

The peroxide value is determined by measuring the iodine released from potassium iodide by peroxide, using sodium thiosulfate solution as the titrant. In acetic acid solution, the reaction scheme for is as follows:

 $R-H+O_{2} \longrightarrow ROOH$ $KI+H_{3}C-COOH \longrightarrow HI+H_{3}C-COO^{-}K^{+}$ $ROOH+2HI \longrightarrow ROH+H_{2}O+I_{2}$ $I_{2}+2Na_{2}S_{2}O_{3} \xrightarrow{starch} 2NaI+Na_{2}S_{4}O_{6}$

The reaction of peroxides of the structures R-O-O-R' follows an analogous pathway. Cyclic peroxides do not react quantitatively under these conditions.

Alternatively, the ion reaction is of more of general applicability:

$$O_2^{2-} + 2I^- + 4H^+ \longrightarrow I_2 + 2H_2O$$

 $I_2 + 2S_2O_3^{2-} \xrightarrow{\text{starch}} 2I^- + S_4O_6^{2-}$

The Reichert Value

The Reichert value is the number of millilitres of 0.1N aqueous sodium hydroxide solution required to neutralize steam volatile water soluble fatty acids distilled from 5g of an

oil/fat under the given conditions. The Reichert value is a measure of water soluble steam volatile fatty acids mainly butyric and caproic acids present in oil or fat (I.S.I. 1984f).

The Polenske Value

The Polenske value is the number of mililiters of 0.1N aqueous alkali solution required to neutralize steam volatile water insoluble fatty acids distilled from 5g of the oil/fat under the prescribed conditions. It is a measure of the steam volatile and water insoluble fatty acids, chiefly caprylic, capric and lauric acids present in oil or fat (I.S.I. 1984f).

Melting Point

Coconut oil is predominantly mixture of triglycerides. Coconut oil does not exhibit either a definite or sharp melting point. For that reason the melting point does not mean the same characteristics that it does with pure crystalline substances.

Melting point is undoubtedly the most widely used measure of the consistency of oils and fats.

Fats pass through a stage of gradual softening before they become completely liquid. Therefore, the melting point depends by the specific conditions of the method by which it is determined. The test is empirical and has a large experimental error. There are several versions of the method. One method is open-tube capillary-slip method. In this method, the melting point is considered the temperature at which the oil or fat softens or becomes sufficiently fluid to slip or run as determined by the open-tube capillary-slip method (I.S.I. 1984c).

Neither Codex, or the Malaysian Standard specify a melting point for coconut oil but a literature survey it were found 24.2-25.0°C ranging values (Gunstone 2011; Firestone 2013).

Other Physical Characteristics

Other physical characteristics of coconut oil (Table 6) such as density, viscosity, heat capacity and heat of fusion, are important for theoretical considerations and for food engineering purposes.

For example, viscosity is an important property. Viscosity of the oil determines the diameter of pipes and the power needed for pumping or stirring of oils. Viscosity is also important for taste and mouth feel, for example, in the case of fried foods.

SPECIFIC CHARACTERISTICS OF COCONUT OIL

Coconut oil is commercially important oil. Coconut oil differs significantly from other fats and oils in that that it pass abruptly from a brittle solid to a liquid, within a narrow temperature range. Coconut oil is a hard brittle solid at ambient temperature (21.1°C), but it

melts sharply and completely below body temperature. The sharp melting characteristics of the coconut oils are derived from the similarity of the melting points of the triglycerides. Coconut oils leave a clean, cool, nongreasy sensation on the palate, which is difficult to match with other oils (Gopala Krishna et al., 2010).

More than 90% of the coconut oil fatty acids are saturated, which accounts for its excellent oxidative stability. It is the richest source of medium-chain triglycerides, which are composed of C-6, C-8 and C-10 fatty acids. Due to the high medium-chain triglycerides content, coconut oil is a major component of infant formulas and medical foods for people who cannot absorb longer chain fatty acids (Erickson 1990).

Parameter	Value or range
Boiling point (°C) at 760mmHg	298.9
Boiling point (°C) at 1mmHg	130.2
Dielectric constant	ca. 3.1
Heat of combustion $(cal \times g^{-1})$	9020
Heat of fusion $(cal \times g^{-1})$	46.2
Heat of vaporization (cal×g ⁻¹)	69
Heat of vaporization (cal×g ⁻¹)	51
Kinematic viscosity (C. Stokes) (37.8°C)	29.79
Kinematic viscosity (C. Stokes) (98.9°C)	6.06
Refractive index (n_D at 40°C)	1.448-1.450
Relative density (40°/20°C)	0.908-0.921
Slip melting point (°C)	23.0-25.0
Specific gravity (20°/4°C)	0.9226
Specific gravity change/°C	$7.13 imes 10^{-4}$
Specific heat (cal×g ⁻¹) 66°C	0.510
Specific heat (cal×g ⁻¹) 97°C	0.530
Vapor pressure (mmHg) 188°C	0.001
Vapor pressure (mmHg) 244°C	0.05

Table 6. Miscellaneous physical properties of coconut oil

Data adapted from (Swern 1979; Codex 2003; Cocks & Van Rede 1966).

Table 7. Malaysian standard MS 239 (1987): requirements for coconut oil

Characteristics	Crude Refined		Refined
	Grade 1	Grade 2	
Free fatty acids (as lauric acid) (%) max.	1.0	3.5	0.1
Moisture and insoluble impurities, (%) max.	0.50	0.50	0.10
Iodine value (Wijs), g I ₂ /100g of oil, range		7.5-10.5	
Color (5 ¼") Lovibond cell, max.	3R	4R	1.5R
Refractive index at 40°C, range	1.4480-1.4490		
Specific gravity at 30/30°C, range	0.915-0.920		
Saponification value (mg KOH×g ⁻¹), range	248-264		
Unsaponifiable matter (%) max.	0.8		0.5

Adapted from Gunstone 2011).

Chemical and physical characteristics	Values
Relative density 40/20°C	0.908-0.921
Refractive index (n _D 40°C)	1.448-1.450
Saponification values (mgKOH×g ⁻¹)	248-265
Iodine value (g $I_2/100$ g of oil)	6.3-10.6
Unsaponifiable matter $(g \times kg^{-1})$	≤ 15
Matter volatile at 105°C	0.2% m/m
Insoluble impurities	0.05% m/m
Soap content	0.005% m/m
Iron (Fe)	virgin oil 0.5 mg/kg
	refined oil 5.0 mg/kg
Copper (Cu)	virgin oil 0.1 mg/kg
	refined oil 0.4 mg/kg
Acid value	virgin oil 4.0 mgKOH/g oil
	refined oil 0.6 mgKOH/g oil
Peroxide value	virgin oil up to 15 me/kg
	refined oil up to 10 me/kg
Reichert value	6-8.5
Polenske value	13-18

Table 8. Codex (2001) draft standard for coconut oil

Adapted from (O'Brien 2008; Gunstone 2011).

Table 9. Coconut oil composition and physicochemical properties

Characteristic	Range
Unsaponificable matter, %	0.1-0.8
Solidification point, °C	14-22
AOM stability, h	30-250
Oxidative stability index (110°C), h	8.5-85

Adapted from (Fritsch et al., 1971; Gunstone 2011).

The color of crude coconut oil varies from a light yellow to brownish yellow. The National Institute of Oilseed Products specification limit is 15.0 maximum Lovibond red color. Normal processing techniques will produce deodorized oils with very pale yellow colors. The odor and taste of coconut oil are largely due to the presence of small quantities of lactones (less than 150 ppm). Because coconut oil is low in unsaturated fatty acids, it has a high resistance to oxidation; however, coconut oil will hydrolyze two to ten time faster than normal oils to produce a disagreeable soapy flavor. Coconut oil hydrolysis proceeds slowly in the presence of free moisture alone, but rapidly when an enzyme lipase is present in the food product. Pastry dough and cake batters are examples of such products (O'Brien 2008).

The typical characteristics for the coconut oil are presented in Tables 7, 8 and 9.

Trade Specifications

The specifications and test values of oils provide a useful insight into their properties and can be used as guides for good quality and/or authenticity.

The quality of commercial coconut oil quality is based on trade specifications which give rounded values for a small number of basic characteristics. The specifications of the Malaysian Edible Oil Manufacturers' Association (MEOMA 2001) for crude and refined coconut oils are shown in Table 10 and those of a major EU refiner for both refined and hydrogenated CNO are shown in Table 11.

Table 10. MEOMA specifications for crude and refined coconut oil: for export market (MEOMA 2001)

Parameter	Crude coconut oil,	Crude coconut	RBD coconut
	Grade 1	oil, Grade 2	oil
	Values		
Free fatty acids (as lauric acid)	1.0% max.	3.5%max.	0.1% max.
Moisture and insoluble matter	0.5% max.	0.5% max.	0.1% max.
Iodine value (Wijs), g I ₂ /100g of oil	7.5-10.5	7.5-10.5	7.5-10.5
Color (5 ¼" Lovibond cell)	-	-	Red 1.5 max.

Adapted from (Gunstone 2011).

Table 11. Selling specification for RBD coconut oil and hydrogenated coconut oil

Characteristics	RBD [*] coconut oil	Hydrogenated
		coconut oil
Color (5 ¼") Lovibond cell, max.	1.2R-4.5Y	1.2R-4.5Y
Free fatty acids (as lauric acid), % max.	0.1	0.1
Peroxide value, milliequivalents O ₂ /kg oil max.	1.0	1.0
Iodine value (Wijs), g I ₂ /100g of oil	7-10	0-2
Saponification value, mgKOH/g oil	255-260	255-260
Slip melting point, °C	24-26	32-34

Adapted from Codex 2009, 2011, 2013.

^{*} RBD - refined, bleached and deodorized coconut oil.

CLASSIFICATION OF COCONUT OILS

Different terms employed for commercial coconut oils (Fife 2004; Gopala Krishna et al., 2010; Kamariah et al., 2008). Some of them are presented in the following sentences.

100% Pure Coconut Oil

The term 100% pure coconut oils are describing two categories of oils: a) 100% purenatural (only minor treatments were used for obtaining pure oil); b) 100% pure-refined (the purity was achieved by refining, bleaching and deodorizing or other treatments).

Virgin Coconut Oil

Virgin coconut oil is the clear oil extracted from fresh coconut milk by coconut wetprocess, without refining. Oil is extracted only from fresh high quality coconuts and extraction and purification processes are through mechanical means only. The resulting coconut oil has a very low free-fatty acid content. Virgin coconut oil has very long shelf life, natural coconut aroma and flavor, light viscosity and non-oily character.

RBD Coconut Oil

RBD refers refined, bleached and deodorized coconut oil. The process for coconut oil obtaining consists in raw material preparation, extraction of oil and purification. Extracted crude oil is treated with alkali to remove free fatty acids. In the subsequent stage oil is treated with steam under vacuum to remove odors and flavors. In the last stage oil is filtered through carbon to decolorize yellow or dark colors. RBD coconut oil is edible oil and can be used in food and cosmetic applications.

AUTHENTICATION AND ADULTERATION STUDIES ON VIRGIN COCONUT OIL

The oils authentication, especially functional food oil such as virgin coconut oil and virgin olive oil is of crucial importance for consumers, food processors and food industries (Rossell et al., 1985; Fakhri & Qadir 2011). The adulteration often involved the replacement of high-cost ingredients with cheaper substitutes. Virgin coconut oil can be adulterated with other vegetable oils of lower commercial value or with similar composition (Fakhri & Qadir 2011).

Although the adulteration is carried out by economic reasons, the action can affect the quality of food, where virgin coconut oil is one of the food components (Flores et al., 2006).

Adulteration has been a problem in the oil trade for a long time. It is sometimes deliberate, sometimes accidental. Indeed, accidental contamination is hard to avoid in modern bulk handling installations. However, it is sometimes remarked that it is the expensive oil that usually gets contaminated with the cheaper one. Numerous tests were developed for the characterization of oils and fats in order to characterize and identify the oils. These include determination of iodine value, to give a measure of oil's unsaturation and of saponification value, which gives a measure of the average molecular weight of the constituent fatty acids (Nollet & Toldra 2012).

The knowledge of vegetable oil characteristics is important not only with regard to the commercial importance of establishing edible oil authenticity, but also with regard to the need to comply with foodstuff labeling legislation in many countries. Descriptive sensory analysis can be carried out to differentiate virgin coconut oil and refined, bleached and deodorized (RBD) coconut oil samples (Villarino et al., 2007).

Palm kernel and coconut oils both contain about 47% lauric acid. This gives the oils close similarities in physical and chemical properties. The oils are different and it is important to be able to distinguish between them (Fakhri & Qadir 2011). The fatty acid profile plays a key role to the physicochemical properties of the oil rather than evaluation of oils based on sensory evaluation especially visual appearance (Sodamade et al., 2013).

Palm kernel and coconut oils are somewhat alike in physical and chemical properties. Nevertheless the slight differences in their properties are real. It is frequently necessary to distinguish the oils, e.g., for labelling purposes, or decide on their proportions in a blend.

In one study, palm kernel and coconut oils samples were analyzed for triglyceride, fatty acid, sterol and tocopherol compositions. The melting points were also determined. Statistical analysis of the results has shown that a combination of values from the carbon number analysis differentiates palm kernel and coconut oils. Furthermore, carbon number analysis can be used to determine the proportion of each in a blend (Rossell et al., 1983).

Other study based on physicochemical properties has shown that coconut oil can be distinguished from similar oils such as palm kernel oil. Significant differences are observed for iodine value (≤ 11 for coconut oil and ≤ 19 for palm kernel oil). Other parameters are related to content of oleic, stearic and linoleic acids, the sum of these values should not be greater than 11.5% for coconut oil and 22% for palm kernel oil. Furthermore, it is a clear difference between coconut oil and palm kernel oil regarding the content of fatty acids with 10 or fewer carbons. The content is higher in coconut oil in comparison with palm kernel oil (Hamilton 2012). Triglyceride analysis was also useful since palm kernel oil contains higher levels of C-46 to C-54 triglycerides than coconut oil (Ashurst & Dennis 2013).

Comparison of the sterols esters of coconut and palm kernel oil showed enough differences to form a basis for distinguishing between the two oils (Yildiz 2009; Gordon & Griffith 1992). Coconut oil had a lower β -sitosterol content and higher Δ^5 -avenasterol content than palm kernel oil (Ashurst & Dennis 2013).

In a study including numerous samples of coconut oil from a variety of geographical origins have been analyzed for their overall fatty acid composition, triglyceride carbon number composition, tocol concentrations, sterol compositions and slip melting points. The use of triglyceride carbon number analysis has been useful for distinguish between coconut oil, palm kernel oil and their blends. The coconut oil samples contained only low levels of the known tocopherols and tocotrieneols in comparison with other oils. Sterol analysis was carried out by Downes (Downes 1982) in a related project on a selection of the same set of samples. The major sterols in coconut oil were β -sitosterol, Δ -avenasterol and stigmasterol. The relatively high, Δ -avenasterol level in coconut oil can be used to distinguish palm kernel from coconut oil.

Virgin coconut oil, a relative newcomer in the fats and oil market and could be valuable oil, comparable with virgin olive oil. For that reason, it is important to establish reliable purity criteria to assure its premium quality. Even physicochemical parameter values are useful for distinguish among oils, the obtaining and the analysis of these are very difficult, time-consuming and require expensive materials and equipments as well as high qualified personal. Therefore, for practical applications in quality control several methods were developed.

Several analytical techniques were used for authentication of virgin coconut oil such as:

- Refractometry
- Fourier transform infrared (FTIR) spectroscopy
- Differential Scanning Calorimetry
- Electronic Nose Technology
- fiber optical sensor
- ⁻¹³C NMR spectroscopy and ³¹P NMR spectroscopy

Refractometry

Refractive index plays an important role in different fields of chemistry, physics and biology. Knowledge of the refractive index of oil is of critical importance in applications of adulteration of oils and purity (Yunus et al., 2009).

Refractive index of oils increases with the increase in unsaturation and also chain length of fatty acids (Sivasankar 2002). Therefore, this parameter can be used for distinguish among oils. The coconut oil and sunflower oil were mixed with palm oil in different percentages. The refractive indices of oils and mixtures were determined by using Abbe's refractometer. The percentage of adulteration of these oils was calculated using refractive index values and the calibration plot. From this study, it was observed adulteration of the coconut oil starting with 30% percentage of adulterant oil in the mixture. This study reveals that the measurement of refractive index can be used as a quality control technique for finding the adulteration of oils but, the sensibility is reduced (Ariponnammal 2012).

In other study, refractive indexes of virgin coconut oil and virgin olive oil were measured and compared. The refractive index for virgin olive oil is higher than the refractive index for virgin coconut oil and this can certificate the purity of the oil (Yunus et al., 2009).

FTIR Spectroscopy

Fourier transform infrared (FTIR) spectroscopy is well known in the analytical field for quantitative analysis of oils and fats (Rohman et al., 2009). FTIR is also becoming popular for authentication analysis.

FTIR spectra of coconut oils present a series of bands with different intensities and forms. Compared with other edible oils and fats, virgin coconut oil has characteristic FTIR spectra. The functional groups responsible for infrared absorption at each frequency in virgin coconut oil spectrum were assigned as follow. The major peak in the region 3,100-2,800 cm⁻¹ is due to C-H stretching mode. The stretching of > C = O is observable in the region 1,700-1,800 cm⁻¹, while wave number region 1,400-900 cm⁻¹ is associated with C-O-C stretching and C-H bending (Rohman et al., 2009; Moigradean et al., 2012).

FTIR spectra of other vegetable oils present some particularities that can be useful in authentication analysis. For discrimination and classification studies discriminant analysis or partial least square (PLS) regression were performed (Manaf et al., 2007; Che Man & Rohman 2013).

In one study, attempts were made to adulterate virgin coconut oil with palm kernel olein. The study was conducted to evaluate the effectiveness of FTIR spectroscopy in detecting adulteration of virgin coconut oil. As adulterant was used palm kernel olein because of its similarity with virgin coconut oil in chemical composition (Manaf et al., 2007.

The spectra of virgin coconut oil and palm kernel oil show the typical characteristic absorption bands for common vegetable oils (Rohman et al., 2009; Moigradean et al., 2012). In the entire range of wave numbers, both virgin coconut oil and palm kernel olein spectra seems almost identical. This is related to lauric acid content of about 47-48% (Rossell et al., 1983). If a meticulous check is carried out some differences between spectra were observed. For instance a peak at 3,006 cm⁻¹ is observed for in palm kernel olein spectrum that is not present in the spectrum of virgin coconut oil. The absorption peakat 3,006 cm⁻¹ are due to the *cis* C = C-H stretching (Rohman et al., 2009; Moigradean et al., 2012). A higher content of unsaturated fatty acids contributes to an absorbance in the region of *cis* C = C-H stretching. This peakcan be used for distinguishing of palm kernel olein from virgin coconut oil.

The band absorbance increases in the range 3,000-3,010 cm⁻¹ upon increase of the percentage of palm kernel oil in a blend.

FTIR spectroscopy was used in combination with partial least square (PLS) to differentiate and quantify these two oils. The calibration plot of PLS regression model was shown a good linearity between the actual value and FTIR predicted value of percentage of palm kernel olein in virgin coconut oil. The differences between the actual adulteration concentration and the calculated adulteration predicted from the model were very small, with a determination coefficient (\mathbb{R}^2) of 0.9973 and root mean error of calibration of 0.0838.

Discriminant Analysis was carried out for pure virgin coconut oil and the blended samples and a classification into two groups, pure virgin coconut oil and adulterated oils was observed. The model demonstrated the classification of pure virgin coconut oil with addition of 1-50% of palm kernel olein using10 principal components. Therefore, detection of adulteration was possible down to 1%.

Other study was carried out to develop a method based on FTIR spectroscopy combined with chemometrics of multivariate calibrations (partial least square and principal component regression) as well as discriminant analysis for quantification and discrimination of canola oil in virgin coconut oil (Che Man & Rohman 2013).

FTIR spectra of virgin coconut oil revealed some differences to other vegetable oils including canola oil, especially in the regions around 3007 cm⁻¹ and 1654 cm⁻¹ and at wave number regions of 1120-1090 cm⁻¹. There was no band at frequencies of 3007 and 1655 cm⁻¹ for virgin coconut oil and the otherwise was observed for canola oil. Furthermore, at frequency region of 1120-1090 cm⁻¹, virgin coconut oil showed one peak and canola oil exhibited two peaks. Therefore, these differences were used for detection and quantification of canola oil in virgin coconut oil.

Quantification of canola oil in virgin coconut oil was performed using multivariate calibrations of partial least square (PLS) and principal component regression (PCR). The relationship between actual and predicted values of canola oil in virgin coconut oil was

Irina Mirela Apetrei and Constantin Apetrei

demonstrated that FTIR spectroscopy combined with PLS can be a promising method for quantification of canola oil in virgin coconut oil.

DA was used to determine the capability of the system to distinguish between pure and adulterated oil. The DA model classified 100% of all samples accurately according to its group therefore the detection limit of this system is 0.5% (volume/volume).

Other study regarding the authentication of virgin coconut oil adulterated with cheaper plant oils such as corn and sunflower oils. Methods involving Fourier transform mid infrared (FT-MIR) combined with chemometrics techniques (PLS and DA) were developed for quantification and classification of corn oil and sunflower oil in virgin coconut oil (Rohman & Che Man 2011). The classification of virgin coconut oil and that adulterated with corn oil was carried out using spectral regions at combined frequencies of 3028-2983, 2947-1887 and 1685-868 cm⁻¹, meanwhile frequencies at 3030-2980 and 1300-1000 cm⁻¹ was exploited for classification of virgin coconut oil adulterated with sunflower oil.

The quality of developed PLS model was evaluated by cross-validation technique. The values of root mean square error of cross validation obtained are relatively low, 1.68% and 1.32% (volume/volume), respectively for corn oil and sunflower. Based on this result, the method developed has a good ability to estimate the percentage of corn oil and sunflower as oil adulterants in virgin coconut oil samples.

Differential Scanning Calorimetry

Differential scanning calorimetry (DSC) provides unique thermal profiling for each oil and can be used to detect cheaper oils or lard adulteration in virgin coconut oil (Biliaderis 1983).

DSC was used to detect adulteration in virgin coconut oil (Marina et al., 2009b). Soybean, sunflower and palm kernel oil were used as adulterants. Each oil was considered representative oil from linolenic, oleic-linoleic and lauric acid group. The heating curves of sunflower and soybean adulterated samples showed that adulteration peak appeared at the lower temperature region starting at 10% adulteration level. For the data analysis, regression analyses using stepwise multiple linear regression (SMLR) was used. SMLR model was able to predict the percent adulterant with a determination coefficient (R^2) of 0.9390 for sunflower oil and 0.9490 for soybean oil. In the case of palm kernel oil, which belongs to the same group as virgin coconut oil (lauric acid), no adulteration peak was observed. For the palm kernel oil, a good relationship between the main exothermic peak height of palm kernel oil and percentage of adulteration was established.

Other work showed the ability of DSC to detect changes in the cooling and heating curves of virgin coconut oil when it is adulterated with lard (Mansor et al., 2012). Lard can be used as an adulterant in virgin coconut oil. The similar physical characteristic of virgin coconut oil to lard makes lard a possible adulterant in virgin coconut oil.

The exothermic and endothermic peaks were changed for adulterated samples comparing with those observed for pure samples. Thus, the DSC curves showed slight changes such as the size increase or reduction of exothermic peaks and the smoothing effect of shoulder peak in the endothermic peak as the percentage of lard adulteration increased.

Two independent DSC parameters were chooses and by means of multiple linear regression analysis prediction of lard percentage adulteration in virgin coconut oil with a

www.Ebook777.com

determination coefficient (R^2) of 0.9582. Therefore, these parameters can be used as a good measurement index in detecting lard adulteration in virgin coconut oil.

In conclusion, DSC is a simple and chemical free method in the study of adulteration in oils.

Electronic Nose

An electronic nose can be useful for analyzing volatile compounds. It works in a way analogous to the way a human nose functions and does not require previous separation of individual volatile compounds. The system provides a pattern of volatile compounds present in the samples. For data analysis chemometrics methods are necessary.

The potential use of electronic nose in detecting adulteration in virgin coconut oil was studied by Marina et al., 2010. An electronic nose (zNoseTM) was applied to the detection of adulteration of virgin coconut oil. The system is based on a surface acoustic wave sensor. Virgin coconut oil was mixed with refined, bleached and deodorized palm kernel olein at a level of adulteration from 1 to 20% (weight/weight).

Principal component analysis (PCA) was used to differentiate between pure and adulterated samples. The PCA provided good separation of samples with 91% of the variation accounted for the PC1 and the PC2. Pure samples formed separated cluster from all adulterated samples.

Excellent results were obtained for the differentiation between pure and adulterated samples down to the 1% detection limit.

The adulterant concentration was quantified using PLS analysis. The correlation between the signals measured with the electronic nose and adulteration percentages of samples was established. The coefficient of determination (R^2) for the model was 0.91, indicating that the model fit the data well.

This study has demonstrated the application of electronic nose as a tool to detect adulteration of virgin coconut oil.

Fiber Optical Sensor

The research field in optical fiber grating technology has opened a new platform for sensor applications (Othonos & Kalli 1999). This technique can be used to discriminate different oils as well as to detect adulteration.

A fiber optic sensing system for the detection of adulteration of coconut oil by paraffin oil was demonstrated (Libish et al., 2011). Paraffin oil is extremely hazardous to human health as and up to 20% of paraffin oil can be mixed with coconut oil without any notable difference in the smell or color of coconut oil. The sensing mechanism is based on the sensitive dependence of the resonance peaks of a long period grating on the changes of the refractive index of the environmental medium surrounding the cladding surface of the grating. The modality of measuring is based on the shift of the attenuation bands. The wavelength shift of the attenuation bands of the long period grating was measured with the sensor immersed in a mixture of paraffin oil and pure coconut oil in different proportions. Detection limit of adulteration was found to be 3% for coconut oil-paraffin oil binary mixture.

¹³C NMR Spectroscopy and ³¹P NMR Spectroscopy

Nuclear magnetic resonance (NMR) techniques were used to quantify and profile refined and virgin oils by analysis of monoacylglycerides (MGs), diacylglycerides (DGs), sterols, free fatty acids (FFAs) and other minor components (Vlahov et al., 2003). These compounds are helpful in adulteration studies.

In one study (Dayrit et al., 2008), phosphorus-31 nuclear magnetic resonance spectroscopy (³¹P NMR) was used to differentiate virgin coconut oil from refined, bleached, deodorized coconut oil. MGs, DGs, sterols and FFAs in virgin coconut oil and refined coconut oil were converted into dioxaphospholane derivatives and analyzed by ³¹P NMR.

As results, it can be mentioned that virgin coconut oil had 40% higher of 1-MGs content than refined coconut oil. On the other hand, virgin coconut had lower DGs content (1.5%) than refined coconut oil (4.1%). Total sterol contents in virgin coconut samples was approximately 0.096%, slightly higher compared to refined coconut oil (0.032%). The FFA contents showed that virgin coconut oil had higher FFA contents (0.127%) than refined coconut oil (0.015%). Principal component analysis showed that the DGs and FFA were the most important parameters for distinguishing virgin coconut oil from refined coconut oil.

³¹P NMR is a useful method for the analysis of constituents in coconut oil and gives a valuable way of distinguishing virgin coconut oil from refined, bleached, deodorized coconut oil.

Another method to detect adulteration is ¹³C NMR spectroscopy Application of ¹³C NMR spectroscopy for detection of castor oil in different edible oils, such as palm oil, coconut oil, groundnut oil and mustard oil, was described (Husain et al., 1993). Quantitative¹³C NMR spectra of oils were recorded in CDCl₃. Characteristic signals observed at different δ values related to some carbons of ricinoleic acid in castor oil were selected for distinguishing it from edible oils. The minimum detection limits for qualitative and quantitative analyses were in the range 2.0-3.0%.

CONCLUSION AND PERSPECTIVES

Virgin coconut oil could be considered functional food oil derived from the kernel of *Cocos nucifera L*. Therefore, the determination of coconut oil authenticity and the detection of adulteration are of great importance for food producers and consumers. The international standards for coconut oil are included in the Codex Alimentarius, the Asian and Pacific Coconut organization and in the national standards of countries that produce coconut oil (India, Malaysia, Philippines, Sri Lanka, Brazil, etc.). The identity and authenticity of the virgin coconut oil can be determined using standard chemical methods of analysis or instrumental analysis. Therefore, these methods can be used for detection of adulteration in virgin coconut oil. Several procedures for authentication of virgin coconut oil were developed based on refractometry, FTIR spectroscopy, differential scanning calorimetry, electronic nose, fiber optical sensor and NMR spectroscopy. For the analysis of data chemometrics techniques were employed. The best methods are able to detect adulterants in virgin coconut oil at levels below 1%. However, the improvement of current methods and the development of novel ones are expected in the future.

The future challenge could consist of developing, validating and harmonizing analytical methods and quality parameters related to authenticity issues. The possible applications are related to the blend of virgin coconut oil with soft treated coconut oil (i.e., soft deodorized) and the blend of virgin coconut oil with other vegetable oil or refined coconut oil.

ACKNOWLEDGMENT

This work was supported by a grant of the Romanian National Authority for Scientific Research, CNCS-UEFISCDI, project number PN-II-ID-PCE-2011-3-0255.

REFERENCES

- A.O.A.C. 17th edn, 2000, Official method 920.212 Specific gravity (Apparent) of Oils, *Pycnometer method/I.S.I. Hand book of Food analysis* (Part XIII) 1984, p. 72.
- A.O.A.C. 17th edn, 2000b, Official method 921.08, *Index of refraction of oils and fats/I.S.I. Handbook of Food analysis* (Part XIII), 1984, p. 70.
- A.O.A.C. 17th edn, 2000c, Official method 920, 159: Iodine absorption number of oils and fats.
- Aitzetmuller, K. (1993). HPLC of triglycerides (fingerprint method). *Fat Sci. Technol.* 95, 361-366.
- Aitzetmüller, K., Wessels, H., Werner, G., (1988). Unterscheidung von Palmölen und Palmöl-Fraktionen durch HPLC der Triglyceride. *Lipid/Fett* 90, 442-447.
- Allen, R. R. (1965). Volatile Flavor Constituents in Coconut Oil. *Chemistry and Industry* 1560.
- AOCS, (1998). Official and Recommended Practices of the American Oil Chemists' Society, Official methods and recommended practices, 5th ed., edited by Firestone, D., Publisher: AOAC Press, Champaign Illinois, USA,
- APCC, Asian and Pacific Coconut Community, (2006), Accessed from http://www.apccsec. org/standards.htm on 30 April 2015.
- Ariponnammal S., (2012). A Novel Method of Using Refractive Index as a Tool for Finding the Adulteration of Oils, *Research Journal of Recent Sciences* 1, 77-79.
- Ashurst, P. R., Dennis, M. J., (2013). Food Authentication, Springer Science & Business Media.
- Baldwin, A. R., (1986). World Conference on Emerging Technologies in the Fats and Oils Industry: Proceedings, The American Oil Chemists Society.
- Biliaderis, C. G., (1983). Differential scanning calorimetry in food research: A review. Food Chemistry 10, 239-265.
- Burger B. V., Munro, Z., Smit, D., Schmidt, U., Wu, C.-L., Tien, F.-C. (1990). Sample introduction in gas chromatography: simple method for the solventless introduction of crude samples of biological origin. *Journal of Chromatography* A 518, 207-214.
- Che Man, Y. B., Marina, A. M., (2006). *Medium chain triacylglycerol*, In F. Shahidi (Ed.), Nutraceutical and specialty lipids and their coproducts, Boca Raton: Taylor & Francis Group.

- Che Man, Y. B., Rohman, A. (2013). Analysis of Canola Oil in Virgin Coconut Oil Using FTIR Spectroscopy and Chemometrics, *Journal of Food and Pharmaceutical Sciences* 1, 5-9.
- Chiavaro, E., (ed.). (2015). *Differential Scanning Calorimetry Applications in Fat and Oil Technology*. CRC Press, Taylor & Francis.
- Cocks L. V., van Rede C., (1966), *Laboratory handbook for oil and fat analysts*, Academic Press.
- Codex Alimentarius Commission. Amended, (2003), Codex Standard for Named Vegetable Oils.
- Dayrit, F. M., Buenafe, O. E. M., Chainani, E. T., Vera, I. M. S. D., (2008). Analysis of monoglycerides, diglycerides, sterols and free fatty acids in coconut (*Cocos nucifera L.*) oil by ³¹P NMR spectroscopy. *Journal of Agricultural and Food Chemistry* 56, 5765-5769.
- Dia, V. P., Garcia V. V., Mabesa R. C., Mendoza E. M. T., (2005). Comparative Physicochemical Characteristics of Virgin Coconut Oil Produced by Different Methods. *Philippine Agricultural Scientist* 88, 462-475.
- Downes, M. J. (1982). Leatherhead Food RA. Technical Circular No. 799.
- ECCR: European Community, Commission Regulation (1991). No. 2568/91 of 11 July 1991 on the characteristics of olive oil and olive pomace oil and on the relevant methods of analysis. *Official Journal of European Communities* L248, 1-83.
- EEC 2472/97. Commission Regulation (EC) No 2472/97 of 11 December 1997 amending Regulation (EEC) No 2568/91 on the characteristics of olive oil and olive-residue oil and on their methods of analysis. *Official Journal of European Communities* L341, 25-39.
- EEC Regulation 2568/91, Commision Regulation (EEC) No. 2568/91 of July 1991. On the characteristics of olive and olive pomace oils and on their analytical methods, *Official Journal of European Communities* L248, 29-32.
- Erickson, D. R., (1990). Edible Fats and Oils Processing: Basic Principles and Modern Practices: World Conference Proceedings. The American Oil Chemists Society.
- European Pharmacopoeia 5.0, 1339-1340.
- Fakhri, N. A., Qadir, H. K., (2011). Studies on Various Physico-chemical Characteristics of Some Vegetable Oils. *Journal of Environmental Science and Engineering* 5, 844-849.
- Fife, B., (2004). The Coconut Oil Miracle, Penguin.
- Firestone, D., (Ed.), (2013). *Physical and Chemical Characteristics of Oils, Fats and Waxes*. Third Edition, AOCS Press.
- Flores, G., Ruiz Del Castillo, M. L., Herraiz, M., Blanch, G. P. (2006). Study of the adulteration of olive oil with hazelnut oil by on-line coupled high performance liquid chromatographic and gas chromatographic analysis of filbertone. *Food Chemistry* 97, 742-749.
- Fritsch, C. W., Weiss, V. E., Anderson, R. H. (1971). Stability of coconut oil in food products. *Journal of the American Oil Chemists Society* 48, 392-393.
- Gardner, J. W., Bartlett, P. N., (1999). *Electronic Noses, Principles and Applications*, Oxford University Press, Oxford.
- Gopala Krishna, A. G., Gaurav, R., Ajit Singh, B., Prasanth Kumar, P. K., Preeti, C., (2010). Coconut Oil: Chemistry, Production and Its Applications: A Review. *Indian Coconut Journal* 15-27.

- Gordon, M. H., Griffith, R. E., (1992). A Comparison of the Steryl Esters of Coconut and Palm Kernel Oils, *Lipid/Fett* 94, 218-221.
- Graciani Constante. E., Janer del Valle M.-L., Maestro Duran, R., (1997). Bibliographical summary on high performance liquid chromatography of lipids: IV 1985-1994. *Grasas y Aceites* 48, 236-247.
- Gunstone, F., (2011). Vegetable Oils in Food Technology: Composition, Properties and Uses, John Wiley & Sons.
- Hajimahmoodi M., Vander Heyden Y., Sadeghi N., Jannat. B., Oveisi, M. R., Shahbazian, S. (2005). Gas-chromatographic fatty-acid fingerprints and partial least squares modeling as a basis for the simultaneous determination of edible oil mixtures, *Talanta* 66, 1108-1116.
- Ham, B., Butler, B., Thionville, P., (2000), Evaluating the Isolation and Quantification of Sterols in Seed Oils by Solid-Phase Extraction and Capillary Gas-liquid Chromatography, *LC-GC Magazine* 18, 1174-1181.
- Hamilton, R. J., (2012). Lipid Analysis in Oils and Fats, Springer Science & Business Media, http://www.apccsec.org/document/VCO-STANDARDS.pdf.
- Hui, Y. H. (ed.), (1996). *Bailey's Industrial Oil & Fat Products*, 5th Edition, Vol. 2, John Wiley & Sons Inc., New York.
- Husain S., Kifayatullah M., Sastry G. S. R., Raju N. P. (1993) Quantitative determination of castor oil in edible and heat-abused oils by ¹³C-NMR. J. Am. Oil Chem. Soc. 70 1251-1254.
- I.S: 548 (Part 1): (1964), Methods of Sampling and Test for Oils and Fats.
- I.S.I. (1984a). Handbook of Food Analysis (Part XIII), p. 62.
- I.S.I. (1984b). Hand book of Food Analysis (Part XIII), p. 75.
- I.S.I. (1984c). Handbook of Food Analysis (Part XIII), p. 68.
- I.S.I. (1984d). Handbook of Food Analysis (Part XIII), p. 78
- I.S.I. (1984e). Handbook of Food Analysis (Part XIII), p. 67
- I.S.I. (1984f). Handbook of Food Analysis (Part XIII), p. 81.
- Itoh T., Tamura, T., Matsumoto, T., (1973), Sterol composition of 19 vegetable oils. *Journal* of the American Oil Chemists Society 50, 122-125.
- IUPAC (1987), Standard Method 2324, *Determination of composition of triacylglycerols in vegetable oils in terms of their partition number by HPLC*, in Standard Methods for the Analysis of Oils, Fats and Derivatives, 7th ed., Oxford, Blackwell.
- Jennings, W., (1987). Analytical gas chromatography, Academic Press.
- Kamariah, L., Azmi, A., Rosmawati, A., Wai Ching, M. G., Azlina, M. D., Sivapragasam, A., Tan, C. P., Lai, O. M., (2008). Physico-chemical and quality characteristics of virgin coconut oil: A Malaysian survey. *Journal of Tropical Agriculture and Food Science* 36, 000-000.
- Khanna, D. R., Gulati, H. R., (1985). *Fundamentals of optics*, R. Chand and Co Publishers, 12th edition.
- Kotani A., Kusu, F., Takamura, K., (2002). New electrochemical detection method in highperformance liquid chromatography for determining free fatty acids. *Analytica Chimica Acta* 465, 199-206.
- Laureles L. R., Rodriguez F. M., Reaño C. E., Santos G. A., Laurena A. C., Mendoza E. M. (2002), Variability in fatty acid and triacylglycerol composition of the oil of coconut (*Cocos nucifera L.*) hybrids and their parentals. *Journal of Agricultural and Food Chemistry* 50:1581-1586.

- Lerma García M. J., (2012). Characterization and Authentication of Olive and Other Vegetable Oils: New Analytical Methods, Springer Science & Business Media.
- Libish, T. M., Linesh, J., Bobby, M.C., Biswas, P., Bandyopadhyay, S., Dasgupta, K., Radhakrishnan, P., (2011), Detection and analysis of paraffin oil adulteration in coconut oil using fiber optic long period grating sensor. *Optik: International Journal for Light and Electron Optics* 122, 1939-1942.
- Lin, F., Wilkins, W., (1970), Volatile Flavor Components of Coconut Meat. *Journal of Food Science* 35, 538-539.
- Maarse, H. (1991), Volatile Compounds in Foods and Beverages. Marcel Dekker, Inc.
- Malaysian Standard MS 2943 (2007). *Virgin coconut oil specification*. Departments of Standards Malaysia. p. 2-3.
- Manaf, M. A., Che Man, Y. B., Hamid, N. S. A, Ismail, A., Abidin, S. Z., (2007), Analysis of adulteration of virgin coconut oil by palm kernel olein using Fourier Transform Infrared Spectroscopy. *Journal of Food Lipids* 14, 111-121.
- Mansor, T. S. T., Che Man, Y. B., Shuhaimi, M. (2012), Employment of Differential Scanning Calorimetry in Detecting Lard Adulteration in Virgin Coconut Oil, *Journal of* the American Oil Chemists Society 89, 485-496.
- Mansor, T. S. T., Che Man, Y. B., Shuhaimi, M., Abdul Afiq, M. J., Ku Nurul, F. K. M., (2012), Physicochemical properties of virgin coconut oil extracted from different processing methods. *International Food Research Journal* 19, 837-845.
- Marina A. M., Man Y. B. C., Amin I. (2010). Use of the SAW sensor electronic nose for detecting the adulteration of virgin coconut oil with RBD palm kernel olein. J Journal of the American Oil Chemists Society 87, 263–270.
- Marina, A. M., Che Man Y. B., Amin, I., (2009). Virgin coconut oil: emerging functional food oil. *Trends in Food Science & Technology* 20, 481-487.
- Marina, A. M., Che Man, Y. B., Nazimah, S. A. H., Amin, I. (2009a), Chemical properties of virgin coconut oil. *Journal of the American Oil Chemists' Society* 86, 301-307.
- Marina, A. M., Che Man, Y. B., Nazimah, S. A. H., Amin, I. (2009b). Monitoring the adulteration of virgin coconut oil by selected vegetable oils using differential scanning calorimetry. *Journal of Food Lipids*, 16, 50-61.
- Miwa, H., Yamamoto, M., (1990). Liquid chromatographic determination of free and total fatty acids in milk and milk products as their 2-nitrophenylhydrazides. *Journal of Chromatography* 523, 235-246.
- Miwa, H., Yamamoto, M., (1996). Rapid liquid chromatographic determination of fatty acids as 2-nitrophenylhydrazine derivatives. *Journal of AOAC International* 79, 493-497.
- MMAF 2005: *Manual of Methods of Analysis of Foods. Oils and Fats.* Directorate General of Health Services Ministry of Health and Family Welfare, Government of India, New Delhi.
- Moigradean, D., Poiana, M.-A., Gogoasa, I., (2012). Quality characteristics and oxidative stability of coconut oil during storage. *Journal of Agroalimentary Processes and Technologies* 18, 272-276.
- Monseur, X., Walravens, J., Dourte, P., Termonia, M. (1981). Determination of volatile carboxylic acids in animal wastes by (GC)2 of their benzyl esters. *Journal of High Resolution Chromatography* 4, 49-53.

- Moreda, W., Pérez-Camino, M.C., Cert, (2003). A. Improved method for the determination of triacylglycerols in olive oils by high performance liquid chromatography. *Grasasy Aceites* 54, 175-179.
- Moura Fe J. A., Brown, W. H., Whiting, F. M., Stull, J. W., (1975), Unsaponifiable matter of crude and processed coconut oil, *Journal of the Science of Food and Agriculture* 26, 523-531.
- Nollet, L. M. L., F. Toldra, (2012), Handbook of Analysis of Active Compounds in Functional Foods, CRC Press.
- O'Brien, R. D., (2008). *Fats and Oils: Formulating and Processing for Applications*. Third Edition, CRC Press.
- Othonos A., Kalli, K., (1999). Fiber Bragg Gratings: Fundamentals and Applications in Telecommunications and Sensing, Artech House, Norwood.
- Padolina, W. G., Lucas, L. Z., Torres, L. G. (1987). Chemical and Physical Properties of Coconut Oil, *Philippine Journal of Coconut Studies* XII, 4-17.
- Pai J. S., Lomanno S. S., Nawar, W.W., (1970). Effect of Heat Treatments on the Volatile Composition of Coconut Oil, *Journal of the American Oil Chemists' Society* 56, 494-497.
- Pande, A., (1974). *Handbook of moisture determination and control,* Marcel Dekker, New York.
- Patel, H. K., Ed., (2014). *The Electronic Nose: Artificial Olfaction Technology*, Springer, New Delhi.
- Regulation (EEC) N8 2568/91, annex X.
- Reske, J., Siebrecht, J., Hazebroek, J., (1997). Triacylglycerol composition and structure in genetically modified sunflower and soybean oils. *Journal of the American Oil Chemist' Society* 74, 989-998.
- Rohman, A., Che Man, Y. B., (2011), The use of Fourier transform mid infrared (FT-MIR) spectroscopy for detection and quantification of adulteration in virgin coconut oil. *Food Chemistry* 129, 583–588.
- Rohman, A., Che Man, Y. B., Ismail, A., Hashim, P., (2011). Monitoring the oxidative stability of virgin coconut oil during oven test using chemical indexes and FTIR spectroscopy. *International Food Research Journal* 18, 303-310.
- Rohman, A., Che Man, Y. B., Sismindari, E. Y., (2009), Quantitative analysis of virgin coconut oil (VCO) in cream cosmetics preparations using Fourier Transform Infrared (FTIR) spectroscopy. *Pakistan Journal of Pharmaceutical Sciences* 22, 415-420.
- Rossell, J. B., King, B., Downes, M. J., (1983). Detection of adulteration. *Journal of the American Oil Chemists Society* 60, 333-339.
- Santos J. L. A., Bispo, V. S., Filho, A. B. C., Pinto I. F. D., Dantas, L. S., Vasconcelos, D. F., Abreu F. F, Melo, D. A., Matos, I. A., Freitas, F. P., Gomes, O. F., Medeiros, M. H. G., Matos, H. R., (2013). Evaluation of Chemical Constituents and Antioxidant Activity of Coconut Water (*Cocus nucifera L.*) and Caffeic Acid in Cell Culture. *Anais da Academia Brasileira de Ciências* 85, 1235-1246.
- Santos, J. E. R., Villarino B. J., Zosa A. R., Dayrit, F. M., (2011). Analysis of Volatile Organic Compounds in Virgin Coconut Oil and their Sensory Attibutes. *Philippine Journal of Science* 140, 161-171.
- Schulte, E. (1981), Separation of triglycerides according to chain length and degree of saturation by HPLC. *Fette, Seifen, Anstrichmittel* 83, 289-291.

- Seneviratne K. N., HapuarachchI, C. D., Ekanayake, S., (2009). Comparison of the phenolicdependent antioxidant properties of coconut oil extracted under cold and hot conditions, *Food Chemistry* 114, 1444-1449.
- Seneviratne K. N., Sudarshana Dissanayake, D. M., (2008). Variation of phenolic content in coconut oil extracted by two conventional methods. *International Journal of Food Science & Technology* 43, 597-602.
- Shahidi, F. (2005). *Bailey's Industrial Oil and Fat Products*, Sixth Edition, John Wiley & Sons.
- Sivasankar, B., (2002). Food Processing and Preservation. PHI Learning Pvt. Ltd.
- Sodamade, A., Oyedepo, T. A., Bolaji, O. S., (2013). Fatty Acids Composition of Three Different Vegetable Oils (Soybean Oil, Groundnut Oil and Coconut Oil) by High-Performance Liquid Chromatography. *Chemistry and Materials Research* 3, 26-29.
- Swern, D. (ed.), (1979). *Bailey's Industrial Oil & Fat Products*, 4th Edition, Vol. 1, John Wiley & Sons Inc., New York.
- Villarino, B. J., Dy L. M., Lizada M. C. C., (2007). Descriptive Sensory Evaluation of Virgin Coconut Oil and Refined, Bleached and Deodorized Coconut Oil. *LWT: Food Science* and Technology 40, 193-199.
- Vlahov, G., Del Re, P., Simone, N. (2003). Determination of geographical origin of olive oils using ¹³C nuclear magnetic resonance spectroscopy. I-Classification of olive oils of the Puglia region with denomination of protected origin. *Journal of Agricultural and Food Chemistry* 51, 5612-5615.
- Yasuda, M., (1931). The determination of the iodine number of lipids. *The Journal of Biological Chemistry* 94, 401-409.
- Yildiz, F., (2009). Advances in Food Biochemistry, CRC Press.
- Young, F. V. K., (1983), Palmkernel and coconut oils: analytical characteristics, process technology and uses. *Journal of the American Oil Chemists' Society*, 374-379.
- Yunus, W. M. M., Fen, Y. W., Yee, L. M., (2009), Refractive Index and Fourier Transform Infrared Spectra of Virgin Coconut Oil and Virgin Olive Oil. American Journal of Applied Sciences 6, 328-331.

Chapter 8

THE EFFECTS OF FATTY ACID DERIVATES FROM CORN AND COCONUT OILS ON MICROBIAL PHYSIOLOGY

Leontina Gurgu, Georgiana Horincar and Gabriela Bahrim^{*} Faculty of Food Science and Engineering, "Dunărea de Jos" University of Galati, Galati, Romania

ABSTRACT

Nowdays, green consumption" tendency promotes natural compounds, such as lipids and their derivatives, like fatty acids and formulated acylglycerols, as ingredients for many food and feed commercial products, nutraceuticals, pharmaceuticals or cosmetics products. Many studies demonstrated that the fats and their bioderivatives are important sources of biological active compounds that have influence on the cells metabolism and their physiology. The role of some fatty acids was intensely studied especially for their antimicrobial activity, in regards to natural microbiota and starter microorganisms development. The positive effects in cell's metabolism stimulation were less investigated but the preliminary results indicated good perspectives.

Keywords: coconut oil, corn oil, antimicrobial activity, microbial metabolism stimulation

INTRODUCTION

Lipids with biological activities have been extracted from various plant and animal species. It has been demonstrated that many lipids from different sources and also their derivatives (glycerides and fatty acids) exerted antimicrobial activity against a large spectrum of Gram-positive and Gram-negative bacteria, yeasts, fungi, viruses and parasites, that cause spoilage or various infections in both animals and humans (Glatz et al., 2015). Furthermore, fatty acids and derivated glycerides are important in the manufacture of emulsions (Thormar

^{*} Corresponding author: Email:Gabriela.Bahrim@ugal.ro

et al., 2006) and drugs (Thormar et al., 2006; Preston et al., 1996; Bergsson et al., 1999; Desbois and Smith, 2010; Kristmundsdottir et al., 1999) as alternative or adjuvant to antibiotic treatment used against resistant bacteria (Nakatsuji et al., 2009; Kitahara et al., 2004; Nair et al., 2005a,b). Nowadays, some fatty acids and their derivatives played an important role in food preservation and food safety assurance by producing natural food preservatives from vegetal and animal lipids (oils and fats) (Botterweck et al., 2000; Skandamis et al., 2001; Thormar, 2011). Beside the important role that fats and oils played in the diet, they are the main source of energy, have antioxidant properties and are also considered as raw materials for obtaining a wide range of products in the food, pharmaceutical, cosmetic and chemical industries (Kabara et al., 1991; Mandal et al., 2009).

Some tropical vegetable oils (coconut and palm oils) contain saturated fatty acids, while monounsaturated fatty acids are found prevalent in olive and corn oils whereas polyunsaturated fatty acids are found in corn, soybean and sunflower oils.

Coconut oil composition consists of 90% saturated fatty acids, 7.5% monounsaturated fatty acids and 1.8% polyunsaturated fatty acids (Zambiazi, et al., 2007) and it is as it follows: 1% caproic acid (C6:0), 2% caprylic acid (C8:0), 4% capric acid (C10:0), 50% lauric acid (C12:0), 18% myristic acid (C14:0), 9% palmitic acid (C16:0), 8% oleic acid (C18:1), 5% stearic acid (C18:0) and 2% linoleic acid (C18:2) (Paiwan et al., 2013). The structured lipids (24%), which usually are difficult to find in nature are obtained by crude oil biotransformation but another possibility to obtain them is by bioconversions in controlled conditions, in order to develop specific target compounds with biological activities that have different effects on microbial cells (Nugrahini et al., 2015).

On the other hand, corn oil composition consists of 14% saturated fatty acids, 24.76% monounsaturated fatty acids and 61.37% polyunsaturated fatty acids. The main profile of corn oil fatty acids is it follows: 10.47% palmitic acid (C16:0), 2% stearic acid (C18: 0), 24% oleic acid (C18: 1), 60% linoleic acid (C18: 2) and 1% linolenic acid (C18: 3) (Zambiazi, et al., 2007).

According to the literature, coconut and corn oils are recognized as the major sources of fatty acids, with active biological properties, that exert influences on microorganisms physiological activities, have microbiostatic and microbicide effects and also stimulate the metabolites production of the cells. (Desbois et al., 2010; DebMandal et al., 2011).

The antimicrobial activity of some fatty acids has been studied over several years through various research strategies in order to produce food additives that can inhibit or kill the microorganisms that have incidence in food spoilage and food safety assurance. The saturated fatty acids, (capric, caprylic, lauric, myristic acids) and the unsaturated fatty acids, (palmitoleic, linoleic and linolenic acids) that are found in the composition of coconut and corn oils have an important antimicrobial activity against a large number of pathogens (Agoramoorthy et al., 2007; Parfene et al., 2013).

The antimicrobial effect on bacteria and viruses of coconut oil is given by its high content of lauric acid (more than 50%), which *in vivo* is converted to monolaurin (Oyi et al., 2010; Parfene et al., 2013). The long chain fatty acids represented by oleic, linoleic and linolenic acids are more active against microorganisms compared to long-chain saturated fatty acids such as palmitic and stearic acid. Also, the medium-chain free fatty acids and their derivatives from coconut oil showed significant antimicrobial activity against Gram-positive and less significant to Gram-negative bacteria (Wille et al., 2003; Georgel et al., 2005; Skrivanova et al., 2006; Drake et al., 2008).

The positive effect of oils and their derivatives on microbial physiology was less studied, but the results obtained so far encouraged the research extention regarding the metabolism regulation, in order to improve the functionality and the biosynthesis potential of the cells (Azimatun Nur, 2015).

The main topic of this chapter is to review the *in vitro* and *in vivo* effects of coconut and corn oils and their derivatives (glycerides and fatty acids) on microorganisms' functionality in order to extend their benefits for increasing the quality of life.

ANTIMICROBIAL POTENTIAL OF VEGETABLE LIPIDS

Mechanism of Action

According to the literature, it was found a correlation between lipid's structure and antimicrobial activity of vegetable oils. Kabara and his collaborators (1978) concluded that, lauric acid, palmitoleic acid and linoleic acids have the most important antimicrobial activity from all the saturated fatty acids, monounsaturated fatty acids and polyunsaturated fatty acids, respectively. Their effect may be microbiostatic or microbicidal (Desbois & Smith, 2010).

It was also demonstrated that medium chain fatty acids and their monoglycerides are the most active in terms of antimicrobial effect than free fatty acids, particularly monolaurin acid. A few Gram-positive bacteria are more susceptible to the bactericidal effect of fatty acids compared with Gram-negative bacteria. On the other hand, the literature data showed that medium chain fatty acids (C10-C12) have a high antimicrobial potential against fungi (Kabara, 1978; Kabara, 1980) and monoesters of fatty acids are more active than diglycerides and triglycerides. It was found that, the antibacterial effect of short chain fatty acids (C6-C10) depends on the degree of their transformation and pH value. It seems like they are more active at pH 6.5 than pH 7.5 (Glatz et al., 2015).

Further, many studies have demonstrated that the antimicrobial potential of long-chain fatty acids increased significantly when the structure is changing. For example, one double bond in oleic acid (C18:1) increase the antimicrobial activity, comparing to the presence of three double bonds, which significantly decrease the antimicrobial potential (Kabara et al., 1972; Conley & Kabara, 1973).

Other studies showed that the fatty acids are able to break the barrier of the outer membrane of the cell wall allowing its access into the inner membrane (Thormar et al., 2006).

There are some differences in the degree of susceptibility of Gram-negative bacteria to fatty acids, especially *Helicobacter pylori* and enterobacteria (Bergsson et al., 2002). This may be due to differences in the bacteria outer membrane composition. For instance, the outer membrane of *Escherichia coli* and *Salmonella* spp. is composed only of lipopolysaccharide and proteins. The cell membrane surface contains hydrophobic chains represented by lipopolysaccharides which are making it difficult to be penetrated. Moreover, lipid molecules may have some difficulties to penetrate the bacteria membrane because of the low fluidity of lipopolysaccharide hydrocarbon chains and due to the strong interaction between lipopolysaccharide molecules. In contrast, some Gram-negative bacteria, particularly *Neisseria gonorrhoeae* (Preston et al., 1996) are easily damaged by medium chain fatty acids

and derivated monoglycerides. This is because the bacterial membrane is composed of a high amount of lipooligosaharides than lipopolysaccharides (Bergsson et al., 1999).

Another study conducted by Desbois and Smith (2010) showed the antimicrobial effect of some lipids against microorganisms. A series of inactivations or cells damage effects are caused by partial solubilization of cell membrane through action of fatty acids which determine the membrane proteins releasing. The antibacterial effect of fatty acids is carried out by penetration of the cell wall and the outer membrane of Gram-negative bacteria. As mentioned previously, the antimicrobial effect of lipids on Gram-negative bacteria takes place due to different outer membrane structure, which can act as a barrier and can provide protection to inner membrane of the bacterial cell.

In the inner membrane of bacteria there is located the electron transport chain with a number of carriers that pass electrons through each other in order to form water at the end of the process. This process generates a proton gradient and an increase in ATP synthesis by the enzyme, ATP synthase, through conversion of ADP. Medium and long-chain saturated and unsaturated free fatty acids after insertion in the cell wall or in the outer membrane of the Gram-negative bacteria are able to bind to the carrier of this chain of electron transport and to alter its function (Peters & Chin, 2003). There were proposed different mechanisms by which free fatty acids can alter the oxidative phosphorylation process as follows: by directly coupling to ATP synthase itself, by increasing the permeability of the membrane to protons or by decreasing the proton gradient and membrane potential (Boyaval et al. 1995; Wojtczak and Więckowski, 1998, 1999; Beck et al., 2007). The ATP production decrease because, the energy created through the electron transport chain is transformed into heat and not used for its production. In consequence, the cells are in the impossibility to achieve their vital functions and cells lysis is occurring (Chamberlain et al., 1991; Stulnig et al., 2001, Beck et al., 2007).

Antimicrobial potential of lipids and their compounds derivated after their hydrolysis depends on the microorganisms sensitivity. Some microorganisms are lyzed in a shorter period of time, at low concentrations of lipids, while others are destroyed after a longer period of time or at higher concentrations. The antimicrobial activity against bacteria, fungi or viruses depends on the lipids nature. In general, polyunsaturated fatty acids and saturated fatty acids with medium chain length are more active against pathogens (bacteria and viruses), while saturated fatty acid monoglycerides with medium chain are more active against fungi.

Antibacterial Activity

Medium chain fatty acids and their derivatives from coconut oil can destroy a wide range of bacteria after the degradation of bacterial cell lipidic bilayer. It was demonstrated the bactericidal effect against bacteria that cause various diseases, such as stomach ulcers, dental caries, food poisoning or urinary tract infections (Thormar, 2011).

The integration of unsaturated fatty acids into the bacterial membrane also causes an increase of its fluidity and permeability which leads to growth inhibition or bacterial cell death (Boyaval et al., 1995; Shin et al., 2007).

The antibacterial effect was demonstrated against both Gram-positive and Gram-negative bacteria. Moreover, the antibacterial effect of monocaprin and monolaurin against Gram-

negative bacteria (*Escherichia coli*) decreased at 30°C (Shibasaki & Kato, 1978) and at pH 7.0 (Bergsson et al., 2002). When *Escherichia coli* was grown in the presence of a mixture of lauric acid, monocaprin and monolaurin, at 50°C for 5 minutes, a strong inactivation of the bacteria function was observed. These results suggested that the temperature affected the external membrane permeability allowing the penetration of lipids in the cell wall of bacteria. When *Escherichia coli* cells were incubated in the presence of a mixture of citric acid and monolaurin, a large amount of monolaurin was found into bacterial cells (Thormar et al., 2006).

In a recent scientific report published by Oyi et al. (2010) it was shown that the coconut oil, in concentrations from 5% to 40%, has a bactericidal activity against *Pseudomnonas aeruginosa*, *Escherichia coli*, *Proteus vulgaris* and *Bacillus subtilis* strains. Bactericidal potential against these bacteria was attributed to the complex of lauric acid and monolaurin (DebMandal & Mandal, 2011).

Literature data showed that the emulsions obtained with 1.25 mM monocaprin acid in citrate-lactate buffer at low pH values (4.0-5.0) caused a greater reduction in the number of viable cells (> 6 to $7 \cdot \log_{10} \text{ CFU}$) of *Salmonella* spp., *Escherichia coli* and *Clostridium jejuni* (Thormar et al., 2006).

Lauric acid, which is the most abundant fatty acid in coconut oil is also present in breast milk and protect the newborns from the harmful action of pathogens. The coconut oil is a good source of fatty acids with bactericidal activity against pathogenic bacteria that cause infection and can be life threatening (Taheri et al., 2010).

Kabara and collaborators (1972) determined the minimum inhibitory concentrations (MIC) of the medium chain fatty acids against Gram-positive and Gram-negative bacteria, after 18 hours of incubation at 35°C. From all the saturated fatty acids (C6 to C18) that were tested, lauric acid had the greatest antimicrobial effect against Gram-positive bacteria. The most sensitive were the group A streptococci, while *Staphylococcus aureus*, *Staphylococcus epidermidis* and the streptococci from group D were the least sensitive (Conley & Kabara, 1973).

Some results presented the MIC of capric acid (C10:0), lauric acid (C12:0), linoleic (C18:2) and linolenic acid (C18:3) on 242 strains of Staphylococcus aureus and 117 strains of groups A, B and C of Streptococcus spp. (pathogenic strains). It was demonstrated that Staphylococcus aureus strains are less sensitive to fatty acids than other pathogenic species found in corn oil. Further, the lauric acid found in coconut oil showed the highest inhibitory action on the tested strains (Heckzko et al., 1979). Caprylic acid (C8:0), myristic acid (C14:0), lauric acid (C12:0), oleic acid (C18:1), monocaprylin (C8:0), monocaprin (C10: 0), monolaurin (C12:0), monomyristin (C14:0), monopalmitolein (C16:1) and monoolein (C18:1) exhibited a high antimicrobial potential against pathogenic strains from group A and group B streptococci and Staphylococcus aureus strains. Capric acid (C10:0) from coconut oil has antimicrobial effect against *Streptococcus* spp. (group A and group B), while the lauric acid (C12:0) from coconut oil and palmitoleic acid (C16:1) from corn oil has antimicrobial effect against Staphylococcus aureus strains (Bergsson et al., 2001). To determine which fat has the fastest bactericidal effect on Gram-positive bacteria, Bergsson et al. (2001) also tested the influence of time of incubation on antibacterial effect. It was demonstrated that after 10 minutes of incubation, in the presence of different fatty acids, the number of viable bacteria decreased up to 99%. Some bacteria that belong to Streptococcus genus, from the group A (GAS) and group B (GBS) have been cultivated for 10 minutes, in the presence of lauric acid and palmitoleic acid, at 37°C. It was found that 5 mM lauric (C12:0) and palmitoleic acid (C16:1) are able to kill all the cells from tested cultures.

Capric acid showed a low antibacterial effect against *Staphylococcus aureus* strains at a 10 mM concentration. A similar effect has been demonstrated against *Streptococcus agalactiae*, *Streptococcus dysgalactiae* and *Streptococcus uberis* strains, which were incubated, for 1 minute, in the milk supplemented with 100 mM caprylic acid (C8:0) (Nair et al., 2005b).

It has been determined that the antibacterial activity of long-chain fatty acids is increasing when their structure changes by double bonds addition (Kabara et al., 1972; Conley and Kabara, 1973). In consequence, linoleic acid showed a strong antibacterial potential against Gram-positive bacteria, such as *Bacillus cereus* and *Staphylococcus aureus* strains. However, it is already known that the antimicrobial activity of linoleic acid can be also improved by the addition of emulsifying agents such as, monolaurin and monomyristin (Lee et al., 2002).

Other scientific reports proved the antibacterial activity of polyunsaturated fatty acids (PUFA) against some Gram-positive bacteria. *Lactobacillus acidophilus* and *Staphylococcus epidermidis* strains were inactivated after one hour of incubation at 37°C with 0.01 mM arachidonic acid (C20:4) (Knapp & Melly, 1986).

The antimicrobial action spectrum of monoglycerides was found to be more limited compared to those of free fatty acids. The glyceraldehyde minimum inhibitory concentration against a large range of Gram-positive bacteria as *Staphylococcus aureus*, *Staphylococcus epidermidis* and *Streptococcus pyogenes* strains was also determined. From all of the monoglycerides tested, monolaurin was the most active. The antibacterial potential of monolaurin was somehow more emphasized than that of monocaprin. The studied *Staphylococcus pyogenes* strain was the most sensitive to the action of monocaprin, monolaurin and monomyristin. Monolaurin and monocaprin were the only monoglycerides that inhibited a strain of *Staphylococcus aureus*. Antibacterial potential of monoglycerides proved to be superior to that of free fatty acids (Guha et al., 2012).

Furthermore, monocaprin had the best bactericidal effect against *Staphylococcus aureus* from all the monoglycerides that were tested. It was also shown that the 1.25 mM monocaprin exhibited similar antibacterial potential against *Staphylococcus aureus* as monoglycerides that were introduced into the milk or milk formula (Guha et al., 2012).

Recent studies have been focusing on the antimicrobial effect of fatty acids with chain length from C6 to C18 (Guha et al., 2012). Thus, it was observed that, at low pH, the caprylic and capric acid were capable to inhibit and kill a considerable number of *Escherichia coli* viable cells. In contrast, the chain length of fatty acids from C2 to C6 and from C13 to C18 showed a negligible inhibitory effect against Gram–negative bacteria. This result was confirmed by studies conducted by Nair et al. (2005b) who showed that, the caprylic acid (C8:0) in concentrations from 25 mM to 100 mM when it was added into milk was able to reduce *Escherichia coli* cells.

In low concentration conditions, short incubation time (up to 30 minutes) and different physiological pH, fatty acid showed no bactericidal effect against *Escherichia coli* (Berbsson et al., 2002). On the other hand, 10 mM caprylic acid exhibited a significant reduction of *Campylobacter jejuni* cells, a pathogen responsible for food poisoning (Thormar et al., 2006). By measuring the minimum inhibitory concentrations (MIC), Skrivanova and collaborators (2006) demonstrated the inhibitory effect of 21 mM caprylic acid against *Escherichia coli* and

Salmonella spp. and of the 29 mM capric acid against *Escherichia coli*. Some reports have shown that lauric acid was the only fatty acid with antimicrobial activity against *Helicobacter pylori* (Petschow et al., 1996). The susceptibility of *Helicobacter pylori* to the action of saturated and unsaturated fatty acids was studied by Bergsson et al. (2001). They showed that 5 mM of lauric, capric and palmitoleic acids decreased the number of viable bacterial cells, after 10 min of incubation at 37°C. Also, they prove the antimicrobial effect of 1.25 mM lauric acid and 0.625 mM of palmitoleic acid against *Helicobacter pylori*, after 10 minutes of incubation (Bergsson et al., 2002).

Another Gram-negative bacterium, *Neisseria gonorrhoeae* was killed by a mixture of capric acid, lauric acid and palmitoleic acid. In addition, *Neisseria gonorrhoeae* was also inhibited by lauric and palmitoleic acids, after 1 minute of incubation (Bergsson et al., 1999).

Polyunsaturated fatty acids from corn oil, especially arachidonic acid (C20:4) has been proven to be effective in killing Gram-negative bacteria, *Neisseria* spp. and *Haemophilus* spp. Contrary, *Escherichia coli* strains showed no sensitivity during 6 hours of incubation with arachidonic acid (C20:4) (Knapp and Melly, 1986).

In agreement with the research conducted by Conley and Kabara (1973), di- and triglycerides of fatty acids showed no antimicrobial activity on Gram-negative bacteria (Kabara et al., 1972; Conley & Kabara, 1973). Similar studies carried out by Isaac et al. (1995) showed that the medium chain fatty acid monoglycerides were added to milk composition or breast milk formula in order to inhibit *Haemophilus influenzae*.

This effect was not available in the case of monoglycerides who had a less effective antibacterial action (Isaacs et al., 1995).

Some Gram-negative bacteria, such as *Escherichia coli* and *Salmonella* spp. strains, showed stability, at neutral pH, to all monoglycerides tested (Bergsson et al., 2002). Nevertheless, by using an acidic buffer (pH 5.0), monocaprin was highly effective on killing both types of bacteria (Thormar et al., 2006). Moreover, Nair et al. (2005a) have shown that, when 50 mM monocaprin is added in milk or 5 mM monocaprin, in apple juice, *Escherichia coli* viable cells were completely destroyed (Nair et al., 2005a,b). In addition, Preuss et al. (2005) demonstrated the bactericidal effect of essential oils, while monocaprin was totally ineffective to kill *Escherichia coli* and *Klebsiella pneumoniae* cells.

A study focused on hydrolyzed coconut fat microemulsions shown that the antimicrobial activity of microemulsion was higher compare to that of crude hydrolyzed extract obtained after enzyme hydrolysis (Parfene et al., 2010).

In conclusion, bactericidal and bacteriostatic effects differ because of the experimental conditions, regards to the microbial growth factors, the growth phases, the lipids type and their derivatives compounds concentration and also, the incubation time.

Antifungal Activity

Scientific reports have identified antifungal fatty acids that can combat fungal infections that affect plants, humans and food sources (Altieri et al., 2009; Thormar, 2011).

Fungal membrane plays an important role for maintaining cell integrity. Generally, fungal cell membrane represents the target for all the antifungal treatments (Avis, 2007). Avis and Belanger (2001) determined the general mechanism of how antifungal fatty acids interact with fungal cell membrane. The antifungal fatty acids which are naturally introduced in the

lipid bilayer physically disrupt the membrane increasing its fluidity. In consequence, the general disruption of the cell membrane leading to conformational changes in the membrane proteins, intracellular components releasing and cell lyses.

There are some factors, like chemical structure of fatty acids and medium pH, that plays an important role in the antifungal effect inhibition (Leyva et al., 2008).

Liu et al. (2008) showed that the fatty acids may replace some antifungal chemicals that are worldwide used in plant diseases combating. The effect of fungal cells life shortening was also studied and confirmed by Avis (2007).

Saturated and unsaturated free fatty acids may have antifungal potential, which effectiveness increases with their chain length (Kamem et al., 2009). An important role on microbial cells activity is played by hydrophobic groups of saturated fatty acids. The hydrophobicity increases with increasing the chain length, thus the solubility of fatty acids in aqueous environments decreases which prevents some interactions between these hydrophobic groups and the acyl chains of the membranes phospholipids.

Fatty acid	Species	References
Butanoic (butyric) acid	Alternaria solani	Liu et al., 2008
(C4:0)	Cucumerinum lagenarium	Liu et al., 2008
	Fusarium oxysporum	Liu et al., 2008
	Myrothecium verrucaria	McDonough et al., 2002
	Trichoderma viride	McDonough et al., 2002
Hexanoic (caproic) acid	Botrytis cinerea	Leyva et al., 2008
(C6:0)	Cucumerinum lagenarium	Liu et al., 2008
	Fusarium oxysporum	Liu et al., 2008
Octanoic (caprylic) acid	Alernaria solani	Liu et al., 2008
(C8:0)	Cucumerinum lagenarium	Liu et al., 2008
	Fusarium oxysporum	Liu et al., 2008
	Kluyveromyces marxianus	Viegas et al. 1989
	Saccharomyces cerevisiae	Viegas et al. 1989
Decanoic (capric) acid	Aspergillus nidulans	Sjögren et al., 2003
(C10:0)	Aspergillus fumigatus	Sjögren et al., 2003
	Candida albicans	Bergsson et al., 2001
	Cucumerinum lagenarium	Liu et al., 2008
	Fusarium oxysporum	Liu et al., 2008
	Kluyveromyces marxianus	Sjögren et al., 2003
	Microsporum gypseum	Chadeganipour et al., 2001
	Myrothecium verrucaria	Gershon et al., 1978
	Penicillium commune	Sjögren et al., 2003
	Penicillium roquefortii	Sjögren et al., 2003
	Pichia anomala	Sjögren et al., 2003
	Rhodotorula mucilaginosa	Sjögren et al., 2003
	Saccharomyces cerevisiae	Stratford et al., 1996

Table 1. Antifungal activity of fatty acids from coconut oil

Fatty acid	Species	References
Undecanoic acid	Myrothecium verrucaria	Gershon et al., 1978
(C11:0)	Saccharomyces cerevisiae	McDonough et al., 2002
	Trichoderma viride	Gershon et al., 1978
	Trichophyton rubrum	Peres et al., 2010
Dodecanoic (lauric)	Aspergillus niger	Altieri et al., 2007
acid (C12:0)	Blumeria graminis	Walters et al., 2003
	Candida albicans	Murzyn et al., 2010
	Colletotrichum gloeosporioides	Yenjit et al., 2010
	Cucumerinum lagenarium	Liu et al., 2008
	Fusarium avenaceum	Altieri et al., 2009
	Fusarium oxysporum	Liu et al., 2008
	Myrothecium verrucaria	Gershon et al., 1978
	Pythium ultimum	Walters et al., 2003
	Rhizoctonia solani	Walters et al., 2003
	Saccharomyces cerevisiae	McDonough et al., 2002
Tetradecanoic (myristic)	Alternaria solani	Liu et al., 2008
acid (C14:0)	Aspergillus niger	Altieri et al. 2007
	Candida albicans	Kabara et al., 1972
	Emericella nidulans	Altieri et al., 2007
	Fusarium oxysporum	Liu et al., 2008
	Penicillium glabrum	Altieri et al., 2007
	Penicillium italicum	Altieri et al., 2007
Palmitic acid (C16:0)	Alernaria solani	Liu et al., 2008
	Aspergillus niger	Altieri et al., 2007
	Aspergillus terreus	Altieri et al., 2007
	Cucumerinum lagenarium	Liu et al., 2008
	Emericella nidulans	Altieri et al., 2007
	Fusarium oxysporum	Liu et al., 2008
Oleic acid (C18:1)	Crinipellis pernicosa	Walters et al., 2004
	Pythium ultimum	Walters et al., 2004
Linoleic acid (C18:2)	Alternaria solani	Liu et al., 2008
	Candida albicans	Kabara et al., 1972
	Crinipellis pernicosa	Walters et al., 2004
	Fusarium oxysporum	Liu et al., 2008
	Pyrenophora avanae	Walters et al., 2004
	Pythium ultimum	Walters et al., 2004
	Rhizoctonia solani	Walters et al., 2004

Antifungal properties of fatty acids and their derivatives were studied a long time ago. In Table 1 and Table 2 there are presented the fatty acids from coconut and corn oils that shown antifungal activity against some pathogens.

Some studies showed antifungal potential of saturated fatty acids against *Aspergillus* spp., *Penicillium* spp. and *Rhizopus* spp. The cultivation of fungi was performed on solid medium at 37.5° C, for 48 hours. All medium chain fatty acids (up to C14) and the unsaturated fatty acids were tested in terms of antifungal activity, in a pH range between 2.0 and 8.0. It was observed that, depending on the pH and chain length, all the tested fatty acids inhibited the molds growth. It was also proved that, the 6-carbon fatty acids were the most active at a pH value lower than 5.0, while the fatty acids with chain length between C7 and C12 were active at neutral pH. Isomeric forms of the unsaturated fatty acids have been shown to be less active than the corresponding fatty acids, but the degree of unsaturation increased the antifungal potential (Thormar, 2011).

Fungistatic effect of fatty acids was tested against *Trichophyton* spp. strains. The results showed that the antifungal activity of fatty acids increased with fatty acids chain length to the point when the solubility became the limiting key factor. Unsaturated fatty acids were more active than the saturated ones. It was noted that, for the optimal inhibitory activity against *Aspergillus* spp. the length of the chain should have 11 carbons, while for the *Trichophyton* spp., it should have 13 to 14 carbons. It was also observed that, particularly for short chain fatty acids, the antifungal activity increased by decreasing the pH till 4.5 (Peres et al., 2010).

Fungicidal and fungistatic differences were presented in various scientific reports but it seems like, these terms were used superficial in the past. According to this approach, fungistatic action is the one that inhibits or prevents fungal growth, while fungicidal activity destroys them (Murzyn et al., 2010). Medium chain fatty acids, by their mechanism of action developed an inhibitory effect on *Microsporum* spp. metabolic pathway, which was more effective with increasing chain length and decreasing the pH (Murzyn et al., 2010).

Capric acid (C10:0) and lauric acid (C12: 0) have antimicrobial activity against *Candida albicans* strains (Bergsson et al., 2001).

Fatty acid	Species	References
Palmitic acid (C16 :0)	Alernaria solani	Liu et al., 2008
	Aspergillus niger	Altieri et al., 2007
	Aspergillus terreus	Altieri et al., 2007
	Fusarium oxysporum	Liu et al., 2008
Oleic acid (C18:1)	Crinipellis pernicosa	Walters et al., 2004
	Pythium ultimum	Walters et al., 2004
Linoleic acid (C18:2)	Alternaria solani	Liu et al., 2008
	Candida albicans	Kabara et al., 1972
	Crinipellis pernicosa	Walters et al., 2004
	Fusarium oxysporum	Liu et al., 2008
Linolenic acid (C18:3)	Crinipellis pernicosa	Walters et al., 2004
	Pyrenophora avanae	Walters et al., 2004
	Pythium ultimum	Walters et al., 2004
	Rhizoctonia solani	Walters et al., 2004

Table 2. Antifungal activity of fatty acids from corn oil

Antiviral Activity

Coconut oil is very effective against a wide range of viruses, such as: *Epstein Barr* virus, influenza virus, leukemia virus and hepatitis C virus. It was demonstrated that the medium chain fatty acids from coconut oil kill these viruses by destroying their membranes. From all of the saturated fatty acids, lauric acid presents the greatest antiviral activity prior to capric acid (C10:0), caprilic acid (C8:0) and myristic acid (C14:0). It seems like, the monoglycerides from the composition of coconut and corn oils are more active against viruses compared to diglycerides and triglycerides. It was demonstrated that the monolaurin solubilizes the lipids and phospholipids from the membrane of the virus causing the disintegration of the viral body membranes (Arora et al., 2011).

Anti-Protozoal Activity

There are few reports which demonstrated the anti-protozoal activity of vegetable oils of *Hypericum* species (Moon, 2010; Zofou et al., 2011) based on the lipophilic components of the oil and its extract. A few studies have shown that long-chain fatty acids and β -sitosterol contributed to inhibit the protozoan organisms (Nweze et al., 2011).

The most important tropical parasitic disease is malaria. World Health Organization (WHO) estimates that there are around 300-500 millions clinical cases of malaria each year, and approximately 1 million deaths occur every year (Thayer, 2005). Malaria is caused primarily by four *Plasmodium* species, but *Plasmodium falciparum* is responsible for the most severe form of the disease, causing 90% of deaths in Africa (Thayer, 2005).

Antimalarial treatment is an area with a continue rising due to the limited number of drugs towards the parasites that are not resistant (Thayer, 2005). The fact that antimalarial fatty acids can inhibit fatty acid synthesis of *Plasmodium falciparum* was considered a pest control strategy.

Scientific results on antimalarial properties showed that the most effective activity in killing the parasite, *Plasmodium falciparum*, are polyunsaturated fatty acids (Kumaratilake et al., 1992). Therefore, with the increasing of unsaturated grade, it's also increasing their toxicity against *Plasmodium falciparum*. The order in which the fatty acids toxicity decrease is: docosahexaenoic acid (C22:6); eicosapentaenoic acid (C20:5); arachidonic acid (C20:4); oleic acid (C18:1); docosahexaenoic acid (C22:0). On the other hand, the concentrations of 20-40 mg/mL of docosahexaenoic acid (C22:6) were the most efficient in the breaking of the malaria parasite. The fatty acids methyl esters were reported as the effective free fatty acids in killing the parasite. These investigations led to the hypothesis that the lipid peroxidation is the responsible mechanism for the antiplasmodial activity of unsaturated fatty acids (Kumaratilake et al., 1992).

Other studies have also shown that the concentrations of 6.2 mg/mL of C18 fatty acids had an inhibitory effect against *Plasmodium falciparum*. The highest inhibitory effect against the parasites was depending on the degree of unsaturation and was given by the 96 μ g/mL linolenic acid (C18:3), followed by 76 μ g/mL linoleic acid (C18:2) and 23 μ g/mL oleic acid (C18:1) (Kumaratilake et al., 1992). Corn oil is considered the most important source of polyunsaturated fatty acids, compounds that have inhibitory effect against *Plasmodium falciparum*.

Leontina Gurgu, Georgiana Horincar and Gabriela Bahrim

These are important results because they highlight once again the highest potential of fatty acids with 18 carbons length on antimalarial activity. In consequence, further exploitation of antiplasmodial function compounds, from various vegetable oils and fats, is still needed to be made.

Practical Implications

Antimicrobial activity and low toxicity composition of fatty acids and their derivated compounds obtained after lipids hydrolysis suggest that they may have a wide range of applications, as ingredients in formulation of a large category of products, as food preservatives, in cosmetics and pharmaceutical industries and also as pests control products.

Antimicrobial compounds from vegetable oils have a great potential in obtaining personal hygiene products, like deodorants. For instance, the commensal microorganisms break down axial secretions in secondary compounds responsible for odors (Labows et al., 1999). Therefore, the use of antimicrobial lipids in antiperspirants composition may create an inhibitory effect on the action mechanism of these microorganisms, thus the amount of odor is significantly reduced.

Fatty acids and their monoglycerides were both used as food preservative and ingredients in cosmetic and pharmaceutical industries (Shibasaki et al., 1978; Bergsson et al., 2001; Drake et al., 2008). The advantage of using lipids and derivates with microbicide action in cosmetic products formula is that they cannot cause skin irritations. The main reason for their use as a preservative in cosmetics is the low toxicity and higher antimicrobial activity. There are many examples of using lipids as preservative system in cosmetics, such as shampoos, conditioners and moisturizers (Puri et al., 2009). Lipids were conventionally used as solvents in many pharmaceutical products composition, as oil phase in creams, as emulsifiers, antioxidants and stabilizers. Usually, these are used to facilitate penetration of the micro- and nano- drugs particles into the skin and as an secondary effect they can also exert antimicrobial activity and preservation strategies (Schafer-Korting et al., 2007).

The development and use of biodegradable packaging materials obtained from natural polymers are important growing tendencies, especially in the food industry. The concept of intelligent packaging with antimicrobial and antioxidative films based on natural polymers is an innovative one (Yam et al., 2005).

A combination of virgin coconut oil (VCO) and chitosan was used in the preparation of an edible composite film, with role in food industry, for biochemical and microbiological stability control, thus being used for packaging dry as well as moist foods (Binsi et al., 2013). The researchers evaluated the effect of VCO incorporation in different concentrations on chitosan emulsion system. Chitosan have different degrees of deacetylation thus, this biopolymer produces water-oil-water emulsions. The results of the study provided that, the addition of VCO in the system increased the film thickness and marginal reduction in film transparency (Binsi et al., 2013). On the other hand, between chitosan and the compounds from VCO different interactions were established, such as hydrogen/covalent bonding (between hydroxyl group of the aromatic ring of polyphenolic compounds and the reactive groups of chitosan) and electrostatically interaction (between short/medium chain fatty acids of VCO oil and cationic molecules of chitosan) (Rohn et al., 2002; Jumaa et al., 2002). The implications of VCO-polymer interactions are that the VCO may improve the barrier

www.Ebook777.com
properties of composite films, decrease the oxygen, carbon dioxide and water vapor permeability, reduce the moisture sorption value of chitosan film, improve the elongation at break values and film flexibility (Binsi et al., 2013). So, when there were combined the effects of medium chain fatty acids from coconut oil with those of chitosan it became possible to obtain an edible film used for packaging dry and moist foods, for spoilage control and food safety assurance.

A new field, in which antimicrobials compounds derivated from vegetal lipids could be used in infant formula (Issac et al., 1995). Presentlly the infant formula contains triglycerides, which are hydrolyzed in the gastrointestinal tract of children in order to produce antimicrobial fatty acids and their monoglycerides, in a manner similar to milk triglycerides (Canas-Rodriguez et al., 1966; Smith et al., 1966).

The antimicrobial activity of medium chain fatty acids (MCFAs) or their derivatives has been known for a long time (Kabara et al., 1972) but no experiments have been concentrated on their use as a potential source of *in situ* system antimicrobial compounds. Decuypere and Dierick (2003) proposed an interesting *in vivo* bioconversion process in order to obtain a progressive enzymatic release of MCFA in stomach and foregut. Following this approach it is easy to avoid taste aversion of non-esterified MCFA, given in high doses over a short period of time and the acid-base balance disturbance, in the animal. Further, a combination of triacylglycerols with an adequate lipase, used in a specific preparation, offers a gradual release, in the foregut, of MCFA, eventually together with their monoacylglycerols. This approach it seems to be preferable for a valuable alternative to in-feed antibiotics (Decuypere & Dierick, 2003). Also, the mixture can offer other advantages such as growth promotion and gastrointestinal disease treatment.

Parfene et al. (2013) also demonstrated the possibility to obtain food biopreservation as an effect of saturated fatty acids released from vegetal fats (coconut, palm and shea), by *in situ* solid state cultivation of *Yarrowia lypolitica* selected strains. There was studied the fatty acids production in restrictive environmental conditions, similarly with those applied for food preservation, i.e., at low temperature (4°C) and reduced level of water activity ($a_w 0.93$). The gradual release of fatty acids from lipids was demonstrated in very simple experimental conditions. These results developed new concepts in food preservation as a consequence of using natural preservatives, without influencing the sensorial and nutritive characteristics thus having a great impact on spoilage control and food safety assurance.

STIMULATIVE EFFECTS OF FATTY ACIDS ON MICROBIOTA

Gene Expression Regulation in Escherichia Coli by Fatty Acids

Studies about the regulation of genes from lipid metabolism by fatty acids were made in both unicellular and multicellular organisms. In *Escherichia coli*, fatty acids metabolism is tightly regulated in correlation with the environment changes. The model microorganism, *Escherichia coli* used long-chain fatty acids (LCFA) that are transported into the cell by the outer membrane protein FadL (Pech-Canul et al., 2011).

Medium (C7 to C11) or short (C4 to C6) chain lengths fatty acids cannot induce synthesis of the enzymes encoded by *fad* in *E. coli*. As a consequence, LCFAs represents the sole

carbon and energy sources and is coupled to the citric acid cycle via β -oxidation. Inside the cells, acyl-CoA synthetase enzyme, which is encoded by *fad*D gene, catalyze the formation of fatty acyl-CoA thioesters that are substrates for β -oxidation, which are further cleavaged into acetyl-CoAs (Campbell et al., 2003). Also, the fatty acyl-CoA thioesters are used in phospholipid biosynthesis and cellular signaling. In the first step of β -oxidation process, the acyl-CoA is converted to enoyl-CoA by FadE. The next steps in the fatty acids degradation are made by a tetrameric protein complex encoded by *fad*A and *fad*B genes, each having a double copy (Henry and Cronan, 1992; Campbell and Cronan, 2001). So, for the growth of *E. coli* cells in medium with LCFAs it is necessary that the transcription of genes from the *fad* regulon to be repressed. This function is made by the transcription factor FadR that is derepressed (Gui et al., 1996). In consequences, FadR regulates both the conversion of fatty acids to acetyl-CoA and the utilization of this final product by the citric acid cycle.

Stimulative Effect of Polyunsaturated Fatty Acids on Lactic Acid Bacteria

The gastrointestinal tract (GI) has the most diverse microbiota that represents the largest source of non-self antigens in the body. The gut microbiota content can vary in the gut regions and can influence susceptibility to chronic disease of the intestinal tract. Probiotic microorganisms help in the maintenance of the digestive tract ecosystem and prevent the colonization with pathogenic bacteria at this level (Semih, 2014). Dysbiosis appears when an imbalance or an alteration of the intestinal gut microbiota function and composition are produced. Some factors such as age, diet, life style and disease are causing intestinal dysbiosis (Agans et al., 2011; Turnbaugh et al., 2009; Seksik et al., 2003).

Probiotics are "live microorganisms which, when administered in adequate amounts confer a health benefit on the host" (Schlundt, 2012). Data existing in the literature offers many information regarding the probiotic influences on population of beneficial microorganisms in terms of growth and stimulatory effect (Vandenbergh, 1993, Dima et al., 2014).

The lipids from diet specifically influence the population of lactic acid bacteria (LAB). *In vivo* studies on gnotobiotic piglets showed that, the oral administration of oils containing polyunsaturated fatty acids (PUFAs) enhances the number of *Lactobacillus paracasei* cells in the small intestine and inhibits digestive tract pathogens (Bomba et al., 2002).

Tween 80 is usually added to the growth media of lactobacilli in order to improve aerobic growth rates, glucosyltransferase secretion (Jacques et al., 1985) and glycine-betaine accumulation in LAB (Guillot et al., 2000). Tween 80 contains up to 90% oleic acid that is incorporated into LAB membrane, as it was experimentally proved (Johnsson et al., 1995; Partanen et al., 2001).

Corcoran and collaborators (2007) studied the effects of various C18 fatty acids with 0-2 bonds in the two configurations, cis and trans, on rate survival of a *Lactobacillus rhamnosus* GG strain, in the simulated acid conditions of gastric juice. The protective effects to the acidic environment was offered only by oleic and vaccenic acids [C18: 1 (11t)] among elaidic, linoleic and stearic acids and two conjugated linoleic acid (CLA) isomers (Corcoran et al., 2007; Kankaanpää et al., 2014). On the other hand, oleic acid increased the survival rate of a *Lactobacillus rhamnosus* strain, in 90 minutes after exposure to simulated gastric juice (by 1500-fold) compare to the control. Moreover, when the culture media was supplemented with

Tween-80, the content of cyclopropane fatty acids (CFAs) increased as a protective function to acidic environmental conditions (Budin-Verneuil et al., 2005; Klaenhammer et al., 2005). It seems like, an increased level of CFA in the membrane composition of lactobacilli is found in other stressful conditions for the cells, such as, higher concentration of NaCl (1M, for *Lactobacillus casei*) or the growth condition below 20°C or above 35°C (for *Lactobacillus fermentum*) (Suutari & Laakso, 1992; Machado et al., 2004). Among the utilization of exogenous oleic acid source by probiotic lactobacilli, in order to increase their acid survival, it was suggested that, the higher content of oleic acid found in the membrane is reduced to stearic acid (Corcoran et al., 2007).

Literature data specified that an important factor in the probiotic strains` growth is PUFA`s concentration levels. *In vitro* studies prove that a low concentration of gammalinolenic acid and arachidonic acid (5 μ g/mL) are essential for growth promoting and mucus adhesion of *L. casei* strains (Kankaanpää et al., 2001; Laparra & Sanz, 2010). The adhesion of probiotics depends on the hydrophobicity of the bacterial cell surface, being favorized the cells with a hydrophobic surface than with a hydrophilic one (Wadstrom, 1990; Kankaanpää et al. 2004). These are valuable information that can be used in order to increase the adhesion ability of bacterial cells if it is taken in account that the environmental growth conditions can affect bacterial membrane fluidity. So, by adhering to epithelial cells, the probiotics block the binding sites and protect the host from pathogens to initiate invasion.

In vitro studies on HT-29 epithelial cells were made in order to test the capacity of *L. johnsonii* strains to decrease *S. enterica* serovar *typhimurium* UK1-lux invasion (Muller et al., 2011). It was shown that the *L. johnsonii* proved its exclusion properties just when the probiotic bacteria are growing in minimal medium supplemented with fatty acids. In these conditions, *L. johnsonii* reduced significantly *Salmonella* spp. adhesion and invasion. If the *L. johnsonii* was grown in a medium supplemented with linoleic and linolenic acids, its adhesion was also possible even when the HT-29 cells were infected with *Salmonella* spp. (Muller et al., 2011). The mechanism proposed in order to explain how probiotics prevent pathogens from infecting cells is that they sterically hinder the adherence of pathogens (Fourniat et al., 1992).

The changes induced in the bacterial cells composition by PUFAs after their addition in the growth medium, can be correlated with the regulation of the fatty acids degree unsaturation, cyclization, proportions of CLA and PUFAs with C20 to C22 length (Kankaanpää et al., 2004).

In order to enhance the efficacy of probiotics, there were made combination between probiotics and components of natural origin, such as oligosaccharides, phyto-components, nutrients and growth factors, proteins, PUFAs, organic acids and bacterial metabolites (Pollmann et al. 1980; Galfi and Bokori, 1990; Gibson and Roberfroid, 1995; Yadava et al. 1995). It seems like the synergic effects of maltodextrin KMS X-70 and PUFAs (oil blend containing 0.1 g total n-6 PUFAs, 1.0 g total n-3 PUFAs, 2.6 g total monounsaturated fatty acids and 0.9 g total saturated fatty acids) is useful for potentiating the probiotic effect in the small intestine of gnotobiotic piglets (Bomba et al., 2002).

Implications of Fatty Acids in Rumen Methanogenesis

In vitro study on rumen fermentation microbiota and methanogenesis emphasized the synergistically influence of medium chain fatty acids, like capric acid (C10:0), lauric acid (C12:0) and myristic acid (C14:0). Lauric (C12:0) and myristic (C14:0) acids showed their potential to decrease rumen methanogenesis and the number of methanogens. When the two fatty acids were not combined the effects exhibited some limitations. *In vivo* incorporation of coconut oil in sheep diet caused a reduction from 26% up to 73% in the methane (CH4) emission and up to 39% after using a oil, in each day, in feed composition for beef heifers (Panyakaew et al., 2013).

Kongmun et al. (2011) investigated the effect of coconut oil and garlic powder supplementation on digestibility of nutrients, rumen fermentation, ecology and microbiota and also methanogens diversity. It was also observed that methane production was dramatically reduced while the number of amylolytic and proteolytic bacteria increased and the protozoal population decreased by 68–75%. The study demonstrated that the coconut oil could be efficiently used in the animal rumen after diet supplementation. Furthermore, coconut oil digestion could provide good fermentation end products and could also decrease the methane gas production without changing nutrient digestibilities.

Stimulation Effect of Corn Oil on Microorganisms' Stabilization

Corn oil was used in the preparation process of water-in-oil emulsions (W/O) with the aim of long-term stabilization of microorganisms by cells encapsulation. In these types of emulsions, there were included the larvacide *Lagenidium giganteum* and the green alga *Chlorella vulgaris* and it was observed that cells had the possibility to remain hydrated during storage and delivery, without being contaminated. The researchers proved that, during room temperature storage, the water-in-oil emulsion prolonged the viability of the microalgae and larvacide, *Lagenidium giganteum* (VanderGheynst, 2013). Water-in-oil (W/O) emulsions formula overcame the problems of industrial microbial agents regarding the contamination and also about the decreasment of cells viability during distribution, storage and delivery.

Effect of Polyunsaturated Fatty Acids and Enthomopatogenic Fungi

Linoleic and oleic acids, the two main components of corn oil can be used to enhance the thermotolerance of enthomopatogenic fungus (Kim et al., 2011) An entomopathogenic fungus is a fungus that kills the parasite of insect by its fixation on the external body surface of insects. The most important enthomopathogens that have been commercially produced were *Beauveria bassiana*, *Metarhizium anisopliae* and *Isaria fumosorosea*. In the mode of action, these entomopathogens act on the cuticle of insects, penetrates it in order to proliferate, to produce the toxins and to kill the insect. It seems like the storage and applications of the fungi as biopesticides have a disadvantage related to exposure to high temperature conditions or to high humidity, which can drastically decrease the shelf life of those fungi. There were made attempts to increase their thermotolerance with unsaturated fatty acids (Kim et al., 2010). The results showed that, the treatments with unsaturated fatty acid from corn oil enhanced conidial

thermotolerance in a dosage-dependent manner. It means that, the more linoleic or oleic acids was added to ground corn, the better thermotolerance of conidia was obtained (Kim et al., 2010). So, from an industrial point of view, the development of a ground corn oil mixture, as a medium in order to obtain more thermotolerant conidia is an inexpensive approach that can allow commercial productions of fungal biopesticides to be more competitive than others.

Effects of Fatty Acids on Exo-Biopolymer Production

Exo-biopolymers produced by some fungi have been recognized as high value biomacromolecules from the last twenty years. The composition and the amount of exopolysaccharides (EPSs) are genetically determined and are dependent to the culture medium composition and to the cultivation conditions during fermentation such as pH, temperature, oxygen concentration and rate of agitation.

The effects of some fatty acids, such as oleic, palmitic and stearic acids on exobiopolymer production and mycelial growth of *Cordyceps militaris* were investigated (Park et al., 2002). C. militaris is the oldest source of some useful chemical constituents but with active principles used for the new herbal biotechnology. These compounds act as antiinflammatory, anti-oxidant/anti-aging, anti-tumor/anti-cancer/anti-leukemic. antiproliferative, anti-metastatic, immunomodulatory, anti-microbial, anti-bacterial, anti-viral, anti-fungal, anti-protozoal, hypolipidaemic, anti-angiogenetic, anti-diabetic, anti-HIV, neuroprotective, liver-protective, reno-protective as well as pneumo-protective (Das et al., 2010). A few long chain fatty acids (oleic, palmitic and stearic acids) with stimulatory effects on the mycelial growth were added in the medium in order to optimize the cost of the process. From the three fatty acids tested, the oleic acid was found to have the best stimulatory effect when added in the concentration of 0.2% (v/v). Palmitic and stearic acids also stimulated the growth mycelia compare to linoleic acid that drastically suppressed it (Park et al., 2002). Moreover, oleic and palmitic acids increased the exo-biopolymer concentration contrary to the effect of linoleic and stearic acids (Yang et al., 2000; Kendrick and Ratledge, 1996).

Enzymes Biosynthesis in Medium with Coconut Oil by Solid State Fermentation

Enzymes are indispensable components of all types of cells that have been extensively used in many industrial sectors and for environmental protection. Enzyme synthesis process was improved over the years in order to reach the best relation cost - product.

Coconut oil cake is a waste by-product obtained after the oil extraction from dried coconut. The major compounds from the coconut oil cake are short chain saturated fatty acids, which were proposed to be used as a carbon substrate source in solid-state fermentation (SSF) for enzymes biosynthesis. This approach in which the coconut oil cake represents a cheap way for the bioproducts synthesis is a new one and was adopted in the last years. The first report on enzymes synthesis using coconut oil cake was published by Ramachandran and coworkers, in 2004. The researchers optimized the production of α -amylase using a fungal culture of *Aspergillus oryzae* strain in SSF cultivation system on a medium based on raw coconut oil cake and it was obtained an α -amylase activity of 1372 U/g dw, in 24 h. Eight

years later, the enzyme, α -amylase was also produced in a SSF process based on the same substrate but using a *Streptomyces* spp. strain (Maheswari and Soundariya, 2012). Another enzyme, L-asparaginase, was obtained under SSF using *Serratia marcescens* (Ghosh et al., 2013). L-asparaginase is an important enzyme that catalyzes the hydrolysis of L-asparagine to aspartic acid and ammonia which was found to be an anti-cancer chemotherapy drug, thus being involved in the treatment of acute lymphocytic leukemia. The researchers used the response surface methodology in order to optimize the L-asparaginase production took into account four variables: substrate amount, moisture of substrate, temperature and pH. They concluded that, the coconut oil is an efficient substrate that can reduce the cost of L-asparaginase production, under optimal conditions (Ghosh et al., 2013).

Oxygenated Fatty Acids Synthesis by Microorganisms

In the industrial materials, the hydroxy, keto or epoxy fatty acids are used in many different types of products including resins, waxes, nylons, plastics, corrosion inhibitors, cosmetics, or coatings (Hou et al., 1998). Moreover, the hydroxy-fatty and keto-fatty acids are important as plasticizer, surfactant and lubricant. The oxygenated fatty acids can be obtained through the oxygen-dependent hydroxylations of some fatty acids, from vegetable oils, added in the growth media of specific microorganisms. A strain of *Torulopsis* spp. was able to hydroxylate palmitic, stearic and oleic acid to 15- or 17-hydroxy-fatty acids (Shanklin and Cahoon, 1998). In the hydroxylation process of palmitic acid by *Bacillus megaterium* ALA2 is involved Cytocrome P 450-dependent monooxygenase complex, in order to obtain three hydroxy palmitic acids, $\omega -1$, ω -2 and ω -3 (Hou, 2005). Through β -oxidation, the yeast *Dipodascopsis uninucleata* converts linoleic acids (9Z, 12Z-octadecadienoic acid) to 3(R)-hydroxylated metabolites of shorter chain length (Venter et al., 1997). Further, through ω -oxidation, the enzyme system of *Pseudomonas oleovorans* strain is able to catalyze biotransformation of medium chain fatty acids, such as octanoate, decanoate, laurate and myristate or long-chain fatty acids, palmitic and stearic acids (Kusunose et al., 1964).

On the other hand, the enzymatic system of *B. megaterium* ALA2 is able to produce compounds with anticancer activity (two tetrahydrofuranyl fatty acids -THFAs), starting from linoleic acid (Hou et al., 1998; Gardner et al., 2000). So, the microbial conversion of medium and long chain fatty acids from vegetable oils sources, into sophorolipids gives new application possibilities in various sectors due to either their emulsifying, antimicrobial activity or other beneficial properties.

CONCLUSION AND PERSPECTIVES

Fatty acids and related metabolites derivated from coconut and corn oils are known to have an excellent biological activity with impact on microorganism metabolism, on cells nutrition and energy uptake. They are also recognized to have important antimicrobial activity. Their accessibility or their limitations, in time and space, in terms of cells contact have an important impact on biological processes regulation. The type and also the concentration of fatty acids are important factors that influences various biological processes that take place into natural microbiota or in the starter cultures.

The antimicrobial potential of vegetable oils and their derivatives (acylglycerols and fatty acids) has practical importance in obtaining biopreservatives, as natural ingredients for commercial products, thus obtaining biochemical and microbiological stability and safety assurance. For practical applications is important to obtain active compounds using cheaper processing conditions like *in situ* bioconversions of lipids with enzymes or with whole cells. The positive effects of coconut and corn oils on microorganism's metabolism is prooved by *in vivo* applications, by stimulating the activity of different microbiota (gut microbiota, epithelial microbiota, rumen microbiota etc.) and also by *in vitro* approaches, in order to induce the biosynthesis of microbial metabolites, to produce active biofilms or to have surfactant role. The practical and scientific perspectives to use oils and bioderivates in microbiology or in applied biotechnology can also emphasize other new and modern practical applications.

REFERENCES

- Abdallah, M. A., Lei, Z. M., Li, X., Greenwold N., Nakajima, S. T., Jauniaux, E. & Rao, C. V. (2004). Human Fetal nongonadal tissues contain human chorionic gonadotropin/ luteinizing hormone receptors. J. Clin. Endocrinol. Metab., 89, 952-956.
- Agans, R., Rigsbee, L., Kenche, H., Michail, S., Khamis, H. J. & Paliy, O. (2011). Distal gut microbiota of adolescent children is different from that of adults. *FEMS Microbiol. Ecol.*, 77(2), 404-412.
- Agoramoorthy, G., Chandrasekaran, M., Venkatesalu, V. & Hsu, M. J. (2007). Antibacterial and antifungal activities of fatty acids methyl esters of the blind-your-eye mangrove from India. *Braz. J.Microbiol.*, 38, 739–742.
- Altieri, C., Cardillo, D., Bevilacqua, A. & Singaglia, M. (2007). Inhibition of Aspergillus ssp. and Penicillium spp. by fatty acids and their monoglycerides. J.Food Protect., 70, 1206-1212.
- Arora, R., Chawla, R., Marwah, R., Arora, P., Sharma, R. K., Kaushik, V., Goel, R., Kaur, A., Silambarasan, M., Tripathi, R. P. & Bhardwaj, J. R. (2011). Potential of complementary and alternative medicine in preventive mamagement of novel H1N1 flu (Swine flu) pandemic: thwarting potential disasters in the bud. *Evid.Based. Complement Alternat. Med.* 1-16.
- Avis, T. J. (2007). Antifungal compound that target fungal membrane: application in plant disease control. *Can. J. Plant Pathol.*, 29, 323-329
- Avis, T. J. & Bélanger, R. R. (2001). Specificity and mode of action of the antifungal fatty acid cis-9-heptadecenoic acid produced by *Pseudozyma flocculosa*. *Appl. Environ. Microb.*, 7, 956-960.
- Azimatun Nur, M. M., Irawan, M. A. & Hadiyanto. (2015). Utilization of coconut milk skim effluent (CMSE) as medium growth for *Spirulina platensis*. *Procedia Environmental Sciences*, 23, 72 – 77.

- Beck, V., Jabůrek, M., Demina, T., Rupprecht, A., Porter, R. K., Ježek, P. & Pohl, E. E. (2007). Polyunsaturated fatty acids activate human uncoupling proteins 1 and 2 in planar lipid bilayers. *FASEB J.*, *21*, 1137-1144.
- Bergsson, G., Arnfinnsson, J., Steingr'imsson, O. & Thormar, H. (2001). Killing of Grampositive cocci by fatty acids and monoglycerides. *APMIS.*, 109, 670–678.
- Bergsson, G., Arnfinnsson, J., Steingrimsson, O. & Thormar, H. (2001). In vitro killing of Candida albicans by fatty acids and monoglycerides. Antimicrob. Agents Chemother., 45, 3209–3212.
- Bergsson, G., Steingrimsson, O. & Thormar, H. (2002). Bactericidal effects of fatty acids and monoglycerides on *Helicobacter pylori*. Int. J. Antimicrob. Agents, 20, 258–262.
- Binsi, P. K. Ravishankar, C. N. & Srinivasa Gopal, T. K. (2013). Development and characterization of an edible composite film based on chitosan and virgin coconut oil with improved moisture sorption properties. J. Food Sci., 78(4), 526-53.
- Boyaval, P., Corre, C., Dupuis, C. & Roussel, E. (1995). Effects of free fatty acids on propionic acid bacteria. *Lait.*, 75, 17-29.
- Bomba, R., Nemcova, S., Gancarcikova, R., Herich, P., Guba, D. & Mudronov, a I. (2002). Improvement of the probiotic effect of micro-organisms by their combination with maltodextrins, fructo-oligosaccharides and polyunsaturated fatty acids. *Br. J. Nutr.*, 88(1), 95–99.
- Botterweck, A. A., Verhagen, H., Goldbohm, R. A., Kleinjans, J. & van den Brandt, P. A. (2000). Intake of butylated hydroxyanisole and butylated hydroxytoluene and stomach cancer risk: results from analyses in the Netherlands Cohort Study. *Food. Chem. Toxicol.*, 38(7), 599-605.
- Budin-Verneuil, A., Maguin, E., Auffray, Y., Ehrlich, S. D. & Pichereau, V. (2005). Transcriptional analysis of the cyclopropane fatty acid synthase gene of *Lactococcus lactis* MG1363 at low pH. *FEMS Microbiol. Lett.*, 250, 189–194.
- Campbell, J. W., Morgan-Kiss, R. M. & Cronan, Jr. J. E. (2003). A new *Escherichia coli* metabolic competency: growth on fatty acids by a novel anaerobic beta-oxidation pathway. *Mol. Microbiol.*, 47, 793-805.
- Campbell, J. W. & Cronan, Jr. J. E. (2001). *Escherichia coli* FadR positively regulates transcription of the *fabB* fatty acid biosynthetic gene. *J. Bacteriol.*, *183*, 5982-5990.
- Canas-Rodriguez, B. & Smith, H. W. (1966). The identification of the antimicrobial factors of the stomach contents of suckling rabbits. *Biochem. J.*, *100*, 79–82.
- Chadeganipour, M. & Haims, A. (2001). Antifungal activities of pelargonic and capric acid on *Microsporum gypseum*. *Mycoses.*, 44, 109-112.
- Chamberlain, N. R., Mehrtens, B. G., Xiong, Z., Kapral, F. A., Boardman, J. L. & Rearick, J. I. (1991). Correlation of carotenoid production, decreased membrane fluidity, and resistance to oleic acid killing in *Staphylococcus aureus* 18Z. *Infect. Immun.*, 59, 4332-4337.
- Conley, A. J. & Kabara, J. J. (1973). Antimicrobial action of esters of polyhydric alcohols. *Antimicrob. Agents Chemother.*, 4, 501–506.
- Corcoran, B. M., Stanton, C., Fitzgerald, G. F. & Ross, R. P. (2007). Growth of probiotic lactobacilli in the presence of oleic acid enhances subsequent survival in gastric juice. *Microbiology.*, 153, 291–299.
- Corn Refiners Association 1701 Pennsylvania Avenuem N.W. (2006). Corn oil. Washington, D.C. 20006- 5805. 5th Edition.

- Das, S. K., Masuda, M. A. & Sakakibara, S. M. (2010). Medicinal uses of the mushroom Cordyceps militaris: Current state and prospects. *Fitoterapia*. *81*(8), 961–968.
- DebMandal, M. & Mandal, S. (2011). Coconut (*Cocos nucifera* L.: Arecaceae): In health promotion and disease prevention. *Asian Pac. J. Trop. Med.*, 4, 241-7.
- Decuypere, J. A. & Dierick, N. A. (2003). The combined use of triacylglycerols containing medium-chain fatty acids and exogenous lipolytic enzymes as an alternative to in-feed antibiotics in piglets: concept, possibilities and limitations. An overview. *Nutr. Res. Rev.*, 16, 193–209.
- Desbois, A. P. & Smith, V. J. (2010). Antibacterial free fatty acids: activities, mechanisms of action and biotechnological potential. *Appl. Microbiol. Biotechnol.*, 85, 1629–1642.
- Dima, St. Bahrim, G. & Iordachescu G. (2014). Sources, Production and Microencapsulation of Probiotics, 25-50. In: Semih, O. (ed.) *Probiotic and Prebiotics in Food Nutrition and Health*, CRC Press Taylor&Francis Group, Boca Raton, FL, USA
- Drake, D. R., Brogden, K. A., Dawson, D. V. & Wertz, P. W. (2008). Thematic review series: skin lipids. Antimicrobial lipids at the skin surface. *J. Lipid Res.*, 49, 4–11.
- Food and Drug Administration (1999) Code of Federal Regulations, Title 21, Vol., *3*, Part 184, Sec. 184.1505, page 505, US Government Printing Office, Washington, DC.
- Fourniat, J., Colomban C., Linxe C. & Karam D. (1992). Heat-killed *Lactobacillus* acidophilus inhibits adhesion of *Escherichia coli* B41 to HeLa cells. Ann. Vet. Res., 23, 361–370.
- Galfi, P. & Bokori, J. (1990). Feeding trial in pigs with a diet containing sodium n-butyrate. *Acta Vet. Hung.*, *38*, 3–17.
- Gardner, H. W., Hou, C. T., Weisleder, D. & Brown, W. (2000). Biotransformation of linoleic acid by *Clavibacter* sp. ALA2: cyclic and bicyclic fatty acids. *Lipids.*, 35, 1055– 1060.
- Georgel, P., Crozat, K., Lauth, X., Makrantonaki, E., Seltmann, H., Sovath, S., Hoebe, K., Du, X., Rutschmann, S., Jiang, Z., Bigby, T., Nizet, V., Zouboulis, C. C. & Beutler, B. (2005). A toll-like receptor 2-responsive lipid effector pathway protects mammals against skin infections with Gram-positive bacteria. *Infect. Immun.*, 73, 4512–4521.
- German, J. B. & Dillard, C. J. (2004). Saturated fats: what dietary intake? Am. J. Clin. Nutr., 80, 550-559.
- Gershon, H. & Shanks, L. (1978). Antifungal activity of fatty acids and derivatives: structureactivity relationships. In: Kabara JJ ed. The Pharmacological Effect of Lipids. Champaign, IL: American Oil Chemists' Society. 51-62.
- Gibson, G. R. & Roberfroid, M. B. (1995). Dietary modulation of the human colonic microbiota: introducing the concept of prebiotics. *J. Nutr.*, *125*, 1401–1412.
- Glatz, J. F.C. & Luiken, J. F.P. (2015). Fatty acids in cell signaling: Historical perspective and future outlook. *Prostaglandins, Leukotrienes and Essential Fatty Acids*, 92, 57–62.
- Guha, S., Klees, M., Wang, X., Li, J., Dong, Y. & Cao, M. (2012). Influence of planktonic and sessile *Listeria monocytogenes* on *Caenorhabditis elegans*. Arch. Microbiol., 195, 19-26.
- Gui, L., Sunnarborg, A. & LaPorte, D. C. (1996). Regulated expression of a repressor protein: FadR activates *iclR. J. Bacteriol.*, 178, 4704-4709.
- Guillot, A., Obis, D. & Mistou, M. Y. (2000). Fatty acid membrane composition and activation of glycine-betaine transport in *Lactococcus lactis* subjected to osmotic stress. *Int. J. Food Microbiol.*, 55, 47–51.

- Heczko, P. B., Lutticken, R., Hryniewicz, W., Neugebauer, M. & Pulverer, G. (1979). Susceptibility of *Staphylococcus aureus* and group A, B, C, and G streptococci to free fatty acids. *J. Clin. Microbiol.*, 9, 333–335.
- Henry, M. F. & Cronan, Jr. J. E. (1992). A new mechanism of transcriptional regulation: release of an activator triggered by small molecule binding. *Cell.*, 70, 671-679.
- Hoffman, C., Schweitzer, T. R. & Dalby, G. (1939). Fungistatic properties of the fatty acids and possible biochemical significance. *Food Res.*, *6*, 539–545
- Hou, C. T., Gardner, H. W. & Brown, W. (1998). Production of multihydroxy fatty acids from linoleic acid by *Clavibacter* sp. ALA2. J. Am. Oil Chem. Soc., 75, 1483–1487.
- Hou, C. T., (2005). Monooxygenase system of *Bacillus megaterium* ALA2: Studies on palmitic acid oxidation products. J. Am. Oil Chem. Soc., 82, 839–843
- Isaacs, C. E., Litov, R. E. & Thormar, H. (1995). Antimicrobial activity of lipids added to human milk, infant formula, and bovine milk. *J. Nutr. Biochem.*, *6*, 362–366.
- Jacques, N. A., Jacques, V. L., Wolf, A. C. & Wittenberger, C. L. (1985). Does an increase in membrane unsaturated fatty acids account for Tween 80 stimulation of glucosyltransferase secretion by *Streptococcus salivarius*? J. Gen. Microbiol., 131, 67– 72.
- Johnsson, T., Nikkila, P., Toivonen, L., Rosenqvist, H. & Laakso, S. (1995). Cellular fatty acid profiles of *Lactobacillus* and *Lactococcus* strains in relation to the oleic acid content of the cultivation medium. *Appl. Environ. Microbiol.*, *61*, 4497–4499.
- Jumaa, M., Furkert, F. H. & Muller, B. W. (2002). A new lipid emulsion formulation with high antimicrobial efficacy using chitosan. *Euro. J. Pharma. Biopharma.*, *53*, 115–123.
- Kabara, J. J. (1978). Fatty acids and derivatives as antimicrobial agents: a review, 1–14. In : Kabara, J.J.(ed.) Symposium on the Pharmacological Effect of Lipids. The American Oil Chemists' Society, Champaign, IL, USA.
- Kabara, J. J. (1980). Lipids as host-resistance factors of human milk. Nutr. Rev., 38, 65–73.
- Kabara, J. J. (1991). Chemistry and biology of monoglycerides in cosmetic formulations. *Cosmet. Sci. Technol. Ser.*, 11, 311–344
- Kamem, S. L., Vannini, L. & Guerzoni, M. E. (2009). Effect of α-linolenic, capric and lauric acid on the fatty acid biosynthesis in *Staphylococcus aureus*. *Int. J. Food Microbiol.*, *129*, 288-294.
- Kankaanpää, P. E., Salminen, S. J, Isolauri, E, Lee, Y. K. (2001). The influence of polyunsaturated fatty acids on probiotic growth and adhesion. *FEMS Microbiol. Lett.*, 194, 149–53.
- Kankaanpaa, P., Yang, B., Kallio, H., Isolauri, E. & Salminen, S. (2004). Effects of Polyunsaturated Fatty Acids in Growth Medium on Lipid Composition and on Physicochemical Surface Properties of lactobacili. AEM. 70(1), 129–136.
- Kitahara, T., Koyama, N., Matsuda, J., Aoyama, Y., Hirakata, Y., Kamihira, S., Kohno, S., Nakashima, M. & Sasaki, H. (2004). Antimicrobial activity of saturated fatty acids and fatty amines against methicillin-resistant *Staphylococcus aureus*. *Biol.Pharm. Bull.*, 27, 1321–1326.
- Kendrick, A. & Ratledge, C. (1996). Cessation of polyunsaturated fatty acid formation in four selected filamentous fungi when grown on plant oils. J. Am. Oil. Chem. Soc., 73, 431– 435.

- Kim, J. S., Je, Y. H. & Roh, J. Y. (2010). Production of thermotolerant entomopathogenic Isaria fumosorosea SFP-198 conidia in corn-corn oil mixture. J. Ind. Microbiol Biotechnol., 37, 419–423.
- Kim, J. S., Je, Y. H., Woo, E. O. & Park, J. S. (2011). Persistence of *Isaria fumosorosea* (*Hypocreales: Cordycipitaceae*) SFP-198 Conidia in Corn Oil-Based Suspension. *Mycopathologya*. 171(1), 67-75
- Klaenhammer, T. R., Barrangou, R., Buck, B. L., Azcarate-Peril, M. A. & Altermann, E. (2005). Genomic features of lactic acid bacteria effecting bioprocessing and health. *FEMS Microbiol. Rev.*, 29, 393–409.
- Knapp, H. R. & Melly, M. A. (1986). Bactericidal effects of polyunsaturated fatty acids. J. Infect. Dis., 154, 84–94.
- Kongmun, P., Wanapat, M., Pakdee, P. & Navanukraw, C., Yu, Z. (2011). Manipulation of rumen fermentation and ecology of swamp buffalo by coconut oil and garlic powder supplementation. *Livestock Science*, 135, 84–92.
- Kristmundsdottir, T., Arnadottir, S. G., Bergsson, G. & Thormar, H. (1999). Development and evaluation of microbiocidal hydrogels containing monoglyceride as the active ingredient. J.Pharm. Sci., 88, 1011–1015.
- Kumaratilake, L. M., Robinson, B. S., Ferrante, A. & Poulos, A. (1992). Antimalarial properties of n-3 and n-6 polyunsaturated fatty acids: *in vitro* effects on *Plasmodium falciparum* and *in vivo* effects on *P. berghei. J. Clin. Invest.*, 89, 961–7.
- Kusunose, M., Kusunose, E. & Coon, M. J. (1964). Enzymatic ω-oxidation of fatty acids. I. Products of octanoate, decanoate, and laurate oxidation. *J. Biol. Chem.*, 239, 1374–1380.
- Labows, J., Reilly, J., Leyden, J. & Preti, G. (1999). Axillary odor determination, formulation, and control, in 59–82. In: Laden, K. (ed.) *Antiperspirants and Deodorants*, 2nd Edition Marcel Dekker, New York, NY, USA.
- Laparra J. M. & Sanz Y. (2010). Interactions of gut microbiota with functional food components and nutraceuticals. *Pharmacol. Res.*, 61, 219–225.
- Lee, J. Y., Kim, Y. S. & Shin, D. H. (2002). Antimicrobial synergistic effect of linolenic acid and monoglyceride against *Bacillus cereus* and *Staphylococcus aureus*. J. Agric. Food Chem., 50, 2193–2199.
- Leyva, M. O., Vicedo, B., Finiti, I., Del Amo, G., Real, M. D., García-Agustin, P. & González-Bosch, C. (2008). Preventative and post-infection control of *Botrytis cinerea* in tomato plants by hexanoic acid. *Plant Physiol.*, 57, 1038-1046.
- Liu, S., Weibin, R., Jing, L., Hua, X., Jingan, W., Yubao, G. & Jingguo, W. (2008). Biological control of phytopathogenic fungi by fatty acids. *Mycopathologia.*, 166, 93-102.
- Machado, M. C., Lopez, C. S., Heras, H. & Rivas, E. A. (2004). Osmotic response in *Lactobacillus casei* ATCC 393: biochemical and biophysical characteristics of membrane. *Arch. Biochem. Biophys.*, 422, 61–70.
- Maheswari, N. U. & Soundariya, S. (2012). Coconut oil cake A novel substrate of solid state fermentation for the production of α -amylase using *Streptomyces* spp. *JOCPR.*, *4*(7), 3383-3386.
- Mandal, S. M., Dey, S., Mandal, M., Sarkar, S., Neto, S. M. & Franco, O. L. (2009). Identification and structural insights of three novel antimicrobial peptides isolated from green coconut water, *Peptides*, 30, 633–637

- Marina, A. M., Che Mana, Y. B. & Amin, I. (2009). Virgin coconut oil: emerging functional food oil. *Trends in Food Sci. Tech.*, 20, 481-487.
- Marounek, M., Skrivanov´a, E. & Rada, V. (2003). Susceptibility of *Escherichia coli* to C2– C18 fatty acids. *Folia Microbiol.*, 48, 731–735.
- McDonough, V., Stukey, J. & Cavanagh, T. (2002). Mutations in erg4 affect the sensitivity of Saccharomyces cerevisiae to medium-chain fatty acids. BBA-Gen Subjects., 1581, 109-118.
- Moon, H. I., (2010). Antiplasmodial and cytotoxic activity of phloroglucinol derivatives from *Hypericum erectum Thunb. Phytother. Res.*, 24, 941–944.
- Muller, J. A., Ross, R. P., Sybesma, W. F.H., Fitzgerald, G. F. & Stanton, C. (2011). Modification of the technical properties of *Lactobacillus johnsonii* NCC 533 by supplementing the growth medium with unsaturated fatty acids. *Appl. Environ. Microbiol.*, 77(19), 6889–6898
- Murzyn, A., Krasowska, A., Stefanowicz, P., Dziadkowiec, D. & Łukaszewicz, M. (2010). Capric acid secreted by S. boulardii inhibits C. albicans filamentous growth, adhesion and biofilm formation. Plos One., 5(8), 1-7.
- Nair, M. K., Abouelezz, H., Hoagland, T. & Venkitanarayanan, K. (2005a). Antibacterial effect of monocaprylin on *Escherichia coli* O157:H7 in apple juice. J. Food Prot., 68, 1895–1899.
- Nair, M. K., Joy, J., Vasudevan, P., Hinckley, L., Hoagland, T. A. & Venkitanarayanan, K. S. (2005b). Antibacterial effect of caprylic acid and monocaprylin on major bacterial mastitis pathogens. J. Dairy Sci., 88, 3488–3495.
- Nakatsuji, T., Kao, M. C., Fang, J. Y., Zouboulis, C. C., Zhang, L., Gallo, R. L. & Huang, C. M. (2009). Antimicrobial property of lauric acid against *Propionibacterium acnes*: Its therapeutic potential for inflammatory acne vulgaris. *J. Invest. Dermatol.*, 129, 2480–2488.
- Nugrahini, A. D. & Soerawidjaja, T. H. (2015). Directed Interesterification of Coconut Oil to Produce Structured Lipid. *Agriculture and Agricultural Science Procedia*, *3*, 248 – 254
- Nweze, N. E., Anene, B. M. & Asuzu, I. U. (2011). In vitro anti-trypanosomal activities of crude extracts, b-sitosterol and a-sulphur from *Buchholzia coriacea* seed. *Afr. J. Biotechnol.*, 10, 15626–15632.
- Oyi, A. R., Onaolapo, J. A. & Obi, R. C. (2010). Formulation and antimicrobial studies of coconut (*Cocos nucifera* Linne) Oil. *Res. J. Appl. S. Eng. Tech.*, 2, 133-137.
- Paiwan, P., Gunjan, G., Marta, L., Chalermpon, Y. & Veerle, F. (2013). Medium-chain fatty acids from coconut or krabok oil inhibit *in vitro* rumen methanogenesis and conversion of non-conjugated dienoic biohydrogenation intermediates, *Anim. Feed Sci. Tech.*, 180, 18– 25.
- Panyakaewa, P., Goel, G., Louren, M., Yuangklang, C. & Fievez, V. (2013). Medium-chain fatty acids from coconut or krabok oil inhibit *in vitro* rumen methanogenesis and conversion of non-conjugated dienoic biohydrogenation intermediates. *Anim. Feed Sci.* and Tech., 180, 18–25.
- Parfene, G., Dima, S., Horincar, V., Dinica, R. & Bahrim, G. (2010). Approaches for the Development of New Food Biopreservatives Obtained by Self-Microemulsifying Formulation of Raw Coconut Fat and Lipase. *Bulletin UASVM Agriculture*, 62, 360-366.

- Parfene, G., Horincar, V. B., Tyagi, A. K., Malik, A. & Bahrim, G. (2013). Production of medium chain saturated fatty acids with enhanced antimicrobial activity from crude coconut fat by solid state cultivation of *Yarrowia lipolytica*. *Food. Chem.*, 136, 1345 – 1349.
- Park, J. P., Kim, S. W., Hwang, H. J., Cho, Y. J. & Yun., J. W. (2002). Stimulatory effect of plant oils and fatty acids on the exo-biopolymer production in *Cordyceps militaris*. *Enzyme Microb. Tech.*, 31, 250–255.
- Partanen, L., Marttinen, N. & Alatossava, T. (2001). Fats and fatty acids as growth factors for *Lactobacillus delbrueckii. Syst. Appl. Microbiol.*, 24, 500–506.
- Pech-Canul, A., Nogales, J., Miranda-Molina, A., Álvarez, L., Geiger, O., Soto, M. J. & López-Lara, I. M. (2011). FadD is required for utilization of endogenous fatty acids released from membrane lipids. J. Bacteriol., 193(22), 6295-6304.
- Peres, N. T. A., Cursino-Santos, J. R., Rossi, A. & Martinez-Rossi, N. M. (2010). *In vitro* susceptibility to antimycotic drug undecanoic acid, a medium-chain fatty acid, is nutrient dependent in the dermatophyte *Trichophyton rubrum*. *World J. Microb. Biot.*, 27, 1719-1723.
- Peters, J. S. & Chin, C. K. (2003). Inhibition of photosynthetic electron transport by palmitoleic acid is partially correlated to loss of thylakoid membrane proteins. *Plant Physiol. Biochem.*, *41*, 117-124.
- Petschow, B. W., Batema, R. P. & Ford, L. L. (1996). Susceptibility of *Helicobacter pylori* to bactericidal properties of medium-chain monoglycerides and free fatty acids. *Antimicrob. Agents Chemother.*, 40, 302–306.
- Pollmann, D. S., Danielson, D. M., Wren, W. B., Peo, E. R. & Shahani, K. M. (1980). Influence of *Lactobacillus acidophilus* inoculum on gnotobiotic and conventional pigs. J. Anim. Sci., 51, 629–637.
- Preston, A., Mandrell, R. E., Gibson, B. W. & Apicella, M. A. (1996). The lipooligosaccharides of pathogenic Gram-negative bacteria. *Crit. Rev. Microbiol.*, 22, 139–180.
- Preuss, H. G., Echard, B., Enig, M., Brook, I. & Elliott, T. B. (2005). Minimum inhibitory concentrations of herbal essential oils and monolaurin for Gram-positive and Gramnegative bacteria. *Mol. Cell. Biochem.*, 272, 29–34.
- Puri, A., Loomis, K., Smith, B., Lee, J. H., Yavlovich, A., Heldman, E. & Blumenthal, R. (2009). Lipid-based nanoparticles as pharmaceutical drug carriers: from concepts to clinic. *Crit. Rev. Ther. Drug Carr. Syst.*, 26, 523–580.
- Ramachandran, S., Patel, A. K., Nampoothiri, K. M., Francis, F., Nagy, V., Szakacs, G. & Pandey, A. (2004). Coconut oil cake—a potential raw material for the production of αamylase. *Bioresource Technol*. 93(2), 169-174.
- Řiháková, Z., Plocková, M. & Filip, V. (2001). Antifungal activity of lauric acid derivatives against Aspergillus niger. Eur. Food Res. Technol., 213, 488-490.
- Rohn, S., Rawel, H. M. & Kroll, J. (2004). Antioxidant activity of protein-bound quercetin. J. Agric. Food Chem., 52, 4725–2479.
- Schlundt, J. (2012). Health and nutritional properties of probiotics in food including powder milk with live lactic acid bacteria. Report of a joint FAO/WHO expert consultation on evaluation of health and nutritional properties of probiotics in food including powder milk with live lactic acid bacteria. FAO / WHO.

- Shanklin, J. & Cahoon, E. B. (1998). Desaturation and related modifications of fatty acids. Annu. Rev. Plant Physiol., 49, 611–641
- Schafer-Korting, M., Mehnert, W. G. & Korting, H. C. (2007). Lipid nanoparticles for improved topical application of drugs for skin diseases. Adv. Drug Deliv. Rev., 59, 427– 443.
- Shibasaki, I. & Kato, N., (1978). Combined effects on antibacterial activity of fatty acids and their esters against Gram-negative bacteria, in Symposium on the Pharmacological Effect of Lipids (ed. J. J. Kabara), The American Oil Chemists' Society, Champaign, IL, USA, 15–24.
- Shin, S. Y., Bajpai, V. K., Kim, H. R. & Kang, S. C. (2007). Antibacterial activity of eicosapentaenoic acid (EPA) against foodborne and food spoilage microorganisms. *LWT*-*Food Sci. Technol.*, 40, 1515-1519.
- Seksik, P., Rigottier-Gois, L., Gramet, G., Sutren, M., Pochart, P., Marteau, P., et al. (2003). Alterations of the dominant faecal bacterial groups in patients with Crohn's disease of the colon. *Gut.*, 52(2), 237-242.
- Semih, O. (2014). Probiotic and Prebiotics in Food Nutrition and Health, CRC Press Taylor&Francis Group, Boca Raton, FL, USA.
- Sjögren, J., Magnusson, J., Broberg, A., Schnürer, J. & Kenne, L. (2003). Antifungal 3-Hydroxy Fatty acids from *Lactobacillus plantarum* MiLAB 14. *Appl. Environ. Microb.*, 69, 7554-7557.
- Skandamis, P. N. & Nychas, G. J. E. (2001). Effect of oregano essential oil on microbiological and physico-chemical attributes of minced meat stored in air and modified atmospheres. *Journal of Applied Microbiology*, 91, 1011-1022.
- Skrivanova, E., Marounek, M., Benda, V. & Brezina, P. (2006). Susceptibility of *Escherichia coli*, *Salmonella* sp. and *Clostridium perfringens* to organic acids and monolaurin. *Vet. Med.*, 51, 81–88.
- Smith, H. W. (1966). The antimicrobial activity of the stomach contents of suckling rabbits. *J. Pathol. Bacteriol.*, *91*, 1–9.
- Stratford, M. & Anslow, P. A. (1996). Comparison of the inhibitory action on Saccharomyces cerevisiae of weak-acid preervatives, uncoulers, and medium-chain fatty acids. FEMS Microbiol. Lett., 142, 53-58.
- Stulnig, T. M., Huber, J., Leitinger, N., Imre, E. M., Angelisová, Nowotny, P. & Waldhäusl, W. (2001). Polyunsaturated eicosapentaenoic acid displaces proteins from membrane rafts by altering raft lipid composition. *J. Biol. Chem.*, 276, 37335-37340.
- Sugiharto S. (2014). Role of nutraceuticals in gut health and growth performance of poultry. *Journal of the Saudi Society of Agricultural Sciences*. Article in press.
- Suutari, M. & Laakso, S. (1992). Temperature adaptation in *Lactobacillus fermentum*: interconversions of oleic, vaccenic and dihydrosterulic acids. J. Gen. Microbiol., 138, 445–450.
- Taheri, J. B., Espineli, F. W., Lu, H., Asayesh, M., Bakshi, M. & Makhostin, M. R. (2010). Antimicrobial effect of coconut flour on oral microflora: An *in vitro* study. *Res. J. Biol. Scs.*, 5, 456-459.
- Thayer, A. M. (2005). Fighting malaria. Chem. Eng. News., 83, 69-82.
- Thayer, A. M. (2005). Preventing malaria. Chem. Eng. News., 83, 85-95.

- Thormar, H., Hilmarsson, H. & Bergsson, G. (2006). Stable concentrated emulsions of the 2monoglyceride of capric acid (monocaprin) with microbicidal activities against the foodborne bacteria *Campylobacter jejuni*, *Salmonella* spp., *Escherichia coli. Appl. Environ. Microbiol.*, 1, 522-526.
- Thormar, H. (2011). *Lipids and Esential Oils as Antimicrobial Agents*, John Wiley&Sons, Ltd., United Kingdom.
- Turnbaugh, P. J., Ridaura, V. K., Faith, J. J., Rey, F. E., Knight, R. & Gordon, J. I. (2009). The effect of diet on the human gut microbiome: a metagenomic analysis in humanized gnotobiotic mice. *Sci. Transl. Med.*, 1(6), 6-14.
- VanderGheynst, J. S., Guo, H. Y., Cheng, Y. S. & Scher, H. (2013). Microorganism viability influences internal phase droplet size changes during storage in water-in-oil emulsions. *Bioprocess. Biosyst. Eng.*, 36, 1427–1434.
- Venter, P., Kock, J. L.F., Botha, A., Coetzee, D. J., Botes, P. J. & Nigam, S. (1997). Production of 3R-hydroxy-polyenoic fatty acids by the yeast *Dipodascopsis uninucleata*. *Lipids.*, 32, 1277–1283.
- Viegas, C. M., Rosa, M. F., Sá-Correia, I. & Novais, J. M. (1989). Inhibition of yeast growth by octanoic and decanoic acids produce during ethanolic fermentation. *Appl. Environ. Microbiol.*, 55, 21-28.
- Walters, D., Raynor, L., Mitchell, A., Walker, R. & Walker, K. (2004). Antifungal activities of four fatty acids against plant pathogenic fungi. *Mycopathologia.*, 157, 87-90.
- Walters, D. R., Walker, R. L. & Walker, K. C. (2003). Lauric acid exhibits antifungal activity against plant pathogenic fungi. *Phytophathology*, 151, 228-230.
- Więckowski, M. R. & Wojtczak, L. (1998). Fatty acid-induced uncoupling of oxidative phosphorylation is partly due to opening of the mitochondrial permeability transition pore. *FEBS Lett.*, 423, 339-342.
- Wille, J. J. & Kydonieus, A. (2003). Palmitoleic Acid Isomer (C16:1∆6) In Human Skin Sebum Is Effective against Gram-Positive Bacteria. Skin Pharmacol. Appl. Skin Physiol., 16, 176–187.
- Wojtczak, L. & Więckowski, M. R. (1999). The mechanisms 1 of fatty acid-induced proton permeability of the inner mitochondrial membrane. J. Bioenerg. Biomembr., 31, 447-455.
- Yam, K. L., Takshistov, P. T. & Miltz, J. (2005). Intelligent packaging: concepts and applications. J. Food Sci., 70(1), 1–10.
- Yang, F. C., Ke, Y. F. & Kuo, S. S. (2000). Effect of fatty acids on the mycelial growth and polysaccharide formation by *Ganoderma lucidum* in shake flask cultures. *Enzyme Microb. Technol.*, 27, 295–301.
- Yenjit, P., Issarakraisila, M., Intana, W. & Chantrapromma, K. (2010). Fungicidal activity of compounds extracted from the pericarp of Areca catachu against *Colletotrichum* gloeosporioides in vitro and in mango fruit. *Postharvest Biol. Tec.*, 55, 129-132.
- Zambiazi, R. C., Przybylski, R., Zambiazi, M. W. & Mendoca, C. B. (2007). Fatty acid composition of vegetable oils and Fats. *B.CEPPA*, *Curitiba.*, 25, 111-120.
- Zofou, D., Kowa, T. K., Wabo, H. K., Ngemenya, M. N., Tane, P. & Titanji, V. P. (2011).Hypericum lanceolatum (Hypericaceae) as a potential source of new antimalarial agents: A bioassay-guided fractionation of the stem bark. *Malaria J.*, *17*, 167–174.

Chapter 9

NUTRITIONAL AND HEALTH BENEFITS OF COCONUT OIL

Ratheesh Mohanan^{1,*}, Asha Sukumarapillai¹ and Sandya Sukumaran²

¹Department of Biochemistry, St.Thomas College, Pala, Kottayam, Kerala, India ²Department of Inorganic and Physical Chemistry, Indian Institute of Science, Bangalore, Karnataka, India

ABSTRACT

Coconut oil is one of the most widely used natural oils which assure good health to mankind. It is rich in saturated fats, especially medium chain triglycerides that enhance digestion and other metabolic functions of the body. Applications of coconut oil extend from the kitchen to the dressing room; it is stable edible oil with unique medicinal properties. It gets easily absorbed in the body and is a nature mimic of the human breast milk fat and hence **it is** used in infant formulae. Coconut oil is extensively used to cure certain dietary disorders, dermatitis, psoriasis etc. The non-edible applications of coconut oil are in the soap industries, pharmaceuticals, cosmetics, rubber substitutes, synthetic resins etc. After a period of controversy, researchers concluded that pure coconut oil promises a healthy heart to the consumers by increasing the total HDL (High-density lipoprotein, 'good cholesterol') content of the body. Thus, coconut oil is a pure food that can play an important role in a well-balanced diet.

Keywords: coconut oil, saturated fats, nutritional benefits, cholesterol

INTRODUCTION

The coconut (*Cocosnucifera*) is the most extensively grown palm with great importance. Coconut tree meets almost all basic needs of mankind including food, shelter, clothing,

^{*} Corresponding author: E-mail address: sivatheertha@gmail.com; Tel. +91-9995012581.

health, wealth, etc. and for this reason, they are known as 'Kalpavirksha' or 'Tree of Life' (Lombard, 2001). Edible product of dried fruit (copra) of coconut is coconut oil which is equally popular just as other coconut products such as coconut water and meat; and indeed, it is the chief source of cooking oil consumed in many South-east and East Asian regions. Additionally, it has also found wide applications in traditional medicines and as carrier oil in pharmaceutical industries besides being widely used in pharmaceuticals, cosmetics andbaking industries. Moreover, it is one of the main ingredients in soap making and infant formulae.

Nutritious products, other than coconut oil, obtained from coconut tree include virgin coconut oil and Neera. Virgin coconut oil (VCO) differs from commercial coconut oil in the way it is processed. The latter is produced from copra or dried coconut meat and undergoes refining processes to make the oil edible. The refined oil produced is called RBD (Refined, Bleached and Deodorized) coconut oil, which is largely used as cooking oil. VCO production does not subject the oil to refining processes since the oil produced is already edible. In effect, the term Virgin refers only to the process and not the chemical properties which are essentially the same in both RBD oil and VCO. Their effect on health would likewise be the same, given the same medium chain fatty acid (MCFA) compositions. Neera is the sweet, oyster white colored sap, tapped from the immature inflorescence of coconut. It is a delicious health drink and a rich source of natural sugars, minerals and vitamins.

Dietary fats and oils are the most concentrated form of energy as they yield 9 calories/g, whereas, carbohydrates and proteins yield only 4 calories/g. The significance of dietary oils is that they provide essential fatty acids (EFA), as their name defines, and are absolute essential nutrients required by the body (Baba et al., 1982). Coconut oil is a dense source of energy; 100g oil provides about 862 kilocalories of energy. Its unique combination of fatty acids can have profound positive effects on health. Coconut oil has low polyunsaturated fatty acid; it is very stable and resistant to oxidation. This makes it excellent cooking oil, thereby protecting our cells against damage. Today, coconut oils available in the market vary dramatically in terms of quality. Low-quality coconut oils, which should be avoided, are processed by chemical extraction, using solvent extracts and chemical residues. Many of them are hydrogenated, bleached and deodorized which produce higher yields and are quicker and less expensive. High-quality coconut oil is a completely different product and is truly the healthiest oil we can consume (Carandang, 2008). It is a much safer alternative to other popular oils such as canola oil, where most of its omega-3s are transformed into trans-fats during the deodorization process, which increases the dangers of chronic diseases.

PHYSICAL CHARACTERISTICS

Coconut oil is a colorless to pale brownish yellow oil and is white in its solid form. Unrefined coconut oil melts at 24-25°C (76°F) and smokes at 177°C (350°F), while refined coconut oil has a higher smoke point of 232°C (450°F). Though coconut oil is known as triglyceride or lipid, it also contains minor proportions of mono- and di-glycerides and has the highest content of glycerol (13.5% to 15.0%). Coconut oil is a highly saturated oil (about 90%), and 60% of its total fatty acid composition are medium-chain fatty acids (MCFA) with a chain length of 6 to 12 carbon atoms (Bach &Babayan, 1982; Bhatnagar et al., 2009; Gopala Krishna et al., 2010; Rosell et al., 1985). The majority of fats in the human diet are

composed almost entirely of long chain fatty acids (LCFA), making coconut oil unique among other dietary fats. Among the most stable of all oils, coconut oil is slow to oxidize and thus resistant to rancidity, lasting up to two years due to its high saturated fat content (Lauretes et al., 2002).

HEALTH AND NUTRITIONAL ASPECTS

Coconut oil is one of the few foods that can be classified as a "super food or functional food." The health and nutritional benefits derived from coconut oil are unique and compelling (Enig, 1996; Dayrit et. al, 2008), they had stated that medium chain triglyceride, a fraction of coconut oil, has been identified as an important, medically efficacious food.

Increases Digestion and Nutrient Absorption

Most of the fatty acids in the diet are long-chain fatty acids (LCFA) whereas coconut oil is predominantly rich in medium-chain fatty acids (MCFA). They are low molecular weight compounds and are highly soluble in biological fluids. These properties make them unique.Medium chain fatty acids are metabolized differently compared to LCFA (Fushiki, et al., 1995). They are absorbed directly into the portal circulation without re-esterification in intestinal cells (Ferreira et al., 2012). The MCFA are partly independent of the carnitine transport mechanism into the mitochondria of the liver and are rapidly oxidized for the production of energy (Rubin et al., 2000) In contrast, the long-chain fatty acid (LCFA) commonly found in most diets are incorporated into chylomicrons after being absorbed in the intestine where they are subjected to re-esterification and then reach the bloodstream via the lymphatic system (Ferreira et al., 2012). Most LCFA are stored in the adipose tissue (Rego Costa et al., 2012). As a result, coconut oil is used in special food preparations for those who suffer digestive disorders and have trouble in digesting fats (Hoahland& Snider, 1943).

Name of fatty acid	Percentage	Type of fat
Lauric acid	45% to 52%	Saturated fat
Myristic acid	16% to 21%	Saturated fat
Caprylic acid	5% to 10%	Saturated fat
Capric acid	4% to 8%	Saturated fat
Caproic acid	0.5% to 1%	Saturated fat
Palmitic acid	7% to 10%	Saturated fat
Oleic acid	5% to 8%	Monounsaturated fat
Palmitoleic acid	in traces	Saturated fat
Linoleic acid	1% to 3%	Polyunsaturated fat
Linolenic acid	up to 0.2%	Unsaturated fat
Stearic acid	2% to 4%	Saturated fat

Table 1. Fatty Acid Composition of Coconut Oil

Free ebooks ==> www.Ebook777.com

Ratheesh Mohanan, Asha Sukumarapillai and Sandya Sukumaran

Medium-chain fatty acids also improve the absorption of many other nutrients. The absorption of minerals (particularly calcium and magnesium), B vitamins, fat soluble vitamins (A, D, E, K and beta-carotene) and also amino acids **has** been found to increase when infants are fed a diet containing MCFA. In addition, coconut oil can be digested by the salivary lipase, getting absorbed very fast to give energy like carbohydrates. All other fats need the pancreatic lipase for digestion that the infants do not have (Murthi et al., 1987).

Reduces Cardiovascular Diseases

Cardiovascular diseases (CVD) are one of the most common diseases in different parts of the world especially in developing countries. Earlier it was believed that as coconut oil contains a high amount of saturated fats, it initiates a rise in blood cholesterol levels and promotes heart disease. Later, Kurup and Rajmohan (1995) conducted a study on 64 volunteers and found no statistically significant alteration in the serum total cholesterol or LDL cholesterol from baseline values. Kaunitz and Dayrit (1992) reviewed epidemiological and experimental data regarding coconut-eating people and noted that the population studies show that dietary coconut oil does not lead to high serum cholesterol or to high coronary heart disease mortality or morbidity. The research over four decades concerning the benefits of coconut oil in controlling heart disease is quite clear; coconut oil has been shown to be beneficial. The saturated fats in most hydrogenated oils are of a far worse kind. These are high chain triglycerides that the body cannot break down as efficiently as medium chains fatty acids. This means that they build up as fatty deposits around heart and arteries increasing the risk of coronary heart disease. Replacing these oils with coconut oil therefore actually decreases the risk of heart disease (Bellenand et al., 1980; Halden&Lieb, 1961; Nagaraju&Lokesh, 2008; Tsuji et al., 2001). Finally researches concluded that natural coconut oil significantly increases HDL(good cholesterol) which promotes healthy heart but the use of hydrogenated coconut oil may increase LDL(low density lipids, "bad cholesterol") (Harris et al., 1993; Dayrit, 2003).

Neurological Effects

The brain is the functionally and metabolically active organ of the body (Fernstrom, 2000). When the blood glucose levels fall, the brain requires an alternative source of fuel instead of glucose for its function. Ketones are high-energy fuel produced in the liver, specifically to nourish the brain. A common feature found in Alzheimer's disease and many other neurological disorders is chronic inflammation which interferes with normal glucose metabolism. This defect in energy conversion starves the brain cells causing them to degenerate and die. Therefore, if enough ketones are available on a continual basis, they could satisfy the brain's energy needs. However, ketones are only produced when food, particularly carbohydrate, consumption is very low. When coconut oil is consumed, a portion of the MCFA is automatically converted into ketones, which in turn, enhance the functioning of the brain. Case histories of Alzheimer's patients receiving coconut oil have demonstrated that it is possible not only to stop the progression of the disease, but also to bring about significant improvement.

www.Ebook777.com

Improves Immune Support

Maintaining a healthy diet, results in a well-balanced immune system. Including unsaturated oils in the diet disturbs the smooth functioning of the immune system. Here, antioxidants are frequently used up to remove free radicals produced by these oils which ultimately slowdown the immune system. In the case of natural coconut oil, saturated fatty acids are stable and do not oxidize easily, which in turn, reduce the use of antioxidants. Caprylic and Capric fatty acids in coconut oil, when broken down, are converted into specific antibodies that enhance the body's defenses against a range of diseases, both bacterial and viral in nature. Having the right antibodies to fight specific bacteria is central to body's wellbeing. Hence, including coconut oil in the diet is an easy way to ensure good health. Consumption of coconut oil enhances the metabolic rate, which in turn, accelerates healing processes, cell regeneration and smooth functioning of the immune system. When applied to infected areas, coconut oil forms a chemical layer that protects the infected body part from external dust, air, fungi, bacteria and viruses. Coconut oil is highly effective on bruises because it speeds up the healing process of damaged tissues.

Coconut oil has been demonstrated to have anti-inflammatory, analgesic and feverreducing properties. This plays a good role in removal of toxins from cells. It is thought to curb inflammation of cells by improving cellular function. The cells become more efficient in removing toxins. Bruce Fife (Fife, 2013), in "Coconut Cures" lists fifteen toxins neutralized in part or whole by coconut oil. These include aflatoxin, *E. coli*endotoxin, and MSG. Thus in different aspects coconut oil supports the immune system and is an ideal food for immune suppressed individuals.

Enhances Antimicrobial Activity

Major saturated fatty acids in coconut oil include 48% lauric acid (an 12 chain saturated fat), 7% capric acid (an 10 chain saturated fat), 8% caprylic acid (an 8 chain saturated fat) and 5% caproic acid (an 6 chain saturated fat) which enhances the antimicrobial properties of coconut oil. Several reports state that these saturated fatty acids and their monoglycerides kill or inactivate microorganisms by lysing the phospholipid bilayer of plasma membrane, interfering in signal transductions, virus reproduction etc. (Kabara, 1978; Hierholzer&Kabara, 1982). Monolaurin, the monoglyceride of lauric acid, possesses antiviral and antibacterial properties. They kill the enveloped bacteria and viruses by solubilizing their phospholipid bilayers. Evidences suggest that MCFA are effective in destroying bacteria like *Staphylococcus, Helicobacter Pylori, Chalamydia trachomatis, Streptococcus, Neisseria* etc., fungi like Candida and yeast, protozoans like giardia, viruses including herpes virus, influenza, Epstein-Barr virus, hepatitis C virus, human immunodeficiency virus(HIV), and others (Isaacs et al., 1992).

Anti-Cancer Effects

Coconut oil resists cancer and tumors to a greater extent compared to other unsaturated oils. The protective effect of coconut oil against chemically induced colon and breast cancer

is more profound than that of other oils. They inhibit the carcinogenic agents of colon and mammary glands. In 1987 Lim-Sylianco (Lim-Sylianco, 1987) published a 50-year literature review showing the anti-cancer effects of coconut oil. Growing evidences show that cancer is a metabolic disease characterized by cellular mitochondrial respiratory insufficiency. Cancer cells can only survive and thrive off of glucose and amino acid fermentation. A ketogenic cleanse has been proposed as a means of starving off cancer cell development (Seyfried, 2012). The ideal fuel source for the ketogenic cleanse is coconut oil and virgin coconut oil with its powerful immune boosting properties. The metabolism of fatty acids in coconut oil results in the liberation of ketones that normal cells can utilize but cancer cells cannot. These ketone bodies provide a great anti-inflammatory fuel source for the body that also starves the cancer cells from their nutrient demands.

Good for Diabetics

Recent studies show that coconut oil provides a good protection against insulin resistance, a major cause of diabetics. Medium chain fatty acids in coconut oil put a lowerdemand on the enzyme production of the pancreas. This lessens the stress on the pancreas during meal time when insulin is produced most heavily, thus allowing the organ to function efficiently. MCFAs in coconut oil have a greater ability compared to other oils in developing binding affinity between insulin and cells (Yost et al., 1998) which improves the secretion of insulin, which in turn, controls blood sugar. It also helps in effective utilization of blood glucose.

Combat Tooth Decay

Dental caries is a commonly overlooked health problem, affecting 60 to 90 per cent of children and the majority of adults in industrialized countries. Present studies by researchers of Athlone Institute of Technology in Ireland shows that coconut oil can act as a natural antibiotic and helps to fight against the sugar loving bacterium that causes tooth decay. Coconut oil which had been treated with enzymes stopped the growth of Streptococcus bacteria, a major sugar loving bacterium that causes tooth decay. With this, coconut oil finds its way into toothpaste and mouthwash as an active ingredient. Since coconut oil facilitates absorption of calcium by the body, it also helps in getting strong teeth (Southward, 2011).

Reduces Body Weight

Coconut oil is a low calorie fat and helps control body weight. In addition, coconut oil stimulates metabolism to get itself metabolized fast to supply quick energy unlike other fats. This also helps control body weight. Most of the coconut oil is medium chain fat, it gets absorbed and metabolized so fast that it rarely gets transported to fat depots like other fats, altering the lipoprotein fractions of blood– another great boon. Changing the food fat to coconut oil could help reduce weight in obese individuals. Additionally, research has demonstrated that due to its metabolic effect, coconut oil increases the activity of thyroid

which is one of the reasons why some people are unable to lose weight (Geliebter et al., 1983; Kaunits, 1970).

Hair Care

Coconut oil is rich in nutrients that are required for hair growth.Regular massage of the head with coconut oil ensures growth of healthy and shiny hair. Coconut oil is a good conditioner and enhances re-growth of damaged hair. It is therefore used for the manufacture of herbal oils, baby shampoos, conditioners and various anti-dandruff products (Rele&Mohite, 2003).

Strengthens Bones

Coconut oil improves the ability of the body to absorb important minerals. This includes calcium and magnesium which are necessary for the development of bones and also reduces the oxidative stress within the bone, which may prevent structural damage in osteoporotic bone. Thus, coconut oil is very useful to women who are prone to osteoporosis (a disease of bone that can cause fractures) after middle age (Hegde, 2006; Hegde, 2009).

SKIN CARE

Coconut oil is an excellent moisturizer for the skin. It prevents the formation of destructive free radicals and can prevent skin from developing blemishes caused by aging and overexposure to sunlight. When oil is applied over theskin, it aids in exfoliating the dead skin cells which make skin smoother and also reduces wrinkle formation. Coconut oil forms a protective coating over the skin which guards against bacteria. Different formulations of coconut oils protect skin from various skin problems like psoriasis, dermatitis, eczema and other skin infections. Using its antimicrobial property, it also protects the skin from microbial infections. Therefore, coconut oil is one of the basic ingredients in many of the body care products like soap, body lotions, baby oils, sun screens etc. (Gopala Krishna et al., 2010)

Prevents Acne or Pimples

Acne is common during the stage of adolescence due to some hormonal changes experienced in puberty. One of these hormonal changes is the abnormally excessive secretion of oil (sebum) by the sebaceous glands. When bacteria mix up with sebum, there is acne eruption. Coconut oil contains medium chain fatty acids that boost the metabolic processes in the body. Ingesting coconut oil can improve metabolism that enhances cell and hormonal functions. As a result of better digestion, there is regulated excretion of oil by the sebaceous glands (Vala&Kapadiya, 2014).

Eczema

Coconut oil cure can also be true to eczema. Eczema, an inflammation of the epidermis, is caused by hereditary factors, allergens like dust, pets, or plant pollens and irritants like smoke or solvents. As a result, a person experiences itching, flaking, dryness, or even bleeding of the skin. Moisturizers are very important treatment for eczema and coconut oil is best for lasting moisture for the skin (Rethinam, 2013).

CONCLUSION AND PERSPECTIVES

Coconut oil contributes to good health and nutrient benefits to mankind. It is one of the healthiest dietary oils on earth. As a functional food, coconut oil has a long shelf life and is used for the manufacturing of various dairy and other food products. It is one of nature's richest sources of medium chain triglycerides with antiviral, anti-bacterial and anti-fungal properties. Recently, the western world has discovered coconut oil for its ability in promoting weight loss and preventing premature aging. Coconut oil contains short and medium-chain fatty acids which enhance digestion and support healthy functioning of the thyroidgland and enzymesystems. Its antioxidant properties have been shown to slow the aging process. It is safe for cooking at high temperatures.

The regular use of coconut oil in the diet would regularize blood fats and is known to increase the HDL cholesterol fraction while decreasing LDL and triglycerides.Coconut oil contains so many anti-oxidants that it resists oxidation even if it is preserved for as long as a year whereas all other fats would have been already oxidized and have become trans-fatty acids by the time they come on the food store shelves! Coconut oil resists oxidation even on boiling at 76 degrees centigrade. So, there are no trans-fats in coconut oil. While fried foods are not good for health, if fried in coconut oil, fried foods are not that bad, after all. With all these health, nutritional and body care benefits that onegets from coconut oil, there is no doubt as to why this has been hailed as the "miracle oil."

REFERENCES

- Baba, N., Bracco, E. F. & Hashim, S. A. (1982). Enhanced thermogenesis and diminished deposition of fat in response to overfeeding with diet containing medium-chain triglycerides. *American Journal of Clinical. Nutrition*, 35, 678-682.
- Bach, A. C. & Babayan, V. K. (1982). Medium-chain triglycerides. American Journal of Nutrition, 36, 950-62.
- Bellenand, J. F., Baloutch, G., Ong, N. & Lecerf, J. (1980). Effects of coconut oil on heart lipids and on fatty acid utilization in rapeseed oil. *Lipids*, *15*, 938-943.
- Bhatnagar, A. S., Prasanth Kumar, P. K., Hemavathy, J. & Gopala Krishna, A. G. (2009). Fatty Acid Composition, Oxidative Stability, and Radical Scavenging Activity of Vegetable Oil Blends with Coconut Oil. *Journal Of American Oil Chemist' Society*, 86, 991-999.

- Carandang, E. V. (2008). Health benefits of virgin coconut oil. *Indian Coconut Journal*, 38(9), 8-12.
- Dayrit, C. S. (2003). Coconut Oil: AtherogenicOr Not? (What Therefore Causes Atherosclerosis?). *Philippine Journal of Cardiology*, 31, 97-104.
- Dayrit, F. M., Buenafe, O. E. M., Chainani, E. T. & Vera, I. M. S. D. (2008). Analysis of monoglycerides, diglycerides, sterols, and free fatty acids in coconut (*Cocosnucifera L.*) oil by ³¹P NMR spectroscopy. *Journal of Agricultural and Food Chemistry*, 56, 5765-5769.
- Enig, M. G. (2010). Health and nutritional benefits from coconut oil and advantages over competing oils. *Indian Coconut Journal*, 9, 9-15.
- Fernstrom, J. D. (2000). Can nutrient supplements modify brain function? *American Journal* of Clinical Nutrition, 71, 1669-75.
- Ferreira, L., Lisenko, K., Barros, B., Zangeronimo, M., Pereira, L. & Sousa, R. (2012). Influence of medium-chain triglycerides on consumption and weight gain in rats: A systematic review. J. Anim. Physiol. Anim. Nutr., 98, 1–8.
- Fife, B. (2013). Health properties of coconut oil. Agro food industry HI-TECH, 6-7.
- Fushiki, T., Matsumoto, K., Inoue, K., Kawada, T. & Sugimoto, E. (1995). Swimming endurance capacity of mice is increased by chronic consumption of medium-chain triglycerides. *Journal of Nutrition*, 125, 531-539.
- Geliebter, A., Torbay, N., Bracco, E. F, Hashim, S. A & Van Itallie, T. B. (1983). Overfeeding with medium-chain triglyceride diet results in diminished deposition of fat. *Journal of Clinical Nutrition*, 37, 1-4.
- Gopala Krishna, A. G., Raj, Gaurav., Bhatnagar, Ajit Singh., Prasanth Kumar, P. K. & Chandrashekar, Preeti (2010). Coconut Oil: Chemistry, Production and Its Applications. *Indian Coconut Journal*, 15-27
- Halden, V. W. & Lieb, H. (1961). Influence of biologically improved coconut oil products on blood cholesterol levels in human volunteers. *Nutritional Diet*, *3*, 75-88.
- Harris, K. B., Cross, H. R., Pond, W. G. & Mersmann, H. J. (1993). Effect of dietary fat and cholesterol level on tissue cholesterol concentrations of growing pigs selected for high or low serum cholesterol. *Journal of Animal Science*, 71, 807-10.
- Hegde, B. M. (2006). Coconut Oil Ideal fat next only to mother's milk.*Indian Academy of Clinical Medicine*, 7, 16-19.
- Hegde, B. M. (2009). Coconut The best food for human beings' health and longevity. *Indian Coconut Journal*, 17-18.
- Hierholzer, J. C. & Kabara, J. J. (1982). In vitro effects of monolaurins on enveloped RNA and DNA viruses. *Journal of Food Safety*, *4*, 1-12.
- Hoahland, R. & Snider, G. G. (1943). Digestability of certain higher saturated fatty acids and triglycerides. *Journal of Nutrition*, 219-225.
- Isaacs, C. E., Litov, R. E., Marie, P. & Thormas, H. (1992). Addition of lipases to infant formulas produces antiviral and antibacterial activity. *Journal of Nutritional Biochemistry*, 3, 304-308.
- Kabara, J. J. (1978). The Pharmacological Effect of Lipids. American Oil Chemist's Society, Champaign IL, 1-95.
- Kaunits, H. (1970). Nutritional properties of coconut oil. Journal of American Oil Chemist' Society, 47, 462A-465A.

- Kaunitz, H. & Dayrit, C. S. (1992). Coconut oil consumption and coronary heart disease. *Philippine Journal of Internal Medicine* 30, 165-171.
- Kurup, P. A. & Rajmohan, T. (1994). Consumption of coconut oil and coconut kernel and the incidence of atherosclerosis, Coconut and Coconut Oil in Human Nutrition, Coconut Development Board, Kochin, India, 35-59.
- Lauretes, L. R., Rodriguez, F. M., Reano, C. E., Santos, G. A., Laurena, A. C. & Tecson Mendoza, E. M. (2002). Variability in fatty acid and triacylglycerol composition of the oil of coconut (*Cocosnucifera L.*) hybrid and their parental. *Journal of Agriculture and Food Chemistry* 50, 1581-1586.
- Lim Sylianco, C. Y. (1987). Anticarcinogenic effects of coconut oil. *Journal of Coconut Studies*, 12, 89-102.
- Lombard, K. (2001). Reviewing the coconut (*CocosNucifera L.*) Tree of Life, Texas Tech University.
- Murthi, T. N., Sharma, M., Devdhara, V. D., Chatterjee, S. & Chakraborty, B. K. (1987). Storage stability of edible oils and their blends. *Journal of Food Science and Technology*. (*India*), 24, 84-87.
- Nagaraju, A. & Lokesh, B. R. (2008). Rat fed blended oils containing coconut oil with groundnut oil or olive oil showed an enhanced activity of hepatic antioxidant enzymes and a reduction in LDL oxidation. *Food Chemistry*, 108, 950-957.
- Rego Costa, A. C., Rosado, E. L. & Soares-Mota, M. (2012). Influence of the dietary intake of medium chain triglycerides on body composition, energy expenditure and satiety; A systematic review. *Nutr. Hosp.*, 27, 103–108.
- Rele, A. S. & Mohile, R. B. (2003). Effects of mineral oil, sunflower oil, and coconut oil on the prevention of hair damage. *Journal of Cosmetic Science*, *54*, 175-92.
- Rethinam, P. (2013). *Health and Nutritional Aspects of Coconut Oil*. Asian and Pacific Coconut Community (Jakarta, Indonesia).
- Rosell, J. B. King, B. B. & Downes, M. J. (1985). Composition of oils. *Journal of the American Oil Chemists' Society*, 62, 221-230.
- Rubin, M., Moser, A., Vaserberg, N., Greig, F., Levy, Y., Spivak, H., Ziv, Y. & Lelcuk S. (2000). Long-chain fatty acids, in long-term home parenteral nutrition: A double-blind randomized cross-over study. *Nutrition*, 16, 95–100.
- Seyfried, T. (2012). Cancer as a Metabolic Disease. John Wiley and Sons Publishing.
- Southward, K. (2011). The systemic theory of dental caries. General Dentistry, 59(5), 367-73.
- Tsuji, H., Kasai, M, Takeuchi, H., Nakamura, M., Okazaki, M. & Kondo, K. (2001). Dietary medium- chain triglycerides suppress body fat accumulation in a double blind, controlled trial in healthy men and women. *Journal of Nutrition*, 131, 2853-9.
- Vala, G. S. & Kapadiya, P. K. (2014). Medicinal benefits of coconut oil. International Journal of Life Sciences Research, 2, 124-26.
- Yost, T. J., Jensen, D. R., Haugen, B. R. & Eckel, R. H. (1998). Effect of dietary macronutrient composition on tissue-specific lipoprotein lipase activity and insulin action in normal-weight subjects. *Am. J. Clin. Nutr.*, 68, 296–302

Chapter 10

BENEFICIAL EFFECTS OF FATTY ACIDS FROM COCONUT OIL ON HUMAN METABOLISM AND HEALTH

Leontina Gurgu^{*}

Faculty of Food Science and Engineering, "Dunarea de Jos" University of Galati, Galati, Romania

ABSTRACT

The aim of this chapter was to underline the most important beneficial aspects of the fatty acids found in the coconut oil on the human metabolism and, as a consequence, on health. Medium-chain triglycerides (MCT) are the most representative fatty acids from coconut oil that are digested in the gastrointestinal tract, and then subjected to the absorption process, easier than long chain fatty acids. Medium-chain fatty acids are faster oxidized in the liver, being an important source of energy. They increase the energy expenditure and reduce the fat mass. Thus, they have been proposed for usage in the obesity prevention and treatment, and also in the prevention of insulin resistance development through the Peroxisome Proliferator-Activated Receptors (PPARs) activation. Medium-chain fatty acids stimulate the gut hormone, grelin, that is acylated by ghrelin-o-acyltransferase, which acts as a stimulus that sends the information regarding the presence of a calorie-rich environment to the central nervous system, in order to optimize the lipid storage. In vivo and in vitro experiments have shown that fatty acids from coconut oils decrease the inflammatory response in patients with inflammatory bowel disease and that they may reduce the side effects related symptoms of chemotherapy on women diagnosed with breast cancer. Nonetheless, the MCTs are considerated to be therapeutic agents used not only in the inhibition of tumoral cells development. Moreover, they have also been proposed to usage in the diet formula of patients with type 1 diabetes and Crohn's disease. Furthermore, they are able to improve the cognitive function and to support the synaptic transmission in case of acute hypoglycemia, without increasing the blood glucose levels.

^{*} Corresponding author: Email: lili_gurgu@yahoo.com.

Keywords: medium-chain fatty acids, coconut oil, metabolic syndromes, therapeutic agents

INTRODUCTION

Coconut oil has been used for many years in different countries, for edible and non-edible interests. The beneficial aspects of coconut oil, which distinguish it from other oils, derive from its chemical composition that consists of a high content of short (SCFAs) and medium-chain fatty acids (MCFAs). Most of the fatty acids of coconut oil are saturated (90%) and are represented by 47% of lauric acid (C12:0), 18.1% of myristic acid (C14:0), 8.8% of palmitic acid (C16:0), 7.8% of caprylic acid (C8:0) and 6.7% of capric acid (C10:0) (Rossell, 1985). The medium-chain fatty acids (C8-C10) are metabolized differently from the long-chain fatty acids (C14 or longer) and this process leads to significant pharmacological effects.

Usually, the intestinal digestion of long-chain fatty acids begins with the emulsification process through the bile salt action, which allows lipases to hydrolase lipids into free fatty acids and monoglycerides. Medium chain triglycerides (MCTs) are rapidly subjected to hydrolysis without the necessity of bile emulsion, but through the action of the intestinal and pancreatic lipases which generate the MCFAs and the specific monoglyceride (Ramirez et al., 2001).

The absorption process is carried out through passive diffusion because the MCFAs have low affinity for fatty acid binding proteins (Bach & Babayan, 1982). The diffusion of MCFAs into portal blood is associated with the albumin that is used for the transport directly to the liver, without the reesterified process (Bloch, 1974; Guillot et al., 1993; Guillot et al., 1994). For these reasons, medium-chain triglyceride has been used as energy source in syndromes having pancreatic-enzyme deficiency, such as cystic fibrosis (Jensen et al., 1995).

In contrast, LCFAa are incorporated into chylomicrons, which are transported through the lymphatic system before hepatic oxidation and then assembled into more complex triglycerides.

The proved potential of MCFAs is that of reducing body fat by reducing lipoprotein secretion and of protecting the intestinal mucosa from injury. On the other hand, it has been demonstrated that the substitution of LCT with MCT acts on energy metabolism and promotes weight reduction. Further, the MCTs enriched diet is a good strategy for patients with inflammatory bowel disease because the medium chain fatty acids decreases the proinflammatory activity and the symptoms. The clinical symptoms of those patients are the results of an excessive expression of the pro-inflammatory cytokines IL-1, IL-6, and IL-8.

This chapter focuses on the changes induced by the MCFAs in the mitochondrial metabolism and on the oxidative stress, changes that confer beneficial metabolic reactions when MCFAs are used in diets conferring the hypocholesterolemic effect. It also presents the remarks made by the researchers in the field regarding the functions of saturated fatty acids in relation to anti-diabetic properties, intestinal inflammation, thrombogenic and atherogenic risk factor improvement, immune system activation, the inhibition of tumor proliferation and the functions of the nervous system.

EFFECTS OF MEDIUM CHAIN FATTY ACIDS ON HUMAN METABOLISM AND OBESITY

Fat is a valuable macronutrient in human health. The types of fatty acids are important because each of them biologically affects the body in different ways. Vast and long-term researches have been carried out on the impact of specific fatty acid groups on public health (American Dietetic Association, 2007).

Metabolic syndromes (Grundy et al., 2004) include a group of metabolic disorders that contribute to increased cardiovascular morbidity and mortality. The plethora of disorders is represented by abdominal obesity, atherogenic dyslipidemia, hypertension, insulin resistance, and/or glucose intolerance, proinflammatory and prothrombotic states (Grundy et al., 2004; Nagao & Yanagita, 2010). The International Diabetes Federation defines a person as having metabolic syndrome if that person presents abdominal obesity waist circumference, WC \geq 88cm), plus two of the following values: higher triacylglycerol level (\geq 150 mg/dL), reduced HDL-cholesterol (80 mg/dL in males and <50 mg/dL in females), higher blood pressure (systolic \geq 130mmHg or diastolic \geq 85mmHg), and higher plasma glucose (\geq 100 mg/dL).

Assunção et al., (2009) studied the effects of dietary supplementation with coconut oil (for a period of 12 weeks) on women with abdominal obesity. The selected women were aged between 20 and 40 years old (pregnant woman excluded), had a low income family (USD\$1/day) and suffered from abdominal obesity. The diet consisted of the ingestion of 30 mL of coconut oil /day, under specific nutritional recommendations and under a fitness trainer's supervision. The authors concluded that the ingestion of coconut oil offers the reduction of body mass index, hip circumference, total abdominal fat, visceral fat, and waist circumference and that it induces an increase in peripheral insulin resistance, when compared with the control group (Assunção et al., 2009; Crozier et al., 1987; Geliebter et al., 1983; St-Onge et al., 2003; Aluko, 2012).

Other experimental studies have demonstrated that dietary MCFAs/MCTs inhibit fat deposition through enhanced thermogenesis and fat oxidation in animal and human subjects. An open-label pilot study has been carried out by Liau et al. (2011), in order to assess the efficacy of virgin coconut oil (VCO) on lipid profile of some obese volunteers, for a four weeks period. The researchers concluded that VCO is helpful for WC reduction, being proposed for safe use in humans.

On the other hand, studies on human subjects have shown that, by replacing dietary LCT with MCT, the daily energy expenditure increases from 100 kJ (Flat et al., 1985) to 669 kJ (Hill et al., 1989) in men and to 138 kJ/day (White et al., 1999) in women. The MCTs were proposed to be used in the treatment or prevention of human obesity because their energy expenditure released after ingestion was higher, in comparison with LCT (Scalfi et al., 1991; Seaton et al., 1986; Dulloo et al., 1996).

For a 4 weeks period, Onge and Jones (2003) examined the relationship between the composition of the body and the thermogenic responsiveness to MCTs, represented by 55% octanoic and 44% decanoic acids, extracted from coconut oil. The body weights have been measured using a standard scale and the body composition has been assessed using magnetic resonance imaging (MRI). Their research showed that, when dietary LCT is replaced with MCT, the energy expenditure and fat oxidation increases and the body composition changes

Free ebooks ==> www.Ebook777.com

Leontina Gurgu

in a beneficial way. Therefore, MCTs improve the clinical nutrition of patients suffering from malabsorption because they supply both energy and essential fatty acids.

The question "How does MCTs act on these two factors in order to obtain good results for health?" arises. It seems that the fat mass is being reduced through the down-regulation of adipogenic genes as well as through the t factor PPAR γ (Peroxisome Proliferator-Activated Receptor γ , as known as NR1C3) transcription (Han et al., 2003). PPAR γ , a member of the nuclear hormone receptor superfamily is highly expressed in adipose tissues (white and brown), especially during adipogenesis and is activated by NEFA (Non-esterified fatty acids).

A more recent study supports the idea according to which dietary with SCFAs induces PPAR γ to switch from lipid synthesis to hydrolysis. By decreasing the adipose and the hepatic PPAR γ expression and activity, it increases the expression of uncoupling protein 2 (UCP2) and stimulates the oxidative metabolism in these tissues, via AMP-activated protein kinase (den Besten et al., 2015).

An important role in the MCFAs metabolism was directed to cytochrome P450 omega hydroxylase (CYP4), which catalyzes the ω -hydroxylation of saturated, branched chain or unsaturated fatty acids that are further metabolized to dicarboxylic acids (DCA), which are metabolized by the peroxisome β -oxidation system (Figure 1).

Under normal conditions, (5-10%) fatty acids are converted to dicarboxylic acids. The short chain fatty acids are transported to the mitochondria for complete oxidation after the peroxisome β -oxidation (L-PBE) process that prevents DCAs accumulation and, consequently, liver failure through the inflammation and fibrosis induction (Hardwick, 2008).

Therefore, the MCFAs increase the energy expenditure, reduce the fat mass and prevent the development of insulin resistance through the PPARs activation (Figure 1).



Figure 1. Schematic representation of dicarboxylic fatty acids (DCAs) synthesis and their degradation through L-PBE peroxisomal system, after medium chain triglycerides (MCT) liver absorption (adapted from Hardwick, 2008; Ding et al., 2013).

200

www.Ebook777.com

The uncoupling proteins (UCPs) are located in the inner mitochondrial membrane and are involved in different processes, such as energy metabolism regulation, body weight control, and glucose level control (Echtay, 2007).

They have been recently found to produce a biological impact on the areas of degenerative, neurological, circulatory, and immunological diseases and aging. The sequence analysis indicated that UCPs are mitochondrial anion carriers. The uncoupling protein-1 (UCP1) is a brown adipocyte-specific protein not coupled to ATP synthesis, but involved in obesity and in energy expenditure modulation.

The UPC2 and UPC3 are expressed in several cell types participating in fatty acid metabolism: adipose cells, skeletal muscle, and macrophages. More exactly, the UCP2 is found in all mammalian tissues, except parenchymal hepatocytes (Fleury et al., 1997) and UCP3 is found in skeletal muscles and brown fat (Boss et al., 1997).

Because the Ucp2 and Ucp3 genes are placed in a region with genes that encodes proteins involved in the hyperinsulinemia and obesity control (Fleury et al., 1997; Pecqueur et al., 1999) it seems that the fatty acids increase the expression of Ucp2 and Ucp3 (Brand and Esteves, 2005). In a few studies, Robbins and Zhao have suggested that the malfunction of the UCP isfollowed by neurodegenerative disease, atherosclerosis, and cancer. Thus, targeting the UCP contributes as a promising experiment for cancer prevention (Robbins and Zhao, 2011).

In order to reach the mitochondrial matrix, MCFAs do not need carnitine palmitoyl transferase (CPT); however, they do require medium-chain acyl CoA synthases in order to be rapidly oxidized. It has been observed that, after high MCT diets consumption, it increases the hepatic lipogenesis, it down regulates the activity of CPT1 and the MCFAs bypass the intramitochondrial transport process. In a human volunteers study, Labros et al. have shown that, under the condition of hyperinsulinemia-hyperglycemia, the oxidation of fatty acids depends on the rate of fatty acid entrance into the mitochondria (Labros et al., 1996). The difference in the oxidation pattern of the LCFAs and MCFAs is just the requirement of the former for CPT I, in order to reach access into the mitochondria, observation that has been proved after the CPT1 inhibition. Their results suggest that glucose plus insulin directly regulate the fatty acid oxidation by controlling the rate of the LCFA entrance into the mitochondria, the MCFAs entrance not being affected (Labros et al., 1996). Nagasaki et al. (2012) also identify the physiological roles of GPR84 in adipose tissue along with its specific ligands, the medium chain fatty acids. GPR84 is a member of the G protein - coupled receptors (GPCRs) that are remarkable molecules, activated by a spectrum of ligands. GPRs represent the largest family of membrane proteins in the human genome.

In 2006, a group of researchers showed that the orphan G protein-coupled receptor, GPR84, functions as a specific receptor for MCFAs with carbon chain lengths of C9–C14 (Wang et al., 2006), the most potent agonists of GPR84 being C10:0, C11:0, and C12:0. The researchers found that the GPR84 is involved in the inflammatory changes that occur in the adipocytes of the person that suffers from insulin-resistance syndrome (Nagasaki et al., 2012).

In the adipose tissue, the adiponectin and the TNF α are important control factors through the MCFA GPR84 systemthat is involved in the use of excess calories and in the control of the insulinresistance syndrome. In the inflammation process of the adipose tissue, which also increases the GPR84 mRNA expression, some specific products such as tumor necrosis factor (TNF)- α , interleukin-6 (IL-6) and adiponectin are secreted. On the other hand, MCFAs (such as C10) suppress the expression of adiponectin mRNA in the TNF α -primed adipocytes (Nagasaki et al., 2012). Adiponectin is a 29-kDa protein, expressed exclusively in adipocytes that have an antagonist function with the TNF α .

INSULINOTROPIC EFFECT OF MEDIUM CHAIN FATTY ACIDS

Medium chain fatty acids can also act as ligands on two other G protein-coupled receptors: GPR120/FFAR4 and GPR40/FFAR1. These fatty acids receptors are involved in the mechanism that links obesity and type 2 diabetes(T2D). Because the GPR40 is abundantly expressed in the pancreatic β -cells, after its activation, the signal is transduced and the secretion of insulin is stimulated. Insulin acts in many stimulatory and inhibitory ways in different tissues mediated by a complex intracellular signaling pathway. Insulin maintains the glucose homeostasis by decreasing the blood glucose levels and by increasing its uptake into various organs. The mechanism by which the MCFAs enhance the glucose-stimulated insulin secretion takes into account the coupling of GPR40 - Gq/11, resulting in the stimulation of protein kinase C activity and in the enhancement of intracellular calcium (Covington et al., 2006).

In addition, signaling through phospholipase C (PLC) and diacylglycerol (DAG) may contribute to exocytosis of insulin in the pancreatic β cells and its release. Some of the relative insulinotropic effects of the fatty acids may also be exerted via an extracellular mechanism. Thus, when the glucose concentration in the blood increases, the transcription of the GPR40 gene is also intensified which transforms these receptors into a new target for the type II diabetes treatment (Kebede et al., 2012). Fatty acids, including the medium chain fatty acids, may act as extracellular signaling molecules at G protein – coupled receptors, to regulate the β - pancreatic cells, in addition to their effects mediated through their intracellular metabolism.

In 2012, Pu and Liu published a very interesting study about the mechanism through which palmitic acid stimulates the glucose uptake and the GLUT 4 translocation from the citosol to the plasma membrane, in rat skeletal muscle cell line L6. The stimulatory effects of palmitic acids were comparable to those of the insulin. The researchers proved that, in the glucose uptake process, the palmitic acid stimulated Akt (protein kinase B) phosphorylation in a manner depending on the palmitic acid concentration. They also tested the activity of some signaling molecules, such as AMP-activated protein kinase (AMPK) and extracellular signal related kinase (ERK1/2), that were activated through the palmitic acid treatment. It seems that after the increase of the palmitic acid level, the glucose uptake occurs via activation of the AMPK and ERK1/2, which leads to GLT4 translocation (Pu & Liu, 2012). The results obtained by the two researchers are physiologically significant in the obesity treatment because they explain how the fatty acids level induces an earlier Akt phosphorylation, leading to abnormal glucose uptakes, in the case of the obese patients (Pu & Liu, 2012).

The anti-diabetic properties of MCTs have been reported both in humans and animals studies. Yost and Eckel have analyzed the effects of a hypocaloric diet containing MCTs and LCTs on obese women. The rate of weight lost, the amount of ketones from the serum, and the nitrogen balance did not show differences between the groups. Anyway, the patients whoconsumed MCTs needed a glucose increase in order to maintain euglycernia during the

exogenous insulin administration after weight loss (Yost & Eckel, 1989). In general, it appears that the MCTs improve the body tolerance to carbohydrate. In a similar study, Eckel et al. have reported the possibility of improving the insulin-mediated glucose metabolism in patients with diabetes mellitus, but non-insulin-dependents, using a 40% fat diet containing 77.5% medium-chain triglycerides (Eckel et al., 1992). The conclusion was that MCT could increase the insulin-mediated glucose metabolism in diabetic patients, effect that appeared to be mediated by the increases in the insulin-mediated glucose disposal (Eckel et al., 1992).

The insulinotropic property was also attributed to lauric acid from coconut oil *in vitro* experimental conditions, with isolated islets of mice. The synergic effect of the lauric acid combined with the polyphenolic content of the coconut oil offer a reduction in blood sugar. When the lauric acid is in the same percent, the hot extracted virgin coconut oil appears to have a better hypoglycemic effect, compared to the cold extracted virgin coconut oil. The result offers a good perspective on the protective effect of the polyphenolic compounds found in the two oils on pancreatic beta cells from apoptosis (Siddalingaswamy et al., 2011).

In a recent study, Iranloye and colleagues have also proved the hypoglycemic effect of the virgin coconut oil in an induced type I diabetes male rats. The diabetes was induced by alloxan, which generates reactive oxygen species, causing pancreatic cells death. The treatment with a dose of 10 mL/kg body of VCO for a four weeks period has a beneficial effect on hyperglycemia, improving both glucose tolerance and insulin secretion (Iranloye et al., 2013). On a long term, the rats' diet with MCT also improved both insulin sensitivity and glucose tolerance, in comparison with the LCTs diet (Han et al., 2003).

MEDIUM CHAIN FATTY ACIDS AND THE SATIETY EFFECT

Hunger and satiety are mechanisms that govern the energy intake. Satiety is a stage of satiation where hunger decreases because of food ingestion and is influenced by the distribution of the dietary macronutrients. Satiety is influenced by the fatty acid chain length and by the intake mode (oral vs. gastro-intestinal infusions). The published results have shown the fact that the triacylglycerols of the medium chain are more satiating than the triacylglycerols of the long-chain (Onge et al., 2003, Samra, 2010). Moreover, when the MCT is taken orally in the case of humans, the satiating effect is more marked than long chain triglycerides (Onge et al., 2003; Samra, 2010). An interesting study tested the effect of a meal rich in medium-chain SFA (from coconut oil: 0.5% caproic acid, 8% caprylic acid, 6% capric acid, 47.5% lauric acid, 18% myristic acid, 9% palmitic acid , 3% stearic acid, 6% oleic acid cis-9-18:1, 2% linoleic acid cis-9, cis-12-18:2.) on postprandial triglyceridemia (PPL) in seventeen relatives (REL) of patients with type 2 diabetes, compared with 17 controls (CON) (Pietraszek et al., 2012). It also evaluated the effects of these fatty acids on incretin, on ghrelin responses, and on the expression of some genes involved in lipid and glucose metabolism in muscle and adipose tissues.

Glucagon-like peptide-1 (GLP-1) is a hormone secreted by the intestinal L cells. It is involved in glucose homeostasis, in improving insulin secretion and the pancreatic beta-cell function. On the other hand, the ghrelin is a hormone that is synthesized in the human stomach, being involved in meal initiation. The biological functions of ghrelin are: to stimulate the appetite and the food intake, to balance the gastric acid secretion and motility, and to modulate the endocrine and exocrine pancreatic secretions (Delporte, 2013).

The results obtained by Pietraszek and co-workers have shown that, during the first 2 hours after the meal, the TG level did not increase and the insulin, the postprandial incretin and ghrelin responses did not differ significantly between REL and CON (Pietraszek et al., 2012).

Ghrelin was discovered by Nishi et al. (2005) as a gut hormone stimulated by the ingestion of medium-chain fatty acids or by medium-chain triglycerides. It seems that these lipids are the direct source of the fatty acids that induce the acyl modification of the ghrelin. The acyl modification mechanism of the ghrelin after ingesting MCT appears to be slightly different than the one used after administrating high-fat diet (Nishi et al., 2005) and is catalyzed by the Ghrelin-O-acyltransferase (GOAT). The GOAT acylates ghrelin with fatty acids ranges from C7 to C12 and it is believed that the process occurs in the endoplasmic reticulum. The GOAT acylates ghrelin reacts more efficiently with octanoyl-CoA than the decanoyl-CoA at Ser3 residue (Akamizu et al., 2012). The ghrelin acylation, via GOAT, is involved in eating behavior. When the food intake is conducted by necessity, due to energy deficiency, the GOAT mice models can show similar or opposing phenotypes, depending on the type of diet. The GOAT knockout mice fed with medium-chain triglycerides rich diet displayed lower body weight and fat mass, despite the increased food intake (Kirchner et al., 2009).

Thus, the main role of the ghrelin–GOAT system is to send information about lipids to the central nervous system and to act as stimulus, in order to alert the same system about the presence of a calorie-rich environment, and to optimize the lipid storage (Janssen et al., 2012).

On the other hand, the medium-chain acyl molecules modify the stomach ghrelin peptides and are easily displaced by ingested MCFAs or MCTs. Therefore, it might be possible to treat appetite and control obesity by simply ingesting MCFAs or MCTs (Nishi et al., 2012). The adipose tissue gene expression can also be modulated by the type of dietary fat. Some differences in gene expression between patients with type 2 diabetes and their relatives were also found (Pietraszek et al., 2012). In the muscle tissues of the patients with type 2 diabetes, the following genes were upregulated: ACSL1 (coding of the long-chain acyl-CoA synthetase), ACCB (coding of the acetyl-CoA carboxylase beta), LIPE (coding the Hormone-sensitive lipase), and UCP3, as a response to the meal reach in medium chain fatty acids.

The protein encoded by the ACSL1 gene is an isozyme of the long-chain fatty-acidcoenzyme A ligase family, being involved in the conversion of the free long-chain fatty acids into fatty acyl-CoA esters. It also has a decisive function in lipid biosynthesis and in fatty acids degradation. Acetyl-CoA carboxylase beta is supposed to be involved in the regulation of fatty acid oxidation, rather than fatty acid biosynthesis. Hormone-sensitive lipase, in the short form, has a key function in the mobilization of the stored fat inhibited by insulin. Altogether, the data suggest that the MCFAs might represent a potential therapeutic instrument for the clinical manipulation of energy metabolism, through the modulation of ghrelin activity.

The effects of MCFAs on some genes expression were also investigated during rat pregnancy, in order to test the susceptibility of offspring to later-life obesity (Dong et al., 2011). The results obtained showed a down–regulated ACC1 (acetyl-CoA carboxylase) and

fatty acid synthase (FAS) genes expression in the liver of the offspring rat from the MCFA group, accompanied by a decrease in body weight and body fat content. Equally, the mRNA and the protein expression of the adenosine monophosphate (AMP)-activated the protein kinase (AMPK), and the carnitine palmitoyltransferase 1 and the UCP3 were increased (Dong et al., 2011). The researchers' conclusion was that, during pregnancy, the diet rich in MCFAs might prevent later-life obesity in the offspring because of the programming fatty acid metabolism gene expression (Dong et al., 2011). To sum up, the dietary supplemented with MCFAs is able to decrease the basal insulin level, to equilibrate the blood glucose level, and to attenuate the symptoms of the metabolic syndrome (Hajri et al., 2001) by gene expression regulation.

THE EFFECTS OF THE MEDIUM CHAIN FATTY ACIDS ON INTESTINAL INFLAMMATION

Inflammatory bowel disease (IBD), alongside Crohn's disease (CD) and ulcerative colitis are characterized by the inflammation of the gastrointestinal tract. The etiology of IBD is induced by a combination of factors that contributes to this pathogenesis.

Even if the etiology of IBD is not yet known, the existing experimental data indicate that dietary fat may play important roles in the pathogenesis and clinical display of the IBD.

The role of MCT in this disorder was proved in various experiments carried out *in vivo* and *in vitro*. It was suggested that the MCTs decrease the inflammatory response and they were proposed for usage in patients with IBD. The clinical symptoms derived from the excessive expression of pro-inflammatory cytokines such as IL-1, IL-6, and IL-8.

In order to assess the mucosal inflammatory responses of the fat, some researchers have comparatively evaluated the effects of the MCFAs and LCFAs on the interleukin (IL-8) secretion in the intestine-407 cells (Andoh et al., 2000). IL-8 is a human signaling molecule that initiates the inflammatory responses when its expression increases by 10 to 100 fold, in response to some proinflammatory factors such as tumor necrosis factor (TNF)-a / IL-1, bacterial or viral products, and cellular stress. Moreover, in the intestinal lesions of the IBD, the neutrophils level is increased because of the IL-8 action.

It was proved that the absorption process of the MCFAs did not affect the basal secretion of the IL-8 in the intestine cells in comparison with LCFAs, results of clinical importance in the maintenance of the intestinal inflammation (Andoh et al., 2000). The replacement of dietary LCT with MCT is a good strategy for CD patients because MCTs decrease the proinflammatory activity and because they reduce "both steatorrhea and diarrhea as well as fecal electrolyte excretion in patients with a reduced small intestinal mucosal area" (Andoh et al., 2000; Jeppensen and Mortensen 1998).

Another chemical property of the MCT, water solubility, makes the MCT colon absorption possible, a very important process for the CD patients. Medium chain fatty acids are also energy sources for the patients which suffer from ulceration and inflammation in the intestine (Andoh et al., 2000). The nutrition plays an important role in the management of patients with IBD. It seems that a dose of MCT oil up to 50 g/day can be a good dose for patients with ileitis or an extensive resection of the small intestine, as osmotic diarrhea (Eiden, 2003).

In another study, it has also been proved that the MCT rich diet induced a relative decrease in the IL-6 levels, factor that is significantly increased in the serum of the CD patients (Papada et al., 2013).

EFFECTS OF MEDIUM CHAIN FATTY ACIDS ON THE IMMUNE SYSTEM

Recent works have demonstrated that fatty acids play a key role in immune responses. This effect can be obtained in different ways, for instance, by modulation of the intestinal permeability or by influence on intestinal epithelial cytokine expression (Yoshida et al., 2001; Hoshimoto et al., 2002). As presented before, the mRNA of GPR84 is mainly found in brain, heart, muscle, colon, thymus, spleen, kidney, liver, intestine, bone marrow and leukocytes (Wittenberger et al., 2001; Venkataraman and Kuo, 2005). So far, it was reported that the MCFAs activated the GPR84 and stimulated the IL-12p40 production in monocytes/macrophages and neutrophils (Wang et al., 2006), its function being distinguished in the immune system, more exactly in the modulating leukocyte functions and host defense.

Il-12p40 is included in the Interleukin-12 (IL-12) family of cytokines, which also includes important mediators of inflammatory disease such as: IL-12, IL-23, IL-27 and IL-35. The IL-12 (also known as IL-12p70) is composed of two subunits: p35 and p40. The p40 (IL12-p40) subunit can exist and act as an antagonist monomer of the IL-12 function. IL-12p40 operates as a chemoattractant for macrophages, being associated with several pathogenic inflammatory responses. The proinflammatory cytokine IL-12 induces and maintains T helper1 (Th1) responses and inhibits T helper 2 (Th2) responses. Because the MCFAs stimulate the GPR84, it is possible to affect the Th1/Th2 equilibrium and could give important information about the correlation between the metabolic and autoimmune diseases (Venkataraman and Kuo, 2005).

EFFECTS OF THE MEDIUM CHAIN FATTY ACIDS ON CARDIOVASCULAR DISEASE

Cardiovascular disease (CVD) is a substantial issue in most of the developing regions of the world. Cardiovascular disease is a general term that includes ischemic myocardium, myocardial infarction, cerebrovascular disease, hypertension, peripheral arterial disease, arrhythmias, congenital heart disease or rheumatic heart disease (Mendis et al., 2011). Traditionally, it has been estimated that low density lipoprotein cholesterol (LDL-C) is an important marker in assessing the risk of evolving CVD. As it was presented, medium chain fatty acids results after coconut oil digestion. In order to produce energy, they are directly transferred through the portal vein into the liver, where they are oxidized. Because the MCFAs are not incorporated into chylomicrons, they do not contribute to hyperlipidemia. Furthermore, the consumption of coconut oil has not been associated with heart attacks or other forms of cardiovascular disease.

Among all saturated fatty acids, the stearic acid (C18:0) has a unique characteristic from a nutritional point of view, even though it is classified as a saturated fatty acid. In 2001, Kelly and coworkers showed that there is no correlation between the rich stearic acid diet (19 g/day)
and the risk of CVD or for those associated with thrombosis. Two years later, Mensink et al. evaluated the effects of different fatty acids on the ratio of total to high density lipoprotein cholesterol (HDL-C) and on serum lipoproteins. Their results suggested that the stearic acid lowered the cholesterol level, and that the lauric acid, the major compound of the coconut oil, increased the cholesterol level, especially the HDL cholesterol. *In vitro* and *in vivo* studies have consistently shown that the HDL molecules have multiple functions which include: capacities to protect vascular endothelium through effluxing the cellular cholesterol, to decrease vascular constriction, to reduce the inflammatory response, to protect against the pathological oxidation, and to improve the glucose metabolism (Chapman, 2005; Francis, 2010; Navab et al., 2009; Khera et al., 2011; Navab et al., 2011).

Other studies have demonstrated that the stearic acid had a neutral effect on blood total and low density lipoprotein (LDL) cholesterol levels. Yu et al. (1995) developed predictive formulas to evaluate the effect of the stearic acid on blood lipid levels. The analyses revealed that the stearic acid had no effect on blood total, LDL-C and HDL-C levels in adults.

The mechanisms through which stearic acid lowers the plasma cholesterol levels are still debatable. One of these is that the absorption of stearic acid is lower than that of other saturated fatty acids, being a very poor substrate for TG and cholesteryl ester formation (Bonanome et al., 1992; Woollett et al., 1992).

Another hypothesis supports the idea that stearic acid is converted by the Stearoyl-CoA-Desaturase (SCD) enzyme into oleic acid, which does not affect blood cholesterol levels (Pai et al., 1997). Moreover, it seems that stearic acid does not suppress LDL receptors activity and, consequently, the plasma cholesterol level decrease. *In vitro* experiments have shown that stearic acid reduces cholesterol absorption by altering the synthesis of bile acids and cholesterol solubility.

Kelli et al. (2001) compared the thrombosis potential of diets rich in stearic acid with palmitic acid rich diets by measuring the platelet aggregation, the platelet volume and other key components of the haemostatic pathways. The researchers found out that the mean platelet volume, the coagulation factor FVII activity and the plasma lipid concentrations were significantly decreased on diet with stearic acid. The FVII factor is one of the proteins that have been proven to be associated with blood clot in conjunction with the tissue factor that is located outside the blood vessels. Following the same idea, Cocchi and coworkers analyzed the biochemical and functional aspects of the stearic acid on platelets in different population groups: supposedly healthy, depressive, young adult, young children, ischemic and pigs (Cocchi et al., 2009). It seems that the stearic acid protects the platelets aggregation, having a protective role in controlling the ischemic stroke, and being a marker for the ICD (Kelli et al., 2001; Cocchi et al., 2009).

Other several studies have consistently shown that the virgin coconut oil (VCO) decreases the lipid levels and the thrombotic risk factors in rats fed with VCO, compared to copra oil and sunflower oil (Nevin & Rajamohan, 2008; Marina et al., 2009). Furthermore, the high density lipoprotein cholesterol in animals fed with VCO was increased, while low density lipoprotein cholesterol levels were significantly decreased, when compared to copra oil (Nevin & Rajamohan, 2004).

In their next study, Nevin and Rajamohan (2006) offered a new perspective on the beneficial aspects of the VCO. They proved that the VCO has an antioxidant property, distinguished by an increased catalase and superoxide dismutase activities and an increased total glutathionine content, the last one being a sensitive indicator of the antioxidant status.

The phenolic content of the VCO can provide disease prevention because it inhibits the formation of free radicals. It is already known that free radicals are responsible for causing a great number of diseases, including cardiovascular disease and atherosclerosis.

THE EFFECTS OF MEDIUM CHAIN TRIGLYCERIDE ON TUMOR CELLS

Nowadays, cancer research is one of the topics that have priority in the world of science. Many experimental protocols have been designed in order to find the key that turns a normal cell into a tumor and to discover the way in which the body can be precisely determined to cure it. Some investigators have shown that there is a bridge between the ketogenic diet and some antioncogenes, although the correlation between the energy cell metabolism and the genes responsible for the tumor development is not completely understood (Otto et al., 2008; Ramanathan et al., 2005). Cancer is caused by multiple factors which affect the normal cell processes such as cellular proliferation, differentiation and death. Otto and colleagues, for example, have compared the effects of a nutritionally balanced carbohydrate - restricted diet, supplemented with oil extracts displaying different levels of polyunsaturated omega-3 fatty acid and MCT with those of a standard diet, on the growth inhibiting tumors of the human gastric adenocarcinoma cell line 23132/87, in a xenograft model (Otto et al., 2008). Their data suggested an antitumor effect for both omega-3 fatty acids and MCT in patients and experimental models, results that were comparable with those obtained by other researchers (Wolters et al., 1994; Wolters et al., 1994; Nebeling & Lerner, 1995).

Other several studies have shown that coconut oil has an inhibitory effect on the carcinogenic agents used to induce tumors in the colon and mammary in the case of test animals (Reddy & Maeura, 1984; Cohen et al., 1986). In their study, Reddy and Maeura (1984) compared the effect of dietary with corn oil, sunflower oil, olive oil, coconut oil and medium-chain triglycerides on 20 mg azoxymethane/kg body weight, factor that induced colon tumors in F344 female rats. The animals were fed 5% of oils 2 weeks before, during, and 1 week after the azoxymethane injection and after that; for some groups, the dose was increased with 23.52% for oils and 17.64% for MCT. It seems that diets containing high levels of olive oil, coconut oilor MCT had no promoting effect on the colon tumor incidence, compared to the other tested oils, the fatty acid composition / the type of fat being an important factor in the determination of the promoting effect in colon carcinogenesis (Reddy & Maeura, 1984). In vivo studies on the N-methylnitrosourea-induced rat mammary tumor model have proved that diets containing MCT: com oil (3:1) can decline the mammary tumorpromoting effect (Cohen et al., 1984). The study shows that the diet containing high levels of medium chain fatty acids does not induce the development of mammary tumors, in comparison with the diet containing long chain fatty acids from corn oil (Cohen et al., 1984).

Other *in vivo* studies have been made on Yoshida-sarcoma-bearing rat model, in order to prove that the fatty acids composition is essential for the inhibition of the tumor growth (Ling et al., 1991). The researchers fed the animals the same fat carbohydrate ratio, with the exception of dietary fat type LCT or 60% MCT /40% fish oil, and analyzed the changes in the tumor volume, body weight, urinary nitrogen, whole-body and tissue protein kinetics (Ling et al., 1991). The results have demonstrated that the feeding rich in MCT/fish oil correlated with

the TNF treatment leads to a tumor growth inhibition, less total urinary nitrogen excretion and a decreasing tumor protein synthesis in Yoshida-sarcoma-bearing rats (Ling et al., 1991).

Medium-chain fatty acids are adjunctive therapeutic agents for the treatment of colorectal cancer. Fauser et al. (2013) treated the Caco-2 and IEC-6 intestinal cell lines with lauric acid and butyrate, in order to compare their potential for apoptosis induction, glutathione levels modification, and changes of the cell cycle phase.. The results presented were promising in terms of antineoplastic properties of the lauric acid. It was shown that this medium chain fatty acid is capable to induce the death of the cells, to decrease glutathione concentration, to generate reactive oxygen and to block the tumoral cells in the S and G2/M phases or to reduce them in G0/G1. Compared with the butyrate, the lauric acid displayed anti-tumoral properties (Fauser et al., 2013).

For a normal cell, G0 is the phase of the cell cycle where the cell stopped dividing. G1 (Gap1) belongs to the Interphase and is a checkpoint control mechanism that guarantees that the DNA synthesis is ready to start. In the Synthesis (S) phase the DNA replication occurs, followed by G2 (Gap 2), another checkpoint that ensures that the cell is ready to enter the Mitosis (M), thus being ready to divide into two daughter cells (figure 2).

The inhibitory effect on the growth of colorectal cancer cells could be explained by the potential of the lauric acid to down-regulate the genes that are involved in the control of cell cycle and to up-regulate the genes that are correlated with the apoptosis process. Apoptosis is an essential physiological process for the homeostasis maintenance in multicellular organisms through cell death programming. Moreover, apoptosis is a coordinated process in which a group of cysteine proteases (caspase) is activated in a cascade of events that have, as final goal, the demise of the cell. Caspase-8 is an initiator that activates caspase-3, which is considered to be the most important element of the caspases (figure 3). Its role is that of activating the endonuclease CAD (Caspase-activated DNase) and of degrading the chromosomal DNA and the chromatin condensation (Martinnvalet et al., 2005) (figure 3).



Figure 2. The cell cycle. P – Prophase, M-Mitosis, A-Anaphase, T-Telophase; G1- Gap1; S- DNA Synthesis; G2-Gap.



Figure 3. Schematic representation of caspase pathways activation in mammalian cells.

Furthermore, the ketogenic diet has been proven to reduce the tumor growth by lowering the levels of circulating glucose and by reducing the blood insulin levels. A ketogenic diet composed of 60% MCT, 20% proteins, 10% carbohydrates and 10% other dietary fats was divided into five or more small feedings and was given to a selected group of children with cancer, for a period of 7 days (Nebeling and Lerner, 1995). The researchers checked the influence of a ketogenic diet on the tumor glucose metabolism. After this period, the blood glucose level decreased to low-normal levels and the blood ketones were elevated 20 to 30 fold (Nebeling & Lerner, 1995). The obtained results promote MCT on a diet formula for pediatric patients with cancer, who are not under chemotherapy treatment or having other compromising nutritional status (Nebeling & Lerner, 1995).

This could be an interesting approach for cancer therapy, if it is considered that the Warburg effect, a phenomenon with elevated aerobic glycolysis, is based on the metabolic alteration in cancer cells.

Another proven benefit of the VCO was the reducing of symptoms related to the side effects of chemotherapy on persons diagnosed with breast cancer influencing the global quality of life (QOL) (Law et al., 2014; Richardson et al., 2007; Lemieux et al., 2008). Chemotherapy is used in order to inhibit metastasis and to prolong the survival rates but it is linked with many uncomfortable effects such as vomiting, hair loss/ alopecia, fatigue, anxiety, and depression. The research conducted in 2014 by Law et. al was based on some questionnaires, all documented, in order to measure QOL in cancer patients regarding the global health status, the functional status, and the cancer-related symptoms (Law et al., 2014). Thus, after the consumption of VCO (10 mL twice/day) throughout six cycles of

210

Leontina Gurgu

www.Ebook777.com

chemotherapy, the women diagnosed with breast cancer had an improved functional status and global QOL, with decreasing systemic therapy side effects.(Law et al., 2014).

EFFECTS OF MEDIUM CHAIN TRIGLYCERIDE ON THE NERVOUS SYSTEM AND ITS DISORDERS

The nervous system is a complex system that, structurally, has two components: the central nervous system and the peripheral nervous system. Between these two there is a complicated network of signals, that are responsible for all the activities, senses, speech, memories and emotions that a human can have. The human brain, being the most metabolically active organ in the body, requires large amounts of energy in order to develop its functions.

Because in normal conditions the most important source of energy for the brain is glucose, in some unfavorable cases, such as a defective capacity of metabolizing glucose or if the glucose supply is broken, the brain is put at risk. Severe hypoglycemia is a cause for cognitive dysfunctions, including disturbances in the cerebral metabolic rate of glucose (CMRglu) which is implicitly a cause for the deterioration of the cognitive function, without warning symptoms. In some diseases, in the type 1 diabetes, the brain can use alternative fuels, such as monocarboxylic acids, lactate, and ketones.

Medium-chain fatty acids can generate ketones after the oxidizing process that takes place in the brain and can offer type 1 diabetic patients a prophylactic treatment strategy and a normal function of the brain during hypoglycaemic episodes, without increasing the blood glucose levels.

A group of researchers (Page et al., 2009) has shown that MCFAs improve the cognitive function in patients treated for type 1 diabetes and also support the synaptic transmission in the case of acute hypoglycemia. They have observed that the most beneficial effects of the MCFAs ingestion were on verbal memory, which is under hippocampus control (Page et al., 2009).

Moreover, the researchers have checked *in vitro* the hypoglycaemic conditions of the hippocampal synaptic transmission of β -hydroxybutyrate and octanoate. It seems that, in these conditions (*in vitro* hippocampal slice preparations and low-glucose conditions), the β -hydroxybutyrate is involved in the synaptic transmission. Instead, octanoate is involved in the synaptic function recovery after the normalization of the glucose concentration (Page et al., 2009). The study brings valuable results to the diabetes management field, offering patients a normal functioning of the brain during the hypoglycaemic episodes (Page et al., 2009).

The favorable effects of the MCFAs on brain homeostasis in patients with mild cognitive impairment (MCI) were also demonstrated by Rebello and collaborators (2015), on six patients enrolled in a randomized placebo-controlled experiment. MCI is one of the earliest stages of Alzheimer's disease (AD) that was characterized, in incipient stages, by learning and memory dysfunctions.

In AD there are a number of associated abnormalities on the cellular and subcellular level that include dystrophic neurites, the activation of genes from the signaling pathways or the deficiency of the mitochondrial function. Diet supplementation with MCT oil favors the increase of the circulating ketone bodies, which are a good secondary energy sources for the brain when glucose concentration is decreased, situation that has been observed in persons with AD.

In the appearance and development mechanism of the Alzheimer's disease, the presence of apolipoprotein Ee4 allele (ApoE4) seems to play a major role. The evaluation test was designed for a period of 24 weeks on serum β -hydroxybutyrate (BHB) concentration, apolipoprotein-E4 status and cognitive performance in MCI subjects (Rebello et al., 2015). The consumption of 56 g/day of MCT oil seems to have a positive effect on the memory of people with negative or positive ApoE4 status, based on the increased concentration of BHB proving that ketones are able to offset the effects of the cerebral energy metabolism.

The effect of the capric acid (C8:0) was checked on 152 subjects diagnosed with mild to moderate AD in a 90 days study (Henderson et al., 2009). The MCT and the capric acid were introduced in a normal diet as it follows: 10 g/day in the first 7 days and 20 g/day, for the next days. The results presented proved the increase of the serum ketone body, in the next two hours after the administration. More exactly, the mean of the BHB concentration increased from the 0.09 mmol/L in the first day to 0.039 mmol/L, in the last day of the study. This effect was more obvious in ApoE4 negative subjects (Henderson et al., 2009).

The efficacy of the extra virgin coconut oil, whose predominant component is MCT has also been checked in subjects with moderate to severe AD (Gandotra et al., 2014). The test has been carried out on 31 patients who were daily given 20 g extra virgin coconut oil, on a 6 weeks trial. The researchers have evaluated the primary cognitive end points using the Alzheimer's Disease Assessment Scale-Cognitive subscale (ADAS-Cog), coupled with the Clinicians' Interview based on the Impression of Change Plus Caregivers input (CIBIC-Plus) (Gandotra et al., 2014). Gandotra et al. (2014) have obtained the same favorable effect as other researchers who have tested the MCT oils on cognitive performance in subjects with AD, by using the equivalent amount of virgin coconut oil.

In vivo experiments have demonstrated that ketones offer neuroprotection in the case of cell culture models of the two most common degenerative neurological diseases, Alzheimer's and Parkinson's and they have also proved that the ketones may offer neuroprotection in the treatment or prevention of both diseases.

Genetic and environmental factors (toxins or infections) may cause the Parkinson's disease, which is characterized by muscle rigidity, bradykinesia, eosinophilic Lewy-body inclusions and the death of the dopaminergic neurons in the pars compact of the substantia nigra (Jankovic, 2008).

In the investigation of Parkinson's disease it has been found out that the complex I of the electron transport chain from the mitochondria is defective. The role of the NADH formed in the citric acid cycle is to send electrons to complex I and further on, the electrons are sent to complex IV undergoing complex III. The gradient proton is produced and induces the formation of ATP, which is used by the cells in every physiological step that requires energy. Thus, the inadequate function of complex I reduces the ATP formation and causes the oxidative stress (Maalouf et al., 2009). Furthermore, by using a ketogenic diet it is possible to fuel the electron transport chain with electrons to the complex II from the FADH2, which results from the fatty acids oxidation.

Consequently, an enriched MCT diet may slow down the progression of the Parkinson` disease, allowing those patients to have normal lives. Another proper aspect of the ketogenic diet is its usage in therapies for dementia, in dietary management of multiple sclerosis and epilepsy (Maalouf et al, 2009).

CONCLUSION AND PERSPECTIVES

The basis of this chapter was the plenitude of information that exists in literature regarding the beneficial properties of the fatty acids from coconut oil on human health. *In vivo* and *in vitro* studies on both human and animal metabolism offered valuable experimental data on the positive effects of the enriched MCT diets, more precisely on the fact that MCFAs inhibit fat deposition, enhance thermogenesis, fat oxidation, and daily energy expediture, respectively. The chapter also emphasizes the fact that the unique physicochemical characteristics of these fatty acids offer facilities in their absorption mode, transport, oxidation degree and utilization. These data provide new perspectives on the mechanisms that contribute to the supportive effects of the MCFAs on energy homeostasis and insulin action. There are highly available energy substrates in different diseases such as type 2 diabetes, Alzheimer's and Parkinson's disorders, in which the hypoglycaemic phases affect the brain function. The anticancer activity of some MCFAs was also proven on cultured human colorectal, skin and breast cancer cells. Many research studies mention the necessity of more genomic experiments, that would offer comprehensive information about the effects of these fatty acids in relation to the anticancer mechanism.

REFERENCES

- Andoh, A., Takaya, H., Araki, Y., Tsujikawa, T., Fujiyama, Y. & Bamba, T. (2000). Medium- and long-chain fatty acids differentially modulate interleukin-8 secretion in human fetal intestinal epithelial cells. J.Nutr., 130(11), 2636-2640.
- American Dietetic Association, Dietitians of Canada. (2007). Position of the american dietetic association and dietitians of canada: dietary fatty acids. J. Am. Diet. Assoc., 107(9), 1599-1611.
- Assuncao, M. L., Ferreira, H. S., dos Santos, A. F., Cabral, Jr. C. R. & Florencio, T. M. M. T. (2009). Effects of dietary coconut oil on the biochemical and anthropometric profiles of women presenting abdominal obesity. *Lipids.*, 44(7), 593-601.
- Bach, A. C. & Babayan, V. K. (1982). Medium-chain triglycerides: an update. Am. J. Clin. Nutr., 36, 950–962.
- Bloch, R. (1974). Intestinal absorption of medium-chain fatty acids. J. Nutr. Sci., 13, 42-49.
- Bonanome, A., Bennet, M. & Grundy, S. M. (1992). Metabolic effects of dietary stearic acid in mice: changes in the fatty acid composition of triglycerides and phospholipids in various tissues, *Atherosclerosis.*, 94, 119–127.
- Boss, O., Samec, S., Paoloni-Giacobino, A., Rossier, C., Dulloo, A., Seydoux, J., Muzzin, P. & Giacobino, J. P. (1997). Uncoupling protein-3: a new member of the mitochondrial carrier family with tissue-specific expression. *FEBS Lett.*, 408, 39–42.
- Bourque, C., St-Onge, M. P., Papamandjaris, A. A., Cohn, J. S. & Jones, P. J. H. (2003). Consumption of a functional oil composed of medium chain triacylglycerols, phytosterols and n-3 fatty acids improves the overall cardiovascular risk profile of overweight women. *Metabolism.*, 52, 771–777.
- Chapman, M. J. (2005). Beyond the statins: new therapeutic perspectives in cardiovascular disease prevention. *Cardiovasc. Drugs Ther.*, 19, 135–139.

- Crozier, G., Bois-Joyeux, B., Chanez, M., Girard, J. & Peret, J. (1987). Metabolic effects induced by long-term feeding of medium-chain triglycerides in the rat. *Metabolism.*, *36*, 807–814.
- Cocchi, M., Tonello, L. & Lercke, G. (2009). Platelet Stearic Acid in different population groups: biochemical and functional hypothesis. *Nutr. Clín. Diet. Hosp.*, 29(1), 34-45.
- Cohen, L. A., Thompson, D. O., Maeura, Y., Choi, K., Blank, M. E. & Rose, D. P. (1986). Dietary fat and mammary cancer. Promoting effects of different dietary fats on Nnitrosomethylurea- induced rat mammary tumorigenesis. J. Natl. Cancer Inst., 77(1), 33– 42.
- Covington, D. K., Briscoe, C. A., Brown, A. J. & Jayawickreme, C. K. (2006). The Gprotein-coupled receptor 40 family (GPR40–GPR43) and its role in nutrient sensing. *Biochem. Soc. Trans.*, 34, 770-773.
- den Gijs, B., Aycha, B., Albert, G., van Eunen, K., Rick, H., van Dijk, T. H., Oosterveer, M. H., Jonker J. W., Groen A. K., Reijngoud D. J. & Bakker, B. M. (2015). Short-Chain Fatty Acids protect against high-fat diet-induced obesity via a PPARγ-dependent switch from lipogenesis to fat oxidation. *Diabetes.*, 64(5).
- Ding, J., Loizides, M. U., Rando, G., Zoete V., Michielin, O., Reddy, J. K., Wahli, W., Riezman, H. & Thorens, B. (2013). The peroxisomal enzyme L-PBE is required to prevent the dietary toxicity of medium chain fatty acids. *Cell Rep.*, 5, 248–258.
- Dong, Y. M., Lia, Y., Ninga, H., Wanga, C., Liua, J. R. & Sun, C. H. (2011). High dietary intake of medium-chain fatty acids during pregnancy in rats prevents later-life obesity in their offspring. J. Nutr. Biochem., 22(8), 791–797.
- Dulloo, A. G., Fathi, M., Mensi, N. & Girardier, L. (1996). Twenty-fourhour energy expenditure and urinary catecholamines of humans consuming low to- moderate amounts of medium-chain triglycerides: a dose-response study in human respiratory chamber. *Eur. J. Clin. Nutr.*, 50, 152–158.
- Eckel, R. H., Hanson, A. S., Chen A. Y., Berman, J. N., Yost, T. J. & Brass, E. P. (1992). Dietary substitution of medium-chain triglycerides improves insulin-mediated glucose metabolism in NIDDM subjects. *Diabetes.*, 41(5), 641-647.
- Echtay, K. S. (2007). Mitochondrial uncoupling proteins what is their physiological role? Free Radical. *Biol. Med.*, 43, 1351-1371.
- Eiden, K. A. (2003). Nutritional considerations in inflammatory bowel disease. Nutrition issues in gastroenterology, *Pract. Gastroenterol.*, *5*, 33-54.
- Fauser, J. K., Matthews, G. M., Cummins, A. G. & Howarth, G. S. (2013). Induction of apoptosis by the medium-chain length fatty acid lauric acid in colon cancer cells due to induction of oxidative stress. *Chemotherapy.*, 59(3), 214-24.
- Flatt, J. P., Ravussin, E., Acheson, K. J. & Jequier, E. (1985), Effects of dietary fat on postprandial substrate oxidation and on carbohydrate and fat balances. J. Clin. Investig., 76, 1019–1024.
- Fleury, C., Neverova, M., Collins, S., Raimbault, S., Champigny, O., Levi-Meyrueis, C., Bouillaud, F., Seldin, M. F., Surwit, R. S. & Ricquier, D. (1997). Uncoupling protein-2: a novel gene linked to obesity and hyperinsulinemia. *Nat. Genet.*, 15, 269–272.
- Francis, G. A. (2010). The complexity of HDL. Biochim. Biophys. Acta., 1801, 1286–1293.
- Gandotra, S., Kour, J. & Van der Waag, A. (2014). Efficacy of Adjunctive Extra Virgin Coconut Oil Use in Moderate to Severe Alzheimer's Disease. *Int. J. Sch. Cog. Psychol.*, *1*(2), 1-10.

- Geliebter, A., Torbay, N., Bracco, E. F., Hashim, S. A. & Van Itallie, T. B. (1983). Overfeeding with medium-chain triglyceride diet results in diminished deposition of fat. *Am. J. Clin. Nutr.*, 37(1), 1-4.
- Greenberger, N. J. & Skillmann, T. G. (1969), Medium chain triglycerides. Physiologic considerations and clinical implications. *New Engl. J. Med.*, 280, 1045–1058.
- Grundy, S. M., Brewer, H. B., Cleeman, J. I., Smith, S. C. & Lenfant, C. (2004). Definition of metabolic syndrome-Report of the National Heart, Lund and Blood Institute/American Heart Association Conference on Scientific Issues Related to Definition. *Circulation.*, 109, 433-8.
- Guillot, B., Vaugelade, P., Lemarchal, P. & Rerat A. (1993). Intestinal absorption and liver uptake of medium-chain fatty acids in non-anaesthetized pigs. *Brit. J. Nutr.*, 69, 431–442.
- Guillot, E., Lemarchal, P. & Dhorne, T. (1994). Intestinal absorption of medium chain fatty acids: in vivo studies in pigs devoid of exocrine pancreatic secretion. *Brit. J. Nutr.*, 72, 545–553.
- Han, J. R., Hamilton, J. A., Kirkland, J. L., Corkey, B. E. & Guo, W. (2003). Medium-chain oil reduces fat mass and downregulates expression of adipogenic genes in rats. *Obes. Res.*, 11, 734–744.
- Hardwick, J. P. (2008). Cytochrome P450 omega hydroxylase (CYP4) function in fatty acid metabolism and metabolic diseases. *Biochem. Pharmacol.*, 75, 2263–2275.
- Henderson, S. T., Vogel, J. L., Barr, L. J., Garvin, F., Jones, J. J. & Costantini, L. C. (2009). Study of the ketogenic agent AC1202 in mild to moderate Alzheimer's disease: a randomized, double-blind, placebo-controlled, multicenter trial. *Nutr. Metab.*, 6, 31.
- Hill, J. O., Peters, J. C., Yang, D., Sharp, T., Kaler, M., Abumrad, N. N. & Greene, H. L. (1989). Thermogenesis in humans during overfeeding with medium-chain triglycerides. *Metabolis.*, 38, 641–648.
- International Diabetes Federation. Worldwide definition of the metabolic syndrome Anthropometric Profiles of Women Presenting Abdominal Obesity. *Lipids.*, 44, 593–60.
- Iranloye, B., Oludare, G. & Olubiyi, M. (2013). Anti-diabetic and antioxidant effects of virgin coconut oil in alloxan induced diabetic male Sprague Dawley rats. *JDM.*, *3*(4), 221-226.
- Jankovic, J. (2008). Parkinson's disease: Clinical features and diagnosis. *J.Neurol. Neurosur. Ps.*, 79 (4), 368–76.
- Janssen, S., Laermans, J., Iwakura, H., Tack, J. & Depoortere, I. (2012). Sensing of Fatty Acids for Octanoylation of Ghrelin Involves a Gustatory G-Protein. *PLoS ONE.*, 7(6), 1-11.
- Jensen, M. M., Christensen, M. S. & Hoy, C. E. (1994). Intestinal absorption of octadecanoic, decanoic and linoleic acid: effect of triglyceride structure. *Ann. Nutr. Metab.*, *38*, 104–16.
- Jeppesen, P. B. & Mortensen, P. B. (1998). The influence of a preserved colon on the absorption of medium chain fat in patients with small bowel resection. *Gut.*, 43, 478–483.
- Kebede, M., Ferdaoussi, M., Mancini, A., Alquier, T., Kulkarni, R. N., Walker, M. D. & Poitout, V. (2012). Glucose activates free fatty acid receptor 1 gene transcription via phosphatidylinositol-3-kinase-dependent O-GlcNAcylation of pancreas-duodenum homeobox-1. *Proc. Natl. Acad. Sci. U.S.A.*, 109 (7), 2376–81.
- Kelly, F. D., Sinclair A. J., Mann N. J., Turner, A. H., Abedin, L. & Li, D. (2001). A stearic acid-rich diet improves thrombogenic and atherogenic risk factor profiles in healthy males. *Eur. J Clin. Nutr.*, 55, 88-96.

- Khera, A.V., Cuchel, M., de la Llera-Moya, M., Rodrigues, A., Burke, M. F., Jafri, K., French, B. C., Phillips, J. A., Mucksavage, M. L., Wilensky, R. L., Mohler, E. R., Rothblat, G. H. & Rader, D. J. (2011). Cholesterolefflux capacity, high-density lipoproteinfunction, and atherosclerosis. *N. Engl. J. Med.*, 364, 127–135.
- Kirchner, H., Gutierrez, J. A. & Solenberg, P. J. (2009). GOAT links dietary lipids with the endocrine control of energy balance. *Nat. Med.*, 15, 741–745.
- Kossoff, E. H., Zupec-Kania, B. A. & Rho, J. M. (2009). Ketogenic diets: an update for child neurologists. J. Child. Neurol., 24, 979-988.
- Law, K. S., Azman, N., Omar, E. A., Musa, M. Y., Yusoff, N. M, Sulaiman, S. A. & Hussain, N. H. N. (2014). The effects of virgin coconut oil (VCO) as supplementation on quality of life (QOL) among breast cancer patients. *Lipids Health Dis.*, 13, 139.
- Lemieux, J., Maunsell, E. & Provencher, L. (2008). Chemotherapy-induced alopecia and effects on quality of life among women with breast cancer: a literature review. *Psycho. Oncology.*, 17(4), 317–328.
- Liau, K. M. Lee, Y. Y., Chen, C. K., Hanum, A. & Rasool, G. (2011). An open-label pilot study to assess the efficacy and safety of virgin coconut oil in reducing visceral adiposity. *ISRN Pharmacol.*, 1-7.
- Ling, P. R., Istfan, W. N., Lopes, M. S, Babayan, K. V., Blackburn, L. G. & Bistrian, R. B. (1991). Structured lipid made from fish oil and medium-chain triglycerides alters tumor and host metabolism in Yoshida-sarcoma-bearing rats. *Am. J. Clin. Nutr.*, 53, 1177-84.
- Liu, Y. M. (2008). Medium-chain triglyceride (MCT) ketogenic therapy. *Epilepsia.*, 49 (8), 33-36.
- Liu, Y. C. & Wang, H. S. (2013). Medium-chain triglyceride ketogenic diet, an effective treatment for drug-resistant epilepsy and a comparison with other ketogenic diets. *Biomedical J.*, 36 (1), 9-15.
- Maalouf, M., Rho, J. M. & Mattson, M. P. (2009). The neuroprotective properties of calorie restriction, the ketogenic diet and ketone bodies. *Brain Res.Rev.*, 59(2), 293-315.
- Marina, A. M., Che Man, Y. B. & Amin, I. (2009). Virgin coconut oil: emerging functional food oil. *Trends Food Sci. Technol.*, 20, 481-487.
- Martin, D., Brand, C. & Telma, C. E. (2005). Physiological functions of the mitochondrial uncoupling proteins UCP2 and UCP3. *Cell Metab.*, 2(2), 85–93.
- Mendis, S., Puska, P. & Norrving, B. (2011). Global atlas on cardiovascular disease prevention and control. World health organization in collaboration with the World heart federation and the World stroke organization. 3–18.
- Nagao, K., Yanagita, T. (2010). Medium-chain fatty acids: Functional lipids for the prevention and treatment of the metabolic syndrome. *Pharmacol. Res.*, *61*, 208–212.
- Navab, M., Reddy S. T., Van Lenten, B. J., Anantharamaiah, G. M. & Fogelman, A. M. (2009). The role of dysfunctional HDL in atherosclerosis. *J. Lipid Res.*, *50*, 145–149.
- Navab, M., Reddy, S. T., Van Lenten, B. J. & Fogelman, A. M. (2011). HDL and cardiovascular disease: atherogenic and atheroprotective mechanisms. *Nat. Rev. Cardiol.*, 8, 222–232.
- Nebeling, L. C. & Lerner, E. (1995). Implementing a ketogenic diet based on medium-chain triglyceride oil in pediatric patients with cancer. J. Am. Diet Assoc., 95, 693-697.

- Neal, E. G., Chaffe, H., Schwartz, R. H., Lawson, M. S., Edwards, N., Fitzsimmons, G., Whitney, A. & Cross, J. H. (2009). A randomized trial of classical and medium-chain triglyceride ketogenic diets in the treatment of childhood epilepsy. *Epilepsia.*, 50, 1109-1117.
- Newport, M. T. (2010). Caregiver Reports Following Dietary Intervention with Medium Chain Fatty Acids in 60 Persons with Dementia. International Symposium of Dietery Interventions for Epilepsy and other Neurological Diseases.
- Nishi, Y., Hiejima, H., Hosoda, H., Kaiya, H., Mori, K., Fukue, Y., Yanase, T., Nawata, H., Kangawa, K. & Kojima, M. (2005). Ingested Medium-Chain Fatty Acids Are Directly Utilized for the Acyl Modification of Ghrelin. *Endocrinol.*, 146(5), 2255–2264.
- Nishi, Y., Mifune, H. & Kojima, M. (2012). Ghrelin Acylation by Ingestion of Medium-Chain Fatty Acids. *Methods Enzymol.*, *514*, 303-315.
- Onge, M. P. & St. Jones, P. J. H. (2003). Greater rise in fat oxidation with medium-chain triglyceride consumption relative to long-chain triglyceride is associated with lower initial body weight and greater loss of subcutaneous adipose tissue. *IJO.*, 27, 1565–1571.
- Otto, C., Kaemmerer, U., Illert, B., Muehling, B., Pfetzer, N., Wittig, R., Voelker, H. U., Thiede, A. & Coy, J. F. (2008). Growth of human gastric cancer cells in nude mice is delayed by a ketogenic diet supplemented with omega-3 fatty acids and medium-chain triglycerides. *BMC Cancer.*, *8*, 1-12.
- Page, K. A., Williamson, A., Yu, N., McNay, E. C., Dzuira, J., McCrimmon, R. J. & Sherwin, R. S. (2009). Medium-Chain Fatty Acids Improve Cognitive Function in Intensively Treated Type 1 Diabetic Patients and Support In Vitro Synaptic Transmission During Acute Hypoglycemia. *Diabetes.*, 58, 1237–1244.
- Pai, T. & Yeh, Y. Y. (1997). Desaturation of stearate is insufficient to increase the concentrations of oleate in cultured rat hepatocytes. J. Nutr., 127, 753–757.
- Papada, E., Kaliora, A. C., Gioxari, A., Papalois, A. & Forbes A. (2013). Anti- inflammatory effect of elemental diets with different fat composition in experimental colitis. *B.J.N.*, 1-8.
- Pietraszek, A., Gregersen, S. & Hermansen, K. (2011). Acute effects of dietary fat on inflammatory markers and gene expression in first-degree relatives of type 2 diabetes patients. *Rev. Diabet Stud.*, 8(4), 477–489.
- Pu, J. & Liu, P. (2012). Fatty Acids Stimulate Glucose Uptake by the PI3K/AMPK/Akt and PI3K/ERK1/2 Pathways. Protein Phosphorylation in Human Health. *Biochem. Genet. Molec. Biol.*, 4, 129-148.
- Ramanathan, A., Wang, C. & Schreiber, S. L. (2005). Perturbational profiling of a cell-line model of tumorigenesis by using metabolic measurements. *Proc. Natl. Acad. Sci. USA.*, 102, 5992-5997.
- Ramirez, M., Amate, L. & Gil, A. (2001). Absorption and distribution of dietary fatty acids from different sources. *Early Hum. Dev.*, 65, 95–101.
- Rebello, C. J., Keller, J. N., Liu, A. G., Johnson, W. D. & Greenway, F. L. (2015). Pilot feasibility and safety study examining the effect of medium chain triglyceride supplementation in subjects with mild cognitive impairment: A randomized controlled trial. *BBA Clinical*, *3*, 123–125.
- Reddy, B. S. & Maeura, Y. (1984). Tumor promotion by dietary fat in azoxymethane-induced colon carcinogenesis in female F344 rats: influence of amount and source of dietary fat. *J. Natl. Cancer Inst.*, 72(3), 745-50.

- Rho, J. M. & Stafstrom, C. E. (2012). The ketogenic diet: what has science taught us? *Epilepsy Res.*, 100 (3), 210-217.
- Richardson, L. C, Wang, W., Hartzema, A. G. & Wagner, S. (2007). The role of healthrelated quality of life in early discontinuation of chemotherapy for breast cancer. *Breast J.*, 13, 581–587.
- Rossell, J. B., King, B. & Downes, M. J. (1985).Composition of oil. J. Am. Oil Chem. Soc., 62(2), 221-230.
- Samra, R. A. (2010). Fat Detection. Taste, Texture, and Post Ingestive Effects. Fats and Satiety. Front. Neurol. Neurosci., 14, 1-37
- Scalfi, L., Coltorti, A. & Contaldo, F. (1991). Postprandial thermogenesis in lean and obese subjects after meals supplemented with medium-chain and long-chain triglycerides. *Am. J. Clin. Nutr.*, 53, 1130–1133.
- Seaton, T. B., Welle, S. L, Warenko, M. K. & Campbell, R. G. (1986). Thermic effect of medium-chain and long-chain triglycerides in man. Am. J. Clin. Nutr., 44, 630–634.
- Siddalingaswamy, M., Rayaorth, A. & Khanum, F. (2011). Anti-diabetic effects of cold and hot extracted virgin coconut oil. *J.DM.*, *1*(4), 118-123.
- Venkataraman, C. & Kuo, F. (2005). The G-protein coupled receptor, GPR84 regulates IL-4 production by T lymphocytes in response to CD3 crosslinking. *Immunol. Lett.*, 101, 144– 153.
- Wang, J., Wu, X., Simonavicius, N., Tian, H. & Ling, L. (2006). Medium-chain Fatty Acids as Ligands for Orphan G Protein-coupled Receptor GPR84. J. Bio. Chem., 281(45), 34457–34464.
- White, M. D., Papamandjaris, A. A. & Jones, P. J. H. (1999). Enhanced postprandial energy expenditure with medium-chain fatty acid feeding is attenuated after 14 d in premenopausal women. *Am. J. Clin. Nutr.*, *69*, 883–889.
- Wittenberger, T., Schaller, H. C. & Hellebrand, S. (2001). An expressed sequence tag (EST) data mining strategy succeeding in the discovery of new G-protein coupled receptors. J. Mol. Biol., 307, 799–813.
- Wolters, U., Keller, H. W., Becker, R., Stickeler, P., Dahlmeyer, M. & Müller, J. M. (1994). Influence of various fatty acids on tumor growth in total parenteral nutrition. *Eur. Surg. Res.*, 26, 288-297.
- Woollett, L. A., Spady, D. K. & Dietschy, J. M. (1992). Regulatory Effects of the Saturated Fatty Acids 6:0 through 18:0 on Hepatic Low Density Lipoprotein Receptor Activity in the Hamster, J. Clin. Invest., 89, 1133–1141.
- Yost, T. J. & Eckel, R. H. (1989). Hypocaloric feedings in obese women: Metabolic effects of medium-chain triglyceride substitution. *Am. J. Clin. Nutr.*, 49, 326-330.
- Yu, S., Derr, J., Etherton, T. D. & Kris-Etherton P. M. (1995). Plasma cholesterol-predictive equations demonstrate that stearic acid is neutral and monounsaturated fatty acids are hypocholesterolemic. Am. J. Clin. Nutr., 61, 1129-1139.

INDEX

#

1,7-diphenyl-2-picrylhydrazyl radical, 21
¹³C NMR, 148, 152
1-oleo -2-linoleo-3-linolein, 8, 9
2,2'-azinobis(3-ethylbenzothiazoline-6-sulfonic acid) diammonium salt cation radical, 22
2,2-diphenyl-1-picrylhydrazyl, 40
2,4-Decadienal, 23
24-Methylene cholesterol, 11
2-Decenal, 23
2-methylpentane, 80
2-Pentyl-furan, 23
³¹P NMR, 118, 148, 152, 154, 195
5α-Stigmasta-7,22-dien-3β-ol, 11, 14

A

abdominal obesity, 199, 213 Abscisic acid, 33 ABTS^{●+}, 21 acetic acid, 68, 82, 119, 141 acetone, 37, 125, 133 acetonitrile, 125, 133 acid index, 53, 60 acid value, 103, 110, 111, 112, 114, 115, 116, 118, 119, 126, 137 acidity, 57, 68, 75, 111, 112, 115, 137 acidity index, 68, 137 acne, 182, 193 acylglycerols, 94, 177 adenocarcinoma. 208 adenosine monophosphate, 205 adhesives, 87 adipose tissues, 200 adjuvants, 41 adsorbable compounds, 54

adulteration, 18, 131, 146, 148, 149, 150, 151, 152, 154, 156, 157 aflatoxin, 126, 128, 191 aflatoxin B1, 126 aflatoxin B2, 126 aflatoxin G1, 126 aflatoxin G2, 126 aflatoxin M1, 126 aflatoxin M2, 126 aflatoxins, 105, 125, 126 Africa, 129, 169 aging, vii, 93, 94, 95, 96, 97, 98, 99, 100, 101, 102, 111, 115, 193, 194, 201 alcholysis, 82 alcohol, 57, 77, 82, 85, 87, 132 aldehydes, 112, 114 aliphatic alcohols, 2 Alternaria solani, 166, 167, 168 amino acid(s), 57, 190, 192 analgesic, 191 animal feed, 54, 79 anisidine value, 69, 103, 114, 115, 126 antagonist function, 202 anti-aging, 175 anti-angiogenetic, 175 anti-bacterial, 175, 194 antibacterial action, 165 antibacterial activity, 164, 184, 195 antibacterial effect, 161, 162, 164 antibiotics, 31, 171, 179 anti-cancer, 175, 176, 192 anti-diabetic, 175, 198, 202 anti-fungal, 175, 194 antifungal activity, 167, 168, 185 antifungal fatty acids, 165 antifungal potential, 166, 168 anti-HIV, 175 anti-inflammatory, 99, 100, 101, 175, 191, 192 anti-leukemic, 175

Free ebooks ==> www.Ebook777.com

220

Index

anti-metastatic, 175 antimicrobial activity, vii, 159, 160, 161, 162, 164, 165, 168, 170, 171, 176, 182, 184 antimicrobial effect, 160, 161, 162, 163, 164, 165 antioxidant, vii, 10, 16, 17, 18, 19, 22, 24, 26, 29, 30, 31, 38, 39, 40, 41, 42, 43, 44, 45, 46, 47, 48, 49, 50, 51, 53, 57, 60, 70, 71, 75, 76, 81, 99, 103, 120, 122, 123, 127, 128, 158, 160, 194, 196, 207, 215 antioxidant activity, 24, 26, 29, 30, 38, 39, 40, 42, 43, 45, 46, 48, 49, 57, 60, 75, 123 antioxidant capacity(s), 24, 31, 38, 39, 40, 45, 47, 48, 50, 53, 60, 70, 122, 123, 127 antioxidant content, 43, 81 antioxidant enzyme, 124, 196 antioxidant potency, 30, 39 antioxidant property(s), vii, 16, 17, 18, 41, 45, 46, 49, 76, 103, 123, 124, 128, 158, 160, 194, 207 antioxidant substances, 123 antioxidants, 10, 17, 20, 24, 29, 30, 38, 39, 41, 42, 44, 48, 49, 60, 70, 71, 122, 124, 126, 128, 170, 191 antiplasmodial activity, 169 anti-proliferative, 175 anti-protozoal, 169, 175 anti-tumor, 175, 209 anti-viral, 175 antiviral activity, 169 anxiety, 210 AOCS, 26, 59, 60, 76, 77, 89, 141, 154 AOM stability, 144 APCC, 110, 111, 116, 117, 127, 132, 153 arachidic acid, 6 Argentina, 1 aroma, 109, 136, 146 aromatic amino acids, 31 arrhythmias, 206 ash, 104, 132 Asian Pacific Coconut Community, 110 Aspergillus flavus, 125 Aspergillus fumigatus, 166 Aspergillus niger, 167, 168, 183 Aspergillus oryzae, 175 Aspergillus parasiticus, 125 Aspergillus spp., 168 Aspergillus terreus, 167, 168 atherogenic dyslipidemia, 199 atherogenic risk factor, 198, 215 atherosclerosis, 19, 196, 201, 208, 216 ATP, 162, 201, 212 ATP synthase, 162 ATP synthesis, 162 authentication, vii, 131, 132, 146, 148, 149, 150, 152

authenticity, 114, 135, 136, 139, 152, 153 autocondensation, 30 azoxymethane, 208

B

baby shampoos, 193 Bacillus subtilis, 163 bacteria, vii, 109, 159, 160, 161, 162, 163, 164, 165, 173, 174, 178, 179, 183, 184, 191, 192, 193 bacterial membrane, 162, 173 baked goods, 80 baking mixes, 80 ball copra, 104, 124 beanv odor, 23 Beauveria bassiana, 174 Behenic acid, 5 beneficial effects, vii, 93, 94, 97, 98, 123, 211 beneficial health effects, 18, 126 benzoquinone, 22, 32 beta-sitosterol, 135 binary mixture, 151 bioderivatives, 159 biodiesel, vii, 1, 79, 82, 83, 84, 85, 86, 87, 88, 89, 90 biodiesel feedstocks, 79, 82 biofuel, 79, 80 biofuel feedstock, 80 biological active compounds, 24, 159 biological properties, 22 biomacromolecules, 175 biomedicine, 87 biopesticides, 174 biopreservatives, 177 biosynthesis, 18, 30, 31, 32, 33, 34, 41, 42, 43, 49, 161, 172, 177, 180, 204 bleached, 110, 119, 132, 145, 146, 147, 151, 188 bleaching, 2, 18, 20, 47, 123, 125, 146 blend, 83, 147, 149, 153, 173 blood, 6, 88, 93, 98, 100, 123, 124, 127, 190, 192, 194, 195, 197, 198, 199, 202, 203, 205, 207, 210, 211 blood pressure, 6, 199 blood vessels, 207 Blumeria graminis, 167 body weight, 96, 97, 201, 204, 205, 217 boiling point, 107 Botrytis cinerea, 166, 181 brain, 98, 100, 190, 195, 211, 212, 213 Brassicasterol, 11, 12, 135 Brazil, 1, 152 breaded fried foods, 80 breast cancer, 6, 191, 197, 210, 213, 216, 218 brownish, 106, 144, 188

www.Ebook777.com

chitosan, 170, 178

butter, 80, 81 buttery, 4

С

caffeic acid, 119 calcium, 190, 192, 193 Campesterol, 11, 12, 14, 122 Campylobacter jejuni, 164 cancer, 6, 18, 26, 30, 39, 42, 93, 97, 178, 191, 201, 208, 209, 210, 214, 216, 217 cancer prevention, 6, 201 cancers, 45 Candida albicans, 166, 167, 168, 178 canola oil. 81, 131, 149, 188 capric acid, 131, 133, 160, 163, 164, 165, 169, 174, 178, 184, 191, 198, 203, 212 caproic acid, 142, 160, 191, 203 caprylic acid, 122, 131, 133, 160, 164, 182, 191, 198, 203 caramelization, 109, 111 carbohydrates, 105, 109, 132, 188, 210 carbon dioxide, 56, 70, 71, 74, 75, 76, 77, 85, 90, 108, 118, 128, 171 carcinogenic compounds, 124 cardiovascular disease, 45, 93, 98, 100, 206, 208, 213, 216 carnitine, 189, 201 carotenoid(s), vii, 1, 3, 18, 19, 29, 30, 33, 34, 35, 36, 38, 39, 41, 42, 43, 44, 45, 46, 47, 48, 49, 50, 51, 119.178 castor oil, 152, 155 catalase, 99, 124, 207 catalyst, 82, 84, 85, 86, 87, 90 cataracts, 19 cellulase, 110, 117 cellulose, 110 centrifugation, 109, 110, 115, 116, 117, 118, 120 Centrifugation Process, 109 cephaline, 10 cereal grain, 37, 47 cerebrovascular disease, 206 chain length, 113, 122, 127, 133, 148, 157, 162, 164, 166, 168, 176, 188, 203, 214 chain reaction, 17 chemical parameters, 60, 116 chemical properties, 114, 147, 188 chemiluminescence, 34 chemometrics, 44, 149, 150, 151, 152 chemopreventive action, 18 chemotherapy, 176, 197, 210, 218 chilling, 117, 120, 122, 123 China, 1, 79

Chlorella vulgaris, 174 chlorophyll(s), 3, 19, 20, 25 Chlorophyll a, 19, 20 Chlorophyll b, 19 cholesterol, 6, 11, 18, 26, 88, 96, 98, 101, 122, 123, 124, 127, 128, 129, 187, 190, 194, 195, 199, 206, 207.218 cholesterol-lowering properties, 18 cholesteryl ester, 207 chorismate, 31 chromane, 32 chromanol phenolic moiety, 38 chronic, 39, 97, 98, 99, 172, 188, 190, 195 chronic diseases, 98, 99 chronic inflammation, 97, 99, 190 cis-phytol, 20, 27 citric acid, 163, 172, 212 citric acid cycle, 172, 212 Clostridium jejuni, 163 closures, 58 cloud point, 3, 80, 85 CO₂, 53, 56, 57, 58, 59, 61, 62, 63, 64, 65, 66, 69, 70, 72, 73, 74, 75, 76, 88, 108, 118 coatings, 56, 79, 80, 87, 176 coconut kernel, 103, 104, 105, 106, 108, 109, 110, 112, 118, 120, 121, 125, 129, 132, 196 coconut milk, 103, 104, 109, 110, 112, 115, 116, 117, 119, 120, 121, 122, 123, 126, 128, 146, 177 coconut oil quality, 136, 145 Cocos nucifera L., 132, 152, 154, 155, 179 Codex Alimentarius, 2, 4, 16, 25, 110, 117, 152, 154 co-enzyme Q, 22 Cold filter plugging point, 85 cold flow properties, 82, 85 Cold Flow Temperatures, 85 cold pressed coconut oils, 119, 122 cold pressing, 107, 119 Colletotrichum gloeosporioides, 167, 185 colon, 26, 191, 205, 208, 214, 215, 217 color, iv, 68, 81, 110, 131, 132, 140, 141, 144, 151 colorless, 105, 110, 118, 132, 188 colors. 2 column extraction, 40 commercial corn oil, 3, 18, 22, 26, 101 compressor, 58 condensation product, 32 conditioners, 170, 193 congenital heart disease, 206 control, 33, 43, 58, 96, 97, 98, 99, 102, 110, 132, 136, 148, 157, 169, 170, 171, 172, 177, 181, 192, 199, 201, 204, 209, 216

conventional extraction, 53, 56, 57, 68, 75

cooking oil, 81, 104, 188 copra, 103, 104, 105, 106, 107, 110, 112, 114, 115, 116, 117, 118, 119, 120, 121, 122, 123, 124, 125, 126, 127, 128, 132, 188, 207 copra oil, 103, 112, 114, 115, 116, 119, 121, 122, 123, 124, 126, 127, 207 Cordyceps militaris, 175 corn germ oil, vii, 1, 2, 16, 18, 26, 47, 53, 54, 55, 56, 57, 60, 62, 63, 64, 65, 66, 68, 69, 70, 71, 72, 73, 74.75.77 Corn kernel, 80 corn oil processing, 68 corn production, 79, 88 corn steeping, 71 corrosion inhibition, 87, 88 cosmetic applications, 54, 146 cosmetics, 157, 159, 170, 176, 187, 188 cosolvent, 57 cottonseed oil, 99 Crinipellis pernicosa, 167, 168 crispy texture, 81 critical point, 55, 56 crude coconut oil, 107, 134, 135, 144 crude corn oil, 1, 2, 54, 94 crude oil, 1, 10, 56, 77, 146, 160 crumb coatings, 80 Cucumerinum lagenarium, 166, 167 cup copra, 104 cyclization, 33 cyclopropane fatty acids, 173 cysteine, 209 Cytocrome P 450, 176

D

DDGS, 80, 88 death, 98, 162, 203, 208, 209, 212 decomposition, 3 decreased risk, 6 Deep fat frying, 81 deep-frying, 112 defatted meal, 57 degree of unsaturation, 8, 112, 133, 137, 168, 169 Delta-5-avenasterol, 135 Delta-7-avenasterol, 135 Delta-7-stigmastenol, 135 dementia. 212 density, 3, 56, 60, 61, 62, 63, 66, 68, 70, 123, 124, 139, 142, 143, 144, 187, 190, 206, 207 dental caries, 162, 196 deodorization, 2, 124, 125, 140, 188 deodorized, 110, 119, 125, 144, 145, 146, 147, 151, 152, 188

depression, 210 dermatitis, 187, 193 desiccated coconut, 104, 105, 106, 110, 125 desmethylsterols, 134, 135 deterioration, 57, 71, 73, 75, 105, 211 determination coefficient, 149, 150, 151 dewaxing, 2 diabetes, 48, 93, 96, 197, 202, 203, 204, 211, 217 diabetes mellitus, 203 diabetics, 192 diacylglycerides, 152 diacylglycerols, 9 dicarboxylic fatty acids, 200 diesel engines, 82 diesel fuel, 79, 82, 83, 86 diet, 93, 95, 96, 97, 98, 100, 101, 102, 127, 129, 160, 172, 174, 179, 185, 187, 189, 190, 191, 194, 195, 197, 198, 199, 202, 203, 204, 205, 206, 207, 208, 210, 212, 214, 215 dietary disorders, 187 dietary intake, 95, 99, 102, 196, 214 dietary management, 212 dietary oils, 188, 194 diets, 44, 93, 96, 97, 99, 101, 102, 123, 189, 198, 201, 207, 208, 217 differential pulse voltammetry, 35 Differential Scanning Calorimetry, 148, 150, 154, 156 diffusional resistance, 53, 61, 74 digestion, 174, 187, 190, 193, 194, 198, 206 dimensionless, 65 diode array detection, 121 discriminant analysis, 131, 149 distillation, 80, 107 divinilbenzene, 87 DNA damages, 98 DNA mutations, 98, 101 DPPH radical scavenging assay, 123 DPPH[•], 21 dressings, 80, 82 dry extraction, 103, 104, 107, 110, 126, 128 dry milling, 54 dry process, 103, 104, 108, 114 Dry-grind, 80 dry-grind process, 80 drying process, 104, 105, 106 dry-milled corn germ, 64, 75 DSC, 150, 151 dysbiosis, 172 dysfunction, 96, 97, 101

Ε

eczema, 193, 194 edible oil, 29, 44, 51, 56, 128, 146, 147, 155, 187 electrochemical methods, 35 electronic nose, 131, 136, 151, 152, 156 Emericella nidulans, 167 emission characteristics, 82, 86, 89, 90 emission spectra, 35 emulsifiers, 9, 170 emulsion, 9, 39, 73, 80, 82, 103, 104, 108, 109, 110, 115, 117, 120, 124, 128, 170, 174, 180, 198 emulsion stability, 82 endogenous antioxidant, 99 endogenous enzymes, 99 endosperm, 30, 42, 54, 104, 120, 132 endothermic peak, 150 energy, 9, 19, 72, 80, 93, 96, 97, 98, 99, 100, 109, 162, 172, 176, 188, 189, 190, 192, 196, 197, 198, 199, 200, 201, 203, 204, 205, 208, 211, 212, 213, 214, 216, 218 energy metabolism, 98, 198, 201, 204, 212 Engine Performance, 86 enthomopatogenic, 174 environmental factors, 212 environmental friendly, 53, 87 enzymatic treatment, 103 enzyme(s), 30, 31, 32, 33, 34, 43, 46, 49, 50, 55, 57, 84, 85, 97, 99, 108, 109, 110, 111, 117, 119, 125, 128, 129, 144, 162, 165, 171, 172, 175, 176, 177, 179, 183, 185, 192, 207, 214 Enzyme assisted extraction, 110, 117 epilepsy, 212, 216, 217 epithelial microbiota, 177 epoxides, 38 EPR, 39 Epstein Barr virus, 169 error limit, 66 Escherichia coli, 161, 163, 164, 165, 171, 178, 179, 182, 184 essential fatty acids, 81, 99, 188 ethanol, 57, 66, 79, 80, 84, 87, 88, 90 ethanolic extracts, 36 ethyl alcohol, 57 exo-biopolymer, 175, 183 exogenous antioxidants, 99 exothermic peaks, 150 expeller process, 116, 118 extra virgin coconut oil, 109, 212 extraction curves, 53, 57, 61, 63, 65, 66, 67, 74, 77 extraction kinetics, 53, 57 extraction method, 54, 103, 108, 118, 126

extraction of coconut oil, 104, 105, 107, 110, 117, 120, 123, 128 extraction pressure, 58, 61, 63, 64, 72, 74 extraction procedure, 68, 117 extraction rate, 62, 63 extraction temperature, 57, 62, 63, 68, 70, 71, 74 extraction time, 54, 57, 65, 108 extraction yield, 54, 57, 61, 62, 63, 64, 65, 77 extractor, 58, 65, 77

F

FADH2, 212 FAO/ALTO standards, 57 fat-soluble vitamins, 139 fatty acid methyl ester, 4, 26, 85, 86 fatty acid profiles, 6 fatty acid synthase, 98, 178, 205 fatty acids methyl esters, 169, 177 fatty acids profile, 60 fatty-acid radicals, 9 feed, 4, 64, 159, 171, 174, 179 feedstocks, 82 fermentation, 80, 103, 116, 118, 119, 120, 121, 123, 174, 175, 181 fermentation method, 123 fermentation process, 109, 116, 118, 119, 120, 121 ferric complexes, 41 Ferric-reducing antioxidant capacity, 22 ferulic acid. 20 FFA, 68, 69, 71, 72, 152 fiber, 26, 47, 80, 104, 131, 148, 151, 152, 156 fiber optical sensor, 131, 148, 152 fibrosis, 198, 200 flame ionization detector, 35 Flash point, 3, 80 flavor, 2, 3, 4, 22, 23, 57, 81, 82, 107, 109, 144, 146 flavor threshold, 23 Flaxseed, 95 florescence, 136 flour, 1, 2, 57, 87, 184 flow rate, 53, 62, 64 fluid phase, 65 fluoranthene, 125 fluorescence, 20, 30, 135 fluorescent detection, 35 fluoropolymer, 56 Folin-Ciocalteu, 21, 136 food consumption, 96 food control, 20 food engineering, 142 food industries, 146 food product, vii, 81, 144, 154, 194

food safety, 129, 160, 171 food samples, 30 foods, vii, 11, 27, 47, 60, 80, 104, 124, 129, 139, 142, 170, 189, 194 foregut, 171 Fourier transform infrared spectroscopy, 131 Fourier transform mid infrared, 150, 157 France, 1 FRAP. 22, 47, 60 free phytosterols, 11 free radical(s), 20, 21, 45, 49, 75, 112, 191, 193, 208 freezing, 103, 124 French fries, 80, 81 freshness, 68, 112 fried foods, 81, 194 frosting, 80 fry oil, 81 frying, 10, 24, 51, 79, 81, 84, 90, 91, 114 frying oil, 51, 79, 81, 84, 91 FTIR, 49, 114, 148, 149, 150, 152, 154, 157 FT-MIR, 150, 157 fuels, 106 functional alterations, 99 fungal, 105, 107, 125, 126, 165, 166, 168, 175, 177 fungal attacks, 105, 125 fungi, vii, 159, 161, 162, 168, 174, 175, 180, 181, 185.191 fungus, 174 furan fatty acids, 7 Fusarium avenaceum, 167 Fusarium oxysporum, 166, 167, 168

G

galactomannans, 110 gallic acid, 21, 46, 119, 120 gallic acid equivalents, 21, 120 gas chromatography, 23, 27, 35, 60, 116, 118, 127 gas chromatography-mass spectrometry, 30 gas-liquid chromatography, 3, 4, 133 gastrointestinal tract, 171, 172 GC, 18, 23, 30, 35, 133, 134, 136, 155, 156 GC-FID, 133, 134 GC-MS, 30, 35, 136 geranyl geranyl transferase, 32 germ, 1, 2, 11, 18, 29, 30, 53, 54, 55, 56, 57, 58, 59, 65, 67, 68, 69, 70, 71, 73, 75, 77, 79, 80, 93, 94 germ separation, 54, 71 germ storage, 71 ghrelin, 203, 204 GLC, 11, 133 glucose, 124, 190, 192, 197, 199, 201, 202, 203, 205, 207, 210, 211, 212, 214

glucose intolerance, 199 glucose level, 190, 197, 201, 202, 205, 210, 211 glucosyltransferase, 172, 180 glutathione, 38, 99, 124, 209 glutathione peroxidase, 99, 124 gluten, 54, 80 glycerides, 159, 161, 188 glycerol, 8, 9, 86, 111, 133, 138, 188 glycerolipids, 4 glycolipids, 2, 3, 72 glycolysis, 31, 210 Gram-negative, vii, 159, 160, 161, 162, 163, 165, 183, 184 Gram-positive, vii, 159, 160, 161, 162, 163, 164, 178, 179, 183 Grapeseed oil, 138 growth factors, 183 **GSH**, 99 gums, 54 gut microbiota, 172, 177, 181

H

Haemophilus influenzae, 165 Haemophilus spp., 165 haemostatic pathways, 207 Hanus method, 137 harmless health effects, 87 headspace, 23, 136 health benefits, vii, 24, 93, 94, 98, 100 healthy, 44, 46, 99, 101, 187, 190, 191, 193, 194, 196, 215 heart attacks, 19, 206 heart protection, 6 heat capacity, 142 heat of fusion, 142 heating and cooling systems, 58 heating curves, 150 Helicobacter pylori, 161, 165, 178, 183 hemicellulase, 110, 117 hepatitis C virus, 169, 191 hepatocellular carcinoma, 125 heptadecenoic acid, 177 herbal oils, 193 herbs, 56 heterogeneous system, 30, 39 hexanal. 114 hexane, 1, 2, 35, 36, 37, 38, 40, 46, 57, 70, 73, 80, 107, 108, 120, 135 Hexanoic acid, 23 high fat diet, 96 High performance liquid chromatography, 17, 48 high pressure, 37, 56, 58, 108

high-density lipoprotein, 123, 216 highly digestible structure, 81 hippocampus control, 211 homogentisic acid, 30, 31, 32 hops, 56 hormone, 177, 200, 203, 204 hot air drying, 106, 115 hot smoke, 106 HPLC, 8, 20, 35, 37, 38, 42, 46, 47, 48, 121, 134, 153 HPLC-DAD, 60, 61 HPLC-MS, 134 human consumption, 104 human diet, 93, 100, 188 human eye, 19, 27 human genome, 201 human health, 94, 151 human serum, 30, 94 human serum cholesterol, 94 humidity, 126 husk, 104, 105, 124 hydrocarbons, 2, 3, 124, 125, 127, 129, 136, 139 hydrolysis, 3, 9, 27, 68, 81, 105, 111, 112, 119, 134, 144, 162, 165, 170, 176, 198, 200 hydrolytic rancidity, 105, 112 hydroperoxide(s), 17, 69 hydrophobic, 9, 31, 161, 166, 173 hydrophobicity, 166, 173 Hypericum, 169, 182, 185 hypertension, 199, 206 hypocholesterolemic effect, 198 hypoglycaemic phases, 213 hypoglycemia, 211 hypolipidaemic, 175

inhibitory effect, 164, 168, 169, 170, 208, 209 initiation, 69, 70, 203 inner membrane, 161, 162 insoluble impurities, 114, 140 Insulating Applications, 88 insulin, 192, 196, 197, 199, 200, 201, 202, 203, 205, 210.213 insulin resistance, 197, 200 intensity of light, 30 intestinal inflammation, 198, 205 intracellular calcium, 202 intracellular metabolism, 202 intracytoplasmic lipid vacuoles, 96 iodine, 36, 103, 110, 112, 114, 116, 117, 120, 126, 129, 136, 137, 138, 140, 141, 146, 147, 158 iodine monobromide, 137 iodine value, 3, 80, 103, 110, 111, 112, 114, 115, 116, 117, 120, 126, 136, 137, 138, 143, 144, 145, 146, 147 IR, 30, 36, 42, 51 Isaria fumosorosea, 174, 181 ischemic myocardium, 206 ischemic stroke, 207 isoelectric region, 110 Isomers, 35

J

jasmonic acid, 33

isoprene-units, 18

K

Karl Fischer method, 140
kernel(s), 2, 30, 26, 37, 45, 47, 54, 79, 80, 94, 103, 104, 105, 106, 108, 109, 110, 118, 120, 121, 124, 127, 131, 132, 138, 147, 149, 150, 151, 152, 156
ketogenic diet, 208, 210, 212, 216, 217
ketone(s), 112, 136, 190, 192, 202, 210, 211, 212, 216
Kiln drying, 106
kiln-dried copra, 114
kinematic viscosity, 85
Kluyveromyces marxianus, 166
KOH, 3, 82, 111, 113, 115, 119, 138, 143

L

L. johnsonii, 173 lactic acid bacteria, 172, 181, 183 Lactobacillus rhamnosus, 172 lactones, 131, 132, 136, 144

Ι

immune system, 191, 198, 206 immune system activation, 198 immunomodulatory, 175 immunosuppression, 125 impurities, 111, 112, 116, 125, 136, 143, 144 incretin, 203, 204 India, viii, 26, 128, 152, 156, 177, 187, 196 industrial plants, 56 industrial scale, 56, 58, 104, 107, 109 infant formulas, 143, 195 inflammable, 88 inflammatory bowel disease, 197, 198 influenza virus, 169 ingredient, 1, 80, 104, 106 inhibition, 39, 60, 162, 166, 197, 198, 208 inhibitory action, 163, 184

Lagenidium giganteum, 174 L-asparaginase, 176 lauric acid, 111, 117, 119, 122, 131, 133, 137, 142, 143, 145, 147, 149, 150, 160, 161, 163, 165, 168, 169, 174, 180, 182, 183, 191, 198, 203, 207, 209, 214 LC-MS. 30 LDL, 98, 123, 127, 190, 194, 196, 207 LDL-cholesterol, 96, 98 lecithins, 9 leukemia, 98, 169 leukemia virus, 169 leukocyte, 206 lifespan, 95, 97, 99 light brown color, 110 lightness, 68 lignoceric acid, 6 linoleate-linoleate-linoleate, 8 linoleic, 4, 6, 7, 26, 54, 68, 69, 72, 94, 133, 147, 150, 160, 161, 163, 164, 169, 172, 173, 175, 176, 179, 180, 203, 215 linoleic acid, 6, 7, 26, 54, 68, 69, 72, 94, 133, 147, 160, 161, 164, 169, 172, 175, 176, 179, 180, 203, 215 linolenic acid, 3, 6, 7, 24, 94, 133, 160, 163, 169, 173, 181 linseed oil, 35, 133 lipase enzyme, 85, 111 lipid, 1, 2, 11, 38, 39, 40, 50, 57, 60, 71, 76, 77, 82, 88, 96, 98, 100, 101, 102, 103, 109, 111, 114, 120, 123, 124, 126, 127, 128, 137, 161, 166, 169, 171, 178, 179, 180, 184, 197, 199, 200, 203, 204, 207, 216 lipid fraction, 103, 114, 126 lipid free radicals, 60 lipid oxidation, 39, 60, 114 lipid peroxidation, 38, 40, 124, 169 lipid peroxides, 71 lipid peroxyl radicals, 38 lipid-free radical, 38 lipophilic antioxidants, 39 lipopolysaccharide, 161 liquid chromatography, 24, 30, 35, 48, 60, 128, 133, 134, 135, 155 liquid chromatography-mass spectrometry, 30, 134 liver-protective, 175 longevity, 93, 95, 98, 99, 100, 102 Lovibond, 140, 143, 144, 145 low-density lipoprotein, 6 low-grade inflammation, 98 lutein, 18, 19, 25, 26, 27, 30, 33, 37, 38, 42, 43, 44, 47.48 lycopene, 33, 34, 43

lymphocytic leukemia, 176

Μ

macronutrient, 196, 199 macular degeneration, 19 magnesium, 20, 193 Maillard reaction, 109, 115 maize, 1, 2, 29, 30, 33, 36, 42, 45, 46, 47, 51, 79 maize germ oil, 30 malabsorption, 200 malaria, 169 Malaysia, 152, 156 malodorous compounds, 54 mannans, 110 margarine(s), 1, 24, 79, 80, 81 marker for authentication, 20 mass spectrometry, 24, 35 mass transfer, 57, 61, 64, 71, 81 mayonnaise, 80, 82, 90 MCFAs, 171, 192, 198, 199, 200, 201, 202, 204, 205, 206, 211, 213 MCT, 197, 198, 199, 200, 201, 203, 204, 205, 206, 208, 210, 211, 212, 213, 216 MDA, 40 mechanical and thermophysical performance, 87 mechanical expelling, 1, 54, 58 mechanism(s), 25, 43, 50, 61, 93, 94, 129, 151, 162, 165, 168, 169, 170, 173, 179, 180, 185, 189, 202, 203, 204, 207, 209, 212, 213, 216 medical foods, 143 medium chain fatty acids, 113, 131, 132, 161, 162, 163, 168, 169, 170, 171, 193, 198, 206, 208 medium chain saturated triglyceride, 103 Medium-chain triglycerides, 194, 197, 213 melting point, 137, 142, 143, 145, 147 membrane, 39, 50, 161, 162, 163, 165, 169, 171, 173, 178, 179, 180, 183, 184, 201 metabolic pathway, 30, 122, 168 metabolic product, 30 metabolic syndromes, 198 metabolism, vii, 20, 45, 98, 100, 101, 122, 127, 131, 159, 161, 171, 177, 192, 193, 197, 198, 200, 203, 205, 207, 208, 213, 214, 215, 216 metal scavengers, 10 Metarhizium anisopliae, 174 methanogenesis, 174, 182 methanol, 20, 21, 36, 82, 85, 86, 87, 140 methanolic extract, 40 methanolysis, 83 methyl ester fatty acid, 82 methylation, 3 Mexico, 1

mice, 93, 95, 96, 97, 98, 99, 100, 102, 185, 195, 203, 204.213.217 microbes, 116 microbial cells, 160, 166 microbial contamination, 117 microbial metabolism, 159 microbicidal, 161, 184 microbiostatic, 160, 161 microbiota, 159, 172, 174, 177 microemulsions, 82, 165 microorganisms, 111, 116, 125, 159, 160, 161, 162, 170, 172, 174, 176, 184, 191 Microsporum gypseum, 166, 178 Microsporum spp., 168 microwave, 46, 50, 118 mild flavor. 81 mineral oil, 88, 196 minerals, 188, 190, 193 mitochondria, 99, 189, 200, 201, 212 mitochondrial anion carriers, 201 model correlation, 67 model parameters, 67 moisture, 54, 59, 61, 104, 105, 106, 115, 116, 118, 126, 132, 136, 140, 144, 157, 171, 176, 178, 194 molecular weight, 111, 113, 124, 137, 138, 146, 189 monoacylglycerides, 152 monocaprin, 162, 163, 164, 165, 184 monoethanolamine, 88 monohydroxylated carotenes, 37 monolaurin, 160, 161, 162, 163, 164, 169, 183 mononuclear cell leukemia, 97, 98 monosaturated fatty acids, 69 Mono-unsaturated fatty acids, 95 mortality rate, 96 **MSFA**, 69 mucilaginous materials, 125 multiple sclerosis, 212 multivariate calibrations, 149 multivariate data analysis tools, 131 muscle, 202, 203, 204, 212 mustard oil, 125 mycotoxins, 125 myocardial infarction, 206 myristic acid, 133, 160, 163, 169, 174, 198, 203 Myrothecium verrucaria, 166, 167

Ν

NADH, 212 NaOH, 82, 87, 113 natural antioxidants, 38 natural constituents, 3 natural origin, 173 natural products, 59, 77 Near-infrared reflectance spectroscopy, 36 Neera, 188 Neisseria gonorrhoeae, 161, 165 Neisseria spp., 165 neurodegenerative disease, 201 neuroprotective, 175, 216 neutral lipids, 53, 60, 69 n-hexan. 34 NIRS, 36, 42 nitrogen balance, 202 NMR, 30, 38, 42, 118, 134, 152, 155 nonanal, 23 non-aqueous capillary electrophoresis, 44 non-hydrogenated corn oil, 81 non-lipid acid, 68 non-lipid fraction, 103 non-lipid substances, 111, 113 nonpolar, 1, 2, 38, 118, 120, 124 normal diet, 95, 96, 212 nuclear magnetic resonance, 30, 131, 152, 158 nutraceuticals, 56, 184 nutrient(s), 98, 173, 174, 183, 188, 190, 192, 193, 194, 195, 214 nutritional, vii, 1, 25, 41, 81, 101, 111, 114, 116, 119, 120, 123, 125, 183, 187, 189, 194, 195, 199, 206.210 nutritional benefits, 187, 189, 195 nutritional damage, 123 nutritional deficiencies, 125 nutritional properties, vii, 81, 183 nutritional quality, 111, 114, 119, 120 nutty flavor, 24

0

octanoic acid, 119 odor, 131, 132, 136, 144, 170, 181 Official Methods., 59 oil fraction, 73, 74 oil quality, 54, 57, 68, 73, 74 oilseeds, 61 older population, 94 oleate-linoleate-linoleate, 8 oleic acid, 6, 8, 26, 68, 111, 160, 161, 163, 169, 172, 175, 176, 178, 180, 203 oleo-dipalmitin, 8 oligosaccharides, 173, 178 olive oil, vii, 18, 35, 37, 98, 101, 111, 119, 127, 129, 133, 147, 148, 154, 157, 158, 196, 208 operating pressure, 67, 70 optical fiber grating technology, 151 **ORAC**, 40

organ disorders, 98 organoleptic properties, 131 osmotic diarrhea, 205 osteoporosis, 193 oxalic acid, 20 oxidation products, 72, 109, 112, 113, 180 oxidation stability, 82 oxidative damage, 99 oxidative phosphorylation, 162, 185 oxidative rancidity, 22 oxidative stability, 9, 53, 60, 69, 71, 73, 77, 81, 91, 112, 114, 119, 127, 143, 156, 157 oxidative stress, 22, 98, 193, 212 oxygen, 3, 33, 43, 71, 99, 139, 175, 203, 209 Oxygen radical absorbance capacity, 40

Р

PAHs, 124, 125, 127 palm kernel oil, 147, 149, 150 Palm oil. 83. 138 Palmitic acid, 5, 167, 168, 189 Palmitoleic acid, 5, 189 pancreatic-enzyme, 198 paraffin oil, 151, 156 partial least square, 131, 149 partial least square regressions, 131 particle size, 57, 59 pathogenesis, 205 pathogenic bacteria, 163, 172 pathological conditions, 99 pathways, 31, 33, 49, 98, 210, 211 p-coumaric acid, 20, 119 Penicillium commune, 166 Penicillium glabrum, 167 Penicillium italicum, 167 Penicillium roquefortii, 166 pentose phosphate, 31 peripheral arterial disease, 206 peroxidase activity, 57 peroxide(s), 38, 53, 60, 70, 71, 103, 110, 112, 113, 114, 116, 117, 118, 126, 136, 141 peroxide value, 53, 60, 103, 110, 112, 113, 114, 116, 118, 126, 136, 141 Peroxisome Proliferator-Activated Receptor, 197, 200 peroxyl radicals, 21, 39, 40 pH, 82, 117, 161, 163, 164, 165, 166, 168, 175, 176, 178 phenolic acids, 20, 21, 119, 127 phenolic antioxidant, 120 phenolic compounds, 20, 21, 49, 119, 124, 126, 129, 136

phenolic hydrogens, 60 phenolic moiety, 40 phenolic substances, 118, 119, 120, 121, 122, 123 phosphatidylcholine, 10 phosphatidylethanolamine, 10 Phosphatidylglycerol, 10 phosphatidylinosite, 10 Phosphatidylinositol, 10 Phosphatidylserine, 10 phosphoinositides, 3 phospholipids, 1, 2, 3, 10, 56, 57, 71, 72, 123, 125, 166, 169, 213 phosphorus, 10, 152 p-hydroxyphenyl pyruvate, 31 p-hydroxyphenylpyruvate dioxygenase, 33, 44 physical parameters, 30, 60, 68 physico-chemical analysis, 132 physicochemical characteristics, 213 physiology, 159, 161 phytohormones, 33 phytol, 20, 32, 45, 50 phytonutrients, 11 phytosterol, 1, 2, 11, 16, 122 phytosterol fatty acid esters, 11 phytosterol fatty acyl esters, 1, 2 phytosterol ferulate esters, 1, 2 phytosterols, 1, 2, 3, 11, 16, 24, 94, 98, 122, 213 phytyl-diphosphate, 30, 31 Pichia anomala, 166 Plasmodium falciparum, 169, 181 PLS, 7, 149, 150, 151 Polenske value, 142, 144 polycyclic alcohols, 11 polycyclic aromatic hydrocarbons, 128 Polycyclic aromatic hydrocarbons, 124, 127, 129 polyethylene glycol, 133 polygalacturonases, 117 polymerization, 3, 81, 140 polyphenolic compounds, 170, 203 polyphenolic content, 203 Polyprenyltransferase, 31 **PPARy**, 200 ppb, 126 ppm, 11, 23, 30, 70, 107, 144 primary oxidation products, 112, 113 principal component analysis, 131 principal component regression, 149 probiotics, 173, 183 process parameters, 55, 58, 61, 77 pro-inflammatory, 93, 97, 98, 99, 100, 198, 205 pro-inflammatory cytokines, 93, 100, 198, 205 pro-inflammatory makers, 98, 99

protein, 38, 48, 57, 75, 76, 98, 102, 109, 110, 132, 171, 172, 200, 201, 202, 204, 205, 208, 218 protein isolates, 57 protein kinase, 102, 200, 202, 205 protein oxidation, 38 protein quality, 75 Proteus vulgaris, 163 pseudo-first order, 85 Pseudomnonas aeruginosa, 163 Pseudomonas oleovorans, 176 psoriasis, 187 PUFA, 69, 93, 94, 97, 98, 100, 164, 173 pyrene, 125 Pyrenophora avanae, 167, 168 pyridine, 140 pyrolisis, 82 Pythium ultimum, 167, 168

Q

quality of corn germ oil, 68
quality of frying oil, 81
quality of life, 96, 210, 216, 218
quality of oil, 70, 108, 114, 118
quality parameters, 60, 103, 110, 114, 115, 118, 126, 153
quenching, 17, 39

R

radical, 39, 40, 43, 47, 48, 49, 123 rancidity, 6, 22, 105, 109, 112, 114, 137, 189 Rancimat test, 73 rapeseed oil, 194 rats, 93, 97, 98, 99, 100, 101, 122, 123, 127, 128, 195, 203, 207, 208, 209, 214, 215, 216, 217 raw material, 59, 68, 146, 183 RBD, 110, 116, 118, 120, 123, 127, 145, 146, 147, 156, 188 reactive oxygen species, 38, 39, 99 reactive species, 39 recovery, 55, 59, 86, 108, 110, 118, 211 redness, 68 reductase, 32, 46, 98 re-esterification, 189 refined coconut oil, 138, 152, 188 refined corn oil, 4, 8, 20, 56, 95 refined sunflower oil, 37 refining, 1, 2, 9, 18, 25, 38, 44, 54, 56, 72, 76, 107, 123, 125, 188 refraction, 68, 139, 153 refractive index, 60, 116, 136, 139, 148, 151

refractometry, 131, 152 Reichert value, 141, 144 relative density, 139 relative humidity, 39, 46 reno-protective, 175 restaurant foods, 80 reverse-phase, 37 rheumatic heart disease, 206 rheumatoid arthritis, 6 Rhizoctonia solani, 167, 168 Rhodotorula mucilaginosa, 166 risk, 19, 30, 93, 99, 190, 206, 207, 213 risk factors, 207 risks of disease, 98 Romania, viii, 1, 29, 131, 159, 197 root mean error of calibration, 149 ROS, 33, 99 ROS scavengers, 99 ROS-antioxidant network, 33 rumen microbiota, 177

S

Saccharomyces cerevisiae, 166, 167, 182, 184 safety elements, 58 salad dressings, 80 salad oil, 79, 80 Salmonella spp., 161, 165, 173 saponification, 27, 35, 37, 47, 49, 82, 103, 113, 114, 115, 116, 126, 136, 138, 146 Saponification number, 80 saponification value, 103, 113, 114, 115, 116, 126, 136, 138, 146 saturated fats, 93, 132, 187, 190 saturated fatty acids, 3, 6, 24, 112, 122, 127, 160, 162, 163, 168, 171, 180, 182, 191, 195, 198 scavenger, 39 scavenging assay, 123 SC-CO₂, 53, 56, 57, 58, 59, 61, 62, 63, 64, 65, 68, 69, 70, 71, 72, 73, 74, 75, 108, 118 secondary oxidation products, 112, 113 semiquinone radical, 22 sensitizers, 20 separator, 72, 73, 74, 108 serum, 46, 81, 96, 100, 101, 123, 190, 195, 202, 206, 207, 212 serum cholesterol, 81, 190 SFA, 69, 94, 203 SFE, 55, 56, 57, 58, 61, 62, 64, 65, 66, 68, 72, 73 shikimic acid pathway, 31 significant difference, 86, 96, 115, 116, 117, 118, 119 singlet oxygen, 17, 44

sitosterol, 3, 11, 13, 16, 135, 147, 182 skin care, 104 skin care products, 104 skin irritations, 170 slight nutty, 4 slip melting point, 136, 147 SMLR, 150 smoke, 106, 125, 188, 194 smoke point, 188 snack foods, 80 soap, 80, 86, 88, 104, 134, 187, 188 SOD, 99 sodium, 85, 137, 139, 141, 179 sodium thiosulfate, 137, 141 soft deodorized, 153 solid matrix, 53, 61, 74 solid phase extraction, 60 solid phase microextraction, 118, 136 solid-phase mass transfer coefficient, 54, 67 solubility value, 53, 66, 67 solvent, 2, 36, 38, 48, 54, 55, 56, 57, 58, 61, 63, 64, 65, 66, 67, 68, 70, 71, 80, 103, 107, 108, 118, 125, 139, 140, 141, 188 solvent capacity, 63 solvent extraction, 38, 48, 54, 68, 107, 108 Solvent Extraction, 107 solvent flow rate, 54, 57, 64, 65, 67 soybean oil, 30, 34, 38, 43, 81, 94, 96, 108, 125, 129, 150, 157 spaghetti sauce, 80 SPE, 60 specific gravity, 80, 116, 136, 139, 80, 143, 153 spectrophotometric assay, 35, 39 spectrophotometric method, 21, 40, 50 spectroscopy, 30, 36, 42, 114, 118, 148, 149, 150, 152.157 spices, 56 SPME, 118, 136 spoilage, vii, 68, 108, 159, 160, 171, 184 Spreads, 81 squalene, 3, 17, 18, 26, 131, 132 Sri Lanka, viii, 103, 126, 128, 129, 152 stability, 2, 9, 22, 24, 25, 38, 43, 57, 60, 69, 71, 72, 73, 75, 76, 77, 81, 82, 91, 112, 119, 123, 125, 127, 128, 144, 156, 157, 170, 177, 196 stability against oxidation, 57, 75 stability to oxidation, 72 stabilizers, 170 standard deviation, 115, 121 Staphylococcus aureus, 163, 164, 178, 180, 181 Staphylococcus epidermidis, 163, 164 starch, 2, 54, 80, 94, 137, 141

stearic acid, 6, 7, 8, 160, 172, 175, 176, 203, 206, 207, 213, 215, 218 stepwise multiple linear regression, 150 stepwise reactions, 85 sterane, 11 sterol esters. 4 sterols, 2, 11, 16, 28, 51, 69, 93, 98, 100, 126, 131, 132, 134, 135, 147, 152, 154, 195 Stick margarine, 81 Stigmasta-8,22-dien-3β-ol, 11, 14 stigmasterol, 3, 16, 147 strains, 163, 164, 165, 168, 173, 180 streptococci, 163, 180 Streptococcus agalactiae, 164 Streptococcus dysgalactiae, 164 Streptococcus pyogenes, 164 Streptococcus spp., 163 Streptococcus uberis, 164 styrene, 87 sugars, 111 sulfur dioxide, 106, 140 sulfuric acid, 83 sun drying, 106, 105 sunflower oil, 35, 37, 75, 77, 102, 133, 148, 150, 207 supercritical carbon dioxide, 56, 75, 76, 77, 85, 90, 108, 118, 128 supercritical CO₂, 53, 73, 77 supercritical crude corn oil, 53 supercritical fluid, vii, 2, 23, 47, 53, 55, 56, 58, 59, 61, 64, 75, 76, 77 supercritical fluid extraction, vii, 2, 53, 55, 58, 59, 61, 64, 76, 77 superoxide dismutase, 99, 124, 207 superoxide radicals, 21 supplements, 100, 195 surface tension. 56 synaptic transmission, 197, 211 synergic effect, 71, 203 synthesis, 49, 56, 98, 169, 171, 175, 200, 207, 209

Т

TAG, 69, 132, 133 TEAC, 22 technology, 56, 73, 76, 108, 158 temperature, 2, 9, 23, 30, 35, 39, 48, 53, 55, 57, 58, 61, 62, 63, 64, 65, 70, 72, 74, 81, 82, 85, 87, 88, 106, 107, 108, 109, 112, 115, 116, 118, 120, 123, 132, 139, 140, 142, 150, 163, 171, 174, 175, 176 termination, 69 terpenoid, 18, 43 testa, 105, 106, 110, 120 texture, 82, 96, 109

therapeutic agents, 197, 198, 209 therapies, 41, 212 thermal profiling, 150 thermal stability, 114, 123 thermodynamic parameter, 64 thermotolerance, 174 thin-layer chromatography, 48, 50, 134 thrombosis, 207 thyroid, 193 tintometer, 140 tissues, 33, 99, 100, 111, 122, 177, 191, 200, 201, 202, 203, 204, 213 TLC, 34, 36, 47 tocochromanol, 33, 48 tocol, 29, 147 tocopherol(s), 1, 3, 10, 16, 17, 18, 25, 30, 31, 32, 33, 34, 35, 36, 38, 39, 42, 43, 44, 45, 46, 47, 48, 49, 50, 51, 56, 57, 60, 61, 70, 71, 72, 73, 74, 75, 76, 94, 95, 100, 101, 121, 124, 135, 147 tocotrienol(s), vii, 1, 2, 17, 18, 25, 26, 30, 32, 34, 39, 42, 43, 45, 46, 48, 49, 50, 76, 135 tooth decay, 192 Torulopsis spp., 176 total antioxidant capacity, 39, 40, 43 total phenol content(s), 119, 120 total sterols, 11, 134, 135 total trapping antioxidant parameter, 40 total-radical-trapping antioxidant parameter, 22 toxicant, 88 toxins, 174, 191, 212 trace amounts, 4, 71, 132 trade specifications, 145 traditional coconut oil, 112, 115, 119, 120, 121, 123 trans fatty acid, 81 transesterification, 82, 83, 84, 85, 86, 87, 90 Transesterification Kinetics. 84 transgenic expression, 33 transgenic plants, 33 trans-phytol, 20 TRAP, 22, 40 triacylglycerides, 68, 72, 73, 138 triacylglycerol, 1, 8, 9, 132, 133, 153, 155, 196, 199 Trichoderma viride, 166, 167 Trichophyton rubrum, 167, 183 Trichophyton spp., 168 triglyceride(s), 2, 3, 57, 72, 81, 83, 84, 98, 105, 111, 113, 114, 118, 119, 122, 126, 132, 143, 147, 153, 157, 165, 171, 187, 188, 190, 194, 195, 196, 198, 200, 203, 204, 208, 213, 214, 215, 216, 217, 218 trilinolein, 72, 133 trilinolenin, 133 triolein. 133 Trolox, 22, 39, 123

Trolox equivalent antioxidant capacity, 22, 39 Tub margarine, 81 tubing, 58 tumor, 101, 198, 201, 205, 208, 210, 216, 218 tumor necrosis factor, 201, 205 tumoral cells, 197, 209 tyrosine, 31, 49

U

ubiquinol, 22 ubiquinone, 22, 94 UHPLC-UV, 37, 47 ultraviolet-visible, 30, 43 uncoupling proteins, 178, 201, 214, 216 United States of America, 1 unsaponifiable matter, 114, 116, 122, 131, 132, 136, 139 unsaponifiables, 3, 24 unsaturated fatty acid, 81, 82, 87, 174 UV detection, 36 UV spectra, 34, 35 UV-Vis, 30, 34, 44, 136 UV-Vis spectroscopy, 34, 44

V

valves, 58, 59 vapor pressure, 70 vapour pressure, 62 VCO, 155, 157, 170, 188, 199, 203, 207, 210, 216 virgin coconut oil, vii, 103, 107, 108, 110, 112, 114, 115, 116, 117, 118, 119, 120, 122, 123, 126, 127, 131, 132, 136, 140, 146, 147, 148, 149, 150, 151, 152, 153, 156, 157, 170, 178, 182, 188, 192, 199, 203, 207, 215, 216 virgin olive oil, 101, 119, 124, 128, 146, 148 viruses, vii, 159, 160, 162, 169, 191 viscosity, 3, 80, 82, 85, 89, 142, 143, 146 visible absorption spectra, 36 vision benefits, 19 vitamin C. 38 vitamin E, 3, 17, 24, 25, 26, 30, 35, 38, 39, 43, 44, 45, 47, 48, 49, 50, 51, 93, 94, 99, 100, 118, 119, 121, 122 vitamins, 31, 45, 99, 122, 126, 188, 190 volatile compounds, 1, 2, 23, 114, 136, 151 volatile organic matter, 116

W

walnut, 93, 94

Free ebooks ==> www.Ebook777.com

waste frying oil, 84 a water, 2, 3, 20, 21, 25, 33, 36, 42, 49, 56, 57, 75, 82, 84, 90, 99, 104, 105, 106, 108, 109, 110, 116, α-amylase, 117, 175, 181 117, 120, 129, 132, 139, 140, 141, 142, 162, 170, a-Tocopherol, 16, 47 171, 174, 181, 185, 188, 205 α-tocopheryl acetate, 35 water-in-oil emulsions, 49, 174, 185 a-Tocotrienol, 17 waxes, 1, 2, 54 wet extraction, 103, 104, 108, 110, 112, 117, 120, 123.126 ß wet milling, 54, 55, 80, 94 wet process, 103, 108, 114, 116 β-apo-8'-carotenal, 37 wheat germ oil, 94 β-carotene, 30, 33, 34, 35, 38, 44 whipped toppings, 80 β-cryptoxanthin, 30, 33 Wijs method, 3, 137 β-hydroxybutyrate, 211, 212 winterization, 2 β-oxidation, 172, 176, 200 World Health Organization (WHO), 125, 169, 183 β-sitosterol, 16, 46, 169 β-Tocopherol, 16 β-Tocotrienol, 17 Х xanthophylls, 37, 47 γ γ -methyltransferase, 32 Y γ-Tocopherol, 16 γ-Tocotrienol, 17 Yarrowia lypolitica, 171 yeasts, vii, 159 yellow, 19, 110, 116, 140, 141, 144, 146, 188 δ yellowish color, 105, 109

δ-Tocopherol, 16

δ-Tocotrienol, 17

 ω -6 fatty acid, 6

ω-9 fatty acid, 6

ω

yellowness, 68

yield percentage, 70

Zea mays L., 1, 2, 47, 51

188

yielding, 57

yield, 2, 54, 70, 84, 85, 87, 108, 109, 116, 117, 118,

Ζ

zeaxanthin, 18, 19, 25, 26, 27, 30, 33, 37, 42, 43, 47

Index